

# AZO TEXTILE DYES WASTEWATER TREATMENT WITH CONSTRUCTED WETLANDS

Design and operation of experimental vertical-flow constructed wetlands applied for the treatment of azo textile dyes (with/without artificial wastewater)

Amjad Muhammed Fadhel Hussein

School of Computing, Science and Engineering

University of Salford

Manchester, UK

Submitted in Partial Fulfilment of Requirements of the Degree of

Doctor of Philosophy, June 2017

### Contents

Contents	II
List of Figures	VIII
List of Tables	XIV
ACKNOWLEDGEMENTS	XVI
DECLARATION: LIST OF ORIGINAL PAPERS AND CONFERENCES	XVII
ACRONYMS AND ABBREVIATIONS	XIX
ABSTRACT	XXII
1. CHAPTER ONE: INTRODUCTION	1
1.1 Overview	1
1.2 Background of the Research	1
1.3 Problem Statement	13
1.4 The novelty of the work	14
1.5 Aim of the study and objectives	15
1.6 Thesis Structure	16
2. CHAPTER TWO: CRITICAL LITERATURE REVIEW	19
2.1 Overview	19
2.2 Historical Development of Constructed Treatment Wetlands	19
2.3 Components of Constructed Wetlands	25
2.3.1 The Water	26
2.3.2 The Media	27

	2.3.3 The Plants	30
	2.3.4 Micro-organisms	32
	2.4 Classification of Constructed Wetlands for Wastewater Treatment	33
	2.4.1 Overall Classification	33
	2.4.2 Free Water Surface Flow Constructed Wetlands	34
	2.4.3 Subsurface Flow	36
	2.4.3.1 Horizontal Subsurface Flow Constructed Wetlands	36
	2.4.3.2 Vertical Subsurface Flow Constructed Wetlands	
	2.4.4 Hybrid Flow Constructed Wetlands	41
2	2.5 Reduction Mechanisms in a Constructed Wetland	45
	2.5.1 Total Suspended Solids Reduction Mechanism	45
	2.5.2 Organic Matter Reduction Mechanism	47
	2.5.3 Nutrients Reduction Mechanism	49
	2.5.4 Heavy Metals Reduction Mechanism	53
	2.5.5 Azo Dye Reduction Mechanism	56
2	2.6 Choice of Vertical-Flow Constructed Wetlands over Horizontal-Flow	61
	2.6.1 Nitrification and Denitrification	61
	2.6.2 Chemical Oxygen Demand (COD)	62
	2.6.3 Heavy Metals	62
	2.6.4 Area Demand	63
4	2.7 Amine Compounds	64

2.7.1 Overview6	54
2.7.2 Classification and Structure of Amines	55
2.7.2.1 Primary Amines	55
2.7.2.2 Secondary Amines	56
2.7.2.3 Tertiary Amines	56
2.8 Textile Wastewater Treatment within Vertical-Flow Constructed Wetlands6	57
2.9 Concluding Remarks	72
3. CHAPTER THREE: MATERIALS AND METHODS	73
3.1 Overview	73
3.2 Experimental Set-up	73
3.2.1 Site Description	73
3.2.2 Wetland Design and Media Composition	75
3.2.3 Phragmites Australis	76
3.2.4 Dye Details	77
3.2.5 Aquarium Plant Nutrient	79
3.2.6 Artificial Wastewater Compositions	79
3.3 Environmental Conditions	30
3.3.1 Operational Conditions	30
3.3.1.1 Operational Conditions for Experimental Constructed Wetland Set-Up	
Treating Azo Textile Dyes Only	31

3.3.1.2 Operational Conditions for Experimental Constructed Wetland Set-Up
Treating Artificial Wastewater Containing Two Azo Textile Dyes
3.3.2 Measurement of Water Quality Parameters
3.3.2.1 Hydraulic Loading Rate
3.3.2.2 Mass Loading Rate (MLR)
3.3.2.3 Percentage Reduction Rate
3.3.2.4 Analytical Methods and Equipment
3.3.2.4.1 Measurement of Physical Parameters
3.3.2.4.2 Measurements of Chemical Parameters
3.3.2.4.3 Statistical Analysis Methods91
3.4 Experimental Research Limitations92
3.5 Concluding Remarks94
4. CHAPTER FOUR: RESULTS AND DISCUSSION FOR TWO AZO TEXTILE
DYES REDUCTION95
4.1 Overview
4.2 Comparison of Unwashed Gravel versus Washed Gravel
4.3 Test of Normality
4.4 Plant Growth Monitoring
4.5 Redox Potential and Dissolved Oxygen101
4.6 Electrical Conductivity, Total Suspended Solids and Turbidity107
4.7 pH Value114

	4.8 Nutrient Reduction	121
	4.9 Dye and Chemical Oxygen Demand Reductions	126
	4.10 Seasonal Treatment Performance of Wetlands on Dye Reduction	132
	4.11 Concluding Remarks	134
5	. CHAPTER FIVE: RESULTS AND DISCUSSION FOR ARTIFICIAL	
V	VASTEWATER CONTAINING TWO AZO TEXTILE DYES REDUCTIONS	135
	5.1 Overview	135
	5.2 Test of Normality	135
	5.3 Plant Growth Monitoring	136
	5.4 Redox Potential and Dissolved Oxygen	137
	5.5. Electrical Conductivity, Total Suspended Solids and Turbidity	144
	5.6 pH Value	150
	5.7 Dye, Colour and Chemical Oxygen Demand Reductions	158
	5.8 Nutrient Reduction	165
	5.9 Seasonal Comparison of Effluent Dyes Reduction	169
	5.10 Aromatic Amines Reduction	172
	5.11 Concluding Remarks	176
6	. CONCLUSIONS AND RECOMMENDATIONS	177
	6.1 Conclusions	177
	6.2 Recommendations	179
	References	180

APPENDIX A	.217
Published Journal Paper	.217
APPENDIX B	.227
Test of normality for plant and water quality variables	.227
B.1 Azo textile Dyes	.227
B.2 Artificial wastewater containing two azo textile dyes	.233

# List of Figures

Figure 1.1 Worldwide renewable water resources (m <sup>3</sup> per capita per year, adapted from
Sultana, 2014)2
Figure 1.2 The effluent of textile industry- China (Wood, 2010)4
Figure 1.3 The effluent of textile industry- India (Ramesh, 2011)5
Figure 1.4 Flowchart for the general steps in textile industry Adopted from Dasgupta et
al. (2015)
Figure 2.1 Schematic representation of vertical-flow constructed wetlands (Tilley et al.,
2014)
Figure 2.2 Classification of constructed wetlands (Stefanakis et al., 2014)
Figure 2.3 Schematic representation of free water surface flow constructed wetland
(Oki and White, 2011)
Figure 2.4 Plants for free water surface flow constructed wetlands (Kamariah, 2006) .36
Figure 2.5 Schematic of the horizontal subsurface flow constructed wetland (Tilley et
al., 2014)
Figure 2.6 Schematic of the vertical subsurface flow constructed wetland (Tilley et al.,
2014)
Figure 2.7 Various combinations of different constructed wetland types in hybrid
constructed wetlands; (a), VSSF-HSSF, (b); HSSF-VSSF, (c); HSSF-VSSF-HSSF, (d);
VSSF-HSSF-HSSF, (e); VSSF-HSSF-FWS, (f); HSSF-FWS-HSSF (Stefanakis et al.,
2014)
Figure 2.8 Schematic diagram showing the main nitrogen reduction mechanisms
(Mustafa, 2010; adapted from Reddy and DeLaune, 2008)
Figure 2.9 Aromatic amine rings
Figure 2.10 Methylamine (Primary Amines) chemical structure

Figure 2.11 Dimethylamine (Secondary Amines) chemical structure
Figure 2.12 Trimethylamine (Tertiary Amines) chemical structure
Figure 3.1 Experimental vertical-flow wetland rig located in greenhouse, Newton
Building (May 2015)
Figure 4.1 Total suspended solids comparison between Wetlands 1, 2 and 397
Figure 4.2 Turbidity comparison between Wetlands 1, 2 and 397
Figure 4.3 Electrical conductivity comparison between Wetlands 1, 2 and 3
Figure 4.4 Temporal variation of temperature throughout the experimental period100
Figure 4.5 Overall variation for redox potential in the influent and effluent for Wetlands
4, 8/9 and 10/11 (low concentration of BR46)
Figure 4.6 Overall variation for redox potential in the influent and effluent for103
Figure 4.7 Overall variation for redox potential in the influent and effluent for Wetlands
14/15 and 18/19 (high concentration of BR46)
Figure 4.8 Overall variation for redox potential in the influent and effluent for Wetlands
16/17 and 20/21 (high concentration of AB113)
Figure 4.9 Overall variation for dissolved oxygen in the influent and effluent for
Wetlands 4, 8/9 and 10/11 (low concentration of BR46)
Figure 4.10 Overall variation for dissolved oxygen in the influent and effluent for
Wetlands 5, 6/7 and 12/13 (low concentration of AB113)105
Figure 4.11 Overall variation for dissolved oxygen in the influent and effluent for
Wetlands 14/15 and 18/19 (high concentration of BR46)
Figure 4.12 Overall variation for dissolved oxygen in the influent and effluent for
Wetlands 16/17 and 20/21 (high concentration of AB113)106
Figure 4.13 Overall variation for total suspended solids in the influent and effluent for
Wetlands 4, 8/9 and 10/11 (low concentration of BR46)

Figure 4.14 Overall variation for total suspended solids in the influent and effluent for
Wetlands 14/15 and 18/19 (high concentration of BR46)110
Figure 4.15 Overall variation for total suspended solids in the influent and effluent for
Wetlands 5, 6/7 and 12/13 (low concentration of AB113)110
Figure 4.16 Overall variation for turbidity in the influent and effluent for Wetlands 4,
8/9 and 10/11 (low concentration of BR46)
Figure 4.17 Overall variation for turbidity in the influent and effluent for Wetlands
14/15 and 18/19 (high concentration of BR46)
Figure 4.18 Overall variation for turbidity in the influent and effluent for Wetlands 5,
6/7 and 12/13 (low concentration of AB113)113
Figure 4.19 Overall variation for turbidity in the influent and effluent for Wetlands
16/17 and 20/21 (high concentration of AB113)114
Figure 4.20 Overall variation in pH of the influent and effluent for Wetlands 4, 8/9 and
10/11 (low concentration of BR46)
Figure 4.21 Overall variation in pH of the influent and effluent for Wetlands 14/15 and
18/19 (high concentration of BR46)
Figure 4.22 Overall variation in pH of the influent and effluent for Wetlands 4, 6/7 and
12/13 (low concentration of AB113)
Figure 4.23 Overall variation in pH of the influent and effluent for Wetlands 16/17 and
20/21 (high concentration of AB113)118
Figure 4.24 Influent and effluent concentrations of ortho-phosphate-phosphorus for
Wetlands 16/17 and 20/21 (high concentration of AB113)122
Figure 4.25 Influent and effluent dye concentrations for Wetlands 4, 8/9 and 10/11 (low
concentration of BR46)126

Figure 4.26 Influent and effluent dye concentrations for Wetlands 5, 6/7 and 12/13 (low
concentration of AB113)
Figure 4.27 Influent and effluent dye concentrations for Wetlands 14/15 and 18/19
(high concentration of BR46)
Figure 5.1 Overall variation for redox potential in the influent and effluent for Wetlands
1, 5 and 7 (low concentration of BR46)
Figure 5.2 Overall variation for redox potential in the influent and effluent for Wetlands
2, 3 and 9 (low concentration of AB113)
Figure 5.3 Overall variation for redox potential in the influent and effluent for Wetlands
4, 6, 8 and 10 (low concentration of the mixture of dyes BR46 and AB113)
Figure 5.4 Overall variation for redox potential in the influent and effluent for Wetlands
11 and 15 (high concentration of BR46)140
Figure 5.5 Overall variation for redox potential in the influent and effluent for Wetlands
13 and 17 (high concentration of AB113)140
Figure 5.6 Overall variation for redox potential in the influent and effluent for Wetlands
12, 14, 16 and 18 (high concentration of the mixture of dyes BR46 and AB113)141
Figure 5.7 Overall variation for dissolved oxygen in the influent and effluent for
Wetlands 1, 5 and 7 (low concentration of BR46)142
Figure 5.8 Overall variation for dissolved oxygen in the influent and effluent for
Wetlands 2, 3 and 9 (low concentration of AB113)
Figure 5.9 Overall variation for dissolved oxygen in the influent and effluent for
Wetlands 11 and 15 (high concentration of BR46)143
Figure 5.10 Overall variation for dissolved oxygen in the influent and effluent for
Wetlands 11 and 15 (high concentration of BR46)144

Figure 5.11 Overall variation for total suspended solids in the influent and effluent for
Wetlands 1, 5 and 7 (low concentration of BR46)146
Figure 5.12 Overall variation for total suspended solids in the influent and effluent for
Wetlands 11 and 15 (high concentration of BR46)147
Figure 5.13 Overall variation for total suspended solids in the influent and effluent for
Wetlands 2, 3 and 9 (low concentration of AB113)
Figure 5.14 Overall variation for total suspended solids in the influent and effluent for
Wetlands 13 and 17 (high concentration of AB113)
Figure 5.15 Overall variation for total suspended solids in the influent and effluent for
Wetlands 4, 6, 8 and 10 (low concentration of the mixture of both of the dyes)148
Figure 5.16 Overall variation for total suspended solids in the influent and effluent for
Wetlands 12, 14, 16 and 18 (high concentration of the mixture of both of the dyes)149
Figure 5.17 Overall variation in pH of the influent and effluent for Wetlands 1, 5 and 7
(low concentration of BR46)
Figure 5.18 Overall variation in pH of the influent and effluent for Wetlands 11and 15
(high concentration of BR46)
Figure 5.19 Overall variation in pH of the influent and effluent for Wetlands 2, 3 and 9
(low concentration of AB113)
Figure 5.20 Overall variation in pH of the influent and effluent for Wetlands 13 and 17
(high concentration of AB113)154
Figure 5.21 Overall variation in pH of the influent and effluent for Wetlands 4, 6, 8 and
10 (low concentration of mixture of both of the dyes)
Figure 5.22 Overall variation in pH of the influent and effluent for Wetlands 12, 14, 16
and 18 (high concentration of mixture of both of the dyes)

Figure 5.23 Overall variation in dye concentration of the influent and effluent for
Wetlands 11 and 15 (high concentration of BR46)
Figure 5.24 Overall variation in dye colour of the influent and effluent for Wetlands 11
and 15 (high concentration of BR46)159
Figure 5.25 Overall variation in dye concentration of the influent and effluent for
Wetlands 13 and 17 (high concentration of AB113)
Figure 5.26 Overall variation in dye colour of the influent and effluent for Wetlands 13
and 17 (high concentration of AB113)
Figure 5.27 Absorbance of ABSA released for Wetlands 2, 3 and 9 (low concentration
of AB113)
Figure 5.28 Absorbance of ABSA released for Wetlands 13 and 17 (high concentration
of AB113)
Figure 5.29 Absorbance of NBNMD released for Wetlands 1, 5 and 7 (low
concentration of BR46)175
Figure 5.30 Absorbance of NBNMD released for Wetlands 11 and 15 (high
concentration of BR46)

## List of Tables

Table 1.1 Some examples of azo dye (listed in order of date) effects on human health
and the environment5
Table 2.1 Wetland media characteristics (Strickland and Korleski, 2007)         29
Table 2.2 Some related studies (listed in order of date) on azo textile dye wastewater
treatment by constructed wetlands
Table 2.3 Examples of area requirements (listed in order of alphabet letters)         63
Table 3.1 Details of azo textile dyes used in the experimental constructed wetlands78
Table 3.2 Details of artificial wastewater compositions used in the second phase of
experimental work
Table 3.3 Packing order of the experimental constructed wetland set-up treating two azo
textile dyes only. Samples were taken at the end of each contact time period, just before
the start of the resting time period between 1 May 2015 and 31 May 2016
Table 3.4 Application of the simplified statistical wetland filter set-up design (Table
3.3) to assess the impact of individual key variables
Table 3.5 Packing order of the experimental constructed wetland set-up treating
artificial wastewater containing two azo textile dyes. Samples were taken at the end of
each contact time period, just before the start of the resting time period between 1 June
2016 and 31 May 2017
Table 3.6 Application of the simplified statistical wetland filter set-up design (Table
3.5) to assess the impact of individual key variables
Table 4.1 The descriptive statistics of the changes in temperature
Table 4.2 Dimensions of Phragmites australis (Cav.) Trin. ex Steud. (Common Reed)
planted in the experimental wetlands101

Table 4.3 Influent and effluent water quality characteristics for general physical and
chemical variables119
Table 4.4 Influent and effluent water quality characteristics for nutrients.    125
Table 4.5 Dye and chemical oxygen demand (COD) reductions.       130
Table 4.6 Dye and chemical oxygen demand (COD) loading rate.    131
Table 4.7 Seasonal dyes reduction (mg/l)
Table 5.1 Dimensions of <i>Phragmites australis</i> (Cav.) Trin. ex Steud. (Common Reed)
planted in the experimental wetlands
Table 5.2 Inflow and outflow water quality characteristics for general physical and
chemical variables related to different wetlands156
Table 5.3 Colour, dye and chemical oxygen demand (COD) reductions for different
wetlands163
Table 5.4 Dye loading rate
Table 5.5 Chemical oxygen demand (COD) loading rate.    164
Table 5.6 Inflow and outflow water quality characteristics for nutrients related to
different wetlands
Table 5.7 Seasonal reduction (mg/l) of dyes from artificial wastewater containing two
azo textile dyes

#### ACKNOWLEDGEMENTS

Alhamdulillah, I am grateful to Allah for giving me the opportunity to complete my thesis.

Many people have greatly assisted in this accomplishment and my being in the position I am in today. I would like to thank my supervisor Professor Miklas Scholz for the time and effort he has given me. He has been a constant source of support and encouragement that I sincerely appreciate. Thank you to the staff at the Cockcroft Building, especially Manisha and David who have been cooperative and helpful along the way. I would like to thank all of my colleagues in room 244-Newton building for their advice and support, especially Rawaa, Ruqayah, Suhail and Suhad.

I would also like to show my appreciation to the entire Environmental Engineering Faculty and fellow students. They have all been supportive and good sources of knowledge.

Finally, I dedicate this work to the memory of my father (Muhammad Hussein), and colleagues, Dr. Mahmood Bashir who was killed by ISIS and Dr. Muhammad Mallah.

Also, many thanks to my mother, brothers (Asaad and Yaser), and sisters, my wife, Isra Hussein, and my daughters, Farah, Tiba, Fatma, Zainab, Ruqayha and Isra (baby), for their encouragement, patience and support during this PhD, for which I am very appreciative. I thank them for everything and may Allah give them all the best in return.

Finally, special thanks go to the Iraqi Ministry of Higher Education (Iraq), for funding this research and deep thanks goes to the Civil Engineering Department, College of Engineering, Almuthanna University in Iraq for supporting me during the period of my study.

### DECLARATION: LIST OF ORIGINAL PAPERS AND CONFERENCES

- Amjad Hussein, Miklas Scholz (2016). Dye wastewater treatment by verticalflow constructed wetlands. *4th International Environment Conference 2016* (2-3 March, 2016). Ajman – United Arab Emirates. Web: <u>www.aiec2016.org/</u>.
- 2- Amjad Hussein, Miklas Scholz (2016). Dye removal in experimental verticalflow constructed wetlands treating textile wastewater. *Salford Postgraduate Annual Research Conference (SPARC)* 14-16 June 2016. University of Salford, Media City UK, Salford. Web: <u>www.pg.salford.ac.uk/sparc\_conference</u>.
- 3- Amjad Hussein, Miklas Scholz (2016). Experimental vertical-flow constructed wetlands treating textile wastewater. School of Computing, Science and Engineering, Postgraduate Symposium 16 (CSE\_PGSym16). University of Salford, Great Manchester, UK.
- 4- Amjad Hussein, Miklas Scholz (2017). Dye wastewater treatment by vertical-flow constructed wetlands. Full research paper. *Ecological Engineering* 101 (2017) 28-38.
  Author contributions: Amjad Hussein undertook the meteorological data collection, investigation, and analysis, results visualization and discussion, and

prepared the draft paper, which was revised by Miklas Scholz.

5- Amjad Hussein, Miklas Scholz (2017). Effect of hydraulic contact time on dye wastewater treating by vertical flow constructed wetlands. *School of Computing, Science and Engineering, Postgraduate Symposium 17* (CSE\_PGSym17). University of Salford, Great Manchester, UK.

- 6- Amjad Hussein, Miklas Scholz (2017). Seasonal assessments of vertical-flow constructed wetlands treating azo textile dyes. *Salford Postgraduate Annual Research Conference (SPARC)* 28-29 June 2017. University of Salford, Media City UK, Salford. Web: <u>www.pg.salford.ac.uk/sparc\_conference</u>.
- 7- Amjad Hussein, Miklas Scholz (2017). Artificial wastewater containing two azo textile dyes treatment by vertical-flow constructed wetlands (peer-reviewed journals, Environmental Science and Pollution Research).
   Author contributions: Amjad Hussein undertook the meteorological data collection, investigation, and analysis, results visualization and discussion, and

prepared the draft paper, which was revised by Miklas Scholz.

# ACRONYMS AND ABBREVIATIONS

AAFA	American Apparel and Footwear Association					
AB113	Acid Blue 113					
ABSA	3-AminoBenzeneSulfonic Acid					
ADMI	American Dye Manufactures Institute					
Al	Aluminium					
AMD	Acid Mine Drainage					
ANOVA	Analysis of Variance					
ANSA	5-Amino-8-(phenylamino) Naphthalene-1-Sulfonic Acid					
AO7	Acid Orange 7					
As	Arsenic					
AY	Acid Yellow					
BB	Basic Blue					
BOD	Biochemical Oxygen Demand					
BR46	Basic Red 46					
Cd	Cadmium					
cm	Centimetre					
COD	Chemical Oxygen Demand					
Cr	Chromium					
Cu	Copper					
CWs	Constructed Wetlands					
DAN	1,4-DiAminoNaphthalene					
DO	Dissolved Oxygen					
DR81	Direct Red 81					
DY	Disperse Yellow					
EC	Electrical Conductivity					
Fe	Iron					
FTIR	Fourier Transform Infrared Spectroscopy					
FWS	Free Water Surface					
FWSF	Free Water Surface Flow					
HF	Horizontal Flow					
Hg	Mercury					

XIX

HLR	Hydraulic Loading Rate				
HPLC	High-Performance Liquid Chromatography				
HRT	Hydraulic Retention Time				
HSSF	Horizontal Sub-Surface Flow				
IBM SPSS	International Business Machine Statistical Package for Social Science				
IVF	Integrated Vertical Flow				
IWA	International Water Association				
Ks	K factor at saturated conditions				
λmax	Maximum Absorbance Wavelength				
MLR	Mass Loading Rate				
mm	Millimetre				
Mn	Manganese				
Mol.	Mole				
Ν	Nitrogen				
N/A	Not Applicable				
N=N	Nitrogen Atom				
$N_2$	Gaseous Nitrogen				
NBNMA	N-Benzyl-N-Methyl Aniline				
NH <sub>4</sub> -N	Ammonia-Nitrogen				
Ni	Nickel				
NBNMD	N-Benzyl-N-Methylbenzene-Diamine				
NO <sub>2</sub> -N	Nitrate-Nitrogen				
NO <sub>3</sub> -N	Nitrate-Nitrogen				
NTU	Nephelometric Turbidity unit				
Pb	Lead				
PE	Person Equivalent				
pН	Potential of Hydrogen				
PO <sub>4</sub> -P	Ortho-phosphate-phosphorus				
POD	Plant Peroxidase				
Pt Co.	Platinum-Cobalt				
RB*	Reactive Blue				
RB**	Reactive Black				
Redox	Reduction-Oxidation Reaction				
	XX				

RZM	Root Zone Method
SD	Standard Deviation
SE	Standard Error
SF	Surface Flow
SS	Suspended Solids
SSF	Subsurface Flow
STWI	Sweden Textile Water Initiative
TDS	Total Dissolved Solids
TN	Total Nitrogen
TOC	Total Organic Carbon
TSS	Total Suspended Solids
UK	United Kingdom
USA	United States of America
USEPA	United State Environmental Protection Agency
VF	Vertical Flow
VFCW	Vertical-Flow Constructed Wetland
VSSF	Vertical Subsurface Flow
VY	Vat Yellow
WHO	World Health Organization
WRc	Water Research council
Zn	Zinc

#### ABSTRACT

Wetlands have long played a significant role as natural purification systems. Textile industry processes are among the most environmentally unsustainable industrial processes, because they produce coloured effluents in large quantities polluting water. The aim of this study is to assess the performance of VFCWs to treat two different azo textile dyes with and without artificial wastewater for long periods of time through different operation modes such as contact and resting times, and loading rates, which has rarely been considered in previous research works. The corresponding key objectives are (a) to assess the role of gravel (as a control wetland) and plants on dye reduction and other pollutants; (b) to determine the influence of two groups of dyes (acid (AB113) and basic (BR46)), each dye having a different molecular weight and chemical structure, at two different concentrations (7 mg/l and 215 mg/l); (c) to evaluate the impact of the mixture of both dyes on the performance of vertical-flow constructed wetlands in terms of dye reduction with/without artificial wastewater; (d) to determine the annual and seasonal reduction; and (e) to assess the influence of operational parameters such as contact time (48h and 84h), resting time and mass loading rate on dye reduction and other pollutants for a long period.

The first phase dealt with treating the two azo textile dyes only during the period between 1 June 2015 and 31 May 2016, while the second phase dealt with artificial wastewater containing the two azo textile dyes during the period between 1 June 2016 and 31 May 2017. According to the first phase, for the low concentration of BR46, there was no significant ( $p \ge 0.05$ ) difference between the wetlands in terms of dye reductions. However, for chemical oxygen demand (COD), the reduction percentages were 50%, 59% and 67% for the control and for the wetlands with short and long contact times, respectively. All reductions were statistically significant (p < 0.05). For the high XXII

concentration of BR46, the reduction percentages for the dyes were 94% and 82%, and for COD, they were 89% and 74% for the long and short contact times, correspondingly. A good reduction was noted for total suspended solids for long and short contact times. For the low concentration of AB113, the percentage reductions for the dye were 71%, 68% and 80%, and for COD, they were 5%, 7% and 16% for the control, and the short and long contact times, respectively. For the high concentration of AB113, the percentage reductions for the dye were 72% and 73%, and for COD, they were 54% and 55% for the 48 h and 96 h contact times in this order. Regarding ortho-phosphate-phosphorous for the low concentrations of BR46 and AB113, the reduction percentages for wetlands, which have high contact times, were significantly (p<0.05) better than those of the control wetlands, as well as wetlands, which have low contact times. In the case of high concentration regarding BR46, the reduction percentages of wetlands with low loading rates were significantly (p<0.05) better than wetlands with high loading rates, while for AB113, the reduction percentages of wetlands with high loading rate were significantly (p<0.05) better than those for wetlands with low loading rates. In the case of ammonianitrogen for the high concentration of dyes, there were no significant ( $p \ge 0.05$ ) differences between wetlands. Regarding nitrate-nitrogen reduction for low and high concentration of BR46 and AB113, the reduction percentages for wetlands with long contact times were better than those for wetlands having short contact times.

In the case of phase two, the presence of plants had no effect on the dye and COD reductions. For the low concentration of BR46, the percentage reductions for the dye were 92%, 89% and 91%, and for COD, they were 69%, 82% and 70% for the control, and the short and long contact times, respectively. All reductions were statistically significant (p<0.05). For the high concentration of BR46, the reduction percentages for the dyes were 73% and 33%, and for COD, they were 56% and 39% for the long and

short contact times, respectively. For the low concentration of AB113, the percentage reductions for the dye were 85%, 77% and 82%, and for COD, they were 76%, 81% and 62% for the control, and the short and long contact times in this order. For the high concentration of AB113, the percentage reductions for the dye were 44% and 54%, and for COD, they were 40% and 56% for the 48 h and 96 h contact times, correspondingly. Regarding ortho-phosphate-phosphorous for the low concentrations in the case of AB113 and BR46 and the mixture of both dyes, the reduction percentage in wetlands with high contact time was significantly (p<0.05) better than those of the control wetlands and wetlands with low contact time. For the high concentration of BR46, AB113 and the mixture of both of them, wetlands with high resting and contact times had lower orthophosphate-phosphorous effluent concentrations when compared with wetlands with low resting and contact times. Regarding ammonia-nitrogen reduction percentages for low concentrations of BR46 and AB113 and the mixture of both dyes, wetlands with high resting times had better reduction percentages (p<0.05) when compared with the control wetlands as well as wetlands with low resting times. In the case of high concentrations for BR46, AB113 and the mixture of both of them, wetlands with low loading rates had a better reduction percentage when compared with wetlands with a high loading rate. Regarding nitrate-nitrogen reductions for low and high concentrations of BR46, AB113 and the mixture of both of them, the reduction percentages for all wetlands were in the range from 75 to 100%. Regarding aromatic amine compound reductions, wetlands with long contact times showed significant (p<0.05) differences when compared with the control and wetlands with short contact times for the low concentrations of BR46 and AB113. For the high concentration of BR46 and AB113, wetlands with low loading rates showed a significant difference (p<0.05) when compared with wetlands with a high loading rate. The researcher recommended that using HPLC combined with FTIR to investigate the reduction in aromatic amines and working on modelling of the results should help the designer in improving the construction of wetlands on an industrial scale.

# 1. CHAPTER ONE: INTRODUCTION

### **1.1 Overview**

The purpose of this chapter is to present the concept of wetlands, including definition, foundation, attributes, significance, types, procedures and standard of their application. Moreover, the chapter is further partitioned into sections. Section 1.2 presents the background of the research. The problem statement, aim of the study and its objectives are presented in sections 1.3 and 1.4, respectively. Finally, the thesis structure is presented in section 1.5.

#### **1.2 Background of the Research**

The environment is composed of water, earth, space and atmosphere. Without pollution, it remains perfect and charming. The complex nature of the environment is changing due to different human activities, including transportation, construction, industrialization and so on. Such activities, although desirable for human development and welfare, lead to environmental degradation and the release of objectionable materials into the environment, leading to loss of quality of life.

Over half of the world's population faces water shortage. Almost 900 million people in the world still do not have access to safe water and somewhere in the range of 2.6 billion, a large portion of the number of inhabitants in the developing world, do not have access to satisfactory sanitation. No less than 1.8 million children under five years old die every year, consistently, due to water-related diseases. Furthermore, 1.2 billion people (one

fifth of the total populace) live in areas of water shortage (Corcoran, 2010; Sultana, 2014; Connor, 2015). Figure 1.1 shows the worldwide renewable water resources distribution.



Figure 1.1 Worldwide renewable water resources (m<sup>3</sup> per capita per year, adapted from Sultana, 2014)

Extreme use and pollution of freshwater resources are nowadays worldwide ecological issues that apply pressure on the supportability of worldwide ecosystems. Most of the activities which renewable freshwater is used for – industrial, agricultural and domestic purposes – lead to pollution of freshwater systems worldwide by thousands of natural and industrial chemical compounds. These chemical compounds accumulate in ground waters, surface waters, plants and substrate, degrading the water quality and causing a negative impact on the receiving streams (Kanu and Achi, 2011; Robles-Molina et al., 2014). Furthermore, Schwarzenbach et al. (2006) stated that around two million tons of waste, derived from all human activities, are released into receiving streams everyday

with incomplete treatment or no treatment at all, and this has been demonstrated by Corcoran (2010) as well.

Any increase in population will be accompanied by an increase in industrial, agricultural and urbanization land. These increases lead to increases in the effluent wastewater, which contains a wide diversity of pollutants including azo dyes, chemical oxygen demand (COD), biochemical oxygen demand (BOD), total suspended solids (TSS) nitrogen compounds, turbidity toxic metals (such as Cr, Ni, Cd and Pb), and faecal coliform. These pollutants have harmful effects on the water environment and make the water unsuitable for irrigation, drinking and aquatic life. All these negative impacts of pollutants have forced governments at different levels to assess guidelines and regulations with regard to treatment of effluent wastewater before disposal to receiving streams (Helmer et al., 1997; Spiegel and Maystre, 1998).

Textile dyeing processes are the most environmentally unfriendly industrial processes, because the chemical reagents used are very rich in chemical compounds ranging in both inorganic and organic products (Juang et al., 1996; Robinson et al., 2001). Furthermore, the presence of colour in the effluent textile wastewater is one of the most important problems. Sultana (2014) stated that coloured wastewaters, produced from dyeing processes, are heavily polluted with chemicals, textile auxiliaries and dyes. The properties of textile wastewater depend on the production, technology and chemicals used (Wang et al., 2011). Textile industries devour gigantic amounts of water and generate vast volumes of wastewater through different steps in dyeing and finishing processes, and the discharged wastewater (Figures 1.2 and 1.3) is an overwhelming blend of various polluting substances such as organic, inorganic, elemental and polymeric products (Babu et al., 2007; Kant, 2012). Dye wastes are the most dominating materials in textile wastewater and these materials are toxic to the biological world as well as the dark colour

of these materials blocking sunlight which causes acute problems in biological communities (Ratna and Padhi, 2012; Dey and Islam, 2015). Some examples of the effects of azo dyes on human health and the environment are shown in Table 1.1.



Figure 1.2 The effluent of textile industry- China (Wood, 2010)



#### Figure 1.3 The effluent of textile industry- India (Ramesh, 2011)

Authors (Year)	Human	Environmental
(Hatch and Maibach, 1995)	Dermatitis	
(Specht and Platzek, 1995)	Eczema and dermatitis	
(Greene and Baughman, 1996)		Algae growth inhibited
(Rizzo et al., 2010)	Alzheimer	
(Gudelj et al., 2011)	Allergenic, carcinogenic, mutagenic and teratogenic properties	Bioaccumulation
(Gopinathan and Nair, 2011) (Kant, 2012)		River pollution Depletion of dissolved oxygen
(Pereira and Alves, 2012)	Bladder cancer, splenic sarcomas and hepatocarcinomas	Hindering photosynthesis
(Akarslan and Demiralay, 2015)	Cancer and allergenic	

Table 1.1 Some examples of azo dye (listed in order of date) effects on human health and the environment

The receiving of azo dye effluent wastewater by the river without any degradation or reduction, will lead later to azo dye degradation by the intestinal tract and release of aromatic amine compounds, which are absorbed by the intestine with carcinogenic effects, as demonstrated by Brown and De Vito (1993). For instance, in April 2013 more than 1100 people were killed in Bangladesh due to untreated textile wastewater that came from garment factories output. Bangladesh's garment and textile industries have contributed heavily to what experts describe as a water pollution disaster. Many rice paddies are inundated with toxic wastewater and fish stocks are dying in the nearby water bodies (Yardley, 2013; Belal et al., 2015). In China, a 2003 survey found that since an industrial zone opened in 1992, almost 60 of the 1,500 residents in the two neighbouring villages have died of cancer – about 3% of the population – accounting for 80% of all deaths (Wei, 2003; Zheng et al., 2017). Polluted water causes 75% of diseases and over

100,000 deaths annually, according to the World Health organization (WHO). Meanwhile, cancer rates among villagers who live along polluted waterways are much higher than the national average (China Water Risk, 2011)

Textile manufacture can be classified into three categories viz., cotton, woollen, and synthetic fibres depending upon the raw materials used (Wang et al., 2011). Pure water in large quantities will be consumed and a large volume of wastewater will be produced by the textile dyeing industry from different steps in the dyeing and finishing processes (Allegre et al., 2004; Parshetti et al., 2006; Ojstršek et al., 2007), with an average 200 l/kg of fibre (Pereira and Alves, 2012). An extensive variety of chemicals and dyestuffs, are required for the operation of dyeing and finishing. These materials are organic substances of complex structure, are found in the effluent wastewater, and cause disposal difficulty (Joshi et al., 2004; Ji et al., 2008).

The main pollutants in textile wastewater are shown in Table 1.2 and Figure 1.4. Furthermore, more than 10,000 different textile dyes with an expected annual production of  $7 \times 10^5$  metric tonnes are commercially available around the world; 30% of these dyes are used in excess of 1000 tonnes per annum (Baban et al., 2010). Azo dyes represent 65-75% of the total textile dyes. In the dyeing process, 10-25% of textile dyes are lost and 2-20% are specifically discharged as fluid effluents in different environmental components (Carmen and Daniela , 2012).

Deremeters			Са	ategories			
Farameters	1	2	3	4	5	6	7
BOD <sub>5</sub> /COD	0.2	0.29	0.35	0.54	0.35	0.3	0.31
BOD <sub>5</sub> (mg/l)	6000	300	350	650	350	300	250
TSS (mg/l)	8000	130	200	300	300	120	75
COD (mg/l)	30000	1040	1000	1200	1000	1000	800
Oil and grease (mg/l)	5500	N/A	N/A	14	53	N/A	N/A
Total chrome (mg/l)	0.05	4	0.014	0.04	0.05	0.42	0.27
Phenol (mg/l)	1.5	0.5	N/A	0.04	0.24	0.13	0.12
Sulphide (mg/l)	0.2	0.1	8.0	3.0	0.2	0.14	0.09
Colour (ADMI)	2000	1000	N/A	325	400	600	600
pH	8.0	7.0	10	10	8.0	8.0	11
Temp. (°C)	28	62	21	37	39	20	38
Water usage (1/kg)	36	33	13	113	150	69	150

Table 1.2 Textile processing categories (adopted from Correia et al., 1994) related with Figure 1.4

Note: BOD<sub>5</sub>, biochemical oxygen demand after five days; COD, chemical oxygen demand; TSS, total suspended solids; ADMI, American dye manufacture institute; °C, degrees Celsius; N/A, not applicable; categories descriptions: 1, Raw wool scouring; 2, Yarn and fabric manufacturing; 3, Wool finishing; 4, Woven fabric finishing; 5, Knitted fabric finishing; 6, Carpet manufacturing; 7, Stock and yarn dyeing and finishing.



Figure 1.4 Flowchart for the general steps in textile industry Adopted from Dasgupta et al. (2015).

Wastewater from printing and dyeing units is rich in colour, containing residues of dyes and chemical materials, such as complex components, high Chroma, and high COD and BOD concentration, as well as materials which are difficult to degrade, leading to their concentrations exceeding standard limits and causing ecological and/or health problems (Scholz, 2006, 2011). The toxic effects on the public of dyestuffs and other organic compounds, as well as acidic and alkaline contaminants, from industrial establishments are widely accepted. At present, the dyes are mainly aromatic and heterocyclic compounds, with colour-display groups and polar groups. Their structure is more complicated and stable, resulting in greater difficulty to degrade printing and dyeing wastewater (Ding et al., 2010).

Textile printing and dyeing processes include pre-treatment, dyeing/printing, finishing and other technologies. Pre-treatment includes de-sizing, scouring, washing, and other processes. Dyeing mainly aims at dissolving the dye in water, which will be transferred to the fabric to produce coloured fabric under certain conditions. Printing is a branch of dyeing which generally is defined as 'localized dyeing', i.e. dyeing that is confined to a certain portion of the fabric that constitutes the design. It is really a form of dyeing in which the essential reactions involved are the same as those in dyeing. In dyeing, colour is applied in the form of solutions, whereas colour is applied in the form of a thick paste of the dye in printing. Both natural and synthetic textiles are subjected to a variety of finishing processes. This is done to improve specific properties in the finished fabric and involves the use of a large number of finishing agents for softening, crosslinking, and waterproofing. All of the finishing processes contribute to water pollution. In addition, in different circumstances, the singeing, mercerizing, base reduction, and other processes may have been carried out before dyeing/printing (Ding et al., 2010). Azo dyes play an important role as colouring agents in the textile, food, and pharmaceutical industry (Parshetti et al., 2006; Zhang et al., 2010; Senthilvelan et al., 2014; Tee et al., 2015). Due to the toxicity, mutagenicity and carcinogenicity of azo dyes and their breakdown products, their reduction in industrial wastewaters has been an urgent challenge (Robinson et al., 2001; Erkurt, 2010; Zhang et al., 2010).

The era of intensifying environmental crises such as pollution, climatic changes, water shortages (Alavian et al., 2009; Ward et al., 2010), rapid population growth and several compelling factors such as pipe vandalization by saboteurs, and underground tank leakages, justify the need for sustainable wastewater treatment technology that is environmentally friendly, easy to operate, less energy-intensive, and cost-effective. Many technology solutions used by the textile industry, such as coagulation-flocculation (Golob et al., 2005) and the advanced oxidation process (Hassaan et al., 2016), to treat the effluent are not feasible in practice, because of too high costs and the complex processes involved (Sivakumar et al., 2013), making their use in field conditions difficult (Ji et al., 2007). Furthermore, some developing countries, such as China (Chen et al., 2007), Bangladesh (Islam et al., 2011) and India (Roy, 1998), which have a strong textile industry, but unreliable energy sources, may benefit from less-energy demanding methods of wastewater treatment. Therefore, it is very important to find an alternative system, which is simple, natural and cost-effective to treat such wastewater, such as constructed wetlands (Knight et al., 2000; Scholz and Lee, 2005; Vymazal, 2010a). For reasons such as this, it is important that a method be used to clean textile effluents to avoid further pollution and the risk of poor health to the local area. Many solutions which have been developed to treat these dyes are not feasible due to the difficulty of

degradation (Parshetti et al., 2006), and high cost and technological complexity of the

processes involved. For countries which have an unreliable energy source, it is necessary for other methods of treatment to be used.

Natural systems such as constructed wetlands (CWs) are engineered systems and commonly used because of their properties such as convenience, mechanical simplicity, low energy requirement, environmental friendliness and low cost of operation. Constructed wetlands have been used to attain wastewater treatment goals by using natural components and processes that significantly reduce the use of energy intensive mechanical devices and technical complexity. Furthermore, constructed wetlands involve natural processes resulting in the efficient conversion of hazardous compounds. Many publications have reported the efficiency of constructed wetlands in treatment of various types of wastewaters, including animal wastewater, urban runoff, and mine drainage (Hunt and Poach, 2001; Mays and Edwards, 2001; Rozema et al., 2016), textile industry wastewater (Davies and Cottingham, 1994; Davies et al., 2009), petroleum wastewater (Al-Isawi et al., 2015a; Al-Isawi et al., 2015b) and domestic wastewater (Sani et al., 2013; Paing et al., 2015).

The values of wetlands are multiple, and they have played a necessary role in the history of humankind from the earliest civilizations, such as those in Mesopotamia and Egypt, who used to be near to the wetlands areas that provided a variety of economic and essential resources to them. Many studies have assessed the economic values of constructed wetlands in different regions of the world (UK, Klein and Bateman, 1998; India, Verma et al., 2001; Greece, Ragkos et al., 2006; China, Yang et al., 2008). Wetlands as a water body include various animals and plant types, provide artificial impoundments for water storage, which can be used for other purposes such as irrigation, and constructed wetlands can be used to provide flood protection by using them as areas
for flood storage (Ghermandi et al., 2010). Furthermore, other values include the absorption of carbon dioxide leading to reduce global warming, and supporting the food chain, indirectly, by fish production and other related edible water animals.

Constructed wetlands (CWs) are considered an alternative efficient method to treat many types of wastewater, including that from the textile industry, due to their green qualities, low investment requirement, ease of operation, low maintenance cost, and low or zero energy input, besides landscape aesthetics (Scholz and Xu, 2002a; Guimaraes, 2016; Sehar et al., 2016). CWs are engineered ecosystems for reduction of pollutants from water by wastewater treatment that resembles the hydraulic conditions and habitat occurring in a swamp. Constructed or artificial wetland systems mimic the treatment that occurs in natural wetlands by relying on heterotrophic micro-organisms and aquatic plants and a combination of naturally occurring biological, chemical and physical processes. Textiles and dyeing wastewater causes significant environmental pollution, the main problems being high concentrations of organic matter and colorants (dyes) that have to be resistant to the effects of sweat, soap, water, light and oxidants (Olejnik and Wojciechowski, 2012).

The performance abilities of constructed wetlands are dependent on the natural environment, microbial communities, plants and the weather of the region (Scholz, 2011; Hernández, 2013; Mahmood et al., 2013). The two major types of CWs are Surface flow (SF) and subsurface flow (SSF) (Kadlec and Knight, 1996; Jørgensen, 2009; Vymazal, 2013; Albalawneh et al., 2016). The main difference between SSF and SF CWs is the water level; SF CWs have an uncovered water surface, while SSF CWs have no clear water surface. Furthermore, SSF CWs can be divided into vertical flow (VF) and horizontal flow (HF) regarding the direction of water movement into the constructed

wetlands system (Vymazal, 2005, 2008b). In general, media in HFCWs is submerged in water, unlike the media in VFCWs, which is ponded and drained due to the type of operation (Stefanakis et al., 2014; Hou et al., 2017).

VFCWs are the state of the art in constructed wetlands technology around the world (Olsson, 2011; Stefanakis et al., 2014). Many publications (Vymazal, 2002, 2005; Saeed and Sun, 2012; Shi et al., 2012; Paing et al., 2015; Butterworth et al., 2016) have reported the ability of this type of constructed wetland to transfer oxygen at a higher rate. Furthermore, the wastewater enters the constructed wetland from the top and drains vertically by gravity through the porous media, and air from the atmosphere follows in its place (Maier et al., 2009; Stefanakis et al., 2014), enhancing the aeration and microbial activity. Many authors (Korkusuz et al., 2004; Scholz, 2011; Sousa et al., 2011) have reported that VFCWs have performed well in reduction of COD, BOD and phosphorus. In addition, regrading nitrification processes, vertical-flow constructed wetlands can achieve a satisfactory level (Platzer, 1999; Song et al., 2015).

## **1.3 Problem Statement**

The release of dye textile wastewater into receiving streams is unacceptable not only for aesthetic reasons and the effect on aquatic life, but also since many of these dyes are toxic and carcinogenic (Lourenço et al., 2003; dos Santos et al., 2007; Dey and Islam, 2015). Current strategies used for treatment of textile wastewaters have technical and economic restrictions. The majority of the physico-chemical methods, which are used to treat this type of wastewater, are costly, produce large amounts of sludge and are wasteful of some soluble dyes (Van der Zee et al., 2001; Sivakumar et al., 2013; Costa et al., 2012). In contrast, biological treatment such as constructed wetlands is much cheaper than the

previous methods, environmentally friendly and does not produce large amounts of sludge (Knight et al., 2000; Scholz and Lee, 2005; Vymazal, 2010a; Tee et al., 2015). Furthermore, the highlights in the above introduction suggest that the sustainable and prospective treatment of textile wastewater is with subsurface flow constructed wetlands to reduce azo dye and other pollutants of different types and concentrations (Davies et al., 2005; Ong et al., 2011; Shehzadi et al., 2014).

Two azo textile dyes (AB113 and BR46) were selected in this research with two different concentrations (low with a target concentration 7mg/l and high with a target concentration 215 mg/l). Typically, textile industry-processing effluents contain dyes in the range between 10 and 200 mg/l (Lavanya et al., 2014; Yaseen and Scholz, 2016). Most textile dyes at a rather low concentration of even < 1 mg/l can be detected by the human eye (Chung, 1983; Lavanya et al., 2014; Pandey et al., 2007). Furthermore, Van der Zee (2002) stated that algal growth was not inhibited at dye concentrations < 1 mg/l. Both of which are commercial dyes extensively used in the textile industry (Chung et al., 1992; Riu et al., 1997; Pervez et al., 1999; Olgun and Atar, 2009; Ong et al., 2010; Deniz and Karaman, 2011; Deniz and Saygideger, 2011). AB113 is an acid dye and BR46 is a basic dye. An acid dye is defined as a negatively charged dye at a chemical level, which contains one or more acidic groups such as a sulphonic group (Akbari et al., 2002; Martínez-Huitle and Brillas, 2009). A basic dye is defined as a positively charged stain at a chemical level (Martínez-Huitle and Brillas, 2009; Brillas and Martínez-Huitle, 2015) which means it reacts well with negatively charged materials (Sun and Yang, 2003).

## 1.4 The novelty of the work

Most of the previous studies on treating azo dye textile wastewater by subsurface flow constructed wetlands focus on studying the role of plant activity on dye degradation, degradation of dyes under aerobic and anaerobic conditions, and using different plants and media or enhancing the media by adding extra material (Pervez et al., 1999; Davies et al., 2005; Mbuligwe, 2005; Davies et al., 2006; Bulc and Ojtrsek, 2008; Ong et al., 2011; Yalcuk and Dogdu, 2014).

Despite the numerous articles published on constructed wetlands during recent decades, there is a notable gap in the literature regarding research on the long-term CW treatment performance under outdoor weather conditions affected by season, and its relationship with other pollutants of vertical-flow constructed wetlands treating azo textile dye wastewaters. It is rarely to find a researcher worked on treatment azo textile dyes with CWs for long term period. Furthermore, working on different contact and resting times, and different hydraulic loading on two different groups of azo dyes with low and high concentration, all together.

## 1.5 Aim of the study and objectives

Constructed wetlands are already widely used to treat wastewaters polluted with various compounds (Scholz, 2006, 2010, 2015). The aim of the project is to assess the efficiency of vertical-flow constructed wetlands in reducing dyes (mixed of them with/without artificial wastewater), aromatic amines compounds, COD, PO<sub>4</sub>-P and other nutrients for long term.

In order to achieve the aim, the research involved in depth investigation covering the following objectives:

a. Undertake critical review of relevant literatures on role of temperature, hydrology and other components of the wetland in the treating efficiency of dye reduction, COD, nutrients (PO<sub>4</sub>-P, NH<sub>4</sub>-N and NO<sub>3</sub>-N) and other water quality parameters such as TSS, pH, etc. in the textile wastewater.

- b. To assess the role of gravel and plants on dye reduction and other pollutants.
- c. To assess the influence of two groups of dyes (acid and basic), each dye having a different molecular weight and chemical structure, at two different concentrations (low and high), and the mixture of both dyes, on the performance of vertical-flow constructed wetlands in dyes reduction, with/without artificial wastewater to evaluate the annual and seasonal reduction.
- d. To assess the influence of operational parameters such as contact time (also known as hydraulic retention time), resting time and mass loading rate on dye reduction and other pollutants for long term. Contact time is defined as the duration the wastewater is in contact with the wetland filter content. In comparison, resting time is the duration when the wetland filter is empty (i.e. no wastewater input). All these operational parameters are linked with the pervious objectives to make the novelty of my work.

## **1.6 Thesis Structure**

This investigation began by reviewing the existing information on CWs treatment of textile wastewater. This study then investigated the performance of VFCWs in a greenhouse applied for dye reduction.

Chapter 1 describes the background of the research, problem statement, aim and objectives, and finally lists the thesis chapters.

Chapter 2 presents the impact of the textile wastewater, an overview of the historical development of the constructed treatment wetland system and components of it (water, media, vegetation and micro-organisms). An overview is also given of the constructed wetlands classification, and the reduction mechanisms of the azo dyes and other pollutants. Furthermore, the choice of vertical-flow over horizontal-flow constructed wetlands is stated in this chapter which also discusses amine compounds and their classification. Finally, an overview is given regarding treating textile wastewater with vertical-flow constructed wetlands.

Chapter 3 describes the materials, the experimental set-up, environmental conditions and operation methods applied for the experimental work. It explains the design of the experimental constructed wetlands, their media and the type of plant. In addition, as an influent, the compositions of the artificial wastewater, the two azo textile dyes and the nutrients for the plant are explained. Furthermore, it includes determinations of water quality parameters and calibration of equipment used for the measuring of water quality parameters. Finally, the type of statistical methods, used for the normality test and the other functions such as significant effect, are described in this chapter.

Chapter 4 presents the results and discussions of the performance efficiency of the vertical-flow constructed wetlands to treat two azo textile dyes. In addition, the other water quality parameters are discussed including pH, redox potential, total suspended solids, turbidity, dissolved oxygen, electrical conductivity, chemical oxygen demand and nutrients (ammonia nitrogen, nitrate nitrogen and ortho-phosphate-phosphorus).

Chapter 5 talks about the performance efficiency of the vertical-flow constructed wetlands in treatment of artificial wastewater containing two azo textile dyes with other

water quality parameters such as COD and nutrients. In addition, other parameters, such as colour and amines compounds reduction, are discussed.

Chapter 6 presents the conclusions and recommendations for further work.

# 2. <u>CHAPTER TWO: CRITICAL LITERATURE</u> <u>REVIEW</u>

#### 2.1 Overview

In this chapter, the historical development of wetlands is described in section 2.2 and the components of constructed wetlands are presented in section 2.3. The classification of constructed wetlands is demonstrated in section 2.4. Section 2.5 includes the reduction mechanisms of pollutants in constructed wetlands. The choice of VSSF over HSSF CW is justified in section 2.6. Amine compounds are documented in section 2.7. Finally, the use of constructed wetland systems for the treatment of textile wastewater is presented in section 2.8 and a concluded remarks of the chapter is given in section 2.9. This chapter addresses the impact of textile wastewater on humans and the environment. The historical development of constructed wetland systems is presented, as well as the early principles of the technology. In addition, this chapter covers water, media, vegetation and microorganisms as key components of constructed wetlands. After that, a classification of constructed wetlands is presented, and applications of intermittent loading of verticalflow treatment systems for different types of wastewater are shown. The mechanisms of pollutant reduction are covered with focus on azo dye reduction mechanisms. Finally, the choice of vertical-flow over horizontal-flow constructed wetlands and types of amine compounds is justified.

#### **2.2 Historical Development of Constructed Treatment Wetlands**

Natural wetlands, in general, are a transitive area between land and water. The boundaries between wetlands and uplands or deep water are therefore not always distinct (Stefanakis

et al., 2014). Wetlands are an ecosystem whose formation and characteristics are mainly controlled by water (Barbour and Billings, 2000). The water saturation level is the most important factor, which determines the nature of the soil and the types of animal species that live in wetlands as well as the kind of plants (Novitzki et al., 2002; Stefanakis et al., 2014). Scholz (2006) reported that wetlands have been appreciated as an important natural resource during human history. Their value has been recognized in their natural state by such people as the Marsh Arabs around the junction of the rivers Euphrates and Tigris in the south of Iraq as well as in managed forms, for example rice paddies, in the south of Asia. Furthermore, wetlands serve as a wildlife conservation resource and can be seen as natural recreational areas for the local community. Typha spp., Phragmites spp. and other macrophytes typical of swamps are widely used in North America and Europe (Knight et al., 2001; Scholz, 2011), and can be found at any place except Antarctica. In most wetlands, the hydrology of the wetlands is either generally one of slow flows and shallow waters or saturated substrates. The slow flows and shallow water depths allow sediments to settle as the water passes through the wetland. The slow flows also provide prolonged contact times between the water and the surfaces within the wetland. The complex mass of organic and inorganic materials and the diverse opportunities for gas/water interchanges foster a diverse community of micro-organisms that breakdown or transform a wide variety of substances. This vegetation slows the water, creates microenvironments within the water column, and provides attachment sites for the microbial community. The litter that accumulates as plants die back in the fall creates additional material and exchange sites, and provides a source of carbon, nitrogen, and phosphorous to fuel microbial processes.

The best definition for natural wetlands was provided by the Ramsar Convention on Wetlands, which was held in Iran (Ramsar, 1971). In this Convention, wetlands are defined as "areas of marsh, fen, peatland or water, whether natural or artificial, permanent or temporary, with water that is static or flowing, fresh, brackish or salt, including areas of marine water, the depth of which at low tide does not exceed six meters." This definition was founded on a wide approach to best describe the main characteristics of wetlands (Mustafa, 2010; Stefanakis et al., 2014). Many publications reported the benefits of natural wetlands to human life (Turner, 1991; Breaux et al., 1995; Knight, 1997; Kirsten and Brander, 2004; Ghermandi et al., 2009; Costanza et al., 2016). This identification of the wide range of economic and ecological benefits has indicated increasing interest in the construction wetlands ecosystems that mimic the functions of natural wetlands in order to support human use (Hammer and Bastian, 1989; Stefanakis et al., 2014).

Constructed wetlands are engineered to optimize the treatment conditions found in natural wetlands by using a complex mixture of water, substrate, plants and a variety of micro-organisms (Langergraber, 2008). The fundamental idea of building constructed wetlands is to duplicate the various natural wetland processes in a way that is more advantageous for humans and under controlled environmental conditions such as temperature and humidity (Eke, 2008). It has more advantages, such as low cost, minimization of biological and chemical sludge and high efficiency (Bulc, 2006; Sivakumar et al., 2013).

Kadlec and Knight (1996) documented that natural wetlands were used for wastewater disposal in 1912. On the other hand, Brix (1994c) and Wallace (2004) documented that the first use of constructed wetlands was in Australia in 1904. The first use of constructed

wetlands with plants as a wastewater treatment was begun in Germany in the 1950s by Seidel (1965a) as reported by Eke (2008), Vymazal (2010a, 2014) and (Stefanakis et al., 2014). Seidel carried out much experimental work to study the effect of the presence of plants on the treatment of different types of wastewater such as dairy wastewater treatment with plant S. lacustris, phenol wastewater treatment with plant Scirpus lacustris, and livestock wastewater (Seidel, 1955, 1961, 1965a, 1966, 1976) as documented by Vymazal (2005). She noticed that growth of plants in pollutant water (wastewater) is better than in uncontaminated water and the plants, which she used in the experimental works (bulrush; kind of reed), have an ability to reduce various pollutants from wastewater (Stefanakis et al., 2014). Moreover, Seidel worked very hard, in the early 1960s, using wetland plants in sludge with different types and wastewater, and improve the performance of rural wastewater treatment, and tried to decentralized the treatment of wastewater by using either pond systems or septic tanks (Vymazal, 2005). After that, she tried to improve her early system by using sandy soil as a media with high hydraulic conductivity with different types of plants. Due to the anaerobic conditions of septic tanks, she worked on improving the system by using a stage of primary sludge as an infiltration bed with a vertical flow and the wastewater flowed as an elimination bed with a horizontal flow, and planting with Phragmites australis (Seidel, 1965b). This system was the foundation for hybrid systems that were re-established at the end of the 20<sup>th</sup> century, as reviewed by Vymazal (2005). Kickuth, one of Seidel's students, in collaboration with her, invented a wetland treatment process known as the Root Zone Method (RZM) (Kickuth, 1977, 1978, 1981; Vymazal, 2009), which was first used at Othfresen, Germany in 1974 (Brix, 1987). The design of the RZM consists of a rectangular bed planted with *Phragmites australis*, and the media in this system differed from Seidel's design in the use of cohesive soil with a high content of clay (Vymazal,

2005). Water flows horizontally through the roots of the plants as a subsurface horizontal flow (Brix, 1994c). Kickuth carried out his experimental work and made this concept popular with his team in Europe, resulting in about 200 industrial and municipal wastewater treatment systems, and attention in the growth of these subsurface (RZM) flow constructed wetlands led to the spread of their use in Europe by the mid-1980s (Bastian and Hammer, 1993).

The first use of CWs in the Netherlands was in 1967 and it was FWSF (Veenstra, 1998; Vymazal, 2010b). In addition, in Hungary, a FWSF CW was built in 1968 near Keszthely to treat the effluent wastewater of the town and set in the place of an existing natural wetland (Lakatos, 1998).

In the USA, research studies on the use of FWSF CWs to treat wastewater started in the late 1960s (Eke, 2008; Sani, 2015). Gessner et al. (2005) reported that the first constructed wetland for engineering purposes was constructed in New York, North America between 1973 and 1976. Nevertheless, in 1901, a constructed wetland as a purposely engineered treatment vessel was used (Monjeau, 1901, Wallace, 2004) as documented by Eke (2008). The research studies became more effective throughout the 1970s and 1980s, and between 1980 and 1990 (USEPA, 2000a). For instance, in 1975, a constructed wetland/pond system was built in North Dakota to treat industrial wastewater (Litchfield and Schatz, 1989) and FWSF CWs were established to treat urban storm water in California in the early 1980s (Chan et al., 1982). In North America, most CWs are used as large-scale facilities for municipal wastewater treatment (Stefanakis et al., 2014). Knight et al. (1993) and Brix (1994c) pointed out that FWSF CWs are the most common type of constructed wetlands used for wastewater treatment. Use of constructed wetlands has been increased widely in the USA after the publication, in 1993 by the Tennessee

Valley Authority, of a design manual aimed primarily for separate houses (Wallace, 2004).

Constructed wetlands were introduced to the UK in the mid-1980s after the UK WRc (Water Research council) visited Kickuth, Seidel's student who was an early advocate of the technology (Boon, 1986). Later, the WRc and UK Water Services Associated established the Reed Bed Treatment System Coordinating Group, which combined the results of the initial technology trials carried out by water utilities with the plan of accelerating development (Cooper and Green, 1998; Murphy and Cooper, 2010; Knowles, 2012). Severn Trent, a member of UK Water Services Associated, confirmed that HSSF CWs have an ability to treat domestic wastewater effluent for communities of <2000 people as a tertiary treatment and as a secondary treatment for communities of <50 people (Green and Upton, 1993). Green and Martin (1996) reported that HSSF CWs were used in 1990 to treat storm water overflow. Therefore, using this type of CWs became more common in small communities. Later, as a result of many problems being found with this type of CW (Cooper et al. 1996), many researchers turned to the use of VSSF CWs as an alternative solution.

In China, the first constructed wetland system was built in 1987 (Ding and Shen, 2006; Liu et al., 2009) and by 2006 there were more than 200 CW systems, almost all of which were established to treat domestic, industrial and agricultural wastewater, and 54.3% of which were VSSF or combined integrated vertical flow (IVF). On the other hand, the volume of wastewater effluent has also increased rapidly. For example, industrial wastewater effluent increased between 2000 and 2005 from 19.4 to 24.3 billion tons and domestic wastewater increased from 15.1 to 28.1 billion tons in the same period (NBS, 1995-2005) as reviewed by Liu et al. (2009). Furthermore, China has the largest industry of aquaculture in the world (Liu and Diamond, 2005) and that's another challenge, especially as the application of CWs to treat this type of wastewater is largely absent (Liu et al., 2009). Furthermore, some researchers have been published in other countries. For example, Kenya (Bojcevska and Tonderski, 2007; Kimani et al., 2012), Tanzania (Haule et al., 2000), Thailand (Klomjek and Nitisoravut, 2005), India (Juwarkar et al., 1995; Sharma and Brighu, 2014) and Egypt (Abou-Elela and Hellal, 2012; Abou-Elela et al., 2013; ElZein et al., 2016).

In the case of textile wastewater, a root zone method (RZM) was used, in 1983 in Germany, for the elimination of sulphur compounds from the textile finishing industry. The system consisted of sand as a media, planted with *Phragmites Communis Trin*. (Winter and Kickuth, 1989). In 1992, HSSF CWs, with gravel as a media and planted with *Phragmites australis* were used in Australia to treat a complex wastewaters, which were the effluents of different processes of textile and dyeing facilities (Davies and Cottingham, 1994). In the USA in 1999, VSSF CWs with gravel as a media and planted with *Phragmites australis* were used to reduce two different dyes from a synthetic wastewater simulating textile wastewater (Pervez et al., 1999). A combination of VSSF and HSSF CWs (hybrid system), with sand as a media and planted with *Phragmites*, was used in 2006 in Slovenia to treat real textile wastewater (Bulc et al., 2006).

#### 2.3 Components of Constructed Wetlands

A constructed wetland consists of a properly designed basin that contains water, a media, plants and micro-organisms. These components can be manipulated in constructing a wetland as shown in Figure 2.1.



Figure 2.1 Schematic representation of vertical-flow constructed wetlands (Tilley et al., 2014)

## 2.3.1 The Water

In general, wetland hydrology refers to influent and effluent of water through the wetland and the communication with the other functions of the wetland: media, vegetation and micro-organisms. Actually, it is the one of most important factors in the design of wetlands because all of the other functions are linking with each other through it (Luise et al., 1993; Carter, 1996). It is often the primary factor in the success or failure of a constructed wetland due to it is role in determining the duration of water biota interactions (Kadlec and Wallace, 2008). Any changes in hydrology can have an effect on a wetland and its treatment performance (Kadlec and Knight, 1996; Okurut, 2000). For instance, Lim et al. (2001) found an increase in pollutant concentration as a result of losses in water.

In constructed wetlands, hydrology engages with the anaerobic condition. Under this condition, constructed wetlands can reduce the pollutant (Scholz, 2011). For example,

denitrification, which is a process to reduce nitrogen, takes place under anaerobic conditions (Merino-Solís et al., 2015). The time for which water remains in constructed wetlands, named hydraulic retention time (HRT) or hydraulic contact time, is an important variable in designing and evaluating the treatment performance of a constructed wetlands system (Ghosh and Gopal, 2010; Merino-Solís et al., 2015). Toet et al. (2005) evaluated the effect of four different hydraulic retention times (0.3, 0.8, 2.3 and 9.3 days) on the pollutant reduction by surface flow wetlands. The results showed the reduction of nitrogen and faecal coliform was enhanced by increasing the HRT.

## 2.3.2 The Media

The media used in constructed wetlands is the substrate. The selection of substrate type is very important to the successful performance of wastewater treatment by the wetland system (Crites et al., 2014). Substrates used to construct wetlands include soil, sand, gravel, rock, and organic materials such as compost (Dordio and Carvalho, 2013). The media in subsurface flow constructed wetlands supports the growth of plant macrophytes and provides a direction for wastewater to pass and at the same time provides surfaces for micro-organisms to live, enabling the reduction of wastewater pollutants via consumption by micro-organisms (Stefanakis et al., 2014; Zidan et al., 2015). The characteristics of the filter media, which define its permeability, should be selected carefully (Stefanakis et al., 2014).

Many researchers have investigated enhancing the constructed wetland ability to reduce pollutants by adding special materials, which are supported by the media (Kadlec and Wallace, 2008; Ballantine and Tanner, 2010; Meng et al., 2014; Austin and Yu, 2016). Zurayk et al. (1997), Shilton et al. (2005), Xu et al. (2006) and Park and Polprasert (2008)

enhanced the reduction of phosphorus by adding limestone, seashells, slag and fly ash to wetland media. García-Pérez et al. (2015) demonstrated that using recycled shredded-tire chips could enhance the reduction efficiency of typical wastewater contaminants, while Prochaska and Zouboulis (2006) found the use of dolomite with sand as a filter media had no significant on the phosphorus reduction rate when they used a synthetic wastewater treated by vertical-flow constructed wetlands. Gorra et al. (2007) used zeolite to enhance the reduction of ammonia-nitrogen from wastewater, the results showed that this material is appropriate for ammonia-oxidizing bacteria and that it led to increased reduction of ammonia-nitrogen, while using zeolite and bauxite together has no significant effect on the reduction of wastewater pollutants by vertical-flow constructed wetlands as pointed out by Stefanakis and Tsihrintzis (2012). Furthermore, Scholz and Xu (2002) used a granular activated carbon as the adsorption media to increase filtration performance of constructed wetlands, the results showed no improvement in the media adsorption capacity. Lavrova and Koumanova (2014) and Shen et al. (2015) used a solid carbon source (methanol and starch blends, respectively) to achieve a higher denitrification in vertical-flow constructed wetlands. Drizo et al. (1997) used shale as a media to enhance the reduction of phosphorus and ammonium by horizontal subsurface constructed wetlands. The results showed an extremely high reduction for both of them.

The size of media, distribution of particle size and the particle shape also affect the treatment performance of constructed wetlands by affecting the hydraulic conductivity (Knowles et al., 2011), Table 2.1. Dordio and Carvalho (2013) and Meng et al. (2014) pointed out that biofilm growth is better established in media of smaller particle size compared with that of excessively larger size, giving microbes a higher biodegradation ability. Furthermore, Austin and Yu (2016) stated that biomass for media of particle size

3.5 mm is three times greater than when the particle size is 10 mm, while the possibility of clogging increases with fine-pore media (Wallace and Knight, 2006; Song et al., 2015). Nevertheless, Sani et al. (2013) and Al-Isawi et al. (2015b) found that COD and BOD results of vertical-flow constructed wetlands with large and small aggregate were similar during the whole period of their experimental work.

Until recently, soil was only utilized as a growth medium for the plants in subsurface flow constructed wetlands (Arias et al., 2001). This practice, however, often resulted in clogging problems due to the relatively low hydraulic conductivity of this material. Therefore, most wetland engineers advise using a fine gravel instead of soil, which reduces the possibility of clogging (Ávila et al., 2014). Many researchers have suggested using a coarser material as a bottom layer in VFCW beds to avoid clogging (Weedon, 2003; Brix and Arias, 2005; Masi et al., 2007; Öövel et al., 2007;Molle et al., 2008) as reviewed by Ávila Martín (2013). Kalyani Kenkar and Kadlag (2016) pointed out that where gravel is used as the growth medium for the microbes, it works as a sieve and determines the hydraulic contact time.

Table 2.1 Wetland media characteristics (Strickland and Korleski, 2007)			
Туре	Effective Size	Porosity (n) %	Hydraulic
	(mm)		Conductivity (Ks;
			ft/day)
Fine Gravel	16	38	7500
Medium Gravel	32	40	10000
Coarse Rock	128	45	100000

#### 2.3.3 The Plants

Today, the significance of plant presence in CWs is recognized and the positive effects on the system operation and performance have been – more or less – proved (Stottmeister et al., 2003; Scholz, 2006, 2010; Gargi Sharma et al., 2014; Villa et al., 2014). Although it was a controversial issue in the past, today it is generally accepted that constructed wetland plants provide a series of benefits and contribute to the creation of the necessary conditions which directly or indirectly affect the system efficiency (Stefanakis et al., 2014). Brix (1994b) documented that plants have an ability to release oxygen into the rhizosphere from their root. This process is very important to help subsurface flow constructed wetlands in aerobic degradation and nitrification. Among the most common plant species used in CWs (*Phragmites spp., Scirpus spp., Typha spp., Carex sp.*), reported values for plant-mediated oxygen transfer are within the range of 0.005-12 g  $O_2/m^2/d$  (Armstrong et al., 1990; Nivala et al., 2012; Stefanakis and Tsihrintzis, 2012). Furthermore, many researchers have demonstrated that plants play an important role in nutrients uptake (Brix, 1994b; Tanner, 1996; Bachand and Horne, 1999; Stottmeister et al., 2003;Vymazal, 2007; Hussein and Scholz, 2017).

The density of vegetation of a constructed wetland strongly affects its hydrology; firstly, by obstructing flow paths as the water finds its sinuous way through the network of stems, leaves, roots, and rhizomes, and secondly, by blocking exposure to wind and sun as stated by Luise et al. (1993). Furthermore, the presence of plants, especially in VFCWs, prevents clogging by dissolving organic matter through the roots, and this process leads to the pervasion of the water through the media of an intermittent or tidal loading vertical-flow system (Brix, 2003). Hua et al. (2014) studied the effect of plants on prevention of clogging between unplanted and planted wetlands, by measuring the coefficient of

permeability and effective porosity of the media within different operational modes. The results of their experimental work suggested that the coefficient of permeability and effective porosity increased in the plant root zone. Moreover, Münch et al. (2005) reported a higher microbial density in planted constructed wetlands.

Many publications have investigated the effect of plant presence on the organic matter reduction by constructed wetlands. For example, Kaseva (2004) reported the reduction percentage rate of COD of domestic wastewater was 56-61% in planted wetlands, while 33% in unplanted wetlands. Karathanasis et al. (2003) found the reduction percentage rate for BOD and TSS was >75% and >88%, respectively, in planted wetlands, and 63% and 46%, respectively, in unplanted wetland, when treating domestic wastewater. However, Akratos and Tsihrintzis (2007) found a slight increase in COD reduction percentage rate in wetland planted with *T. latifolia* (89.3%) when compared with unplanted wetland (87.2%).

. Ong et al. (2009) found a slight increase in COD reduction percentage rate with the presence of plants, when they used UFCWs to treat azo dye AO7 (contact time 3 d). The influent COD was 383.8±13.7 mg/l, the reduction percentage rate was 78% and 79% for the unplanted and planted wetlands, respectively. For the dye reduction, there was no difference in dye reduction rate between planted and unplanted wetlands; the influent of dye AO7 was 50.5±2.0 mg/l and the reduction percentage rate was 96% for planted and unplanted wetlands. Furthermore, Yalcuk and Dogdu (2014) treated azo dye Acid yellow 2G by vertical-flow intermittent-feeding constructed wetlands with 3.74 d as a contact time. The influent COD was 75 mg/l, the reduction percentage rate was 59% and 61-65% for unplanted and planted wetlands, respectively. In the same study, the influent colour

was 22900 Pt Co. and the reduction percentage rate was 87% and 98% for the unplanted and planted wetlands, respectively.

#### 2.3.4 Micro-organisms

A micro-organism, also known as a microbe, is defined as a life organism which is difficult to see with the naked eye, such as fungi, bacteria and protozoa, but is visible under a microscope (Madigan et al., 2008). Many researchers have pointed out that micro-organism communities exist in wetlands of both aerobic and anaerobic conditions (Kadlec and Knight, 1996; Ottova et al., 1997; Scholz et al., 2001; Meng et al., 2014). Micro-organisms are one of the key factors on which functions of constructed wetlands are largely dependent to treat wastewater (Wetzel, 1993; Brix, 2003), and constructed wetlands provide an ideal environmental to support micro-organism growth (Saeed and Sun, 2012). Micro-organisms play a major role in reduction of the organic contaminants of wastewater, which are present because of interaction of the processes (physical, chemical and biological) which occur in constructed wetlands in the purification of wastewater and transformation of nitrogen and phosphorus (Kayombo et al., 2005). For instance, photosynthesis is one of the biological processes, which is performed by the wetland plant and algae (one species of micro-organism), with the process adding carbon and oxygen to the wetlands, both of which are very important in the nitrification process (Stottmeister et al., 2003; Kayombo et al., 2005). Richardson and Craft (1993) and Cooper and Findlater (2013) demonstrated that microbial uptake is one of the major mechanisms in the reduction of phosphorus. Furthermore, Kosolapov et al. (2004) and Sheoran and Sheoran (2006) stated that microbial uptake is one of the reduction mechanisms of heavy metals.

In textile wastewater treatment by constructed wetlands, micro-organisms play an important function in decolourization of azo dyes under aerobic, anaerobic and anoxic conditions (Kodam et al., 2005; Pandey et al., 2007). Under aerobic conditions, a type of microorganism known as *P. aeruginosa* works on decolourization of azo dyes due to its ability to depend on azo groups as a solo carbon source (Edwards, 2000; Pandey et al., 2007), while under anaerobic conditions, many types of micro-organisms work on decolourization of azo dyes with external carbon as an energy source (Şen and Demirer, 2003; Haroun and Idris, 2009). Under anoxic conditions, other types of micro-organisms, such as *Pseudomonas luteola*, work on this process (Grekova-Vasileva et al., 2005).

## 2.4 Classification of Constructed Wetlands for Wastewater Treatment

#### 2.4.1 Overall Classification

The purpose of these engineered wetlands is to receive and purify wastewater of various types, based on the naturally occurring treatment processes (Stefanakis et al., 2014; Scholz, 2015). Constructed wetlands for wastewater treatment can be further divided into other categories. Knight (1997), Fonder and Headley (2010) and Vymazal (2010a, 2014) reported three categories of constructed wetlands. The first category depends on the water level on the bed, the second depends on the type of wetland plants and the third depends on the direction of the water movement in the constructed wetlands are classified into free water surface flow (FWSF CWs) and subsurface flow (SSF CWs). Depending on the direction of horizontal flow (HFCWs) and vertical flow (VFCWs). The hybrid system is a combination of horizontal-

flow and vertical-flow constructed wetlands to achieve the maximum benefits from both flows in the reduction of the wastewater pollutants (Vymazal, 2013).



Figure 2.2 Classification of constructed wetlands (Stefanakis et al., 2014)

## 2.4.2 Free Water Surface Flow Constructed Wetlands

FWSF CWs are defined as constructed wetlands where the water surface is exposed to the atmosphere (USEPA, 2000b), the wastewater enters at the top and moves horizontally on the top of the constructed wetland media as shown in Figure 2.3, so the wastewater infiltrates the media or is evaporated to the atmosphere (Bendoricchio et al., 2000). FWSF CWs are similar to natural wetlands in that they are basins planted with emergent, submerging and/or floating wetland plants as shown in Figure 2.4 (Jadhav and Buchberger, 1995; Lim et al., 2001; Ellis et al., 2003b; Stefanakis et al., 2014). The influent passes as free surface flow (and/or at shallow depths) and at low velocities above the supporting substrates, which provide a promoting environment for biological,

chemical and physical reduction processes to take place (Hamilton et al., 1993; Xianfa and Chuncai, 1995; Stefanakis et al., 2014). FWSF CWs are very effective for reduction of nitrogen, biochemical oxygen demand, pathogenic organisms, suspended solids and heavy metals (DeBusk, 1999; Choudhary et al., 2011; Midhun et al., 2016), while they are not effective for phosphorus reduction as wastewater does not tend to come into contact with the media, which precipitates or adsorbs phosphorus (Taylor et al., 2006). Use of FWSF CWs is more common in North America (Tousignant et al., 1999) where they are applied for municipal wastewater treatment only, as stated by Sani (2015). However, although they have advantages of simple technology and low cost of operation (Economopoulou and Tsihrintzis, 2004), FWSF CWs require more land than SSF CWs, carry increased risk of human exposure to surface flowing wastewater, and odour and insects might be a problem because of the free water surface (Halverson, 2004; Stefanakis et al., 2014).



Figure 2.3 Schematic representation of free water surface flow constructed wetland (Oki and White, 2011)



Figure 2.4 Plants for free water surface flow constructed wetlands (Kamariah, 2006)

## 2.4.3 Subsurface Flow

Subsurface flow systems operate with the influent flowing below the surface of the soil or gravel substrate. Purification occurs during contact with the plant roots and substrate surfaces, which are water-saturated and can be considered oxygen-limited. The overlying vegetation and litter layer thermally insulates the substrate in these systems and so the wetland performance is insignificantly reduced during the winter. There are three types, horizontal flow, vertical flow and hybrid flow (Vymazal, 2010a).

## 2.4.3.1 Horizontal Subsurface Flow Constructed Wetlands

The wastewater enters HSSF CWs (Figure 2.5) through an inlet and flows slowly through the substrate under the surface of the bed, through the pores of the porous media and the roots of the plants, in a horizontal path until it reaches the outlet where it is collected (Vymazal et al., 1998a). This means there is no water surface exposed to the atmosphere, as in FWSF CWs, the surface flow is invisible and the water level is about 5-15 cm below the surface of the media (Vymazal et al., 2006). Moreover, mosquito breeding is not favoured and the health risk for humans and wildlife habitat is minimized (Kadlec and Wallace, 2008). Usually, gravel or a mixture of sand and gravel is used as a media with a depth between 30 and 80 cm. This media supports the growth of the plants (Vymazal et al., 2006; Akratos and Tsihrintzis, 2007). The value of the media depth depends on the depth of the plant roots, and an impermeable geo-membrane covers the bottom of the media bed. The slope of the bottom varies from 1-3%, meaning gravitational wastewater flow is promoted (Kadlec and Wallace, 2008). During this passage, the wastewater is exposed to a network of aerobic, anoxic and anaerobic zones. The aerobic zones occur around roots and rhizomes that leak oxygen into the substrate (Cooper et al., 1996; Vymazal, 2014).

Many publications have demonstrated that the presence of plant roots and porous media in this type of subsurface flow constructed wetland supports the development of the biofilm, which enhances the reduction of suspended solids and organic matter, while nitrogen and phosphorus reduction is low (Brix, 1994a; Vymazal et al., 1998a; Rousseau et al., 2004b; Kadlec and Wallace, 2008; Gikas and Tsihrintzis, 2010; Jegatheesan, 2011; Vymazal, 2013).

HSSF CWs are commonly used in the USA and Europe (Vymazal et al., 2006; Vymazal, 2008, 2011) and occupy a similar area when compared with FWSF CWs but the cost is higher (Tsihrintzis et al., 2007; Kadlec and Wallace, 2008).



Figure 2.5 Schematic of the horizontal subsurface flow constructed wetland (Tilley et al., 2014)

## 2.4.3.2 Vertical Subsurface Flow Constructed Wetlands

VSSF CWs have a structure made from several layers of aggregates (gravel and sand) as a media and planted with macrophytes with increasing gradation with depth (Cooper et al., 1996; Vymazal et al., 2006). The wastewater is dosed over the media surface and collected by an underdrain at the bottom of the media surface, so the wastewater is forced to flow perpendicular to the length of the wetland (Tousignant et al., 1999). The depth of the media varies from 45-120 cm and the slope of the bottom varies from 1-2%, which helps to collect the effluent (Vymazal et al., 2006) as shown in Figure 2.6.

In VSSF CWs, the wastewater disperses on the surface of the wetland media, thereby creating an aerobic environment (Scholz, 2006) by driving out the trapped air and sucking fresh air into the bottom of the media (Stefanakis et al., 2014), this enhances the nitrification process and decomposition of the organic matter, when compared to the

horizontal subsurface flow system (Vymazal et al., 2006; Kadlec and Wallace, 2008; Garcia et al., 2010; Abou-Elela et al., 2013; Kantawanichkul and Wannasri, 2013). Ye et al. (2012) found that 50% of atmospheric oxygenation was supplied to the upper layer of the media. Furthermore, many researchers have demonstrated that VSSF CWs have a good ability for the reduction of wastewater pollutants such as BOD, COD and suspended solids (Brix and Arias, 2005; Mimis and Gaganis, 2007; Bulc and Ojtrsek, 2008; Ong et al., 2010; S Wu et al., 2015). On the other hand, this mode of operation does not support the denitrification process and the reduction of phosphorus is limited when compared to other pollutants, because the contact time between the media and the wastewater is insufficient, as founded by Rustige et al. (2003), Prochaska and Zouboulis (2006) and Stefanakis and Tsihrintzis (2012). Vymazal and Kröpfelová (2008) related the reduction of phosphorus with sorption capacity of the media. However, VSSF CWs with tidal loading can achieve a higher phosphorus reduction, as proved by Hussein and Scholz (2017) when they treated azo textile dyes wastewater without any modification or adding any extra materials to the gravel, which was used as a media in their study. Furthermore, some researchers have stated that vertical-flow constructed wetlands with an intermittent operation mode can denitrify well with amendment (Gross et al., 2007; Weedon, 2010; Song et al., 2015).

VFCWs are mainly used in Europe, especially in Denmark, Austria, Germany, France, and the United Kingdom, but are also used in the USA (USEPA, 2000a; Kadlec and Wallace, 2008), as stated by Stefanakis et al. (2014) and Brazil (Philippi et al., 2006). Another countries in Asia and Africa have started to use this type of subsurface constructed wetland (Kivaisi, 2001; Hoffmann et al., 2011; Von Muench and Winker, 2011; Gargi Sharma et al., 2014).

VSSF CWs are mainly used to treat municipal and domestic wastewater, when the effluent limits are set for ammonia-nitrogen (Vymazal and Kröpfelová, 2008). However, many publications have reported on the use of VSSF CWs in treatment of other types of wastewater such as textile (Davies et al., 2005; Ong et al., 2011), domestic greywater (Pillai and Vijayan, 2012), artificial (Białowiec et al., 2011), refinery (Sharma and Brighu, 2014; Mustapha et al., 2015), dairy (Dąbrowski et al., 2017), combined raw wastewater and storm-water (Lopez, 2014) and mari-culture (Sousa et al., 2011).

VSSF CWs can be classified into further variants depending on flow direction, level of saturation and the duration of saturation (Stefanakis et al., 2014), and also VSSF CWs with intermittent loading (down-flow), recirculating VSSF CWs, tidal flow, saturated vertical up-flow, saturated vertical down-flow and integrated VSSF CWs. VSSF CWs with tidal loading are widely used, especially in Europe. The main advantage of this operation mode is the creation of a range from 3-5 cm of temporary wastewater ponding at the top of the media surface, this condition is known as a saturated (anaerobic) mode (Stefanakis et al., 2014) and the time for which this anaerobic mode takes place is known as the contact or retention time. The trapped air moves downward because of the wastewater force, then the wastewater drains vertically by gravity through the porous media and air from the atmosphere follows in its place, this condition is known as an unsaturated (aerobic) mode (Maier et al., 2009) and the time for which this aerobic mode takes place is known as the resting time. In this type of operation, the bed aeration increases and accelerates the growth of micro-organisms (Du et al., 2016), improving the nitrification process and oxidation of the organic matter to prevent the clogging (Kadlec and Wallace, 2008; Hua et al., 2013; Stefanakis et al., 2014).



Figure 2.6 Schematic of the vertical subsurface flow constructed wetland (Tilley et al., 2014).

#### **2.4.4 Hybrid Flow Constructed Wetlands**

In hybrid flow constructed wetlands, the advantages of various constructed wetlands (surface flow and subsurface flow) can be combined to complement each other (Sayadi et al., 2012; Vymazal, 2013; Ayaz et al., 2016), mainly VSSF and HSSF (Cooper et al., 1999; Liu et al., 2006; Melián et al., 2010; Vymazal and Kröpfelová, 2011; Vymazal, 2013) as shown in Figure 2.7. The idea of this combination is to make the most of the advantages of the one type to offset the disadvantages of the other. For example, the limited oxygen transfer capacity of HSSF constructed wetlands results in this system having a low nitrification process, while VSSF constructed wetlands have better efficiency in this process due to the higher oxygen transfer capacity. Conversely, the denitrification process in HSSF constructed wetlands is better than in VSSF constructed wetlands. By combining HSSF and VSSF constructed wetlands as a hybrid system, good

conditions for nitrification and denitrification processes can be produced (Liu et al., 2006; Mena et al., 2008; Vymazal, 2013). Seidel was the first one who tried to combine the various constructed wetlands, parallel VSSF CWs followed by HSSF CWs in series, in Germany in the 1960s, as reported by Vymazal (2005, 2013) and Stefanakis et al. (2014) and that was the foundation for the hybrid constructed wetlands, which was refreshed at the end of the 20<sup>th</sup> century.

In general, a stage of VSSF units followed by HSSF units in series and a stage of HSSF units followed by VSSF units are the most popular types of hybrid system (Vymazal, 2013). The first combination (Figure 2.7a) includes VSSF units to reduce organic matter and suspended solids, and provide good conditions for the nitrification process. These units are followed by HSSF units, which provide another process to reduce organic matter and suspended solids, and provide good conditions for the denitrification process. This type of hybrid system is widely accepted in treatment of municipal sewage wastewater (Foladori et al., 2012; Vymazal, 2013; Lee et al., 2015). The second combination (Figure 2.7b) includes HSSF units to reduce organic matter and suspended solids, and provide good conditions for the denitrification process followed by VSSF units for the nitrification process, and additional reduction of organic matter and suspended solids. In this type of hybrid, depending on the results of effluent analysis, if there is an increase in concentration of nitrate, which occurs in the wastewater effluent after VSSF because of ammonia nitrogen oxidation, the effluent has to be fed back as an influent in HSSF again or goes to another HSSF unit as shown in Figure 2.7c. These hybrid types are common in treatment of municipal wastewater (Gajewska, 2011; Vymazal, 2013) and can be used in treatment of mixed industrial wastewater (Justin et al., 2009). Vymazal (2005) reported this type of hybrid system, designed by Seidel, consisted of two stages of VSSF units

followed by two or three HSSF units in series (Figure 2.7d). The combination of VSSF units followed by HSSF and then FWS as a last stage (Figure 2.7e) was used in treatment of winery wastewater in Italy (Zanieri et al., 2010). Finally, the combination of HSSF units followed by FWS and then HSSF as a last stage (Figure 2.7f) was used in treatment of sewage wastewater in China (Ye and Li, 2009).

The design of hybrid system depends on the type of wastewater, population and the area demand. For instance, Cerezo et al. (2001) used a combination of different constructed wetlands in three stages and four series. FWS units without media and planted with *Phragmites australis*, HSSF CWs with sand as a media, unplanted and planted with *Typha dominguensis*, up-flow VSSF units with gravel as a media, unplanted and planted with *Phragmites australis* and down-flow VSSF units with gravel and stone as a media, unplanted and planted with *Phragmites australis* and down-flow VSSF units with gravel and stone as a media, unplanted and planted with *Phragmites australis* and down-flow VSSF units with gravel and stone as a media, unplanted and planted with *Phragmites australis*.



Figure 2.7 Various combinations of different constructed wetland types in hybrid constructed wetlands; (a), VSSF-HSSF, (b); HSSF-VSSF, (c); HSSF-VSSF-HSSF, (d); VSSF-HSSF-HSSF, (e); VSSF-HSSF-FWS, (f); HSSF-FWS-HSSF (Stefanakis et al., 2014).

#### 2.5 Reduction Mechanisms in a Constructed Wetland

Constructed wetlands have an ability to reduce many contaminants, including organic compounds, suspended solids, nutrients, heavy metals (Vymazal et al., 1998b; Sheoran and Sheoran, 2006), dye and colour (Nilratnisakorn et al., 2009; Olejnik and Wojciechowski, 2012).

### 2.5.1 Total Suspended Solids Reduction Mechanism

Total suspended solids is defined as all particles in wastewater which can be trapped using a filter, they include solid particles which may be organic or inorganic, such as silt, industrial waste and sewage (Stansbury and Branigan, 2013). A high concentration of total suspend solids leads to light being blocked from reaching plants which then affects the rate of photosynthesis (Bilotta and Brazier, 2008), causing a reduction in dissolved oxygen that should be released into the water from the plants. This is not the only physical effect, it is also leads to temperature changes. Furthermore, chemicals alterations, as a result of the higher concentration of total suspended solids, will release contaminants such as heavy metals (Dawson and Macklin, 1998; Bilotta and Brazier, 2008) and nutrients such as phosphorus (Haygarth et al., 2006) into the receiving stream. Also, the higher concentration of TSS may affect biological properties such as reducing the growth of organisms or reducing the population size (Wagener and LaPerriere, 1985; Shaw and Richardson, 2001), as reported by Bilotta and Brazier (2008).

Many researchers have stated that constructed wetlands have the mechanical ability to remove TSS with a high percentage (Green et al., 1997; Manios et al., 2003; Yirong and Puetpaiboon, 2004; Kadlec and Wallace, 2008; Vymazal, 2010a; Sehar et al., 2013). The

main parameters affecting the reduction of TSS in vertical-flow constructed wetlands are the hydraulic and microbiological characteristics of the media, and this has been confirmed by many researchers (Kadlec and Wallace, 2008; Mburu et al., 2008; Garcia et al., 2010). However, total suspended solids are reduced primarily by physical settling and filtration (Karathanasis et al., 2003; Mburu et al., 2008). Manios et al. (2003) stated that the ability of a gravel bed to remove total suspended solids is better than any other substrate bed (soil, sand and other compost) due to its physical and chemical structure. The other substrate materials mentioned are compactable, and the applied pressure reduces the large porosity of these substrates, while gravel is less compactible. The presence of plants has a positive effect on reduction of the total suspended solids by increasing retention time (Karathanasis et al., 2003; Kadlec and Wallace, 2008).

Some researchers have stated that the main problem linked to physical processing to reduce total suspended solids is clogging (Manios et al., 2003; Rousseau et al., 2004a; Caselles-Osorio et al., 2007). Xu et al. (2013) studied the effect of the combination of two out of three possible types of loading (hydraulic, organic load and total suspended solids) in different rate (high, medium and low) on clogging in vertical-flow constructed wetlands with a soil media. The results showed that the two loading combinations of high hydraulic and high organic, and high hydraulic and high suspended solids, caused the most clogging. Ye et al. (2008) stated that the main reason for clogging is the accumulation of non-filter materials; their results showed that the clogging occurred when the accumulation concentration of this non-filter materials was more than 18.23 mg/l. This problem can be solved by using a backwashing treatment (Ma et al., 2010), hydrogen peroxide treatment (H<sub>2</sub>O<sub>2</sub>) (Nivala and Rousseau, 2009) or by using the anaerobic digesters method (De la Varga et al., 2013).

Vertical constructed wetlands are highly efficient in total suspended solids reduction (Tietz et al., 2007; Kadlec and Wallace, 2008), many publications have shown a high reduction percentage value for total suspended solids (Gikas et al., 2007, 93.2%; Gross et al., 2007, 98%; Kadlec and Wallace, 2008, 87%; Paing et al., 2015, 85%).

In the case of textile wastewater treated by constructed wetlands, Bulc and Ojtrsek (2008) used a hybrid system consisting of vertical-flow followed by horizontal-flow constructed wetlands to treat RB5<sup>\*\*</sup>, DY211 and VY46 dyes. The results showed that the reduction percentage rate for TSS was 93%. Hussein and Scholz (2017) stated that TSS reduction depends on the type of azo dye. They studied treatment of two azo textile dyes wastewaters (AB113 and BR46) by vertical-flow constructed wetlands. They stated, in the case of BR46, the reduction percentage rate for TSS was 86%, while there was an increase in TSS reduction percentage rate, in the case of AB113 of around 0.07%. The TSS in the influent was 365 mg/l while in the effluent it was 392 m/l.

#### 2.5.2 Organic Matter Reduction Mechanism

In general, organic matter refers to any compounds that contain carbon formed by living organisms. In wastewater effluent, the organic composition is approximately 50% protein, 40% carbohydrate and 10% oil, fat, and other pollutants (Shon et al., 2007). Any increase in organic matter will lead to an increase the decomposers and these decomposers will grow rapidly consuming a great amount of oxygen (Naushad and AL-Othman, 2012). A lack of oxygen causes aquatic life in the receiving stream to be destroyed. The release of organic matter should be reduced to protect the life of downstream users (Ellis, 2004). In constructed wetlands, the reduction of organic matter is achieved through filtration, adsorption, and aerobic, anaerobic and microbial
metabolism (Cooper et al., 1996; Vymazal et al., 1998b; Stefanakis et al., 2014). This reduction in organic matter can be assessed by measuring the COD and BOD concentrations (Quayle et al., 2009; Lee and Nikraz, 2014).

The accumulation of organic matter is very important in constructed wetlands. It serves two functions for the micro-organisms, providing energy for micro-organisms to grow and supplying carbon for the long term (Kröpfelová, 2008; Magdoff and van Es, 2010; H Wu et al., 2015). Supplying carbon helps in the denitrification process (nitrogen reduction) and this has been confirmed by many researchers (Van, 1995; Kallner, 2006; Reddy and DeLaune, 2008; Songliu et al., 2009). On the other hand, an increase in the accumulated organic matter plays a significant role in causing clogging (Lianfang et al., 2009; Hua et al., 2010; Fu et al., 2013).

In textile wastewater, chemical oxygen demand is an important index to measure the organic pollution in the wastewater. It is defined as the amount of oxygen equivalent that would be required to oxidize the organic materials in the wastewater (Ding et al., 2005; Chai et al., 2006; Latif and Dickert, 2015). Many researchers have investigated COD concentration in textile wastewater treated by subsurface flow constructed wetlands. Davies et al. (2005), Ojstršek et al. (2007), Ong et al. (2011) and Yalcuk and Dogdu (2014) used vertical-flow constructed wetlands to treat different azo textile dyes; COD reduction percentage rates were 64, 88, 95 and 64%, respectively. Real textile wastewater was treated by horizontal-flow constructed wetlands in Tanzania (Mbuligwe, 2005). The results showed that the COD reduction percentage rate was 72%. Bulc and Ojtrsek (2008) used a hybrid system to treat a mixture of three different dyes. The hybrid system consisted of vertical-flow followed by horizontal-flow constructed wetlands; the COD reduction percentage rate was 84%. Hussein and Scholz (2017) treated two different azo textile dyes (AB113 and BR46) by vertical-flow constructed wetlands.

showed that COD reduction percentage rates for AB113 and BR46 reached 55% and 89%, respectively.

### 2.5.3 Nutrients Reduction Mechanism

The presence of nitrogen and phosphorus in wastewater causes eutrophication in natural water and that will lead to lack of oxygen (Sathasivan, 2009; Mazumder, 2013; Yamashita and Yamamoto, 2014). In constructed wetlands, phosphorus occurs as organic and inorganic phosphate (Heal et al., 2005; Vymazal, 2007). Most phosphorus converts to orthophosphate as a result of biological oxidation (Cooper et al., 1996). Vymazal (2007) demonstrated that the reduction mechanisms of phosphorus in constructed wetlands are: fragmentation, desorption, dissolution, leaching, plant and microbial uptake, mineralization, adsorption, precipitation, burial and sedimentation. However, the major reduction mechanisms take place through adsorption, precipitation, and plant and microbial uptake (Richardson and Craft, 1993; Cooper and Findlater, 2013). This means the best reduction mechanism for phosphorus is a combination of physical, chemical and biological treatments (Mazumder, 2013). Orthophosphate promptly amasses in wetland vegetation and media, therefore promoting natural take-up and substance bonding. Kayombo et al. (2004) pointed out that storing and/or reduction of phosphorus is dependent on the type of wetland and its operational mode. Phosphorus can leave the wetland by coming from the media through the water column, if the anaerobic condition is not reversed due to the lack of oxygen demand (Reddy et al., 1999). The filter media of constructed wetlands plays a major role in the reduction of phosphorus (Karczmarczyk, 2004; Seo et al., 2005), and its capacity to reduce the phosphorus depends on the metal content of the wastewater (Brix et al., 2001; Vohla et al., 2011), and the value of pH as well (Cui et al., 2008) through adsorption and precipitation mechanisms. At pH>6, the reactions are a combination of physical adsorption to Fe and Al oxides, and precipitation as calcium phosphates. At pH<6, precipitation as Fe and Al phosphates becomes more important (Priya and Urmila, 2013). These mechanisms might cause clogging in constructed wetlands (Knowles et al., 2011) especially when the wastewater contains a large amount of industry materials in high concentrations (Fleming et al., 1999; Kadlec and Wallace, 2008). Lantzke et al. (1999) conducted a study regarding the percentage reduction of orthophosphate from wastewater by planted vertical-flow constructed wetlands. The results showed the percentage reduction for the plant, gravel and biofilms were 70%, 20% and 10%, respectively, in the long-term.

In general, the main reduction mechanisms for nitrogen are microbial nitrification and denitrification (Tanner et al., 1995; Obarska and Gajewska, 2003), plant uptake (Koottatep and Polprasert, 1997) and accumulation in soil (Lee et al., 2009), as shown in Figure 2.8. Nitrification is the process of oxidizing ammonium or ammonia to nitrite followed by the oxidation of the nitrite to nitrate (Kessel et al., 2015). Denitrification is the process of reducing nitrate to gaseous nitrogen (Schaechter, 2009). Nitrogen enters the wetland as a dissolved organic and/or inorganic form (Sumner, 1999). The relative proportions of the nitrogen composition depend on the source of the wastewater (Vymazal and Kröpfelová, 2009a). Nitrite, nitrate, ammonium and ammonia are the major forms of inorganic nitrogen (Likens, 2010), while amines, urea, amino acids and purine are the forms of organic nitrogen (Bastian and Hammer, 1993; Kadlec and Wallace, 2008). Several publications have reported other processes to reduce nitrogen in constructed wetlands such as ammonification, nitrogen fixation, ammonia volatilization, and anammox and ammonia adsorption (Vymazal, 2007; Mustafa, 2010; Coban et al., 2015). However, the main mechanism to eliminate nitrogen is the combination of nitrification and denitrification (Green, 1996; Lee et al., 2009; Scholz, 2011).

Nitrification and denitrification mechanisms need both aerobic and anaerobic conditions. The nitrification process needs anaerobic conditions to oxidize ammonia and/or ammonium to nitrite as a first step, and aerobic conditions to oxidize nitrite to nitrate as a second step (Kadlec and Knight, 1996; Kyambadde, 2005). Denitrification needs aerobic/anaerobic conditions to reduce nitrogen compounds (nitrate or nitrite) to gaseous nitrogen (Ji et al., 2014; Winkler et al., 2015), and this process needs a carbon source which can be provided either from the decaying plant detritus and/or as COD present in the incoming wastewater (Kallner, 2006; Cakmak and Apaydin, 2010). Hu Jie et al. (2009) studied the effect of the C/N ratio (C, organic carbon; N, inorganic nitrogen) on both the nitrification and denitrification processes. The results showed that increasing the organic carbon leads to a decrease in the nitrification rate, while denitrification was detected after increasing the organic carbon. These results were also confirmed by Ding et al. (2012).

A change in temperature has an effect on the rate of nitrogen reduction. Many publications point out the rate of reduction decreases in the winter period (Kadlec and Reddy, 2001; Kuschk et al., 2003; Mietto et al., 2015). USEPA (2000a) stated that a temperature of >12.9 °C is better for both nitrification and denitrification when compared to <12.9 °C. Chapanova et al. (2007) pointed out that the reduction rate of ammonianitrogen at 5 °C is low when compared to that at temperatures between 15-20 °C. This means that temperature is an important factor in designing constructed wetlands. Vymazal (2007) stated that the effect of temperature is greater on nitrification than on denitrification. For developing countries (especially when the degree of temperature is over 40 C°, will be effect on the life of the plants and that depend on them growth cycle and that need to be consult with ecologists. In the case of textile wastewater treated by constructed wetlands, Bulc and Ojtrsek (2008) found that the reduction percentage rates for TN, N<sub>organic</sub> and NH<sub>4</sub>-N were 52%, 87% and -331% when they treated dyes (RB5, DY211 and VY46) by a hybrid system. This system consisted of vertical-flow followed by horizontal-flow constructed wetlands. Ong et al. (2011) used vertical-flow constructed wetlands to treat dye AO7. The results demonstrated that the reduction percentage rate for NH<sub>4</sub>-N was 86%. Yalcuk and Dogdu (2014) studied treatment of azo dye (AY 2G E107) by vertical-flow constructed wetlands. The results showed that the reduction percentage rates for PO<sub>4</sub>-P and NH<sub>4</sub>-N were 94% and 77%, respectively. Furthermore, Hussein and Scholz (2017) treated textile wastewater of two azo textile dyes (AB113 and BR46) by vertical-flow constructed wetlands. The results stated that the reduction percentage rates of PO<sub>4</sub>-P, NH<sub>4</sub>-N and NO<sub>3</sub>-N for AB113 were 78%, 53% and 73%, respectively and 81%, 70% and 75% in the case of BR46.



Figure 2.8 Schematic diagram showing the main nitrogen reduction mechanisms (Mustafa, 2010; adapted from Reddy and DeLaune, 2008).

Songliu et al. (2009) investigated the ability of vertical-flow constructed wetlands with different plants to remove nitrate. The constructed wetlands consisted of three parallel wetlands built with concrete. The media was slag at the bottom in a 50 cm layer and 10 cm of local soil above the slag layer. The wetlands included a layer of river gravel with a 30 cm depth and 10-20 mm diameter, as a top layer. The wetlands were planted with three different plants. These constructed wetlands were operated during February 2006 to December 2006 in China. The main conclusion for this study was that the constructed wetland cannot provide enough organic carbon sources to remove the nitrate when it is rich in the effluent and there is a poor organic chemistry contamination in the effluent. The temperature and pH were also investigated in this study and the results showed a strong ability of the constructed wetland to maintain a suitable pH for denitrification, and that the temperature may affect the nitrate reduction in winter.

### 2.5.4 Heavy Metals Reduction Mechanism

Heavy metals refer to any metallic elements which have a relatively high density compared to water (Fergusson, 1990). Järup (2003) stated that the specific density of heavy metals is more than 5 g/cm<sup>3</sup>, for example, cadmium (Cd), mercury (Hg), arsenic (As), thallium (Tl) and lead (Pb) (Martin, 2011). Many publications have stated that most heavy metals ions are known to be carcinogenic or toxic (Singh et al., 2007; Nagajyoti et al., 2010; Fu and Wang, 2011; Lakherwal, 2014), are not biodegradable and are accumulated by living organisms (Gadd, 1990; Barakat, 2011). Serious health problems for humans are caused by heavy metals, such as skin irritation, anaemia, cirrhosis and stomach cramps (Babel and Kurniawan, 2004; Paulino et al., 2006; Oyaro et al., 2007). Therefore, the reduction of heavy metals is very necessary.

In constructed wetlands, there are a number of physical, chemicals and biological processes which reduce heavy metals including settling, sedimentation and adsorption as physical processes; sorption, adsorption, oxidation, precipitation and co-precipitation, metal carbonates and metal sulphides as chemical processes; and microbial and plant uptake as biological processes (Patrick and Verloo, 1998; Cheng et al., 2002; Kosolapov et al., 2004; Sheoran and Sheoran, 2006; Marchand et al., 2010; Sukumaran, 2013; Arivoli et al., 2015). In settling and sedimentation processes, heavy metals associated with particulate matter significantly reduced (Ellis et al., 2003a; Sheoran and Sheoran, 2006) because of dynamic transformation, which occurs in wetland systems whether the water is static or mobile (Johnston, 1993). Heavy metals can be transported from the inflow wastewater to wetland media or microbes (Dunbabin and Bowmer, 1992; Matagi et al., 1998). Sheoran and Sheoran (2006) point out that the sedimentation process is a principle process for the reduction of heavy metals. Sinicrope et al. (1992) founded that a higher reduction rate occurs by plant uptake as a result of the sedimentation process. Walker and Hurl (2002) pointed out that the occurrence of the sedimentation process after another process, like precipitation, leads to the production of larger particles, which act as a trap for heavy metals. Patrick and Verloo (1998) and Lesage (2006) reported that heavy metals can be reduced by physical adsorption (electrostatic attraction). Evangelou (1998) demonstrated that the reduction of heavy metals by chemisorption is better than that by physical adsorption.

Ones of the most important chemical mechanisms of heavy metals reduction is sorption (Sheoran and Sheoran, 2006). This mechanism includes a group of processes: precipitation and adsorption (Marchand et al., 2010). It consists of a transfer of ions from solution phase (inflow wastewater) to solid phase (wetland substance) (Crini and Badot, 2010). The adsorption of heavy metals depends on the type of element, its concentration and media condition (Matagi et al., 1998; Walker and Hurl, 2002; Zeb et al., 2013). For instance, most trace metals such as Pb, Cd, Ni, Zn and Cu exist as cations. The capacity for retention of them depends on the type of wetland media (Sheoran and Sheoran, 2006). Mengzhi et al. (2009) studied the effect of two types of media, coke and gravel, on the reduction rate of three heavy metals (Cu, Pb and Zn). The results showed different absorption efficiencies. Sahu (2014) investigated the effect of different retention times (1-8 days) on the reduction of Cr, Ni, Fe and Hg. the reduction rates increased with increasing retention time. Some authors reported the reduction of heavy metals occurs by the co-precipitation mechanism and cation exchange capacity (Sharma, 2014). For instance, iron can act as a co-precipitation agent for As (Stottmeister et al., 2006). Hedin et al. (1994) and Noller et al. (1994) pointed out that heavy metals co-precipitate with secondary minerals in wetlands. Metals such as Cu, Ni, Zn, and Mn co-precipitated in iron oxides and Co, Fe, Ni and Zn co-precipitated in manganese oxides, as reported by Tumelo Seadira et al. (2014). Hallberg and Johnson (2005) reported that the soluble Mn from AMD can be removed via oxidation and precipitation mechanisms.

Heavy metals reduction by microbial and plant uptake is known as a biological mechanism and it's the most important pathway to reduce heavy metals (Sheoran and Sheoran, 2006; Rai, 2008; Yeh, 2008; Liu et al., 2015). The reduction of heavy metals in constructed wetlands by plant uptake depends on the type of plant itself, concentration of the heavy metals, pH of sedimentation, temperature, sediment chemical properties and organic matter content (Gambrell, 1994; Siedlecka, 1995; Greaney, 2005; Garcia et al., 2010). For instance, Lim et al. (2001) found there is no significant effect of the plant *Typha angustifolia* on reduction of Cu, Zn, Cd and Pb. Allende et al. (2014) pointed out that the reduction rate of heavy metals was <3% by plant uptake, while >85% by the wetland media. On the other hand, Sinicrope et al. (1992) found a significant effect of the

plant *Scirpus lacustris* on the reduction of Cd and Zn. The results showed approximately 13% and 35% as reduction percentages of Zn and Cd, respectively, accumulated in the fine root of the plant. Cooper and Findlater (2013) demonstrated that the higher concentration of heavy metals will affect plant growth. For instance, Wong and Bradshaw (1982) found the average concentrations of 1.9 mg/l for Zn and 1.6 mg/l for Pb would undoubtedly affect plant growth. J Li et al. (2015) indicated that a slightly acidic inflow, with pH= 5.6-6.5, helps the plant to uptake heavy metals, while in the case of an alkaline inflow, pH= 7.5-8.5, the plants hardly uptake any heavy metals. Although the reduction of heavy metals by plant uptake is small, according to some authors, it has an indirect effect on the reduction. The roots of plants provide an environmental culture for the growing micro-organisms (rhizosphere) (Williams, 2002). Also, Dunbabin and Bowmer (1992) found the exudates of roots help to increase bacteria activity, give more space to buffer capacity and keep pH at a neutral level.

### 2.5.5 Azo Dye Reduction Mechanism

Numerous dyes containing azo compounds are known as azo dyes (Zollinger, 2003; Podczeck and Jones, 2004; Hatice, 2010). Azo compounds are defined as any organic chemical compounds that have the general formula (R-N=N-R'), where (N=N) is a nitrogen atom, R' and R, are either alkyl (aliphatic) or aryl (aromatic) (Arcos et al., 2013; Castro et al., 2016; Ekambaram, 2016).

Azo dye produces colour in dye pollutant water, and this colour blocks sunlight, which is detrimental to many photo-initiated chemical reactions necessary for aquatic life of water bodies (Pereira and Alves 2012; Yadav et al., 2012). Azo dye decolourization is achieved under aerobic, anaerobic or anoxic conditions (O'neill et al., 2000; Sponza and Işik, 2002; Van der Zee, 2002; Davies et al., 2006). Hatice (2010) stated that there is no correlation between decolourization rate and molecular weight but the rate is dependent on the dye structure and the added organic carbon source. At anaerobic conditions, azo bond (N=N) cleaves and this process releases aromatic amine, which resists any further anaerobic treatment (Brown and Hamburger, 1987; Chung and Stevens, 1993). Aromatic amine can be reduced under aerobic treatment (Weber and Lee Wolfe, 1987; Pinheiro et al., 2004; Ong et al., 2011), while azo dye colour reduction can be achieved under aerobic, anoxic or anaerobic conditions (Pandey et al., 2007). In constructed wetlands, many publications have stated a high colour reduction under anaerobic treatment (Mbuligwe, 2005; Bulc and Ojtrsek, 2008; Ong et al., 2009; Ong et al., 2010).

The biodegradation mechanism works to achieve decolourization of azo dyes under all three conditions (Kodam et al., 2005; Moosvi et al., 2005; Pandey et al., 2007). Under anaerobic conditions, micro-organisms require external carbon as an energy source to degrade azo dye (Carliell et al., 1994; Şen and Demirer, 2003; Haroun and Idris, 2009). Under aerobic conditions, very few micro-organisms (aerobic bacteria, for example: *P. aeruginosa*) are able to grow depending on azo groups as a sole carbon source (Yang et al., 1998; Edwards, 2000; Pandey et al., 2007). These bacteria cleave the nitrogen atom to release the amines as a source of carbon and an energy to help them to grow. Under anoxic conditions, other types of bacteria, *Pseudomonas luteola* as an example, can decolourize azo dye (Zissi and Lyberatos, 1996; Yu et al., 2001; Grekova-Vasileva et al., 2009). These bacteria need a complex organic carbon source to reduce azo dyes (Khehra et al., 2005b). Pandey et al. (2007) stated that the reduction of azo dye is achieved under anaerobic condition only, although many micro-organisms (bacteria) can be grown aerobically under anoxic conditions.

Aromatic amines are released as a result of decolourization of azo dyes under anaerobic conditions. The reduction of azo dyes under aerobic biodegradation by some groups of

micro-organisms (bacteria) is more easily achieved (Ekici et al., 2001;Arora, 2015), in the presence of oxygen (Stolz et al., 1992; Jothimani et al., 2003). In spite of this, the reduction of a few simple types of aromatic amines (for example, 5-aminosalicylic acid) has been reported under methanogenic conditions (Kalyuzhnyi et al., 2000; Jothimani et al., 2003; Yemashova et al., 2004). In wetlands, this type of bacteria are known as methanogens and are very common in natural wetlands (Walter and Heimann, 2000; Jerman et al., 2009) as well as in constructed wetlands (Tanner et al., 1997; Inamori et al., 2007; Vymazal and Kröpfelová, 2009b).

As a textile wastewater treatment in the USA, Pervez et al. (1999) studied the colour reduction for two azo textile dyes (AB113 and RB171) by vertical-flow constructed wetland with gravel and sand as media, and planted with *Phragmites australis*. The results showed that the colour reduction rate was 98%. In Portugal, Davies et al. (2005) worked on treatment of an azo textile dye (AO7) by vertical-flow constructed wetland with gravel and sandy clay soil as media, and planted with *Phragmites australis*. The results showed that the colour reduction was 74%. In Tanzania, Mbuligwe (2005) achieved a 77% colour reduction when he treated real textile wastewater by horizontal-flow constructed wetland with gravel and sand as media, and planted with *Typha* and *cocoyam*. Furthermore, in Slovenia, Bulk and Ojtrsek (2008) used a hybrid system to treat three azo textile dyes (RB5, DY211 and VY46). The hybrid system consisted of vertical-flow followed by horizontal-flow constructed wetlands with gravel, sand and tuff as media. The results showed that the colour reduction achieved wetlands with gravel, sand and tuff as

A survey regarding previous related works studying the reduction of various types of azo dyes is presented in Table 2.2. The survey is categorized according to the dye used, types and design characteristics of wetlands, plant used, reduction performance, duration of experiment and its location and references.

Dye used	Type of wetland	Design characteristics	Plants used	Reduction performance	Duration (days)	Country of	References
						operation	
AB113, RB171 <sup>*</sup>	VF	Gravel-sand	P. australis	98% colour	70	USA	Pervez et al. (2000)
AO7	VF	Gravel-sandy clay soil	P. australis	74% colour, 64% COD and 71% TOC	77	Portugal	Davies et al. (2005)
Various dyes in real wastewater	HF	Gravel-sand	<i>Typha</i> and <i>cocoyam</i>	77% colour, 72% COD and 59% sulphate	84	Tanzania	Mbuligwe (2005)
AO7	VF	Gravel-sandy clay soil	P. australis	99% colour, 93% COD and TOC	48	Portugal	Davies et al. (2006)
RB5**, DY211, VY46	VF-HF	Gravel-sand- tuff	P. australis	90% colour, 84% COD, 93% TSS, 52% T- N, 87% N <sub>organic</sub> , -331% NH <sub>4</sub> -N, 88% sulphate, 80% anion surfactant and 93% TSS	60	Slovenia	Bulc and Ojstršek (2006)
BB41	VF	Sand	M. spicatum/C. demersum	94-96% dye	50	Turkey	Keskinkan and Lugal Göksu (2007)
RR22, VR13, RB5**	VF	Gravel-Sand- Zeolite-Peat	Without plant	70% dye, 60% EC, 88% COD and TOC	90	Slovenia	Ojstršek et al. (2007)
AO7	VF	Gravel	P. australis	68% dye, 69% COD, 67% TOC	8	Portugal	Davies et al. (2009)
RR141	VF	Gravel-sand	Typha	49% colour, 60% COD, 86% TDS	N/A	Thailand	Nilratnisakorn et al., (2009)
A07	UF	Glass beads- gravel	P. australis/ Manchurian wild rice	98% dye, 86% COD, 96 NH <sub>4</sub> -N, 86% NO <sub>3</sub> -N, 67% T-N, 26% T-P	42	Japan	Ong et al. (2009)

Table 2.2 Some related studies (listed in order of date) on azo textile dye wastewater treatment by constructed wetlands.

Table 2.2 continued

AO7	UF	Gravel-glass	P. australis	98% dye, 90% COD, 67% T-N, 28% T-P,	365 <sup>(a)</sup>	Japan	Ong et al. (2010)
		beads		98% NH <sub>4</sub> -N, 100 NO <sub>3</sub> -N			
AO7	VF	Gravel-sludge	P. australis	94% colour, 95% COD and 86% NH <sub>4</sub> -N	27	n/a	Ong et al. (2011)
	FWS-SSF	Shale	P. australis	98% COD, 97%, colour	N/A	Thailand	Cumnan and
	SSF-FWS			91% COD, 99% colour			Yimrattanabovorn,
							(2012)
Various dyes	VF	Coconut	G. pulchella	70% COD, 74% TOC, 70 BOD	3 and 4	India	Kabra et al. (2013)
in real textile		shavings-Soil					
wastewater		with bacteria					
Various dyes	VF	Coconut	Portulaca	59% COD, 38% BOD, 37% TOC,	3	India	Khandare et al.
in real textile		shavings-sand -	grandifora	41% turbidity, 71% TDS, 60% TSS			(2013)
wastewater		gravel-soil with	0 0	•			
		bacteria					
Various dyes	VF	Coconut	Typha	79% COD, 77% BOD, 59% TDS,	3	Pakistan	Shehzadi et al.
in real textile		shavings-gravel-		27% TSS			(2014)
wastewater		sand-soil					
AY 2G E107	VF	Gravel-sand-	Canna and	95% colour, 64% COD, 94% PO <sub>4</sub> -P	90	Turkey	Yalcuk and Dogdu
		zeolite	Typha	and 77% NH <sub>4</sub> -N		-	(2014)

Note: AB, acid blue; RB<sup>\*</sup>, reactive blue; VF; vertical flow; AO, acid orange; COD; chemical oxygen demand; TOC, total organic carbon; HF; horizontal flow; RB<sup>\*\*</sup>, reactive black; DY, disperse yellow; VY, vat yellow; TSS, total suspended solid; T-N, total nitrogen; N, nitrogen; NH<sub>4</sub>-N, ammonium nitrogen; BB; basic blue, RR; reactive red; VR, vat red; EC, electrical conductivity; UF, upper flow; NO<sub>3</sub>-N; nitrate nitrogen; TDS; total dissolved solids; FWS; free water surface; SSF; subsurface flow; BOD; biochemical oxygen demand; AY; acid yellow; PO<sub>4</sub>-P, Ortho-phosphate-phosphorus; <sup>(a)</sup> the experimental work was under controlled conditions (indoor); N/A, not applicable.

### 2.6 Choice of Vertical-Flow Constructed Wetlands over Horizontal-Flow

The use of vertical and horizontal-flow constructed wetlands as a substitutional means of treating wastewater is widely accepted (Haberl et al., 1995; Brix et al., 2001; Bezbaruah and Zhang, 2003; Abou-Elela and Hellal, 2012; Abou-Elela et al., 2013; Fan et al., 2013b; C Li et al., 2015; Sani, 2015) due to the advantages of their use, such as low cost of setup and operation, ability to tolerate fluctuations in flow, and simplicity of operation and maintenance (Luise et al., 1993; Scholz and Hedmark, 2010). Aslam et al. (2010) concluded that constructed wetlands can be used as an economical option to treat wastewater to help developing countries. Konnerup et al. (2011) stated that vertical-flow and horizontal-flow constructed wetlands have a good ability to improve the quality of the river water, in Vietnam, which is polluted due to aquaculture practices. On the other hand, the vertical-flow constructed wetlands are better than horizontal-flow ones in treatment of wastewater in some water quality variables, according to the results of some research.

### 2.6.1 Nitrification and Denitrification

Nitrification is the biological oxidation of ammonium-nitrogen (NH<sub>4</sub>-N) to nitritenitrogen (NO<sub>2</sub>-N) and then oxidation of the nitrite-nitrogen (NO<sub>2</sub>-N) to nitrate-nitrogen (NO<sub>3</sub>-N) by two groups of micro-organisms (Schmidt et al., 2001; Paul, 2006). VFCWs provide excellent conditions for nitrification because their ability to provide oxygen transfer capacity is greater than that of HFCWs (Vymazal, 2002; Vymazal, 2005; Saeed and Sun, 2012; Shi et al., 2012; Paing et al., 2015; Butterworth et al., 2016).

Denitrification is the biological process of production of gaseous nitrogen  $(N_2)$  by microbial reduction of nitrite-nitrogen  $(NO_2-N)$  (Tiedje, 1982). Although some

researchers have been less concerned about nitrate reduction, and denitrification is usually low in VFCWs (Vymazal, 2005; Fuchs, 2009), other researchers have recently demonstrated that VFCWs can denitrify well with some modifications. For example, recycling the treated effluents enhanced the ability of VFCWs to remove the total nitrogen (TN) (Platzer, 1999; Marti et al., 2003; Arias et al., 2005; Gross et al., 2007; Prost-Boucle and Molle, 2012). Lavrova and Koumanova (2014), and Shen et al. (2015) used a solid carbon source (methanol and starch blends, respectively) to achieve a higher denitrification.

### 2.6.2 Chemical Oxygen Demand (COD)

Chemical oxygen demand is an important index to measure the organic pollution in wastewater. It is defined as the amount of oxygen equivalents required to oxidize the organic materials in the wastewater (Latif and Dickert, 2015). Many researchers have stated that vertical-flow constructed wetland systems perform satisfactorily in the reduction of chemical oxygen demand (COD) (Brix and Arias, 2005; Prochaska et al., 2007; Sharma and Brighu, 2014; Abdelhakeem et al., 2015; Alisawi et al., 2015a).

### 2.6.3 Heavy Metals

The expression "heavy metals" refers to any metal elements that have a relatively high density compared to water and are toxic or poisonous (Fergusson, 1990), for example Cadmium (Cd), Copper (Cu), Nickel (Ni), Lead (Pb) and Zink (Zn) (Salomons et al., 2012). Yalcuk and Ugurlu (2009) showed that the ability of vertical-flow constructed wetlands to remove heavy metals found in the effluent of leachate is better than that of horizontal-flow constructed wetlands. Arivoli et al. (2015) used a vertical-flow system to

treat heavy metals from the effluent of pulp and paper industry. Their results indicated higher reduction efficiencies for iron, copper, manganese, nickel, zinc and cadmium.

### 2.6.4 Area Demand

The vertical-flow constructed wetlands system is the state of art design technologies used in wastewater treatment and its use is increasing rapidly (Stefanakis et al., 2014; Scholz, 2015) due to the advantage of its lower area requirement compared to the horizontal-flow constructed wetlands system (Vymazal et al., 1998a; Luederitz et al., 2001; Scholz, 2011 ; Saeed and Sun, 2012; Stefanakis et al., 2014). The required area for VFCW systems varies from 0.4-2 (m<sup>2</sup>/PE), while in the case of HFCW systems the required area is from 3-10 (m<sup>2</sup>/PE) as shown in Table 2.3 (Cooper and Green, 1995; Vymazal et al., 1998b).

Country	Required A	rea (m <sup>2</sup> /PE)	Deference	
Country	VFCWs	HFCWs	Kelefence	
Austria	1.7		(Hoffmann et al., 2011)	
Canary Islands	1.5		(Vera et al., 2013)	
Czech Republic		4.1-4.5	(Vymazal, 1996)	
Czech Republic	1-1.5		(Vymazal and Kröpfelová, 2011)	
Germany	1.6,1.7		(Olsson, 2011; Hoffmann et al.,	
			2011)	
Greece	1-1.5		(Stefanakis and Tsihrintzis, 2012)	
India	1.2	3	(Hoffmann et al., 2011)	
Philippines	0.9		(Hoffmann et al., 2011)	
Syria	0.4		(Hoffmann et al., 2011)	
United Arab	1.9		(Hoffmann et al., 2011)	
Emirates				
United Kingdom		5.6	(Cooper, 2001)	
United Kingdom	2		(Cooper, 2005)	
United Kingdom	1-2		(Weedon, 2010)	

Table 2.3 Examples of area requirements (listed in order of alphabet letters)

Note: PE, Person Equivalent= 60g BOD/capita/day

### 2.7 Amine Compounds

### 2.7.1 Overview

An aromatic amine is an amine in the molecule of which there are one or more aromatic rings directly on the nitrogen atom (Gunawardena, 2015) as shown in Figure 2.9.



Figure 2.9 Aromatic amine rings

Dyes are the one of major sources of aromatic amines in the environment, (Robinson et al., 2001, Snyderwine et al., 2002 and Pinheiro et al., 2004). In addition, human blood cancer and the other cancers continue to be linked with the present day level of aromatic amines (O'neill et al., 2000; Pielesz et al., 2002; Snyderwine et al., 2002) and they are toxic to aquatic life (Pinheiro et al., 2004; Kirandeep et al., 2015). Each dye has one or more types aromatic amines (Pielesz et al., 2002; Pinheiro et al., 2004). Wetlands can degrade aromatic amines under aerobic conditions (Mbuligwe, 2005; Ong et al., 2010; Ong et al., 2011).

In this study, the two azo textile dyes selected, AB113 and BR46, are toxic to aquatic life (Pinheiro et al., 2004; Mohan et al., 2007; Kirandeep et al., 2015) and carcinogenic (Chung, 2000; Gadaleta et al., 2016). Three types of amines were released as a result of dye AB113 degradation, 3-AminoBenzeneSulfonic Acid (ABSA), 1,4-

DiAminoNaphthalene (DAN) and 5-Amino-8-(phenylamino) Naphthalene-1-Sulfonic Acid (ANSA) (Senthilvelan et al., 2014). The wavelength for each being 288 nm, 255 nm and 225 nm, respectively (Koepernik and Borsdorf, 1983; Paul et al., 1990). In the case of BR46, two aromatic amines were released as a results of its degradation, *N*benzyi-*N*-mehtylaniline (NBNMA) and *N*-benzyl-*N*-methylbenzene-1,4-diamine (NBNMD), with the wavelength of maximum absorbance for each being 254 nm and 290 nm, respectively (Fihtengolts, 1969; Küçükgüzel et al., 1999).

### 2.7.2 Classification and Structure of Amines

Amines are classified into three categories (primary, secondary and tertiary) in which one, two or three hydrogen of  $NH_3$  are replaced by aryl or alkyl groups (Bacon and Adams, 1968; Murry and John, 1992; Dobo et al., 2006; Sankar and Nasar 2009).

### 2.7.2.1 Primary Amines

Primary amines are compounds of amine, which have one group attached to the nitrogen atom as shown in Figure 2.10.

## $CH_3 - NH_2$

Figure 2.10 Methylamine (Primary Amines) chemical structure

### 2.7.2.2 Secondary Amines

Secondary amines are compounds of amine, which have two groups attached to the nitrogen atom as shown in Figure 2.11.

Figure 2.11 Dimethylamine (Secondary Amines) chemical structure

### 2.7.2.3 Tertiary Amines

Tertiary amines are compounds of amine, which have three groups attached to the nitrogen atom as shown in Figure 2.12.



Figure 2.12 Trimethylamine (Tertiary Amines) chemical structure

Amines can be further divided as aliphatic or aromatic: aliphatic, when the nitrogen is bonded to one alkyl group only; and aromatic, when the nitrogen is bonded to one or more aryl groups (Vogt and Gerulis, 2000).

### 2.8 Textile Wastewater Treatment within Vertical-Flow Constructed Wetlands

The use of constructed wetlands in dye wastewater treatment is still at an experimental stage and is not well established (Nawab et al., 2016). Although many researchers have investigated the performance of CWs to treat textile wastewater (dye reduction, COD reduction, phosphorus and nitrogen compounds), the results rarely cover all seasons and also rarely it is found a researchers work on the mix of dyes and investigate the efficiency of constructed wetlands on the reduction of aromatic amine compounds (Table 2.2). for examples, Pervez et al. (1999) used VFCWs, located in the USA, consisting of nine filters: three replicates were planted-up with *Phragmites australis*, three were replicated with air pumped into the substratum and three replicated without plants used as controls to remove two azo textile dyes (RB\*171 and AB113) from a synthetic wastewater simulating textile wastewater over a 70-day time period. The media of the wetlands were filled with layers of gravel of various sizes and sand. The percentage reduction was 98% for both dyes. They observed that the majority of the dye (57%) was removed in the initial period (12 h) by adsorption onto charged surfaces in the substratum. By the addition of peat or other suitable organic substratum, the efficiency was substantially enhanced; however, the period for the experimental work was very short and the researchers did not assess the VFCWs performance in reduction of the COD, phosphorus and nitrogen compounds. Furthermore, the researchers did not investigate the ability of the constructed wetlands system to reduce the amines compounds, which were released due to dye degradation.

Davies et al. (2005) selected AO7 to study the peroxidases (POD) activity of *Phragmites australis* in its degradation in a VFCW. They found a crude plant extract with  $H_2O_2$ 

(which was produced by the plants) to degrade this kind of dye and its aromatic amines. At the concentration of AO7 (130 mg/l), POD activity was found to be 12.9, 4.3 and 2.1-fold for roots, stems and leaves, respectively. POD activity inhibition occurred immediately at an AO7 concentration of 700 mg/l, but 2 days later, it returned to the previous level. The total period was 312 h in an outside environment. Carias et al. (2008) demonstrated the same fact that the enzymes of *Phragmites australis* are involved in the degradation of AO7. Although the results were important to prove that *Phragmites australis* has a good value to use as a plant with VFCWs, 312 h as a time period is not enough to confirm a final result.

Davies et al. (2006) used aerobic degradation of AO7 (127 mg/l concentration) in a VFCW. The media for this wetland consisted of a 10 cm gravel bottom layer and a 77 cm sandy-clay soil top layer. It was planted with *Phragmites sp.* and the duration of this study was 48 days in an outside environment. In this study, the results showed the percentage of colour reduction reached 99% and COD reduction reached 93%. The time period (48 days) was not long enough to evaluate the effect of plants and micro-organisms on pollutant reduction. In addition, the results depended on the aerobic condition only and did not show the ability of CWs, according to this condition, to remove the phosphorus and nitrogen compounds. Furthermore, the researchers did not investigate the ability of the constructed wetlands system to reduce the amines compounds, which were released due to dye degradation.

Keskinkan and Lugal Göksu (2007) used 20 glass aquaria (4 as control wetlands without plants) as vertical-flow constructed wetlands, located in Turkey, to treat dye (BB41; 11 mg/l influent concentration) with different hydraulic retention times (3, 6, 9 and 18 days). River sand was used as a media and to hold the plant (*P. australis/ Manchurian wild rice*)

vertically in selected wetlands. The obtained results indicated that the dye reduction percentage rate in HRTs of 9 and 18 days were higher than those at HRTs of 3 and 6 days. Furthermore, they found that the presence of plants plays a major role in dye reduction. The percentage dye reduction rates for control wetland, wetland planted with *M. spicatum* and wetland planted with *C. demersum* were 96%, 94% and 71%, respectively.

Bulc and Ojtrsek (2008) proved that VFCWs offer an optimal solution to treat textile wastewater and meet the environmental legislation. Their VFCW was packed with sand and gravel, and planted with *Phragmites australis*. Three artificially prepared dye-bath wastewaters were examined (Reactive Black 5, Disperse Yellow 211 and Vat Yellow 46). The results showed that the average treatment efficiency of the VFCW was 90% for colour, 84% for COD, 93% for TSS and -331% for NH<sub>4</sub>-N at different hydraulic loads. The time period for this study was five months.

Davies et al. (2009) used two pilot-scale VFCWs, located in Portugal, to treat a 700 mg/l dye AO7 during an eight-day period for the experimental work. One of the pilots was planted with *Phragmites australis* and another one was used as a control, with both using gravel as a media. The obtained results as reduction percentage rates for dye, COD and TOC were 68%, 69% and 67%, respectively. Furthermore, their experimental results indicated that the azo dye (AO7) is a xenobiotic that acts as a chemical stressor agent for the plant which was used in their study.

Ong et al. (2009) used five UFCWs with different plants, aeration and non-aeration to assess the treatment of AO7 and nutrients. According to the study, the reduction efficiencies in organic matter, aromatic amines and NH<sub>4</sub>-N were enhanced with supplementary aeration. In the aerated case, the percentage reductions for COD and NH<sub>4</sub>-

N were 86% and 96% respectively, while in the case of non-aerated, it was 78-82% for the COD and 41-48% for NH<sub>4</sub>-N. In addition, it was found that the percentage reduction for AO7 by the gravel was 0.5% by shaking 100 g gravel with 100 ml AO7, 20 mg/l concentration for 2 days as a contact time. Six weeks was the period of this study and the concentration of AO7 was 50 mg/l.

Ong et al. (2010) used an up-flow constructed wetland (two-parallel laboratory scale) to examine and compare the efficiency treatment of wastewater containing Acid Orange (AO7) at different concentrations and different hydraulic retention times (50 and 100 mg/l and 3 and 6 days, respectively) with the option of supplementary aeration. The constructed wetlands were located indoors under a temperature of  $23\pm3$  °C during the period from October 2007 to September 2009. The size of the wetlands was 70 cm in height and 18 cm in diameter, the media consisted of 5 mm glass beads at the bottom with a depth of 6 cm and gravel (5.7 mm in diameter) on top of the glass. At 30 cm below the bed surface, there were two areas air inlets as a supplementary aeration in the aerated constructed wetland and the *Phragmites australis* was selected as the plant. The results showed that the organic matters, aromatic amine and NH<sub>4</sub>-N were removed with the aerated wetland more effectively than with the non-aerated wetland, while the colour and NO<sub>3</sub>-N reduction through the non-aerated were better than through the aerated wetland.

Kabra et al. (2013) used four reactors, located in India, as VFCWs, to treat real textile dyes. Every reactor had two units (A and B); unit A was a plastic container of 40 cm length, 20 cm height and 30 cm width. The solution was pumped from unit B (used as a tank for dye solution to be treated) to unit A, then percolated through the media of unit A and drained into unit B through the porous base. The media consisted of coconut shavings as a bottom layer and soil with bacteria as a top layer, and was planted with *G. pulchella*.

A 10-ml sample was collected from all reactors every 2 h during the 72-h total cycle period. The obtained results from this experimental work were 70% for COD, 74% for TOC and 70% for BOD as reduction percentage rates.

Khandare et al. (2013) treated real textile wastewater containing various dyes by verticalflow constructed wetlands systems, located in India. These systems consisted of four different reactors planted with *Portulaca grandifora* and the media was fine gravel as a bottom layer of 8 cm depth followed by a 2-cm layer of coconut shavings above which another of 5 cm soil with bacteria was placed. The obtained results from this experimental work were 59% COD, 38% BOD, 37% TOC, 41% turbidity, 71% TDS and 60% TSS as reduction percentage rates.

Yalcuk and Dogdu (2014) used a VFCW to treat Acid Yellow 2G E107 Dye-containing wastewater. The constructed wetland consisted of three filter vertical wetlands, one filled with fine gravel, sand and zeolite as a control and the others planted with Cannaidica and Typha Angustifolia, respectively. The initial dye concentration was 259 mg/l, the loading rate was 0.075kg/m<sup>3</sup>/day with a flow rate of 1.2 l/day and the retention time was 3.75 days. The period of operation was three months. The results showed that the average colour reduction percentages were 87% for the control and 98% for the other wetlands. For the NH<sub>4</sub>-N, the average percentage reductions were 43%, 61% and 46% for the control, *Canna* and *Typha*, respectively. The average percentage reductions of PO<sub>4</sub>-P for the control, *Canna* and *Typha* were 84%, 87% and 88% respectively. For COD, the average percentage reductions were 58%, 61% and 64% respectively. The total time of operation was 90 days including the planning and acclimation periods.

Shehzadi et al. (2014) used endophytic bacteria with VFCWs to enhance the textile effluent degradation and plant growth. The constructed wetlands consisted of four

different vertical-flow reactors providing different treatments: reactor with tap water and plant, reactor with textile effluent without plant, reactor with textile effluent and plant and the final one with textile effluent and plant, and endophytic bacteria. The plant used was *T. domingensis*. The results showed that the partnership between the plant and the endophyte could be used to enhance the VFCWs ability to degrade textile wastewater.

Nilratnisakorn et al. (2009) used the plant, *Typha angustifolia Linn*, with a VFCW operating from bottom to top, (HRT= 15 days with the flow rate 0.02 m<sup>3</sup>/day), in two pilot scale experiments in a greenhouse. The media consisted of washed gravel (particle size 10-50 mm) as a bottom layer and washed soil (particle size 10-30 mm) as a top layer. The obtained results indicated that the VFCWs achieved good reduction for colour, COD and TDS, at 49%, 60% and 86%, respectively.

### 2.9 Concluding Remarks

The contribution to performance made by plants in the reduction of COD depends on the type of dye itself (basic or acid; basic dye has two azo atoms while acid dye has three azo atoms), and the structure (some dyes have less or more carbon; carbon is very important for the denitrification process and micro-organisms used it as an energy source) and size of the dye molecule. There was no significant (p>0.05) effect of plants on dye BR46 reduction. The influent was  $6.9\pm1.33$  mg/l and the reduction percentage rate was 96% and 97% for unplanted and planted wetlands, respectively. There was a significant (p<0.05) effect on dye reduction in the case of dye AB113; the influent was  $6.6\pm1.82$  mg/l and the reduction percentage rates were 71% and 80% for unplanted and planted wetlands, respectively.

### **3. CHAPTER THREE: MATERIALS AND METHODS**

### 3.1 Overview

This chapter presents a brief description of the system design, construction and analysis. Section 3.2 describes the experimental set-up, while the sub-sections cover wetland design, media composition, dye details and the ingredient composition of the fertiliser used. Section 3.3 presents operational conditions and other processes. In addition, the sub-sections describe operational conditions for the experimental constructed wetland set-up treating two azo textile dyes only and artificial wastewater containing the two azo textile dyes. The limitations of the experimental research and the concluding remarks are presented in sections 3.4 and 3.5, respectively.

### **3.2 Experimental Set-up**

### **3.2.1 Site Description**

This study was conducted between 1 May 2015 and 31 May 2017. The first month may be viewed as the start-up period. The first period (phase one) between 1 June 2015 and 31 May 2016 was for treatment of azo dyes only, while the second period (phase two) between 1 June 2016 and 31 May 2017 was for treatment of artificial wastewater containing two azo dyes. An experimental constructed wetland rig (Figure 3.1) treating textile wastewater, was constructed and operated within a greenhouse located on the top of the Newton Building at Salford University, England, UK. The experimental rig was designed to assess the system performance by simulating processes occurring within fullscale constructed wetlands. The rig comprised of 21 vertical-flow wetland filters, allowing wastewater to drain vertically, enhancing aerobic biodegradation of organic matter and nitrogen (Fuchs, 2009). The rig was operated in the greenhouse under natural environmental conditions to assess seasonal changes. The experimental wetland filters were located randomly in the experimental rig to minimize random impacts of parameters such as sunlight direction and temperature differences on the wetland performance.



Figure 3.1 Experimental vertical-flow wetland rig located in greenhouse, Newton Building (May 2015)

The impact of operational parameters such as contact time, resting time and hydraulic loading rate on dye reduction were assessed. Contact time is defined as the duration for

which the wastewater is in contact with the wetland filter content. In comparison, resting time is the duration when the wetland filter is empty (i.e. no wastewater input).

### **3.2.2 Wetland Design and Media Composition**

Round, black plastic drainage pipes were used to construct the vertical-flow wetlands (Figure 3.1). All 21 wetlands were designed according to the following dimensions: height of 100 cm and inner diameter of 10 cm. One pipe was filled with water, another one was filled to a depth of 90 cm with unwashed gravel and the other wetlands were filled to a depth of 90 cm with gravel. Two different layers of gravel were used (as a filter media) varying in diameter. The diameters of the gravel were: large gravel, as a bottom layer to prevent the clogging of the outlet; and pea gravel, located at the top layer. These types of gravel are commonly available and suitable for constructed wetlands (Crites et al., 2014).

The main characteristics for the vertical-flow constructed wetlands system, used in the experimental work, were as follows: the total number of wetlands was 21 filters with eight replicated filters, the weight of plant (*Phragmites australis*) was 125 g for each selected wetland. These plants were bought from a local source (Botanica Plant Nursery, Chantry Farm, Campsea Ashe, Woodbridge, Suffolk). The total height for every filter was 100 cm with total available volume for the influent of 2.8 l. The media consisted of gravel of two different sizes: gravel with diameter 10-20 mm and height 20 cm as a bottom layer to prevent clogging, and pea gravel with diameter 5-10 mm and height 70 cm as a top player. The outlet valves were located at the centre of the bottom plate of each wetland with 1 cm internal diameter vinyl tubing, and were used for the regulation of flow and sampling. Selected wetlands were planted with *Phragmites australis* and

replicated to reduce the variability in experimental results and help to increase both generality and realism of the experimental data findings (Fréchette and Schotter, 2015).

### 3.2.3 Phragmites Australis

The presence of *Phragmites australis* in VFCWs has a significant impact on promotion of sedimentation of suspended solids and prevention of erosion, by decreasing the rate of water flow through increasing the length per surface area of the hydraulic pathways through the wetland (Lee and Scholz, 2007), and on the reduction of organic matters, aromatic amines and NH<sub>4</sub>-N (Ong et al., 2011). The growth cycle of *Phragmites australis* traditionally completes between May and September in Britain (Haslam, 1972). Ferreira et al. (2014) detected normal growth for *Phragmites australis* with absence of toxic signs or depletion of leaf nitrogen content, when they used vertical-flow wetlands to treat an effluent comprising Diazo dye (DR81).

When temperatures begin to decrease in the winter period, plants exhibit yellowing as the part above ground dies and is inactivated. It is common practice, in constructed wetland systems, that the above-ground dried plants are cut at the level of 10-15 cm height only (Stefanakis et al., 2014). The rhizomes were bought in mid-April, planted in the selected wetlands and fertilized regularly. Water levels were dropped to encourage deep rooting and penetration of the gravel matrix by the rhizomes (Brix and Arias, 2005).

### **3.2.4 Dye Details**

The pollutant features of textile wastewater differ widely among various organic substances, such as dyes, detergents and starches, causing chemical and biological changes which eat up dissolved oxygen from the receiving stream and destroy aquatic life' (Robinson et al., 2002; Deniz and Karaman, 2011; Shirzad-Siboni et al., 2014). Azo dyes AB113 and BR46 were selected as an artificial model because they are commercial dyes and used extensively in the textile industry, (Chung et al., 1992; Riu et al., 1997; Pervez et al., 1999; Olgun and Atar, 2009; Ong et al., 2010; Deniz and Karaman, 2011; Deniz and Saygideger, 2011; Yeddou et al., 2012), which discharges a large quantity of coloured effluents into the water resources and these effluents contain toxic and carcinogenic materials (Chung, 2000; El Qada et al., 2006; Mohan et al., 2007). AB113 is an acid dye with 681.65 g/mol. as a molecular weight and BR46 is a basic dye with 401.3 g/mol. as a molecular weight (Table 3.1). Lade et al. (2012) and Holkar et al. (2014) stated there is a higher degradation rate for dyes with a molecular weight of less than 500 g/mol. and a lower degradation rate for dyes with a molecular weight higher than 500 g/mol. Acid dye is defined as a negatively charged dye at a chemical level, which contains one or more acidic groups such as a sulphonic group (Akbari et al., 2002; Martínez-Huitle and Brillas, 2009). Basic dye is defined as a positively charged stain at a chemical level (Martínez-Huitle and Brillas, 2009; Brillas and Martínez-Huitle, 2015) which therefore reacts well with negatively charged materials (Sun and Yang, 2003).

These dyes were tested at two different concentrations (target concentrations of 7 mg/l and 215 mg/l) for different contact times (approximately 48 h and 94 h) to assess their influence on the performance of vertical-flow constructed wetlands. The details for each dye can be found in Table 3.1. The textile dye BR46 had a wavelength for maximum

absorbance ( $\lambda_{max}$ ) of 530 nm (Khataee, 2009) and was obtained from Dystar (Am Prime Park, Ruanheim, Germany). The dye AB113 had a  $\lambda_{max}$  of 566 nm (Shirzad-Siboni et al., 2014) and was obtained from Sigma Aldrich (The Old Brickyard, New Road Gillingham, Dorset, United Kingdom) and both of them were used without further purification.

The wavelength for the maximum absorbance of a mixture of both the dyes was determined experimentally by using a Biowave II (WPA) Spectrophotometer. Firstly, the  $\lambda_{max}$  of the mixture of the dyes was determined by scanning the absorption of different mixture dye concentrations between 200 and 800 nm wavelength. The  $\lambda_{max}$  for the mixture dye was found to be 511 nm. The dye stock solutions were prepared by dissolving weighted amounts of each dye (10g) in 1000 ml distilled water and the experimental solutions were obtained by diluting the stock solution to the required concentrations and then mixing.

Table 3.1 Details of azo textile dyes used in the experimental constructed wetlands								
Dye	BR46	AB113						
Molecular weight	401.3	681.65						
(g/mol.)								
Molecular	$C_{18}H_{21}BrN_6$	$C_{32}H_{21}N_5Na_2O_6S_2$						
formula								
Source	Dystar	Sigma Aldrich						
CASRN	12221-69-1	3351-05-1						
Purity of dye (%)	70–80	Approximately 50						
Chemical		NaO <sub>3</sub> S						
structure	CH <sub>3</sub>							
		N-N-N-N-NHC <sub>6</sub> H <sub>5</sub>						
	N CH <sub>3</sub> C <sub>6</sub> H							
	CH <sub>3</sub> Br							

Note: mol., mole; CASRN, chemical abstracts service registry number; BR, basic red; AB, acid blue; C, Carbon; H, hydrogen; N, nitrogen; Na, sodium; O, oxygen; S, sulphur; Br, barium.

### **3.2.5 Aquarium Plant Nutrient**

The fertilizer TNC Complete, which is an aquatic plant nutrient bought from TNC Limited (Spotland Bridge Mill, Mellor Street, Rochdale, United Kingdom), was used in the experimental work as a nutrient for the plant. The corresponding ingredient composition was as follows: phosphorus (0.2%), nitrogen (1.5%), iron (0.08%), manganese (0.018%), potassium (5%), magnesium (0.08%), copper (0.002%), molybdenum (0.001%), boron (0.01%) and zinc (0.01%). TNC Complete also provides ethylene diamine tetra acetic acid (EDTA), which is used as a source for copper, iron, manganese and zinc. The fertilizer was used at a concentration of 1 ml of fertilizer in 10 l of tap water.

### 3.2.6 Artificial Wastewater Compositions

At the second phase of experimental work in this study, vertical-flow constructed wetlands were operated to treat artificial wastewater containing two azo textile dyes. All the compositions of artificial wastewater (Wießner et al., 2005; Ong et al., 2009) were purchased from Scientific Laboratory Supplies (Wilford Industrial Estate, Wilford, Nottingham, United Kingdom). Details of each composition including its concentration, used in this experimental work and others, are shown in Table 3.2.

experimental work						
Material	Chemical	Molecula	CAS	Purity of	Concentratio	
	structure	r weight	number	dye (%)	n (mg/l)	
		(g/mol.)		• • •		
Sodium acetate	CH <sub>3</sub> COONa	82.03	127-09-3	≥99	107.1	
anhydrous pure						
Sodium benzoate	C <sub>6</sub> H <sub>5</sub> COONa	144.11	532-32-1	≥99	204.9	
Ammonium nitrate	NH <sub>4</sub> NO <sub>3</sub>	80.04	6484-52-2	≥99	76.1	
pure						
Sodium chloride	NaCl	58.44	7647-14-5	≥99	7.0	
pure						
Magnesium chloride	MgCl <sub>2</sub> ·6H <sub>2</sub> O	203.3	7791-18-6	<u>≥</u> 99	3.4	
hexahydrate						
Calcium chloride	CaCl <sub>2</sub> ·2H <sub>2</sub> O	147.01	10035-04-8	<u>≥</u> 99	4.0	
dehydrate						
Potassium	K <sub>2</sub> HPO <sub>4</sub> ·3H <sub>2</sub> O	228.22	16788-57-1	<u>≥</u> 99	36.7	
phosphate dibasic						
trihydrate						

Table 3.2 Details of artificial wastewater compositions used in the second phase of experimental work

Note: mol., mole; CAS, chemical abstracts service; C, Carbon; H, hydrogen; O, oxygen; Na, sodium; N, nitrogen; Mg, Magnesium; Cl, chloride; K, potassium; P, phosphor.

### **3.3 Environmental Conditions**

### **3.3.1 Operational Conditions**

The wetland system was designed to operate in batch flow mode to avoid expenses such as pumping and automatic control costs. Wastewater was poured directly into the wetland filter from the top. The wastewater stayed within the wetland for the duration of the contact time. The treated wastewater was released from the wetland through an outlet pipe located in the centre of the wetland bottom. The duration for which the wetland filter was empty (i.e. no wastewater input) was the resting time. The packing order of the experimental constructed wetland set-up treating dye only and an outline of the application of the simplified statistical wetland filter set-up design to assess the impact of individual key variables are shown in Tables 3.3 and 3.4, respectively. The packing order of the experimental constructed wetland set-up treating artificial wastewater containing two dyes and the impact of its individual key variables are shown in Tables 3.5 and 3.6, respectively.

### 3.3.1.1 Operational Conditions for Experimental Constructed Wetland Set-Up

### **Treating Azo Textile Dyes Only**

Wetland 1 was deliberately left empty (without media and plants). The mean tap water values for dissolved oxygen (DO), pH, electrical conductivity (EC), redox potential, total suspended solids (TSS) and turbidity were 10.1 mg/l, 7.6,  $85.1 \, \mu$ S/cm,  $-32 \, m$ V, 1 mg/l and 1.3 NTU, respectively. Wetland 2 was filled with unwashed gravel, while Wetland 3 comprised washed gravel and tap water as shown in Table 3.3. Wetlands 4, 8, 9, 10 and 11 were filled with the washed gravel as a media and planted with the plant (*Phragmites* australis), except Wetland 4, which was used as a control, the rhizomes of this plant were washed free of sediments and planted in the selected gravel-filled wetlands. Wetlands 8, 9 and 10, 11 were replicated, respectively. The contact time for Wetlands 8 and 9 was 94 h and the resting time was 2 h, while for Wetlands 10 and 11, the contact time was 48 h and the resting time was 48 h. The replicated wetlands are important to ensure accurate values of their parameters. The purpose of the different contact and resting times is to assess the ability of the wetlands to remove the dye under different design conditions and compare the results with Wetland 4 to assess the importance of the presence of the plant on the dye reduction and the other parameters. The influent was tap water mixed with BR46 ( $6.9\pm1.33$  mg/l). The same previously explained set-up was used for the Wetlands 5, 6, 7, 12 and 13, but with the dye AB113 at a concentration of about  $6.6 \pm 1.82$  mg/l.

For the high concentration of BR46 ( $209.3\pm4.48$ ), Wetlands 14, 15, 18 and 19 were filled with the same media and the plant. Wetlands 14 and 15 were replicated with 48 h as the

contact time and 48 h as the resting time, while Wetlands 18 and 19 were replicated with 96 h as the contact time and a resting time of 96 h. Wetlands 16, 17, 20 and 21 are similar to the previous wetlands, just described, but the influent dye is AB113 at a concentration of about  $221.5\pm31.86$  mg/l.

Table 3.3 Packing order of the experimental constructed wetland set-up treating two azo textile dyes only. Samples were taken at the end of each contact time period, just before the start of the resting time period between 1 May 2015 and 31 May 2016.

Wetland	Media	Plants		Dye		Resting	Contact
number			Туре	Mean	SD	time (h)	time (h)
				(mg/l)	(mg/l)		
1	Only Water	No	None	N/A	N/A	2	94
2	Unwashed	No	None	N/A	N/A	2	94
	gravel						
3	Washed gravel	No	None	N/A	N/A	2	94
4	Washed gravel	No	BR46	6.9	1.33	2	94
5	Washed gravel	No	AB113	6.6	1.82	2	94
6,7	Washed gravel	Yes	AB113	6.6	1.82	2	94
8,9	Washed gravel	Yes	BR46	6.9	1.33	2	94
10,11	Washed gravel	Yes	BR46	6.9	1.33	48	48
12,13	Washed gravel	Yes	AB113	6.6	1.82	48	48
14,15	Washed gravel	Yes	BR46	209.3	4.48	48	48
16,17	Washed gravel	Yes	AB113	221.5	31.86	48	48
18,19	Washed gravel	Yes	BR46	209.3	4.48	96	96
20.21	Washed gravel	Yes	AB113	221.5	31.86	96	96

Note: SD; standard deviation, N/A; not applicable, BR, basic red; AB, acid blue.

Comparison of two	wetland systems with each	Impact to be assessed
	other	
First wetland with	Second wetland with	
number	number	
1	2	Unwashed gravel
2	3	Washed gravel
3	4	BR46
3	5	AB113
4	5	Difference between BR46 and AB113
5	6,7	Phragmites australis
6,7	8,9	Difference between BR46 and AB113
8,9	10,11	Decrease in contact time
		(or increase in resting time)
10,11	12,13	Difference between BR46 and AB113
10,11	14,15	Increased BR46 concentration
12,13	16,17	Increased AB113 concentration
14,15	18,19	Increased contact and resting times
16,17	20,21	Increased contact and resting times
14,15	16,17	Difference between BR46 and AB113
18,19	20,21	Difference between BR46 and AB113

Table 3.4 Application of the simplified statistical wetland filter set-up design (Table 3.3) to assess the impact of individual key variables.

Note: BR46, basic red; AB, acid blue

# **3.3.1.2 Operational Conditions for Experimental Constructed Wetland Set-Up Treating Artificial Wastewater Containing Two Azo Textile Dyes**

All wetlands in this phase have the same media, which is washed gravel (Table 3.5). The influent for Wetlands 1, 5 and 7 was artificial wastewater containing dye BR46 and planted with *Phragmites australis* except for Wetland 1, which was used as a control. The resting and contact times for Wetlands 1 and 5 were 2 h and 94 h, respectively, while for Wetland 7 both resting and contact times were 48 h. The concentration for the influent dye was 6.2 mg/l. Wetlands 2, 3 and 9 followed the same scenario as the previous wetlands but the artificial wastewater contained dye AB113 with an influent concentration 7.5 mg/l. Wetlands 4, 6, 8 and 10 followed the same scenario as the
previous wetlands (where Wetlands 8 and 10 follow the same scenario as 7 or 9) but the influent was the artificial wastewater containing the mixture of both of the previous dyes (BR46 and AB113).

Wetlands 11 and 15 were the same as Wetlands 5 and 7 but the dye (BR46) concentration was approximately 206 mg/l with 48 h resting and contact times for Wetland 11 and 96 h resting and contact times for Wetland 15. Wetlands 13 and 17 followed the same scenario as Wetlands 11 and 15 but the influent artificial wastewater contained dye AB113 with an approximate concentration of 207 mg/l. Wetlands 12 and 14 followed the same scenario as the previous Wetlands 11 and 13 but the influent artificial wastewater was the mixture of both of the dyes (BR46 and AB113). The same influent artificial wastewater containing the mixture of the two dyes was used for Wetlands 16 and 18 but with 96 h for both the resting and contact times.

Table 3.5 Packing order of the experimental constructed wetland set-up treating artificial wastewater containing two azo textile dyes. Samples were taken at the end of each contact time period, just before the start of the resting time period between 1 June 2016 and 31 May 2017.

Wetland	Wetland	Plants		Dye		Resting	Contact
number <sup>(a)</sup>	number <sup>(b)</sup>		Type Mean SD		time (h)	time (h)	
				(mg/l)			
4	1	No	BR46	6.15	0.75	2	94
5	2	No	AB113	7.50	1.66	2	94
6	3	Yes	AB113	7.50	1.66	2	94
7	4	Yes	Mix <sup>(c)</sup>			2	94
8	5	Yes	BR46	6.15	0.75	2	94
10	6	Yes	Mix			2	94
11	7	Yes	BR46	6.15	0.75	48	48
12	8	Yes	Mix			48	48
13	9	Yes	AB113	7.50	1.66	48	48
14	10	Yes	Mix			48	48
15	11	Yes	BR46	206	9.60	48	48
16	12	Yes	Mix			48	48
17	13	Yes	AB113	207	13.70	48	48
18	14	Yes	Mix			48	48
19	15	Yes	BR46	206	9.60	96	96
20	16	Yes	Mix			96	96
21	17	Yes	AB113	207	13.70	96	96
22	18	Yes	Mix			96	96

Note: <sup>(a)</sup>, wetland used to treat azo textile dye only; <sup>(b)</sup>, wetland used to treat artificial wastewater containing two azo textile dye(s); SD, standard deviation; <sup>(c)</sup>, mixture of BR46 and AB113; BR, basic red; AB, acid blue.

Comparison of two we each other	etland systems with	Impact to be assessed
First wetland with	Second wetland	-
number	with number	
1	2	Difference between BR46 and AB113
1	5	Phragmites australis on BR46
2	3	Phragmites australis on AB113
4	6	Mixing dyes (low concentration)
5	7	Decrease in contact time (or increase in resting time on BR46)
3	9	Decrease in contact time (or increase in resting time on AB113)
7	9	Difference between BR46 and AB113
8	10	Mixing dyes (low concentration)
7	11	Increased BR46 concentration
9	13	Increased AB113 concentration
12	14	Mixing dyes (high concentration)
11	15	Increased contact and resting times
13	17	Increased contact and resting times
11	13	Difference between BR46 and AB113
15	17	Difference between BR46 and AB113
16	18	Mixing dyes (high concentration)

Table 3.6 Application of the simplified statistical wetland filter set-up design (Table 3.5) to assess the impact of individual key variables.

Note: BR46, basic red; AB, acid blue.

It is clear that the general rule of a loading period followed by a resting period is used by all researchers/engineers when dealing with this type of constructed wetlands. Applying intervals between the wastewater loadings and allowing the wastewater to drain vertically enhances the aerobic biodegradation of several substances (organic matter, nitrogen, etc.). However, it is also obvious that there is no widely accepted regime concerning the frequency and duration of the loading and resting periods' (Stefanakis et al., 2014).

Two different concentrations for the dyes were studied to find out the effect of concentration of the textile dye effluent on the percentage of dye removed and the other parameters. The low concentration enabled study of the behaviour of the constructed wetland system with different wetlands (filled by media with and without plant), and different contact and resting times. The high concentration enabled assessment of the

efficiency of the vertical wetland in treating the effluent of the textile wastewater and assessment of the effect of long-term performance of the constructed wetlands system in treating the textile wastewater (Robinson et al., 2002; Davies et al., 2006; Tee et al., 2015).

### **3.3.2 Measurement of Water Quality Parameters**

Samples were taken regularly at the end of each contact time period, just before the start of the resting time period between 1 May 2015 and 31 May 2017 as shown in Table 3.3 and Table 3.5, and brought to the laboratory (G19) in the Cockcroft Building for testing. The samples were taken at the same time (from 10:00 to 11:00 am) to ensure that the different conditions which occur throughout the day, such as varying pH, would not affect the results. The laboratory tests measured dye concentration, colour, COD, PO<sub>4</sub>-P, NO<sub>3</sub>-N, NH<sub>4</sub>-N, EC, DO, redox, pH, TSS, turbidity and amines, according to the procedures in the standard method (APHA, 1995), unless stated otherwise. Both influent and effluent were analysed. The effluent was obtained from the bottom of each wetland filter. The inflow wastewater was freshly prepared before it was poured into the wetland from the top. These parameters were used to give an indication of the vertical-flow constructed wetland's effectiveness in treating the textile wastewater and water quality.

### **3.3.2.1 Hydraulic Loading Rate**

Hydraulic loading rate (HLR) was calculated according to Equation 3.1 (Tyler, 2001):

$$HLR = \frac{Q}{A}$$
 Equation 3.1

where HLR is hydraulic loading rate (m/d), Q is flow rate ( $m^3/d$ ), and A is surface area of constructed wetland ( $m^2$ ).

### 3.3.2.2 Mass Loading Rate (MLR)

Mass loading rate (MLR) was calculated according to Equation 3.2 (Timmons et al., 2002): MLR=C<sub>i,e</sub>×HLR Equation 3.2

where MLR is mass loading rate  $(mg.m^{-2}.d^{-1})$ , and  $C_{i,e}$  is influent or effluent concentration (mg/l).

# **3.3.2.3 Percentage Reduction Rate**

The percentage reduction rate of any selected variable was calculated according to

Equation 3.3 (Kadlec and Wallace, 2008).

$$R = \frac{C_{e} \cdot C_{i}}{C_{i}} \times 100$$
 Equation 3.3

where R percentage is reduction rate (%),  $C_e$  is effluent value (mg/l, nm, Pt Co.), and  $C_i$  is influent value (mg/l, nm, Pt Co.).

# 3.3.2.4 Analytical Methods and Equipment

The methods and equipment used to measure different parameters during the experimental work are presented below.

## **3.3.2.4.1 Measurement of Physical Parameters**

The physical parameters included dye concentration, colour, TSS, DO, Turbidity, pH, redox potential, EC and temperature. Dye concentration, colour and TSS were measured using a Spectrophotometer Hach Lange DR2800 (www.hach.com). Dye concentration

was measured through the selective wavelength at maximum absorbance for each dye. Colour was measured with a unit Pt Co scale. TSS was measured with mg/l as a unit. Samples were filtered using Whatman Grade 1 Qualitative Filter Paper (Standard Grade, Circle, 320 mm), which was bought from Scientific Laboratory Supplies (Wilford Industrial Estate, Wilford, Nottingham, United Kingdom).

The dissolved oxygen (DO) was measured using a hand-held Hach Lange HQ30d Flexi Meter (Pacific Way, Salford, United Kingdom); the unit for the measuring of DO is mg/l. The DO was measured promptly after taking the specimens, to limit, as far as possible, the time for which samples were in contact with air. Turbidity was measured using a Turbicheck Portable Turbidity Meter (Lovibond Water Testing, Tintometer Group, Division Street, Chicago, IL, USA); the unit for the measuring of turbidity is NTU.

The pH and redox potential for all samples were measured by using a hand-held WTW VARIO pH Meter (Wissenschaftlich-Technische Werkstätten, Weilheim, Germany); the unit for the measuring of redox is (mV) while pH is without units, the meter is calibrated with standard buffer solution of pH 4.0, 7.0 and 9.0 every two weeks or whenever required. pH is a value of the amount of free hydrogen ion in water. A value pH of 7 is deemed to be neutral and the range of pH for water quality standards is from 6.5 to 9 (Boyd and Gautier, 2000). For any increase in pH value above the neutral value, alkalinity will increase, while for any decrease below the neutral value, acidity will increase. The measuring of pH value is very important due to its effect on the solubility of many toxic and nutritive chemicals. When the acidity increases, most metals become more toxic and more water-soluble. For example, a slight increase in pH value leads to ammonia becoming more toxic (Thurston et al., 1981). In contrast, the toxicity of sulphides and cyanides increases when the value of pH decreases. Electrical conductivity (EC) was

measured using a hand-held Mettler Toledo Education Line Conductivity Meter (Boston Road, Leicester, United Kingdom). The unit for the measurement of EC is micro Siemens per centimetre ( $\mu$ S/cm). Although EC itself is not an aquatic health or human concern, its value gives an indication if there are any other water quality problems. A suddenly increase in EC value indicates that there is a source of dissolved ions in the wetland filter (Kumar and Chopra, 2012). The temperature at the site was recorded each day using a thermometer placed alongside the constructed wetlands.

These hand-held easy to use, robust and waterproof instruments measure, with low costs, the most important parameters for wastewater monitoring. The meters come complete with sensors, and calibration and maintenance solutions for measurement.

### **3.3.2.4.2** Measurements of Chemical Parameters

The chemical parameters measured included COD, NH<sub>4</sub>-N, NO<sub>3</sub>-N, PO<sub>4</sub>-P and amines. A Spectrophotometer Hach Lange DR2800 (www.hach.com) was applied for the water quality analysis, for variables including chemical oxygen demand (COD), orthophosphate-phosphorus (PO<sub>4</sub>-P), nitrate-nitrogen (NO<sub>3</sub>-N) and ammonium-nitrogen (NH<sub>4</sub>-N) with milligrams per litre (mg/l) as a unit. The aromatic amines were measured as an absorbance using a WPA Biowave II UV/Visible Spectrophotometer (Cambourne, Cambridge, United Kingdom), via the wavelength for the selected absorbance for every type of aromatic amines. Also, samples were filtered using Whatman Grade 1 Qualitative Filter Paper (Standard Grade, Circle, 320 mm).

Chemical oxygen demand (COD) does not differentiate between biologically available and inert organic matter and it is a measure of the total amount of oxygen required to oxidize all organic material into water and carbon dioxide. COD was determined with a Palintest Tube tests system. Palintest Tube tests are integrated with the Palintest heater and photometer system to provide a complete system for COD measurement. COD analysis was performed with three Palintest Tube tests with product codes; LCK 114 (range from 150-1000 mg/l), LCI 400 (range from 0-1000 mg/l) and LCK 049 (range from 50-300 mg/l).

Nutrients such as phosphorus and nitrogen are very important for the growth of plants and algae. Aquatic life in a stream or river depends upon these photo-synthesizers, which exist at surface water in low levels. Any extremes in nutrients concentration lead to a rapid growth for algae and aquatic plants, however, this rapid growth leads to a reduction in dissolved oxygen concentration, depriving invertebrates and fish of available oxygen in the water. Nutrients were determined by automated precision colorimetry methods using a Palintest Tube test with product code LCK 339 for nitrate (range from 0.23-13.5 mg/l), LCK 305 for ammonia (range from 1-12 mg/l) and LCK 049 (range from 1.6-30 mg/l) for ortho-phosphate-phosphorus (Allen et al., 1974). The coloured complexes formed were measured spectrometrically at 540, 655 and 882 nm, respectively. All the tube tests were bought from Hach Lange Ltd. (Pacific Way, Salford, UK).

## **3.3.2.4.3 Statistical Analysis Methods**

After data collection, data were subjected to a normality test before validation and subsequent analysis. Microsoft Excel (www.microsoft.com) was used for general data analysis unless stated otherwise. The IBM SPSS version 23 (IBM Corp., 2013) package was applied to perform the correlation analysis between variables and to assess statistical differences between treatments. In order to investigate statistically significant differences, the Shapiro-Wilk's test (Shapiro and Wilk, 1965; Razali and Wah, 2011) was

used to investigate data normality. If  $p \ge 0.05$ , data are normally distributed; if p < 0.05, data are not normally distributed. A one-way analysis of variance (ANOVA) test using Statistical Package for the Social Science software was applied to analyse normal distributed data ( $p \ge 0.05$ ), while the Mann-Whitney test was used to analyse non-normal data (p < 0.05) (Stoline, 1981; Kasuya, 2001). These ANOVA and Mann-Whitney tests were computed using IBM SPSS Statistics Version 23 and used to compare means between different treatments, such as the ones highlighted in Tables 3.4 and 3.6. Significant findings have been outlined and discussed. Surprising and/or important insignificant findings have occasionally been highlighted as well.

### **3.4 Experimental Research Limitations**

Highlights and discussions of limitations in this experimental design and methods are presented in this section.

In spite of the reality that the experimental constructed wetlands used are very small in comparison to large-scale constructed wetland systems used in industries, previous studies based on the same design idea as this one proved that the results obtained were applicable in field scale and therefore accepted by the scientific community (Scholz, 2004; Zhao et al., 2004; Sani et al., 2013; Al-Isawi et al., 2015b; Almuktar et al., 2015).

The experimental constructed wetlands in this study, which was carried out in a greenhouse under semi-controlled conditions, may not correspond with other wetlands operated in real field scale because of the variability in environmental factors. However, designers can use the results obtained from this research as a guide in designing and upscaling field-scale wetlands operated in different climates. In additional, the use and operation of constructed wetland systems in the field require larger land areas and natural

energy inputs to achieve to self-maintaining treatment constructed wetlands systems, which will provide a suitable environment for many more types of micro-organisms as a result of the diversity of the micro-environment in a wetland. These facts lead to the conclusion that using small-scale constructed wetlands, as operated in this research, cannot represent the actual requirement of the large land area involved in actual fieldscale operation. Moreover, because of the large surface area required for a constructed wetland in the field, many problems are expected to occur. For example, a larger area could be home to many types of animals, affecting the treatment performance and causing health problems for humans, while in these small-scale constructed wetlands, as used in this study, the effects of the presence of animals are not considered.

In the first phase of the experimental work (treating azo textile dyes only), all the planted constructed wetlands were replicated but the control wetlands (unplanted) were not replicated due to the lack of sufficient resources and space to cover the required number of replications needed. Although the obtained results from this experimental work don't mimic the actual set-up for the constructed wetlands with full replication, many publications based on the same idea as this experimental work have been accepted by the scientific community (Al-Isawi et al., 2015a; Sani, 2015; Hussein and Scholz, 2017).

In the second phase of the experimental work (treating artificial wastewater containing two azo textile dyes), one out of each of the replicates used in the first phase, as shown in Table 3.5, was used for the influent of mixed dyes as well as artificial wastewater. This new step depended on the value of significance between each replication.

# **3.5 Concluding Remarks**

The main factor, which has to be considered in case of scaling-up the constructed wetlands is the effluent of the textile industry. If the dye effluent concentration is below 7 mg/l, it will not impact on the performance of VFCW systems. In contrast, a relatively high concentration of 215 mg/l will affect the performance of the system.

# 4. CHAPTER FOUR: RESULTS AND DISCUSSION

# FOR TWO AZO TEXTILE DYES REDUCTION

Most of the results and discussion in this chapter have been published in the paper shown below:

Hussein, A. & Scholz, M. 2017. Dye wastewater treatment by vertical-flow constructed wetlands. *Ecological Engineering*, 101, 28-38, doi: 10.1016/j.ecoleng.2017.01.016. (Appendix A).

### 4.1 Overview

This chapter includes the results and discussion of the dye reduction rates and other physical and chemicals water quality parameters for the period between 1 May 2015 and 31 May 2016. A comparison between wetland filters 1, 2 and 3 is shown in section 4.2, while section 4.3 deals with the test of normality for all variables. The results for plant growth monitoring, redox potential, dissolved oxygen, electrical conductivity, total suspended solids, turbidity, pH, nutrients, dyes and chemical oxygen demand are presented in sections 4.4 to 4.9. Section 4.10 is concerned with the seasonal treatment performance of wetlands in terms of dye reductions. Furthermore, statistical differences between the variables are also presented. Concluding remarks are given in section 4.11.

### 4.2 Comparison of Unwashed Gravel versus Washed Gravel

Many publications have suggested using a washed gravel in wetlands media. For instance, (Kopec, 2007) stated that using washed gravel is very important to prevent clogging. (Reuter et al., 1992) believed the drop in reduction efficiency of phosphorus

has to -38% during the winter period was a result of using unwashed gravel as media. Furthermore, (Rosolen, 2000) stated that sediments in unwashed gravel affect the porosity, which determines hydraulic conductivity. The effluent values of TSS for Wetlands 1, 2 and 3 were 1.0, 1550 and 30 mg/l, respectively. After less than six months, the effluent values became approximately similar, as shown in Figure 4.1, as a result of washing it during the contact time. In the case of turbidity, the effluent values for Wetlands 1, 2 and 3 were 1.88, 870 and 42 NTU, respectively. Within six months, the effluent values were approximately similar, as shown in Figure 4.2, as result of the same previous reason. For the electrical conductivity, the effluent values for Wetlands 1, 2 and 3 were 101, 328 and 151  $\mu$ S/cm, respectively. After less than six months, the effluent values were approximately similar, as shown in Figure 4.3. For the other variables, pH, redox and DO, there were no significant (*p*≥0.05) differences between them.



Figure 4.1 Total suspended solids comparison between Wetlands 1, 2 and 3.



Figure 4.2 Turbidity comparison between Wetlands 1, 2 and 3.



×Wetland (1)  $\bigcirc$ Wetland (2)  $\triangle$ Wetland (3)

Figure 4.3 Electrical conductivity comparison between Wetlands 1, 2 and 3.

### **4.3 Test of Normality**

Tests of normality results for dimensions of *P. australis* and for effluent water quality characteristics regarding general physical and chemical variables are discussed in Appendix B.1.

### 4.4 Plant Growth Monitoring

The presence of *Phragmites australis* in vertical-flow constructed wetlands has a significant impact on the reduction of organic matters, aromatic amines and NH<sub>4</sub>-N (Ong et al., 2011). The growth cycle of *P. australis* traditionally completes between May and September in Britain (Haslam, 1972). (Ferreira et al., 2014) detected normal growth for *Phragmites* with absence of toxic signs or depletion of leaf nitrogen content, when they used vertical-flow wetlands to treat an effluent comprising Diazo

dye (DR81). When temperatures started to decrease in winter (Figure. 4.4), the plants in this experiment began to yellow. The above-ground plant parts died, and were cut and returned to the wetland filters. The authors followed common practice to cut the above-ground plant parts down to between 10 and 15 cm in height according to (Stefanakis et al., 2014).

Based on the experimental results, the most critical observation is that for all considered concentrations and contact times, the plants that were subjected to the dye AB113 revealed more effective growth compared to those that were fed by the dye BR46. Taking into consideration the potential impact of retention time, it has been noted that plants associated with a long contact time showed much better growth than those plants linked to a short one (Table 4.1). These findings support those by (Pagter et al., 2005), who investigated the effect of water stress tolerance of *P. australis* grown in the laboratory by examining effects of different levels of required water. The results showed that a water deficit reduces the leaf biomass per plant and the leaf area.

Regarding the growth of plants, there is no significant ( $p \ge 0.05$ ) difference between experimental wetlands with respect to the length and diameter at the low concentration of the dye AB113, while for the high concentration, there is a significant (p < 0.05) difference for the diameter but no significant ( $p \ge 0.05$ ) difference for the length. In the case of plants grown in the presence of the dye BR46, there is a significant (p < 0.05) difference for both the length and diameter at the low concentration of the dye, while no significant difference ( $p \ge 0.05$ ) for either parameter was noted at the high concentration of the dye. It follows that no generic interpretation of the data can be made for either dye, indicating further research is needed by ecologists.



Figure 4.4 Temporal variation of temperature throughout the experimental period (first phase).

Season	Descriptive Statistics								
-	Minimum	Maximum	Mean	SD					
Summer	13.6	25	22.8	3.25					
Autumn	5	19	11.6	3.09					
Winter	2	24	8.6	5.59					
Spring	13.1	32.8	20.5	5.3					

Table 4.1 The descriptive statistics of the changes in temperature

Note: (a)21/06/2015 to 21/9/2015; (b)22/09/2015 to 20/12/2015; (c)21/12/2015 to 9/03/2016; (d)20/03/2016 to 31/05/2016; BR, basic red; AB, acid blue; SD, standard deviation.

The temperature variations cause changes in microbial activity, which in turn creates changes in microbial-mediated water quality improvement (Fan et al., 2016; Xie et al., 2016). The study of Xie et al. (2016) showed that the bacterial numbers and species responsible for ammonification, nitrification and denitrification varied seasonally, with denitrifying bacteria changing the most and nitrifying bacteria changing the least,

with also higher numbers in the warm seasons (summer and autumn) and lower numbers in the cold seasons (spring and winter).

Characteristics			]	Length	(cm)	Diameter (mm)			
Dye	Wetland	Number	Min.	Max. Mean		Min.	Max.	Mean	
	numbers	of stems			±SD			$\pm SD$	
BR46	8 and 9	40	40	110	86±25.0	2.5	3.8	3.6±0.16	
	10 and 11	28	17	44	30±9.0	1.1	1.8	$1.5\pm0.30$	
	14 and 15	34	14	50	27±13.0	1.0	1.2	$1.1\pm0.07$	
	18 and 19	23	12	34	23±8.1	0.9	1.5	1.2±0.25	
AB113	6 and 7	45	110	141	131±9.5	3.4	3.8	3.6±0.16	
	12 and 13	30	23	41	32±4.5	2.1	2.4	2.2±0.10	
	16 and 17	36	28	46	35±6.0	2.1	2.3	$2.2\pm0.06$	
	20 and 21	28	22	52	39±10.7	1.8	2.5	2.1±0.25	

Table 4.2 Dimensions of *Phragmites australis* (Cav.) Trin. ex Steud. (Common Reed) planted in the experimental wetlands.

Note: BR, basic red; AB, acid blue; Min., minimum; Max., maximum; SD, standard deviation.

# 4.5 Redox Potential and Dissolved Oxygen

Redox potentials of less than -100 mV indicate anaerobic environments, while values greater than 100 mV indicate aerobic environments (Suthersan, 2001). DO is an important function in constructed wetlands as it is essential in aerobic respiration for micro-organisms and it regulates the oxidation-redox potential in wastewater (Boyd, 2000). S Wu et al. (2011) and Hou et al. (2016) reported that the main pathways for oxygen transfer in constructed wetlands such as the system in this research (tidal flow) are: wetland macrophytes release via their roots, contact transfer at the interface of biofilm and atmosphere, and dissolved oxygen associated with influent wastewater. Redox potential values for the effluents of both dyes (BR46 and AB113) in the case of low concentration were in the range between -32 and -6 mV as shown in Table 4.2, and the influent and effluent values during the whole period of the experimental work are shown in Figures 4.5 and 4.6, respectively. In the case of high concentration for both dyes (BR46 and AB113), redox potential values for the effluents were in the range between -32 and 4 mV (Table 4.2) and all values for the influents and effluents are shown in Figures 4.7 and 4.8, respectively. Wetlands with low loading rate (18/19 and 20/21 for BR46 and AB113, respectively) have a lower redox potential when compared to Wetlands 14/15 and 16/17 for BR46 and AB113, respectively, which have high loading rate during the period of experimental work. These results can be linked with the dye reductions which are shown in Table 4.4, where dye reduction rate is highest, when the redox potential of the system is at its most negative, findings confirmed by Singh (2014).



Figure 4.5 Overall variation for redox potential in the influent and effluent for Wetlands 4, 8/9 and 10/11 (low concentration of BR46).



Figure 4.6 Overall variation for redox potential in the influent and effluent for Wetlands 5, 6/7 and 12/13 (low concentration of AB113).



Figure 4.7 Overall variation for redox potential in the influent and effluent for Wetlands 14/15 and 18/19 (high concentration of BR46).



Figure 4.8 Overall variation for redox potential in the influent and effluent for Wetlands 16/17 and 20/21 (high concentration of AB113).

Effluent DO values for BR46 and AB113, in the case of low concentration, were in the range between 5.4 and 8.3 mg/l, as shown in Table 4.2, and were between 6.0 and 7.5 mg/l for the high concentration, as shown in Table 4.2. Furthermore, planted Wetlands 8/9 (Figure 4.9) and 6/7 (Figure 4.10), have the lowest DO values during the whole period of experimental work when compared with unplanted Wetlands 4 and 5 (low resting and high contact times), and planted Wetlands 10/11 and 12/13 (high resting and low contact times), respectively, because of the presence of the plants and the higher period of contact time. In the case of high concentration for both dyes, BR46 and AB113, Wetlands 18/19 and 20/21 (high resting and contact times) have the lower effluent DO values when compared with Wetlands 14/15 and 16/17 (low resting and contact times), as shown in Figure 4.11 and Figure 4.12, respectively. However, there is the possibility that the effluent samples became aerated between taking the samples

and the corresponding measurements. Nevertheless, these findings indicate that dye degradation may have taken place in both aerobic and anaerobic environments.



Figure 4.9 Overall variation for dissolved oxygen in the influent and effluent for Wetlands 4, 8/9 and 10/11 (low concentration of BR46).



Figure 4.10 Overall variation for dissolved oxygen in the influent and effluent for Wetlands 5, 6/7 and 12/13 (low concentration of AB113).



Figure 4.11 Overall variation for dissolved oxygen in the influent and effluent for Wetlands 14/15 and 18/19 (high concentration of BR46).



Figure 4.12 Overall variation for dissolved oxygen in the influent and effluent for Wetlands 16/17 and 20/21 (high concentration of AB113).

### 4.6 Electrical Conductivity, Total Suspended Solids and Turbidity

Electrical conductivity can be used as an indirect measure of the charge (or the ioncarrying species) in the wetland outflow (Islam et al., 2011). EC value can be used as an indicator of other water quality problems. Any sudden increase in EC value indicates that there is a source of dissolved ions in the wetland filter (Kumar and Chopra, 2012). In comparison, all the effluent values for all wetlands in the case of low and high concentrations for both dyes were compliant with the national effluent discharge quality standards set by the Government of Bangladesh, which state that the maximum effluent of EC for inland surface water, public sewer secondary treatment plant and irrigated land is 1200 µS/cm (Ahmed et al., 2002). Furthermore, Sri Lanka central environmental authority (2008) state that the maximum EC discharge on land for irrigation purposes is 2250 µS/cm. For the low concentration of both dyes (BR46 and AB113), as shown in Table 4.2, an inconsiderable increase was noted for planted Wetlands 10/11 and 12/13 (contact time of 48 hours) respectively, when compared to planted Wetlands 8/9 and the unplanted control Wetland 4 as well as planted Wetlands 6/7 and the unplanted control Wetland 5 (contact time of 94 h for each), respectively. For the high concentration, as shown in Table 4.2, in the case of dye AB113, the average value of EC in the effluent of Wetlands 20/21 (high resting and contact times) is the same as for the influent, while for Wetlands 16/17 (low resting and contact times), there is a decrease in the average value, if compared to the influent. In the case of dye BR46, there is a statically insignificant (p < 0.05) increase in the average value for the effluent of Wetlands 14/15 (low resting and contact times;  $389\pm99.73 \mu$ S/cm),

and Wetlands 18/19 (high resting and contact times;  $385\pm83.21 \ \mu\text{S/cm}$ ), if compared to the influent ( $357\pm112.3 \ \mu\text{S/cm}$ ), as shown in Table 4.2. Yalcuk and Dogdu (2014) have reported similar findings.

The measurement of TSS is an important factor for the design of water treatment facilities (Dzurik, 2003). The Clean Water Act (CWA) lists it as a conventional pollutant (Bell et al. 2011). For the dye BR46 in the case of low concentration, there is an increase in the effluent for unplanted Wetland 4 and planted Wetlands 10/11 (high resting and low contact times), while there is no increase in the average value for planted Wetlands 8/9 (low resting and high contact times), if compared with the influent, as shown in Table 4.2.

In the case of the high concentration scenario, good reduction was noted for Wetlands 14/15 and 18/19. The reduction in Wetlands 18/19 (high resting and contact times) is better than that in Wetlands 14/15 (low resting and contact times), as shown in Table 4.2. For the dye AB113 in the case of low concentration, increased values of TSS in the effluent were noted for the unplanted Wetland 5, and planted Wetlands 6/7, and 12/13. Only a slight increase was noted for Wetlands 6/7 (low resting and high contact times), while a higher increase was recorded for Wetlands 12/13 (high resting and low contact times), as shown in Table 4.2. In comparison during the whole period of study, as shown in Figures 4.13, 4.14 and 4.15 for low concentration of BR46, high concentration of BR46 and low concentration of AB113, respectively, all the effluent values for the wetlands were compliant with the national effluent discharge quality standards set by the Government of Bangladesh, which state the maximum effluent of TSS for inland surface water, public sewer secondary treatment plant and irrigated land are 150 mg/l, 500 mg/l and 200 mg/l, respectively (Ahmed et al., 2002). In the case of the high concentration of AB113, the effluent values for Wetlands 16/17 and

20/21 (Table 4.2) were compliant only with the standard for the public sewer secondary treatment plant, which is 500 mg/l and 600 mg/l as stated by the Government of Bangladesh (Ahmed et al., 2002) and the Government of India (1986), respectively. An increased value was noted for Wetlands 16/17 (low resting and contact times) and 20/21 (high resting and contact times). Increased values for Wetlands 20/21 were noted ( $392\pm253.13$  mg/l), if compared to Wetlands 16/17 ( $371.7\pm177.57$  mg/l), as shown in Table 4.2.



Figure 4.13 Overall variation for total suspended solids in the influent and effluent for Wetlands 4, 8/9 and 10/11 (low concentration of BR46).



Figure 4.14 Overall variation for total suspended solids in the influent and effluent for Wetlands 14/15 and 18/19 (high concentration of BR46).



Figure 4.15 Overall variation for total suspended solids in the influent and effluent for Wetlands 5, 6/7 and 12/13 (low concentration of AB113).

Turbidity is often used to interpret the degree of clarity of water. It is a variable often applied as an indicator of the amount of suspended sediments and/or larger micro-organisms in water. A high turbidity of surface water may also indicate elevated concentrations of TSS, reduced algal populations, and potential harm to fish and other aquatic fauna (Postolache et al., 2007). Furthermore, a higher turbidity value can also increase the temperature of surface water as a result of increased absorption of heat from sunlight as well as leading to reduce light penetration, which affects photosynthesis (Håkanson, 2006).

For the dye BR46 (p < 0.05) in the case of the low concentration scenario, reduction was noted for planted Wetlands 8/9 (low resting and high contact times, 3.0±1.33 mg/l), while there was an increase linked to unplanted Wetland 4 (8.8±7.12 mg/l) and planted Wetlands 10/11 (high resting and low contact times, 3.6±1.89 mg/l), if compared to the influent  $(3.5\pm2.89 \text{ mg/l})$ , as shown in Table 4.2. The overall variation is shown in Figure 4.16. In the case of high concentrations, relatively good reduction was noted for Wetlands 14/15 (low resting and contact times;  $6.6\pm2.20$  mg/l) and 18/19(high resting and contact times; 6.8±3.90 mg/l), when compared with the influent  $(16.6\pm3.10 \text{ mg/l})$ , as shown in Table 4.2. The overall variation is shown in Figure 4.17. For the dye AB113 in the case of low concentration (p < 0.05), increased values in the effluent were noted for unplanted Wetland 5 (39.2±27.40 mg/l), planted Wetlands 6/7  $(19.1\pm16.11 \text{ mg/l})$  and planted Wetlands 12/13 ( $46.8\pm32.67 \text{ mg/l}$ ), if compared to the influent (7.4 $\pm$ 5.67 mg/l). A greater increase was recorded for Wetlands 12/13 (high resting and low contact times), if compared to Wetlands 6/7 (low resting and high contact times), as shown in Table 4.2. The overall variation is shown in Figure 4.18. In the case of the high concentration scenario, a reduction was noted for Wetlands 16/17 (low resting and contact times;  $208.3 \pm 196.67$  mg/l; p < 0.05), while there was an increase in the average value for Wetlands 20/21 (high resting and contact times;  $224\pm202.69 \text{ mg/l}$ ;  $p\geq0.05$ ), if compared to the influent ( $213.4\pm157.62 \text{ mg/l}$ ; p<0.05), as shown in Table 4.2. The overall variation is shown in Figure 4.19.



Figure 4.16 Overall variation for turbidity in the influent and effluent for Wetlands 4, 8/9 and 10/11 (low concentration of BR46).



Figure 4.17 Overall variation for turbidity in the influent and effluent for Wetlands 14/15 and 18/19 (high concentration of BR46).



Figure 4.18 Overall variation for turbidity in the influent and effluent for Wetlands 5, 6/7 and 12/13 (low concentration of AB113).



Figure 4.19 Overall variation for turbidity in the influent and effluent for Wetlands 16/17 and 20/21 (high concentration of AB113). Many researchers confirmed that vertical-flow wetlands have a relatively poor ability to remove TSS and turbidity (Lin et al., 2005, Bulc and Ojtrsek, 2008). Regarding this study, for the low concentration of BR46, a long contact time (94 h) was better than a short (48 h) one in reduction of TSS and turbidity. While for high concentrations, the short contact time was better than the long contact time. In case of low and high concentrations for AB113, the four-day contact time was not enough to reduce TSS and turbidity.

# 4.7 pH Value

The measuring of pH is very important due to pH conditions having a sensitive impact on the outflow quality including: nutrients, COD and TSS in constructed wetlands. The sensitive impact comes from its effect on the ability of microbial populations to degrade pollutants (Eke and Scholz, 2008; Lavrova and Koumanova, 2013; Paing et al., 2015). In general, all the effluent pH values for all wetlands in the case of low and high concentrations of both dyes (BR46 and AB113) during the whole period of experimental work, as shown in Figure 4.20, Figure 4.21, Figure 4.22 and Figure 4.23, respectively, were compliant with the national effluent discharge quality standards set by all consortia and brand guidelines such as the Government of Bangladesh and STWI, which state the range of the effluent pH value for inland surface water, public sewer secondary treatment plant and irrigated land is between 6 and 9 (Ahmed et al., 2002; STWI, 2012). For the low concentration of the dye BR46, the average influent pH value was 7.2. A slight increase in the effluent pH values of 0.16 and 0.03 was noted for unplanted control Wetland 4 and planted Wetlands 10/11 (low resting and high contact times), respectively, while there was a decrease of 0.31 for planted Wetlands 8/9 (high resting and low contact times), as shown in Table 4.2. For the high concentration, the average influent value was 6.49. Increases of 0.19 and 0.57 were noted for Wetlands 14/15 (low resting and contact times) and Wetlands 18/19 (high resting and contact times), respectively, as shown in Table 4.2. In the case of the dye AB113 for the low concentration scenario, the average influent pH value was 7.32. A slight pH increase of 0.11 was noted for unplanted control Wetland 5, while the pH value decreased for planted Wetlands 6/7 and Wetlands 12/13 by 0.44 and 0.15, respectively, as shown in Table 4.2. For the high concentration, the average influent pH value was 8.39. The pH value decreased for Wetlands 16/17 (low resting and contact times) and 20/21 (high resting and contact times) by 1.10 and 1.06, respectively, as shown in Table 4.2. Regarding the effect of plants on the pH value for both dyes BR46 and AB113, differences of 0.47 and 0.55 were found, as shown in Table 4.2, between the unplanted control Wetlands 4 and 5, and planted Wetlands 8/9 and 6/7 (all these wetlands have the same conditions), respectively. These results suggest that presence of plants has an effect on pH modification in vertical-flow constructed wetland systems used in this study, contrary to the suggestion of Kadlec and Wallace (2000) that the pH modification in this type of constructed wetlands is most likely as a result of interaction between the media and its biofilm, rather than of the plants. Furthermore, there were no changes in pH values in contrast to the findings obtained by Wieder (1989), in which he surveyed 128 constructed wetlands treating acid coal mine wastewater and found the difference between the effluent and the influent to be 0.11 (the influent pH was 2.5). Mitsch and Wise (1998) corroborated this finding, when they found that the difference between the influent and the effluent was 0.52 (the influent pH was 2.83). Furthermore, Kadlec and Wallace (2008) stated that the available pH value range for most degraded bacteria is 4-9.5. Nevertheless, these results indicate the ability of macrophytes such as *P. australis* to modify pH conditions in the rhizosphere, confirming results by (Brix et al., 2002). However, these researches used a different wetland plant (*Typha angustifolia L*.).



Figure 4.20 Overall variation in pH of the influent and effluent for Wetlands 4, 8/9 and 10/11 (low concentration of BR46).



Figure 4.21 Overall variation in pH of the influent and effluent for Wetlands 14/15 and 18/19 (high concentration of BR46).



Figure 4.22 Overall variation in pH of the influent and effluent for Wetlands 4, 6/7 and 12/13 (low concentration of AB113).



Figure 4.23 Overall variation in pH of the influent and effluent for Wetlands 16/17 and 20/21 (high concentration of AB113).

Characteristics			pH			Redox potential (mV)			Dissolved oxygen (mg/l)			
Dye	Туре	Wetland	No. of	Min.	Max.	Mean ±SD	Min.	Max.	Mean	Min.	Max.	Mean
	of	number(s)	samples						±SD			$\pm SD$
	flow											
BR46	In	N/A	70	6.80	7.54	$7.2\pm0.147$	-39	5	-25±7.7	9.0	10.9	9.7±0.46
	Out	4	70	7.11	7.75	7.36±0.137	-43	-14	$-29\pm6.35$	5.5	10.0	$8.2 \pm 1.12$
	Out	8 and 9	70	6.67	7.21	$6.89 \pm 0.148$	-22	8	-5±6.79	3.7	8.1	$5.4\pm0.92$
	Out	10 and 11	70	6.85	8.18	$7.23 \pm 0.212$	-40	-7	-22±9.25	2.9	9.5	$7.5 \pm 1.54$
	In	N/A	70	6.04	6.83	$6.49 \pm 0.179$	-6	28	$12\pm 8.06$	8.7	10.7	9.6±0.41
	Out	14 and 15	70	6.35	7.11	6.68±0.217	-17	17	4±7.53	3.8	9.2	6.3±1.34
	Out	18 and 19	35	6.65	7.37	7.06±0.163	-24	4	-13±7.58	3.2	8.7	$6.0{\pm}1.29$
AB113	In	N/A	70	7.14	7.65	$7.32 \pm 0.122$	-43	-21	$-30\pm5.4$	8.9	11.2	$9.6 \pm 0.55$
	Out	5	70	7.20	7.77	7.43±0.165	-51	-19	$-32\pm6.5$	5.1	10.1	8.3±1.38
	Out	6 and 7	70	6.62	7.4	$6.88 \pm 0.184$	-33	5	-6±8.2	3.7	6.9	$5.4\pm0.77$
	Out	12 and 13	70	6.78	7.57	7.17±0.199	-40	-9	-23±8.2	3.5	10.4	8.1±1.45
	In	N/A	70	7.25	9.04	$8.39 \pm 0.432$	-129	-46	-88±18.2	8.2	10.7	$9.5 \pm 0.49$
	Out	16 and 17	70	7.04	7.6	$7.29 \pm 0.118$	-46	-10	$-28\pm9.6$	5.4	9.9	$7.5 \pm 1.19$
	Out	20 and 21	35	7.10	7.99	7.33±0.202	-74	-12	-32±14	3.9	10.0	7.1±1.76

Table 4.3 Influent and effluent water quality characteristics for general physical and chemical variables.
Table 4.2 co	ntinued
--------------	---------

Characteristics			Total suspended solids				Turbid	ity (NTU)	Electrical conductivity ( $\mu$ S/cm)			
				(mg/l)								
BR46	In	N/A	70	0	5.1	$2.0{\pm}1.51$	1.6	18.5	$3.5 \pm 2.89$	90	270	159±51.6
	Out	4	70	0	23.0	$6.8 \pm 5.81$	1.9	32	8.8±7.12	70	483	193±79.1
	Out	8 and 9	70	0	6.2	$2.0{\pm}1.88$	1.5	6	3.0±1.33	103	749	257±97.4
	Out	10 and 11	70	0	10.2	$3.4 \pm 3.33$	1.9	12	$3.6 \pm 1.89$	82	419	$180\pm62.4$
	In	N/A	70	37.1	59.0	$47.9 \pm 5.11$	8.9	24	16.6±3.1	147	701	357±112.3
	Out	14 and 15	70	1.1	21.3	$7.6\pm4.70$	2.9	13	$6.6 \pm 2.2$	253	647	389±99.73
	Out	18 and 19	35	2.3	41.0	$6.9 \pm 8.41$	2.9	20.1	$6.8 \pm 3.9$	251	535	385±83.21
AB113	In	N/A	70	5.5	25.0	$14.1 \pm 5.40$	1.3	21.0	$7.4 \pm 5.67$	70	350	$220 \pm 104.1$
	Out	5	70	3.4	85.0	$41.9 \pm 23.07$	5.8	134.0	$39.2 \pm 27.40$	75	396	$242\pm87.9$
	Out	6 and 7	70	1.4	64.0	$18.8 \pm 15.92$	2.8	57.0	19.1±16.11	114	448	$294 \pm 85.4$
	Out	12 and 13	70	3.5	133.0	46.5±33.60	2.3	128.0	46.8±32.67	83	430	235±99.6
	In	N/A	70	119.0	843.0	364.8±195.47	21	660	213.4±157.62	465	976	635±123.2
	Out	16 and 17	70	121.0	706.0	371.7±177.57	12	650	208.3±196.67	456	854	624±99.1
	Out	20 and 21	35	67.0	863.0	392±253.13	9	675	224±202.69	502	844	635±93.3

Note: BR, basic red; AB, acid blue; Min., minimum; Max., maximum; SD, standard deviation; N/A, not applicable.

## **4.8 Nutrient Reduction**

The reduction of ortho-phosphate-phosphorous is controlled by chemical and physical adsorption, sedimentation, plant uptake, precipitation and microbial uptake in constructed wetland systems (Brix, 1997; Vymazal, 2007, 2010b; Johari et al., 2016). Moreover, many researchers reported that reduction efficiency of phosphorous compounds is generally poor with constructed wetlands (Choudhary et al., 2011; Lavrova and Koumanova, 2013; Ge et al., 2016). For the low concentrations of BR46 and AB113, the reduction percentages in planted Wetlands 8/9 (89%) as well as planted Wetlands 6/7 (30%), which have low resting and high contact times, were significantly (p < 0.05) better than those of control unplanted Wetlands 4 (78%) and 5 (27%) as well as planted Wetlands 10/11 (68%) and 12/13 (14%), which have high resting and low contact times, as shown in Table 4.3. For the high concentrations of BR46, the reduction percentages of Wetlands 18/19 (high resting and contact times; 81%) were significantly (p < 0.05) better than those for Wetlands 14/15 (low resting and contact times; 70%), while for AB113, the reduction percentages of Wetlands 16/17 (low resting and contact times; 78%) were also significantly (p < 0.05) better than those for Wetlands 20/21 (high resting and contact times; 71%), as shown in Figure 4.24.



Figure 4.24 Influent and effluent concentrations of ortho-phosphate-phosphorus for Wetlands 16/17 and 20/21 (high concentration of AB113).

A typical standard set by environment agencies for  $PO_4$ -P reduction concerning secondary wastewater treatment is 2 mg/l (Royal Commission on Sewage Disposal, 1915). Effluent ortho-phosphate-phosphorus values were compliant with the standard in the case of low concentration of BR46 and the values were relatively high in the case of low concentration of AB113. In the case of high concentration for both dyes, BR46 and AB113, the effluent values of PO4-P were much higher than the standard value.

Nitrification and denitrification are the main reduction mechanisms of nitrogen in constructed wetlands and these mechanisms include two-step processes: firstly, ammonium is oxidized to nitrite followed by oxidation of nitrite to nitrate, these are known as nitrification process. Secondly, the denitrification process includes nitrate being reduced to gaseous nitrogen (Schaechter, 2009; Kessel et al., 2015; Song et al.,

2015; Yang et al., 2016). As indicated in Table 4.3, the concentrations of NH<sub>4</sub>-N for both dyes are very low due to their chemical structures (Yalcuk and Dogdu, 2014). For the high concentration of dyes, there are no significant ( $p \ge 0.05$ ) differences between wetlands. For the low concentration of the dye BR46, the reductions for planted Wetlands 8/9 (low resting and high contact times) were significantly (p < 0.05) better than those for the corresponding reduction of unplanted Wetland 4, while there were increases in reduction for planted Wetlands 10/11 (high resting and low contact times). For the low concentration of the dye AB113, the reduction percentages with respect to planted Wetlands 12/13 (high resting and low contact times) were better than those for unplanted Wetland 4, and planted Wetlands 6/7 (low resting and high contact times), which was statistically insignificant ( $p \ge 0.05$ ). The previous results indicate that aeration availability plays a major function in determining the performance of higher nitrogen reduction. These findings are agreed by many researchers (Vymazal, 2007; H Wu et al., 2011; Fan et al., 2013c).

Regarding NO<sub>3</sub>-N reduction for the high concentration of both dyes (BR46 and AB113), the reduction percentages for wetlands with long contact times are better than those for wetlands having short contact times, confirming findings by (Kampf and Claassen, 2005). Regarding the low concentration of BR46, the reduction percentages for planted Wetlands 8/9 (low resting and high contact times) were better than those for unplanted Wetland 4, and planted Wetlands 10/11 (high resting and low contact times). For the low concentrations of AB113, there were significantly (p<0.05) better reduction percentages for planted Wetlands 6/7 (low resting and high contact times), but there were increases in the effluents regarding the NO<sub>3</sub>-N reduction of planted Wetlands 12/13 (high resting and low contact times) due to the presence of carbon under aerobic conditions (Jie et al., 2009). The reduction percentages for the high

concentrations of dyes were better than those for the low concentrations of dyes due to a corresponding increase in carbon percentage (Songliu et al., 2009; Lavrova and Koumanova, 2014; Shen et al., 2015), as shown in Table 4.3. The effluent NO<sub>3</sub>-N values for all wetlands in the case of low and high concentration for BR46 and AB113 were compliant with the traditional UK standard, which states that NO<sub>3</sub>-N reduction from secondary wastewater is 50 mg/l (Royal Commission on Sewage Disposal, 1915).

Characteristics			Ammonia-nitrogen (mg/l)					Nitrate-nitrogen (mg/l)				Ortho-phosphate-phosphorus (mg/l)			
Dye	Type of	Wetland	No. of	Min.	Max.	Mean ±	Reductio	Min.	Max.	Mean ±	Reductio	Min.	Max.	Mean ±	Reductio
	flow	number(s)	samples			SD	n (%)			SD	n (%)			SD	n (%)
BR46	In	N/A	20	0.54	0.75	$0.62 \pm 0.07$	N/A	17.9	18.0	$18.0 \pm 0.03$	N/A	5.3	6.3	5.9±0.33	N/A
	Out	4	20	0.45	0.68	$0.55 \pm 0.09$	11	4.5	16.9	$9.8 \pm 3.98$	45	0.9	1.7	1.3±0.27	78
	Out	8 and 9	20	0.21	0.51	$0.36 \pm 0.11$	42	2.5	13.9	$7.5 \pm 3.69$	58	0.19	1.3	$0.7 \pm 0.4$	89
	Out	10 and 11	20	0.49	0.90	$0.68 \pm 0.12$	-10	3.5	16.0	$8.9 \pm 3.83$	50	0.5	2.8	$1.9\pm0.83$	68
	In	N/A	20	0.98	1.04	$1.02 \pm 0.02$	N/A	19.3	20.0	$19.7 \pm 0.18$	N/A	55.0	87.0	$64.0 \pm 8.50$	N/A
	Out	14 and 15	20	0.25	0.47	$0.35 \pm 0.08$	66	1.5	16.2	$6.2 \pm 5.43$	69	6.5	47.0	19.3±13.20	70
	Out	18 and 19	20	0.21	0.46	$0.31 \pm 0.08$	70	0.6	12.1	$4.9 \pm 4.46$	75	1.9	43.6	$12.1 \pm 14.60$	81
AB113	In	N/A	20	0.63	0.70	$0.66 \pm 0.02$	N/A	8.4	08.5	$8.5 \pm 0.03$	N/A	6.9	8.3	7.6±0.43	N/A
	Out	5	20	0.37	0.62	$0.48 \pm 0.08$	27	10.5	16.3	$12.1 \pm 1.38$	-42	4.8	6.4	5.6±0.49	27
	Out	6 and 7	20	0.35	0.81	$0.52 \pm 0.14$	21	3.9	6.5	$4.9 \pm 0.98$	42	4.1	6.4	5.3±0.58	30
	Out	12 and 13	20	0.35	0.56	$0.47 \pm 0.06$	29	10.0	15.1	$11.6 \pm 1.45$	-36	5.7	7.7	6.6±0.51	14
	In	N/A	20	2.21	2.77	$2.33 \pm 0.16$	N/A	19.4	20.0	$19.8 \pm 0.21$	N/A	148.0	154.0	$151.0 \pm 2.10$	N/A
	Out	16 and 17	20	0.73	1.54	$1.00\pm0.24$	57	0.9	15.2	$6.0{\pm}5.8$	70	15.0	50.4	33.1±11.30	78
	Out	20 and 21	20	0.80	1.97	$1.10\pm0.32$	53	0.8	15.2	$5.3 \pm 5.10$	73	23.3	70.3	$43.5 \pm 15.30$	71

Table 4.4 Influent and effluent water quality characteristics for nutrients.

Note: BR, basic red; AB, acid blue; Min., minimum; Max., maximum; SD, standard deviation; N/A, not applicable.

# 4.9 Dye and Chemical Oxygen Demand Reductions

The percentage reduction of dye concentrations is shown in Table 4.4. The degradation of azo dyes takes place both in aerobic and anaerobic conditions via various processes involving enzymes and/or chemical reduction (Pandey et al., 2007; Khehra et al., 2005a; Saratale et al., 2011). BR46 is theoretically easier to degrade than AB113 due to its lower molecular weight.

There is no significant ( $p \ge 0.05$ ) difference in dye reduction for the low concentration of BR46 between control Wetland 4, planted Wetlands 8/9 (low resting and high contact times) and planted Wetlands 10/11 (high resting and low contact times), as shown in Figure 4.25. It follows that the presence of plants does not affect dye reduction. In comparison, for AB113, the dye reduction in planted Wetlands 6/7 was significantly (p < 0.05) better than that for unplanted control Wetland 5 (Figure 4.26) due to the presence of plants (Keskinkan and Lugal Göksu, 2007).



Figure 4.25 Influent and effluent dye concentrations for Wetlands 4, 8/9 and 10/11 (low concentration of BR46).



Figure 4.26 Influent and effluent dye concentrations for Wetlands 5, 6/7 and 12/13 (low concentration of AB113).

For high concentrations, the best percentage reductions for BR46 were noted for Wetlands 18/19 (high resting and contact times), which were statistically significantly (p < 0.05) different, if compared to those of Wetlands 14/15 (low resting and contact times), as shown in Figure 4.27. Wetlands with a low loading rate performed better than those with a high loading rate in terms of a dye reduction percentage rate but the influent mass loading rates for Wetlands 14/15 (high loading rate) and Wetlands 18/19 (low loading rate) were  $582.9\pm12.5$  g.m<sup>-2</sup>.d<sup>-1</sup> and  $291.5\pm6.3$  g.m<sup>-2</sup>.d<sup>-1</sup>, respectively, as shown in Table 4.5. The final decision about which one is better (low or high loading rate) depends on the design conditions of the constructed wetlands in the field. For the dye AB113, there was no significant ( $p \ge 0.05$ ) difference in the reduction of dyes between Wetlands 16/17 (low resting and contact times), and Wetlands 20/21 (high

resting and contact times), as shown in Table 4.4. The influent mass loading rates for Wetlands 16/17 (high loading rate) and Wetlands 20/21 (low loading rate) were  $616.9\pm88.8 \text{ g.m}^{-2}.\text{d}^{-1}$  and  $308.4\pm44.4 \text{ g.m}^{-2}.\text{d}^{-1}$ , respectively, as shown in Table 4.5. These results indicate that wetlands with a high loading rate are better than wetlands with a low loading rate.



Figure 4.27 Influent and effluent dye concentrations for Wetlands 14/15 and 18/19 (high concentration of BR46).

In textile wastewater, measurement of COD is very important to evaluate organic matter concentration in constructed wetlands. Its reduction mechanisms include anaerobic, filtration, aerobic, adsorption and microbial metabolism (Vymazal, 1998; Song et al., 2006; Stefanakis et al., 2014). For COD reduction concerning low concentrations of dyes (AB113 and BR46), as shown in Table 4.4, planted wetlands with a long contact time had the best COD reduction, if compared to the control and

other wetlands having short contact times. Cheng et al., 2011) operated two series of vertical-flow constructed wetlands (one planted and the other unplanted) under different C: N: P ratios. The results they obtained showed that the reduction of COD depended on the presence and development of plants. The reduction rate was higher in the planted wetlands.

Concerning high concentrations of dye BR46, the best reduction percentages for COD were noted for Wetlands 18/19 (high resting and contact times), which were statistically significantly (p<0.05) different from those of Wetlands 14/15 (low resting and contact times). Wetlands with a low loading rate performed better than those with a high loading rate as a COD concentration (Table 4.4), but the influent mass loading rate values for Wetlands 14/15 (high loading rate) and Wetlands 18/19 (low loading rate) were 726.9±30.6 g.m<sup>-2</sup>.d<sup>-1</sup> and 363.4±15.3 g.m<sup>-2</sup>.d<sup>-1</sup>, respectively, as shown in Table 4.5. The final decision about which one is better depends on the design conditions of the constructed wetlands in the field. Furthermore, the effluent COD values for both of them, low and high loading rate, were compliant with the national effluent discharge quality standards set by the Government of Bangladesh (most of the textile industries were built in developing countries), which state the maximum COD values for inland surface water, public sewer secondary treatment plant and irrigated land are 200 mg/l, 400 mg/l and 400 mg/l, respectively (Ahmed et al., 2002).

In the case of the dye AB113, there was no significant ( $p \ge 0.05$ ) difference in the reduction of COD between Wetlands 16/17 (low resting and contact times) and Wetlands 20/21 (high resting and contact times), as shown in Table 4.4. The influent mass loading rate values for Wetlands 16/17 (high loading rate) and Wetlands 20/21 (low loading rate) were 1626.4±94.7 g.m<sup>-2</sup>.d<sup>-1</sup> and 813.2±47.3 g.m<sup>-2</sup>.d<sup>-1</sup>, respectively, as shown in Table 4.5. These results suggest that wetlands with a high loading rate are

better than wetlands with a low loading rate. Furthermore, effluent COD values for both of them, low and high loading rate, were compliant for the purposes of public sewer secondary treatment plant and irrigated land (400 mg/l), and not compliant with the purpose of inland surface water (200 mg/l), according to the national effluent discharge quality standards set by the Government of Bangladesh (Ahmed et al., 2002).

Characte	centration (mg	/1)					
Dye	Type	Wetland	No. of	Min.	Max.	Mean	Reduction
5	of	number(s)	samples			±SD	(%)
	flow		1				
BR46	In	N/A	70	4.4	10.0	6.9±1.33	N/A
	Out	4	70	0.1	0.9	$0.3\pm0.2$	96
	Out	8 and 9	70	0.1	0.7	$0.2\pm0.1$	97
	Out	10 and 11	70	0.1	0.5	$0.2\pm0.1$	97
	In	N/A	70	200.0	218.0	209.3±4.5	N/A
	Out	14 and 15	70	2.0	56	37.7±19	82
	Out	18 and 19	35	2.0	29.8	12.5±9	94
AB113	In	N/A	70	3.5	10.5	$6.6 \pm 1.8$	N/A
	Out	5	70	0.1	4.6	$1.9 \pm 1.0$	71
	Out	6 and 7	70	0.1	4.5	$1.3 \pm 1.1$	80
	Out	12 and 13	70	0.3	4.5	$2.1{\pm}1$	68
	In	N/A	70	155.0	292.0	221.5±31.9	N/A
	Out	16 and 17	70	17.1	191.5	63±48	72
	Out	20 and 21	35	26.91	187.5	59±40	73
Characte	eristics				COD con	centration (mg	g/l)
BR46	In	N/A	20	11	32	22±6	N/A
	Out	4	20	2.8	19.5	11±5.4	50
	Out	8 and 9	20	0.3	13.8	7.2±4.3	67
	Out	10 and 11	20	0.7	16.3	9.1±5.1	59
	In	N/A	20	249	280	261±11	N/A
	Out	14 and 15	20	6	100	69±34	74
	Out	18 and 19	20	14	40	28±9	89
AB113	In	N/A	20	36.9	86.4	48.2±12.2	N/A
	Out	5	20	27.6	73	46±14.7	5
	Out	6 and 7	20	14	78	40.6±19.4	16
	Out	12 and 13	20	24	72	44.6±12.9	7
	In	N/A	20	543	610	584±34	N/A
	Out	16 and 17	20	117	454	271±103	54
	Out	20 and 21	20	126	362	$262 \pm 86$	55

Table 4.5 Dye and chemical oxygen demand (COD) reductions.

Note: BR, basic red; AB, acid blue; Min., minimum; Max., maximum; SD, standard deviation; N/A, not applicable.

Characte	eristics				D	ye loading rate	$(g.m^{-2}.d^{-1})$
Dye	Туре	Wetland	No. of	Min.	Max.	Mean	Reduction
	of	number(s)	samples			±SD	(%)
	flow						
BR46	In	N/A	70	12.3	27.9	$19.2 \pm 3.7$	N/A
	Out	4	70	0.3	2.5	$0.8\pm0.6$	96
	Out	8 and 9	70	0.3	1.9	$0.6\pm0.3$	97
	Out	10 and 11	70	0.3	1.4	0.6±0.3	97
	In	14 and 15	70	557	607	$582.9 \pm 12.5$	N/A
	Out	14 and 15	70	5.6	156	$105.0\pm52.9$	82
	In	18 and 19	35	278	303	291.5±6.3	N/A
	Out	18 and 19	35	2.8	41.5	$17.4 \pm 12.5$	94
AB113	In	N/A	70	9.7	29.2	$18.4 \pm 5.0$	N/A
	Out	5	70	0.3	12.8	$5.3 \pm 2.8$	71
	Out	6 and 7	70	0.3	12.5	3.6±3.1	80
	Out	12 and 13	70	0.8	12.5	$5.8 \pm 2.8$	68
	In	16 and 17	70	431	813	616.9±88.8	N/A
	Out	16 and 17	70	47.6	533.3	175.5±133.7	72
	In	20 and 21	35	215	406	$308.4 \pm 44.4$	N/A
	Out	20 and 21	35	37.5	261	82.2±55.7	73
Characte	eristics				CO	OD loading rate	$e(g.m^{-2}.d^{-1})$
BR46	In	N/A	20	30.6	89.1	61.3±16.7	N/A
	Out	4	20	7.8	54.3	30.6±15.0	50
	Out	8 and 9	20	0.8	38.4	20.1±12.0	67
	Out	10 and 11	20	1.9	45.4	$25.3 \pm 14.2$	59
	In	14 and 15	20	694	780	$726.9 \pm 30.6$	N/A
	Out	14 and 15	20	16.7	278.5	192.2±94.7	74
	In	18 and 19	20	347	390	363.4±15.3	N/A
	Out	18 and 19	20	19.5	55.7	39.0±12.5	89
AB113	In	N/A	20	102.8	240.6	$134.2 \pm 34.0$	N/A
	Out	5	20	76.9	203.3	$128.1 \pm 40.9$	5
	Out	6 and 7	20	39.0	217.2	$113.1\pm54.0$	16
	Out	12 and 13	20	66.8	200.5	$124.2\pm35.9$	7
	In	16 and 17	20	1512	1699	1626.4±94.7	N/A
	Out	16 and 17	20	328	1264	754.7±286.9	54
	In	20 and 21	20	756	849	813.2±47.3	N/A
	Out	20 and 21	20	176	504	364.9±119.8	55

Table 4.6 Dye and chemical oxygen demand (COD) loading rate.

Note: BR, basic red; AB, acid blue; Min., minimum; Max., maximum; SD, standard deviation; N/A, not applicable.

#### 4.10 Seasonal Treatment Performance of Wetlands on Dye Reduction

Azo dye decolourization is achieved under aerobic, anaerobic or anoxic conditions (O'neill et al., 2000; Sponza and Işik, 2002; van der Zee, 2002; Davies et al., 2006) through the biodegradation mechanism (Kodam et al., 2005; Moosvi et al., 2005; Pandey et al., 2007). Several publications have stated that micro-organisms are affected by the temperature; they are not active and energetic in cold periods in constructed wetland systems with the best treatment performance occurring during higher temperatures (Song et al., 2006; Imfeld, 2009; Sani et al., 2013). These findings agree with results obtained in this study in the case of low concentration of AB113 and high concentration of BR46 (p < 0.05), the best reduction percentage rates were in the summer season, as shown in Table 4.6. In the case of high concentration of AB113, the best reduction percentage rates (p<0.05) were in the autumn season as a result of well-established microbial populations and plants, as confirmed by many publications (Scholz et al., 2002; Al-Isawi et al., 2015a; Scholz, 2015). In the case of low concentration of BR46 (Wetlands 4, 8/9 and 10/11), as shown in Table 4.6, although the effluent dyes in all seasons were less than 1 mg/l and these effluents do not affect algal growth as stated by Van der Zee (2002), the maximum influent concentration was in the autumn season (7.71 mg/l) and the effluent concentrations for Wetlands 4, 8/9 and 10/11 were 0.33, 0.17 and 0.13 (mg/l), respectively. These results suggest that autumn is better than the other seasons for reduction of BR46 dye.

Characteristics		Summer <sup>(a)</sup>		Autumn <sup>(b)</sup>		Winte	er <sup>(c)</sup>	Spring <sup>(d)</sup>		
Dye	Туре	Wetland	$Mean \pm SD$	Reduction	Mean $\pm$ SD	Reduction	$Mean \pm SD$	Reduction	Mean $\pm$ SD	Reduction
	of	number(s)		(%)		(%)		(%)		(%)
	flow									
BR46	In	N/A	$6.26 \pm 1.07$	N/A	7.71±1.38	N/A	6.12±1.11	N/A	$5.6\pm0.72$	N/A
	Out	4	$0.38\pm0.12$	94	$0.33 \pm 0.12$	96	$0.27 \pm 0.03$	96	$0.24 \pm 0.08$	96
	Out	8 and 9	$0.37 \pm 0.28$	94	$0.17 \pm 0.08$	98	$0.11 \pm 0.02$	98	$0.14\pm0.12$	98
	Out	10 and 11	$0.26 \pm 0.18$	96	$0.13 \pm 0.04$	98	$0.15 \pm 0.04$	98	$0.16\pm0.12$	97
	In	N/A	$210.05 \pm 4.6$	N/A	212.63±4.16	N/A	211.52±3.45	N/A	$207.88 \pm 5.65$	N/A
	Out	14 and 15	$25.4{\pm}12.98$	88	46.73±3.59	78	$46.09 \pm 1.35$	78	$41.41 \pm 1.50$	80
	Out	18 and 19	$5.91 \pm 3.28$	97	$14.95 \pm 5.36$	93	$25.43 \pm 3.62$	88	$21.44 \pm 2.24$	90
AB113	In	N/A	$7.25 \pm 1.72$	N/A	$7.37 \pm 1.48$	N/A	$6.12 \pm 1.28$	N/A	6.59±1.79	N/A
	Out	5	$1.76\pm0.73$	76	$2.69 \pm 0.85$	64	$2.69 \pm 0.97$	56	$2.55 \pm 0.93$	61
	Out	6 and 7	$0.79 \pm 0.35$	89	$1.64 \pm 0.46$	78	$2.29 \pm 0.85$	63	$2.08\pm0.69$	68
	Out	12 and 13	$1.95 \pm 0.69$	73	$2.95 \pm 0.65$	60	2.51±0.54	59	$2.3\pm0.38$	65
	In	N/A	$217.44 \pm 36.08$	N/A	228.57±33.55	N/A	$220.55 \pm 15.88$	N/A	$217.18 \pm 10.13$	N/A
	Out	16 and 17	$97.88 \pm 55.54$	55	32.71±6.04	86	45.11±8.19	80	51.31±3.32	76
	Out	20 and 21	90.43±48.65	58	$37.84 \pm 8.76$	83	46.51±9.15	79	$50.78 \pm 4.65$	77
Temperature										
°C			22.8		11.6		8.6		20.5	

Table 4.7 Seasonal dyes reduction (mg/l)

Note: <sup>(a)</sup>21/06/2015 to 21/9/2015; <sup>(b)</sup>22/09/2015 to 20/12/2015; <sup>(c)</sup>21/12/2015 to 19/03/2016; <sup>(d)</sup>20/03/2016 to 31/05/2016; BR, basic red; AB, acid blue; N/A, not applicable; SD, standard deviation.

#### 4.11 Concluding Remarks

The presence of *P. australis* did not affect dye removal (p>0.05), while it had a minor but insignificant impact on COD removal (p>0.05). The use of plants concerning the short contact time scenario for NH4-N and a low concentration of AB113 is linked to a better removal compared to other cases apart from that for all nutrients; particularly PO4-P and NO3-N. In case of low dye concentrations, the presence of plants for the long contact time scenario impacted significantly (p < 0.05) positively on the removal efficiencies of PO4-P, NH4-N and NO3-N. The vertical-flow constructed wetlands showed significantly (p<0.05) good dye removals for low and high concentrations of BR46, and low concentrations of AB113 during all seasons under greenhouse conditions. The COD removal was a function of the dye used. This was explained by the complexity of the chemical structure of the dye and seasonal variations in temperature. For high concentrations of BR46, wetlands with a low hydraulic rate were significantly (p < 0.05) better that those with a high hydraulic rate for the removal of PO4-P. For AB113, wetlands with a high loading rate removed PO4-P significantly (p < 0.05) well. In case of the dye AB113, the four-day contact time was not enough to remove or reduce TSS and turbidity.

# 5. <u>CHAPTER FIVE: RESULTS AND DISCUSSION FOR</u> <u>ARTIFICIAL WASTEWATER CONTAINING TWO</u> <u>AZO TEXTILE DYES REDUCTIONS</u>

## **5.1 Overview**

This chapter includes the results and discussion of the artificial wastewater containing the two textile dye reduction rates, and the other physical and chemical water quality parameters for the period between 1 June 2016 and 31 May 2017. Section 5.2 deals with the test of normality for all variables. The results concerning the variables plant growth, redox potential, dissolved oxygen, electrical conductivity, total suspended solids, turbidity, pH, dyes, chemical oxygen demand and nutrients are discussed in sections 5.3 to 5.8. Section 5.9 covers the seasonal treatment performance of wetlands on dye reductions. Aromatic amine compound reductions discussed in section 5.10. Furthermore, statistical differences between the variables are also presented in these sections. Concluding remarks are given in section 5.11.

#### **5.2 Test of Normality**

Tests of normality results for dimensions of *P. australis* and for effluent water quality characteristics regarding general physical and chemical variables are discussed in Appendix B.2.

## **5.3 Plant Growth Monitoring**

When temperatures started to decrease in winter, the plants in this experiment began to yellow. The above-ground parts died, were cut and returned to the wetland filters. The authors followed common practice to cut the above-ground plant parts down to between 10 and 15 cm height according to Stefanakis et al. (2014).

Based on the experimental results, the most critical observation is that for all considered concentrations and contact times, the plants that were subjected to the dye AB113 revealed less effective growth compared to those that were fed by the dye BR46. Taking into consideration the potential impact of contact time, it has been noted that plants associated with a long contact time showed much better growth than those plants linked to a short one (Table 5.1). These findings support those by Pagter et al. (2005), who investigated the effect of water stress tolerance of *P. australis* grown in the laboratory by examining effects of different levels of required water. The results showed that a water deficit reduces the leaf biomass per plant and the leaf area.

The ANOVA and Mann-Whitney tests were applied for normal and non-normal distributed data, respectively. Regarding the growth of plants, there was a significant (p<0.05) difference with respect to the length and diameter at the low and high concentration of the dye AB113 (Wetlands 3, 9, 13 and 17). In the case of plants grown in the presence of the dye BR46, there was a significant (p<0.05) difference for both the length and diameter at the low concentration of dye (Wetlands 5 and 7), while no significance (p≥0.05) for either parameter was noted at the high concentration of dye (Wetlands 11 and 15). In the case of the mixed dye, there was no significant (p≥0.05) difference to the length at the low and high concentrations for all wetlands (Wetlands 4, 6, 8, 10, 12, 14, 16 and 18). While with respect to the diameter, there was a significant (p<0.05) difference at the high concentration (Wetlands 12, 14, 16 and 18).

136

Characteristics				Length (	cm)	Diameter (mm)			
Dye	Wetland	Number	Mini-	Maxi-	Mean	Mini-	Maxi-	Mean	
	numbers	of	mum	mum	$\pm SD$	mum	mum	±SD	
		stems							
BR46	5	68	69	142	$108 \pm 17.8$	1.1	3.1	$2.2 \pm 0.58$	
	7	20	60	122	93±16.9	0.8	2.3	$1.1\pm0.39$	
	11	17	45	90	67±11.2	0.8	1.1	$0.9 \pm 0.08$	
	15	14	44	90	60±12.9	0.8	1.4	$1.0\pm0.16$	
AB113	3	40	79	140	$107 \pm 18.0$	1.0	3.4	2.1±0.63	
	9	15	55	98	79±14.8	0.8	1.2	$1.0\pm0.11$	
	13	20	45	98	70±16.3	0.7	2.9	$1.9\pm0.59$	
	17	10	29	56	41±9.1	0.7	1.6	1.1±0.26	
The	4	48	78	142	109±15.9	1.0	3.9	2.4±0.73	
mixture	6	55	70	134	109±16.2	1.1	3.7	$2.3\pm0.59$	
of the	8	7	85	101	94±5.5	0.9	2.0	1.5±0.39	
dyes	10	10	80	110	95±8.2	1.1	2.3	$1.8\pm0.37$	
	12	10	45	87	64±11.3	0.8	1.6	$1.0\pm0.22$	
	14	16	46	80	66±9.7	1.3	3.1	$2.1\pm0.48$	
	16	7	45	61	$54 \pm 5.8$	0.8	1.9	1.3±0.37	
	18	9	44	67	58±7.5	1.0	2.9	1.8±0.63	

Table 5.1 Dimensions of *Phragmites australis* (Cav.) Trin. ex Steud. (Common Reed) planted in the experimental wetlands.

Note: BR, basic red; AB, acid blue; SD, standard deviation.

## 5.4 Redox Potential and Dissolved Oxygen

Redox potentials of greater than 100 mV indicate aerobic environments, while values less than -100 mV indicate anaerobic environments (Suthersan, 2001). DO is an important function in constructed wetlands since it is essential in aerobic respiration for microorganisms and it regulates the oxidation-redox potential in wastewater (Boyd, 2000). S Wu et al. (2011) and Hou et al. (2016) reported that the main pathways for oxygen transfer in constructed wetlands such as the system in this research (tidal flow) are: wetland macrophytes release via their roots, contact transfer at the interface of biofilm and atmosphere, and dissolved oxygen associated with influent wastewater. In the case of low concentration, redox potential values (Table 5.2) for the effluent of BR46, AB113 and the mixture of both of them were in the range between -34 mV and -64 mV and the whole values for the influent and effluents are shown in Figures 5.1, 5.2 and 5.3, respectively, for the effluent high concentrations, the value were in the range -56 mV to -95 mV and the whole values for the influent and effluents are shown in Figures 5.4, 5.5 and 5.6, respectively. These findings indicate that dye degradation may have taken place in between aerobic and anaerobic environments.



Figure 5.1 Overall variation for redox potential in the influent and effluent for Wetlands 1, 5 and 7 (low concentration of BR46).



Figure 5.2 Overall variation for redox potential in the influent and effluent for Wetlands 2, 3 and 9 (low concentration of AB113).



Figure 5.3 Overall variation for redox potential in the influent and effluent for Wetlands 4, 6, 8 and 10 (low concentration of the mixture of dyes BR46 and AB113).





Figure 5.4 Overall variation for redox potential in the influent and effluent for Wetlands 11 and 15 (high concentration of BR46).



Figure 5.5 Overall variation for redox potential in the influent and effluent for Wetlands 13 and 17 (high concentration of AB113).



Figure 5.6 Overall variation for redox potential in the influent and effluent for Wetlands 12, 14, 16 and 18 (high concentration of the mixture of dyes BR46 and AB113).

Regarding dissolved oxygen, in case of low concentration, in both of dyes BR46 and AB113, the lowest effluent values (Table 5.2) were in planted Wetlands 5 and 3 (low resting and high contact times; 2.97 mg/l and 3.37 mg/l, respectively) when compared with unplanted Wetlands 1 and 2 (low resting and high contact times), and planted Wetlands 7 and 9 (high resting and low contact times), respectively, during the whole period, as shown in Figures 5.7 and 5.8, respectively, as a result of the higher contact time leading to consumption of more dissolved oxygen by the microbial community. The same was found in the case of high concentration of BR46 and AB113, as shown in Figures 5.9 and 5.10, respectively. In the case of the low concentration of the mixture of the two dyes, the value of DO for Wetlands 4 and 6 (low resting and high contact times) was lower than Wetlands 8 and 10 (high resting and low contact times) as a result of the higher contact times).

community. The same was found for the DO between Wetlands 12 (low resting and contact times), and 16 (high resting and contact times), in the case of high concentration for the mixture of the two dyes, while the result was opposite to this between Wetlands 14 (low resting and contact times) and 18 (high resting and contact times). Furthermore, in the case of low and high concentration of the two dyes and the mixture of both of them during spring time, wetlands with higher resting time started consuming more dissolved oxygen when compared to wetlands with lower resting time because the increase in aerobic micro-organisms was greater than that of anaerobic macro-organisms.



Figure 5.7 Overall variation for dissolved oxygen in the influent and effluent for Wetlands 1, 5 and 7 (low concentration of BR46).



Figure 5.8 Overall variation for dissolved oxygen in the influent and effluent for Wetlands 2, 3 and 9 (low concentration of AB113).



Figure 5.9 Overall variation for dissolved oxygen in the influent and effluent for Wetlands 11 and 15 (high concentration of BR46).



Figure 5.10 Overall variation for dissolved oxygen in the influent and effluent for Wetlands 11 and 15 (high concentration of BR46).

## 5.5. Electrical Conductivity, Total Suspended Solids and Turbidity

Electrical conductivity can be used as an indirect measure of the charge (or the ioncarrying species) in the wetland outflow (Islam et al., 2011). EC value may be used as an indicator of other water quality problems. Any sudden increase in EC value indicates that there is a source of dissolved ions in the wetland filter (Kumar and Chopra, 2012). In comparison, all the effluent values for all wetlands in the case of low and high concentrations for both of the dyes and the mixture of both of them were compliant with the national effluent discharge quality standards set by the Government of Bangladesh, which state that the maximum effluent of EC for inland surface water, public sewer secondary treatment plant and irrigated land is 1200  $\mu$ S/cm (Ahmed et al., 2002). Furthermore, Sri Lanka central environmental authority (2008) state that the maximum EC discharge on land for irrigation purposes is 2250  $\mu$ S/cm. For the low concentration of both dyes (BR46 and AB113), as shown in Table 5.2, a higher increase was found in planted Wetlands 5 and 3 (contact time 94 h), when compared to unplanted control Wetlands 1 and 2 (contact time 94 h), respectively, while a decrease in EC effluent values were found in Wetlands 7 and 9 (contact time 48 h), respectively. For the high concentration for both dyes (BR46 and AB113), EC effluent values for all wetlands were less than influent values. Furthermore, Wetlands 15 and 17 (long resting and contact times) had EC values less than Wetlands 11 and 13 (low resting and contact times), respectively. In the case of the mixture of both dyes for both low and high concentrations, all the values of the effluent were less than the influent values, as shown in Table 5.2. Nevertheless, all the previous results indicated no sudden increase in EC values for all wetlands.

The measurement of TSS is important for the design of water treatment facilities (Dzurik, 2003) and the Clean Water Act (CWA) lists it as a conventional pollutant (Bell et al. 2011). In the case of low concentration for dyes BR46 and AB113, there was an increase in effluent for all wetlands when compared to the influent, as shown in Table 5.2. A lower increase and decrease were found in planted Wetlands 7 and 9 (high resting and low contact times) when compared with unplanted Wetlands 1 and 2, and planted Wetlands 5 and 3 (low resting and high contact times), respectively. For the mixture of both dyes, a slight decrease was found in Wetlands 4 and 6 while, a decrease was found in Wetlands 8, and 10. In the case of high concentration for both dyes (BR46 and AB113) and the mixture of the two dyes, a good reduction was recorded in all wetlands, as shown in Table 5.2. Wetlands with high resting and contact times had a lower effluent when compared with wetlands which have low resting and contact times.

In comparing effluents of low and high concentrations of BR46, AB113 and the mixture of both of the dyes during the whole period, as shown in Figures 5.11, 5.12, 5.13, 5.14, 5.15 and 5.16, respectively, it was found that all the effluent results for wetlands were

compliant with the national effluent discharge quality standards set by the Government of Bangladesh, which state the maximum effluent values of TSS for inland surface water, public sewer secondary treatment plant and irrigated land are 150 mg/l, 500 mg/l and 200 mg/l, respectively (Ahmed et al., 2002). Furthermore, in the case of low and high concentration of the two dyes and the mixture of both of them during spring time, wetlands with higher resting time started consuming more dissolved oxygen when compared to wetlands with lower resting time because the increase in aerobic microorganisms was greater than that of anaerobic macro-organisms.



Figure 5.11 Overall variation for total suspended solids in the influent and effluent for Wetlands 1, 5 and 7 (low concentration of BR46).



Figure 5.12 Overall variation for total suspended solids in the influent and effluent for Wetlands 11 and 15 (high concentration of BR46).



Figure 5.13 Overall variation for total suspended solids in the influent and effluent for Wetlands 2, 3 and 9 (low concentration of AB113).



Figure 5.14 Overall variation for total suspended solids in the influent and effluent for Wetlands 13 and 17 (high concentration of AB113).



Figure 5.15 Overall variation for total suspended solids in the influent and effluent for Wetlands 4, 6, 8 and 10 (low concentration of the mixture of both of the dyes).



Figure 5.16 Overall variation for total suspended solids in the influent and effluent for Wetlands 12, 14, 16 and 18 (high concentration of the mixture of both of the dyes).

Turbidity is often used to interpret the degree of clarity of water. It is a variable often applied as an indicator of the amount of suspended sediments and/or larger microorganisms in water. A high turbidity of surface water may also indicate elevated concentrations of TSS, reduced algal populations, and potential harm to fish and other aquatic fauna (Postolache et al., 2007). Furthermore, a higher turbidity value can also increase the temperature of surface water as a result of increased absorption of heat from sunlight, as well as leading to reduced light penetration, which affects photosynthesis (Håkanson, 2006).

For the low concentration of the dye BR46, there was an increase in all effluent wetlands when compared with the influents. Planted Wetland 7 (high resting and low contact times) has a smaller increase when compared with unplanted Wetland 1 and planted Wetland 5 (low resting and high contact times). In the case of dye AB113, a slight increase was recorded in the mean value of planted Wetland 3 (low resting and high contact times) while, a slight decrease was recorded in unplanted Wetland 2 (low resting and high contact times) and planted Wetland 9 (high resting and low contact times). In

the case of the mixture of the two dyes, an increase was recorded in Wetlands 4 and 6 (low resting and high contact times), while a decrease was recorded in Wetlands 8 and 10 (high resting and low contact times). For the high concentration for both of the dyes (BR46 and AB113) and the mixture of the two dyes, all the wetlands had a good effluent reduction when compared with the influent. Wetlands 15, 17, 16 and 18 (high resting and contact times) had a greater reduction when compared with Wetlands 11, 13, 12 and 14 (low resting and contact times), respectively.

Lin et al. (2005) and Bulc and Ojtrsek (2008) stated that the ability of vertical-flow constructed wetlands to reduce TSS and turbidity is relatively poor. Regarding this study, for low concentration of AB113, a short contact time (48 h) was better than a long (94 h) one in reduction of TSS and turbidity as well as in the case of the mixture of both of the dyes (BR46 and AB113). While for high concentrations of both dyes and the mixture of the two dyes, the long contact time was better than the short contact time. The percentage TSS reduction rates for BR46, AB113 and the mixture of both dyes were 69%, 47% and 71%, respectively.

# 5.6 pH Value

The measuring of pH is very important due to pH conditions having a sensitive impact on the outflow quality including: nutrients, COD and TSS in constructed wetlands. The sensitive impact comes from its effect on the ability of microbial populations to degrade pollutants (Eke and Scholz, 2008; Lavrova and Koumanova, 2013; Paing et al., 2015). For the low concentration of the dye BR46, the average influent pH value was 7.47, a slight decrease in the pH effluent value of 0.3 was noted in planted Wetland 5 (low resting and high contact times), while there was a slight increase of 0.25 and 0.1 in unplanted Wetland 1 (low resting and high contact times) and planted Wetland 7 (high resting and low contact times), respectively. In the case of dye AB113, there was a slight effluent increase of 0.24, 0.22 and 0.16 for the unplanted control Wetland 2, planted Wetland 3 (low resting and high contact times), and planted Wetland 9 (high resting and low contact times), respectively, when compared to the influent value of 7.35, as shown in Table 5.2. In the case of the mixture of both dyes, there was a slight decrease of 0.03, 0.23 and 0.05 for Wetlands 4, 6 (low resting and high contact times) and Wetland 10 (high resting and low contact times), respectively, when compared with the influent value of 7.32, while for Wetland 8 (high resting and low contact times), there was a slight increase of 0.01. For the high concentration for both dyes (BR46 and AB113) and the mixture of the two dyes, a slight increase was found ranging between 0.08 and 0.65 for Wetlands 11, 13, 12, 14 (low resting and contact times), 17, 16 and 18 (high resting and contact times), while an increase of 1.04 was recorded in Wetland 15 (high resting and contact times) when compared with the influent value of 6.94. This increase in effluent pH values is due to the formation of basic aromatic amine metabolites (Chandra, 2015).

Regarding the effect of plants on the pH value for the low concentration of dye AB113, there was a slight difference of 0.02 between unplanted control Wetland 2 and planted Wetland 3 (both of them have the same conditions). This result suggests that the pH modification in vertical-flow constructed wetlands is probably as a result of interactions between the media and its biofilms, rather than due to the plants, this result was confirmed by Kadlec and Wallace (2008). Unlike the result for the dye BR46, there was a difference of 0.55 between unplanted control Wetland 1 and planted Wetland 5 (both of them have the same conditions). The different results regarding the role of plants on pH value are most likely due to each dye having a different chemical structure and molecular weight, as shown in Table 3.1. Furthermore, there were no changes in pH values in contrast to

the findings obtained by Wieder (1989), in which he surveyed 128 constructed wetlands treating acid coal mine wastewater and found the difference between the effluent and the influent to be 0.11 (influent pH was 2.5). Mitsch and Wise (1998) corroborated this finding, when they found that the difference between the influent and the effluent was 0.52 (influent pH was 2.82). Furthermore, Kadlec and Wallace (2008) stated that the available pH value range for most degraded bacteria is 4-9.5. Nevertheless, these results indicate the ability of macrophytes such as *P. australis* to modify pH conditions in the rhizosphere, confirming results by Brix et al. (2002). However, these studies used a different wetland plant (Typha angustifolia L.). Furthermore, all the effluent pH values for all wetlands in the case of low and high concentrations of both dyes (BR46 and AB113) and the mixture of the two dyes during the whole period, as shown in Figures 5.17, 5.18, 5.19, 5.20, 5.21 and 5.22, respectively, were compliant with the national effluent discharge quality standards set by all consortia and brand guidelines such as the Government of Bangladesh and STWI (Sweden Textile Water Initiative), which state the range of the pH effluent for inland surface water, public sewer secondary treatment plant and irrigated land is between 6 and 9 (Ahmed et al., 2002; STWI, 2012).



Figure 5.17 Overall variation in pH of the influent and effluent for Wetlands 1, 5 and 7 (low concentration of BR46).



Figure 5.18 Overall variation in pH of the influent and effluent for Wetlands 11and 15 (high concentration of BR46).



Figure 5.19 Overall variation in pH of the influent and effluent for Wetlands 2, 3 and 9 (low concentration of AB113).



Figure 5.20 Overall variation in pH of the influent and effluent for Wetlands 13 and 17 (high concentration of AB113).



Figure 5.21 Overall variation in pH of the influent and effluent for Wetlands 4, 6, 8 and 10 (low concentration of mixture of both of the dyes).



Figure 5.22 Overall variation in pH of the influent and effluent for Wetlands 12, 14, 16 and 18 (high concentration of mixture of both of the dyes).
Characteristics			pH			Re	dox pot	ential (mV)	Dissolved oxygen (mg/l)			
Dye	Туре	Wetland	No. of	Min.	Max.	Mean	Min.	Max.	Mean ±SD	Min.	Max.	Mean ±SD
	of flow	number	samples			$\pm SD$						
BR46	In	N/A	82	7.22	7.77	$7.47 \pm 0.17$	-56	-41	-47.72±3.74	8.64	9.66	9.24±0.22
	Out	1	82	7.37	8.14	7.72±0.23	-67	-49	$-55.98 \pm 3.81$	2.13	5.51	$3.73 \pm 0.92$
	Out	5	82	6.94	7.36	$7.17 \pm 0.08$	-45	-26	-35.29±4.31	2.01	4.06	$2.97 \pm 0.46$
	Out	7	82	7.08	8.32	$7.57 \pm 0.42$	-51	-30	-41.80±4.74	2.38	5.21	3.71±0.66
	In	N/A	81	6.79	7.25	$6.94 \pm 0.14$	-24	-14	$-19.43 \pm 2.48$	8.8	9.74	9.21±0.18
	Out	11	81	7.16	8.01	$7.49 \pm 0.24$	-97	-38	$-56.28 \pm 14.03$	2.58	4.73	$3.65 \pm 0.46$
	Out	15	41	7.81	8.21	$7.98 \pm 0.09$	-87	-67	$-72.90 \pm 4.22$	2.41	4.01	$3.03 \pm 0.36$
AB113	In	N/A	82	7.19	7.57	$7.35 \pm 0.08$	-59	-39	$-44.85 \pm 3.21$	8.41	9.89	$9.40 \pm 0.35$
	Out	2	82	7.37	7.72	$7.59 \pm 0.07$	-66	-50	-60.71±3.08	2.42	5.57	$4.14 \pm 0.78$
	Out	3	82	7.21	7.77	7.57±0.16	-71	-46	-64.24±4.90	2.62	4.76	$3.37 \pm 0.54$
	Out	9	82	7.26	7.71	$7.51 \pm 0.07$	-63	-44	$-57.90 \pm 3.42$	2.36	5.45	$3.59 \pm 0.79$
	In	N/A	81	7.99	8.17	$8.07 \pm 0.04$	-90	-75	-82.12±4.27	9.00	9.51	9.31±0.11
	Out	13	81	7.83	8.29	8.16±0.12	-102	-74	-92.24±6.78	1.95	5.21	$4.07 \pm 0.61$
	Out	17	41	7.80	8.31	8.15±0.16	-106	-78	$-95.27 \pm 8.27$	1.98	4.41	$3.62 \pm 0.73$
The mixture	In	N/A	82	7.2	7.39	$7.32 \pm 0.03$	-52	-35	-45.67±3.01	8.78	9.62	9.36±0.19
of the dyes	Out	4	82	7.01	7.49	$7.29 \pm 0.09$	-46	-28	$-40.62 \pm 3.02$	1.67	4.47	$3.12 \pm 0.59$
	Out	6	82	6.78	7.18	$7.09 \pm 0.07$	-39	-17	-33.61±3.06	2.21	4.01	$2.98 \pm 0.44$
	Out	8	82	7.21	7.51	$7.33 \pm 0.07$	-59	-41	$-50.80 \pm 3.81$	2.01	5.01	$3.85 \pm 0.63$
	Out	10	82	7.01	7.51	$7.27 \pm 0.14$	-55	-30	-46.66±5.25	1.89	4.84	$3.55 \pm 0.68$
	In	N/A	81	7.2	7.46	$7.35 \pm 0.04$	-47	-26	-34.91±5.16	8.84	9.72	9.24±0.15
	Out	12	81	7.71	7.9	$7.78 \pm 0.03$	-81	-60	-72.64±4.67	2.7	5.05	$4.29 \pm 0.62$
	Out	14	81	6.35	7.75	$7.60 \pm 0.15$	-72	-48	-61.23±4.68	2.19	4.52	$2.98 \pm 0.63$
	Out	16	41	7.75	8.01	$7.86 \pm 0.06$	-86	-65	-75.32±4.54	1.56	5.19	$3.52 \pm 0.87$
	Out	18	41	7.74	8.17	$8.00 \pm 0.08$	-89	-71	$-82.07 \pm 4.58$	1.42	5.50	3.86±1.06

Table 5.2 Inflow and outflow water quality characteristics for general physical and chemical variables related to different wetlands.

Table 5.2	continued	1											
Character	ristics			Total s	uspende	d solids (mg/l)		Turbidity	y (NTU)	Elec	trical co	onductivity	
										$(\mu S/cm)$			
BR46	In	N/A	82	1	2	$1.07 \pm 0.26$	2.01	6.13	3.91±0.85	545	575	560±9.35	
	Out	1	82	0	8	$4.48 \pm 1.56$	3.43	6.61	5.24±0.69	451	558	$483 \pm 18.04$	
	Out	5	82	2	45	$10.68 \pm 9.52$	4.11	76.5	$11.81 \pm 17.02$	487	670	644±35.79	
	Out	7	82	0	6	$2.73 \pm 1.45$	3	6.53	4.07±0.77	336	421	397±16.62	
	In	N/A	81	35	56	$45.01 \pm 4.51$	12.6	15.9	14.62±0.63	685	765	735±13.46	
	Out	11	81	16	46	$23.68 \pm 5.67$	7.23	27.1	13.11±3.43	563	684	604±16.09	
	Out	15	41	11	32	$14.95 \pm 3.19$	8.04	23.4	$12.09 \pm 3.94$	559	644	581±16.62	
AB113	In	N/A	82	4	8	$5.87 \pm 1.04$	5.06	6.71	7.51±0.42	521	575	543±12.85	
	Out	2	82	3	12	$7.96 \pm 1.72$	3.91	6.99	5.37±0.87	468	601	$559 \pm 22.48$	
	Out	3	82	2	22	6.71±4.04	4.86	10.49	$7.03 \pm 1.18$	497	622	583±21.49	
	Out	9	82	0	7	3.16±1.28	2.87	6.42	$4.59 \pm 0.78$	408	511	484±16.42	
	In	N/A	81	97	126	$110\pm8.59$	42.31	62.34	51.79±3.95	855	892	870±5.63	
	Out	13	81	37	78	$62.64 \pm 8.47$	9.95	34.21	24.21±6.61	702	909	765±51.28	
	Out	17	41	40	97	$58.29 \pm 14.40$	12.45	25.41	$18.62 \pm 4.24$	672	899	$742 \pm 60.84$	
The	In	N/A	82	6	10	$8.32 \pm 0.89$	4.23	5.92	5.14±0.34	510	535	522±6.20	
mixture	Out	4	82	7	62	$12.99 \pm 10.90$	3.85	64.5	$10.67 \pm 10.95$	478	586	508±14.22	
of the	Out	6	82	1	72	$11.40 \pm 17.02$	4.14	87.4	$11.67 \pm 17.43$	475	568	491±12.96	
dyes	Out	8	82	0	6	$2.95 \pm 1.22$	3.54	6.66	4.41±0.55	339	460	399±18.59	
	Out	10	82	0	6	$3.07 \pm 1.38$	3.42	6.93	4.65±0.61	335	439	413±24.16	
	In	N/A	81	242	294	263±16.54	131	157	$144 \pm 5.97$	760	785	775±4.92	
	Out	12	81	67	239	100±44.63	14.2	71.3	43.66±18.77	678	734	710±9.83	
	Out	14	81	57	128	80.48±11.56	11.3	55	31.87±11.95	693	760	738±9.44	
	Out	16	41	67	252	$109 \pm 46.20$	20.4	68.3	44.59±13.99	670	732	717±11.03	
	Out	18	41	42	112	75.27±18.37	14.9	51.2	32.64±10.31	696	785	746±13.33	

Note: BR, basic red; AB, acid blue; SD, standard deviation; Min., minimum; Max., maximum; N/A, not applicable.

#### 5.7 Dye, Colour and Chemical Oxygen Demand Reductions

The degradation of azo dyes takes place both in aerobic and anaerobic conditions via various processes involving enzymes and/or chemical reduction (Khehra et al., 2005; Pandey et al., 2007; Saratale et al., 2011). The first contaminant to be recognized in an effluent textile wastewater is the colour, it adsorbs and reflects the sunlight entering the water, thereby interfering with the aquatic species growth and hindering photosynthesis (Pereira and Alves, 2012; Yadav et al., 2012).

For dye and colour reductions concerning low concentration of the dyes (BR46 and AB113), and the mixture of the two dyes, wetlands with a long contact time (Table 5.3) have the best dye and colour reductions (regardless of the planting regime), when compared to wetlands having short contact times. For the high concentration of BR46 and AB113 (Figures 5.23, 5.24, 5.25 and 5.26, respectively), and the mixture of both of them, wetlands which have a low loading rate (high resting and contact times), have the better dye and colour reductions (p < 0.05) when compared to wetlands with high loading rates (low resting and contact times), as shown in Table 5.3. Although, in terms of dye concentration, wetlands, which have a low loading rate, have the better reduction when compared with wetlands which have a high loading rate, the influents as a mass loading rate for Wetlands 11 and 13 (high loading rate) were 573.71±26.74 and 576.49±38.15 g.m<sup>-2</sup>.d<sup>-1</sup>, respectively, while for Wetlands 15 and 17 (low loading rate) they were  $286.86\pm13.37$  and  $288.25\pm19.08$  g.m<sup>-2</sup>.d<sup>-1</sup>, respectively, as shown in Table 5.4. The final decision about which one is better (low or high loading rate) depends on the design conditions of the constructed wetlands in the field. The effluent colour for the wetlands in the case of low concentrations of BR46, AB113 and the mixture of the two dyes were compliant with the national effluent discharge quality standards set by the Government

of India (1986), which state the maximum colour value is 400 (Pt Co.), while in the case of high concentration of BR46, AB113 and the mixture of the two dyes values were not compliant even when compared to maximum value of colour (550 Pt Co.), which is stated by the Government of Taiwan (2003).



Figure 5.23 Overall variation in dye concentration of the influent and effluent for Wetlands 11 and 15 (high concentration of BR46).



Figure 5.24 Overall variation in dye colour of the influent and effluent for Wetlands 11 and 15 (high concentration of BR46).



Figure 5.25 Overall variation in dye concentration of the influent and effluent for Wetlands 13 and 17 (high concentration of AB113).



Figure 5.26 Overall variation in dye colour of the influent and effluent for Wetlands 13 and 17 (high concentration of AB113).

In textile wastewater, measurement of COD is very important to evaluate organic matter concentration in constructed wetlands. Its reduction mechanisms include anaerobic, filtration, aerobic, adsorption and microbial metabolism (Vymazal et al., 1998; Song et al., 2006; Stefanakis et al., 2014). The effluent COD values for low concentration of BR46, AB113 and the mixture of both dyes (Table 5.3) were compliant with the national effluent discharge quality standards set by the Government of Bangladesh, which state the maximum COD values for inland surface water, public sewer secondary treatment plant and irrigated land are 200 mg/l, 400 mg/l and 400 mg/l, respectively, while in the case of high concentration of BR46, AB113 and the mixture of both dyes (Table 5.3) values were compliant with public sewer secondary treatment and irrigated land (Ahmed et al., 2002). For COD reduction concerning low concentration of dyes (BR46 and AB113) and the mixture of both of the dyes, the results showed that all wetlands demonstrated good COD reduction, as shown in Table 5.3. Furthermore, wetlands with a long resting time had the best COD reduction, if compared to the control (unplanted wetlands) and/or other wetlands having short resting times. These results indicate that both aerobic and anaerobic environments are the best for COD reduction. Wetlands 7, 9, 8 and 10 have effluent DO values higher than Wetlands 5, 3, 4 and 6. For the high concentration, COD reduction in wetlands which have a low loading rate (high resting and contact times), is better than in the wetlands that have a high loading rate (low resting and contact times) in terms of COD concentration (Table 5.3), but the influent mass loading rate values for Wetlands 11 and 13 (high loading rate) were 1423.14±102.27 and 1668.22±132.73 g.m<sup>-2</sup>.d<sup>-1</sup>, respectively as shown in Table 5.5 while, for Wetlands 15 and 17 (low loading rate) were 711.57 $\pm$ 51.14 and 834.11 $\pm$ 66.37 g.m<sup>-2</sup>.d<sup>-1</sup>. The final decision about which one is better depends on the design conditions of the constructed wetlands in the field. All the previous findings regarding low and high concentration for BR46,

AB113 and the mixture of the two dyes indicate that setting aerobic and anaerobic conditions will improve the COD reduction and this has been confirmed by many researchers (Vymazal et al., 1998; Li et al., 2012; Lehl et al., 2016).

Characte	eristics	<i>.</i>			(	Colour (Pt Co.)			Dye c	oncentration (m	g/l)	Chemical oxygen demand			(COD) <sup>(a)</sup>
Dye	Туре	Wetland	No. of	Min.	Max.	Mean	Reduction	Min.	Max.	Mean	Reduction	Min.	Max.	Mean	Reduction
	of	number	samples			$\pm SD$	(%)			$\pm SD$	(%)			$\pm SD$	(%)
	flow														
BR46	In	N/A	82	410	439	422±7.67	N/A	4.3	8.41	6.15±0.75	N/A	200	297	248±20.89	N/A
	Out	1	82	16	187	70.73±35.27	83	0.0	1.18	$0.52 \pm 0.22$	92	49.5	93	76.17±11.79	69
	Out	5	82	26	356	96.96±71.42	77	0.0	0.99	$0.57 \pm 0.24$	91	54	110	73.68±16.44	70
	Out	7	82	9	272	$120\pm80.47$	72	0.11	1.87	$0.69 \pm 0.40$	89	20.1	79.5	44.62±13.89	82
	In	N/A	81	12240	12900	12574±142.19	N/A	185	224	$206 \pm 9.60$	N/A	478	576	511±36.72	N/A
	Out	11	81	5670	10210	8604±1053.03	32	53.8	185	139±39.18	33	196	385	311±52.37	39
	Out	15	41	3200	7520	5442±1459.49	57	25.1	100.6	$55.75 \pm 24.00$	73	135	301	$225 \pm 48.42$	56
AB113	In	N/A	82	530	589	555±16.14	N/A	5.32	10.93	$7.5 \pm 1.66$	N/A	234	310	$275 \pm 18.12$	N/A
	Out	2	82	39	386	$135 \pm 100.27$	76	0.48	4.80	$1.15 \pm 0.87$	85	53.9	125	66.47±16.28	76
	Out	3	82	89	400	$182\pm82.52$	67	0.61	4.49	$1.37 \pm 0.84$	82	88.8	176	$105.75 \pm 19.41$	62
	Out	9	82	224	490	310±71.22	44	0.57	5.31	$1.74{\pm}1.08$	77	39.7	84.6	53.31±10.30	81
	In	N/A	81	13080	13990	13561±279.67	N/A	174	231	207±13.70	N/A	541	710	599±47.66	N/A
	Out	13	81	7230	13420	10699±1338.87	21	50.0	197	$115 \pm 46.97$	44	199	495	$357 \pm 75.48$	40
	Out	17	41	7290	12820	8707±1432.62	36	30.2	169	95.82±43.72	54	190	345	265±43.31	56
The	In	N/A	82	323	450	$400 \pm 20.82$	N/A	$0.108^{(b)}$	$0.182^{(b)}$	$0.154^{(b)} \pm 0.022$	N/A	254	350	$292 \pm 27.99$	N/A
mixture	Out	4	82	145	311	$192 \pm 37.48$	52	0.031	0.156	$0.069 \pm 0.025$	55	85.6	270	$132.63 \pm 43.60$	55
of both	Out	6	82	80	263	$119 \pm 33.75$	70	0.014	0.157	$0.050 \pm 0.029$	68	58.4	276	$116.81 \pm 54.38$	60
dyes	Out	8	82	114	339	$233 \pm 34.02$	42	0.039	0.136	$0.067 \pm 0.021$	56	36.0	91.2	56.79±15.73	81
	Out	10	82	206	372	273±31.43	32	0.044	0.138	$0.073 \pm 0.017$	53	31.5	120	62.71±24.78	79
	In	N/A	81	16090	16190	16130±29.94	N/A	4.652	5.781	$5.339 \pm 0.310$	N/A	480	594	551±43.48	N/A
	Out	12	81	8620	14520	12199±1593.48	24	3.040	5.333	4.371±0.606	18	145	450	$347 \pm 76.40$	37
	Out	14	81	10790	13720	12020±668.23	25	1.884	5.073	$3.587 \pm 0.936$	33	215	431	340±64.17	38
	Out	16	41	9490	13690	11933±1096.63	26	1.521	4.892	$3.429 \pm 0.997$	36	225	397	$310 \pm 47.10$	44
	Out	18	41	6430	12260	10250±1680.25	36	1.100	4.511	$2.819 \pm 1.032$	47	230	373	294±34.61	46

Table 5.3 Colour, dye and chemical oxygen demand (COD) reductions for different wetlands.

Note: BR, basic red; AB, acid blue; SD, standard deviation; Min., minimum; Max., maximum; <sup>(a)</sup> No. of samples is 30; <sup>(b)</sup> All measurements for the mixture are as a wavelength; N/A, not applicable.

Characte	eristics				Dye load	ding rate (g.m <sup>-2</sup> .d	<sup>-1</sup> )		
Dye	Type	Wetland	No. of	Min.	Max.	Mean	Reduction		
	of	number(s)	samples			$\pm SD$	(%)		
	flow								
BR46	In	N/A	82	11.98	23.42	17.13±2.09	N/A		
	Out	1	82	0.00	3.29	$1.45 \pm 0.61$	92		
	Out	5	82	0.00	2.76	$1.59 \pm 0.67$	91		
	Out	7	82	0.31	5.21	$1.92 \pm 1.11$	89		
	In	11	81	515.22	623.84	573.71±26.74	N/A		
	Out	11	81	149.83	515.23	387.12±109.12	33		
	In	15	41	257.61	311.92	286.86±13.37	N/A		
	Out	15	41	34.95	140.09	77.63±33.42	73		
AB113	In	N/A	82	14.82	30.44	$20.89 \pm 4.63$	N/A		
	Out	2	82	1.34	13.39	$3.20 \pm 2.42$	85		
	Out	3	82	1.69	12.50	$3.82 \pm 2.34$	82		
	Out	9	82	1.59	14.79	4.85±3.01	77		
	In	13	81	484.59	643.34	576.49±38.15	N/A		
	Out	13	81	139.25	548.65	320.28±130.81	44		
	In	17	41	242.29	321.67	$288.25 \pm 19.08$	N/A		
	Out	17	41	42.05	235.33	133.43±60.88	54		

Table 5.4 Dye loading rate.

Note: BR, basic red; AB, acid blue; Min., minimum; Max., maximum; SD, standard deviation; N/A, not applicable.

Characte	eristics				COD load	ling rate (g.m- <sup>2</sup> .d <sup>-1</sup>	)
Dye	Туре	Wetland	No. of	Min.	Max.	Mean	Reduction
	of	number(s)	samples			$\pm SD$	(%)
	flow						
BR46	In	N/A	30	557.00	827.14	$690.68 \pm 58.18$	N/A
	Out	1	30	137.86	259.01	212.13±32.84	69
	Out	5	30	150.39	306.35	205.19±45.79	70
	Out	7	30	55.98	221.41	124.27±38.68	82
	In	11	30	1331.23	1604.16	1423.14±102.27	N/A
	Out	11	30	545.86	1072.23	866.14±145.85	39
	In	15	30	665.62	802.08	711.57±51.14	N/A
	Out	15	30	187.99	419.14	313.31±67.42	56
AB113	In	N/A	30	651.69	863.35	$765.88 \pm 50.46$	N/A
	Out	2	30	150.11	348.13	185.12±45.34	76
	Out	3	30	247.31	490.16	294.51±54.06	62
	Out	9	30	110.56	235.61	$148.47 \pm 28.69$	81
	In	13	30	1506.69	1977.35	1668.22±132.73	N/A
	Out	13	30	554.22	1378.58	994.25±210.21	40
	In	17	30	753.35	988.68	834.11±66.37	N/A
	Out	17	30	264.58	480.41	369.02±60.31	56

Table 5.5 Chemical oxygen demand (COD) loading rate.

Note: BR, basic red; AB, acid blue; Min., minimum; Max., maximum; SD, standard deviation; N/A, not applicable.

## **5.8 Nutrient Reduction**

The reduction of ortho-phosphate-phosphorous is controlled by chemical and physical adsorption, sedimentation, plant uptake, precipitation and microbial uptake in constructed wetland systems (Brix, 1997; Vymazal, 2007, 2010; Johari et al., 2016). Moreover, many researchers have reported that reduction efficiency of phosphorous compounds is generally poor with constructed wetlands (Choudhary et al., 2011; Lavrova and Koumanova, 2013; Ge et al., 2016).

For the low concentrations, in the case of AB113 and BR46, the reduction percentage in planted Wetlands 3 and 5 (low resting and high contact times) was significantly (p<0.05) better than those of control unplanted Wetlands 2 and 1 (low resting and high contact times) and planted Wetlands 7 and 9 (high resting and contact times), as shown in Table 5.6. In the case of the mixture of both dyes (BR46 and AB113), Wetlands 4 and 6 (low resting and high contact times) had better reduction percentages when compared with Wetlands 8 and 10 (high resting and low contact times), respectively, as shown in Table 5.6. For the high concentration for BR46, AB113 and the mixture of the two dyes, Wetlands 15, 17, and 16 and 18 (high resting and contact times) had lower PO<sub>4</sub>-P effluent when compared with Wetlands 11, 13, and 12 and 14 (low resting and contact times), respectively, as shown in Table 5.6. The previous results for low and high concentrations, indicate that the reduction efficiency for PO<sub>4</sub>-P was relatively good especially with wetlands, regardless of planting regime, which have the longer contact times (the lower in resting times).

Nitrification and denitrification are the main reduction mechanisms of nitrogen in constructed wetlands and these mechanisms include two-step processes: firstly,

ammonium is oxidized to nitrite followed by oxidation of nitrite to nitrate, these are known as nitrification process. Secondly, the denitrification process includes nitrate being reduced to gaseous nitrogen (Schaechter, 2009; Kessel et al., 2015; Song et al., 2015; Yang et al., 2016). Regarding NH<sub>4</sub>-N reduction percentages for low concentration of BR46 and AB113, as shown in Table 5.4, planted Wetlands 7 and 9 (high resting and low contact times) have better reduction percentages when compared with unplanted control Wetlands 1 and 2 as well as planted Wetlands 5 and 3 (low resting and high contact times), respectively. In the case of the mixture of both dyes, Wetlands 8 and 10 (high resting and low contact times) had better reduction percentages in comparison with Wetlands 4 and 6 (low resting and high contact times), respectively, as shown in Table 5.6. In the case of high concentration for BR46, AB113 and the mixture of both dyes, Wetlands 15, 17, 16 and 18 (high resting and contact times) have better reduction percentages when compared them with Wetlands 11, 13, 12 and 14 (low resting and contact times), respectively. The previous results indicate that aeration availability plays a major function in determining the performance of higher nitrogen reduction. These findings have been confirmed by many researchers (Vymazal, 2007; H Wu et al., 2011; Fan et al., 2013c). The effluent NH<sub>4</sub>-N values for all wetlands in case of low and high concentration for BR46, AB113 and the mixture of both dyes were complaint with the traditional UK standard (Royal Commission on Sewage Disposal, 1915), which stated that NH<sub>4</sub>-N reduction from the secondary wastewater is 50 mg/l. Furthermore, both of Government of India (1986) and Government of Bangladesh (Ahmed et al., 2002) stated that 50 mg/l is also accepted as an effluent for the surface waters.

Regarding NO<sub>3</sub>-N reduction for low concentration for BR46, AB113 and the mixture of both dyes, as shown in Table 5.6, the influent NO<sub>3</sub>-N values were in the range (23.53-25.37) mg/l. The reduction percentages for all wetlands were in the range of 83%-100%. For the high concentration for BR46, AB113 and the mixture of both dyes the influent values were approximately 33.45 mg/l and the reduction percentages for all wetlands were in the range of 75%-86%, as shown in Table 5.6. These results for NO<sub>3</sub>-N reduction percentages indicate that VFCWs have good ability to reduce the nitrogen in high percentages especially when there is a source of organic carbon, and both of the dyes have carbon in their chemical structure as shown in Table 3.1. These findings have been confirmed by Lavrova and Koumanova (2014) and Shen et al. (2015). Furthermore, Lavrova and Koumanova (2013) demonstrated that VFCWs can effectively reduce NO<sub>3</sub>-N with/without plants with a sufficient organic carbon source. The effluent NO<sub>3</sub>-N values for all wetlands in case of low and high concentration for BR46, AB113 and the mixture of both dyes were complaint to the traditional UK standard, which stated that NO<sub>3</sub>-N reduction from the secondary wastewater is 50 mg/l (Royal Commission on Sewage Disposal, 1915).

Character	ristics				Ammor	nia-nitrogen (mg	g/l)		Nitrat	e-nitrogen (mg/l	.)	Or	Ortho-phosphate-phosphorus (mg			
Dye	Туре	Wetland	No. of	Min.	Max.	$Mean \pm SD$	Reductio	Min.	Max.	$Mean \pm SD$	Reductio	Min.	Max.	Mean $\pm$ SD	Reductio	
	of flow	number	samples				n (%)				n (%)				n (%)	
BR46	In	N/A	30	16.9	25.7	$21.98 \pm 2.43$	N/A	21.9	30.2	$25.07 \pm 2.38$	N/A	5.41	7.65	6.36±0.68	N/A	
	Out	1	30	11	22.1	$18.72 \pm 2.48$	15	0.0	0.49	0.21±0.18	99	2.11	5.91	$3.49 \pm 0.98$	45	
	Out	5	30	3.67	19.0	$14.41 \pm 3.53$	34	0.0	0.17	$0.02 \pm 0.05$	100	0.0	3.31	$1.76 \pm 0.83$	72	
	Out	7	30	2.55	16.1	$6.19 \pm 4.78$	72	0.0	0.67	0.21±0.20	99	3.44	5.87	$4.54 \pm 0.69$	29	
	In	N/A	30	21.3	29.7	$26.99 \pm 2.41$	N/A	28.7	38.1	33.47±1.75	N/A	62.5	68.3	65.78±1.75	N/A	
	Out	11	30	12.5	31.4	$21.67 \pm 4.12$	20	4.03	16.6	8.41±3.99	75	33.6	52.4	45.33±6.32	31	
	Out	15	30	9.2	26.3	$20.29 \pm 4.79$	25	1.4	13.9	6.23±3.61	81	11.9	35.8	19.09±6.70	71	
AB113	In	N/A	30	20.4	24.4	23.36±1.02	N/A	19.9	28.23	23.53±2.19	N/A	7.55	11.8	10.34±1.13	N/A	
	Out	2	30	9.86	24.3	19.23±4.16	18	0.0	0.95	0.11±0.23	100	2.25	7.27	$4.00 \pm 1.62$	61	
	Out	3	30	13.50	36.2	$24.48 \pm 7.59$	-5	0.03	1.51	$0.55 \pm 0.43$	98	2.01	6.21	3.91±1.07	62	
	Out	9	30	9.10	17.4	$13.20 \pm 2.25$	43	0.0	1.1	0.22±0.31	99	5.12	8.34	$6.22 \pm 0.94$	40	
	In	N/A	30	26.9	29.0	$28.16 \pm 0.68$	N/A	30.2	37.6	$33.05 \pm 2.10$	N/A	144	158	156±3.27	N/A	
	Out	13	30	13.5	27.7	$20.89 \pm 4.74$	26	3.14	7.29	$5.59 \pm 0.89$	83	8.92	95.4	46.15±28.22	70	
	Out	17	30	9.65	24.7	$18.60 \pm 4.98$	34	2.85	6.32	4.57±0.99	86	21.2	67.0	42.29±16.47	73	
The	In	N/A	30	27.2	34.2	30.27±1.81	N/A	22.6	29	25.37±1.72	N/A	6.0	8.7	$7.56 \pm 0.81$	N/A	
mixture	Out	4	30	17.42	34.97	26.01±4.75	14	0.0	0.34	$0.08 \pm 0.09$	100	2.45	8.31	$4.47 \pm 1.48$	41	
of both	Out	6	30	7.15	22.24	$15.66 \pm 4.39$	48	0.0	2.1	0.13±0.39	100	0.44	4.22	$2.03 \pm 1.02$	73	
dyes	Out	8	30	5.10	17.03	9.78±3.21	68	0.05	13.90	3.71±4.79	85	4.2	8.15	$5.85 \pm 1.19$	23	
-	Out	10	30	2.30	17.88	$10.75 \pm 5.39$	65	0.0	2.54	$0.39 \pm 0.59$	99	3.9	8.9	$6.14 \pm 1.54$	19	
	In	N/A	30	23.2	33.6	$30.75 \pm 2.81$	N/A	31.2	35.1	33.91±1.03	N/A	112.1	123	$118 \pm 2.72$	N/A	
	Out	12	30	10.7	24.5	$20.72 \pm 3.68$	33	4.32	9.73	$6.45 \pm 1.44$	81	5.21	86.8	$58.37 \pm 24.80$	51	
	Out	14	30	9.8	26.4	19.87±3.85	35	3.71	7.51	5.79±1.04	83	4.71	97.2	56.34±26.13	52	
	Out	16	30	7.95	25.2	$17.41 \pm 5.47$	43	2.87	6.11	$4.89 \pm 0.86$	86	19.5	74.3	57.31±15.98	51	
	Out	18	30	8.59	27.7	16.84±5.25	45	2.43	8.41	4.71±1.23	86	26.3	69.7	50.48±13.16	57	

Table 5.6 Inflow and outflow water quality characteristics for nutrients related to different wetlands.

Note: BR, basic red; AB, acid blue; SD, standard deviation; Min., minimum; Max., maximum; N/A, not applicable.

## 5.9 Seasonal Comparison of Effluent Dyes Reduction

The overall seasonal comparison of the influent and effluent dyes concentration for all wetlands are shown in Table 5.7. In the case of low concentration of BR46 and AB113, the best reduction percentage rates (p < 0.05) were in the spring season as a result of well-established microbial populations, favourable operating conditions achieved over time, and plants, as confirmed by many publications (Scholz et al., 2002; Al-Isawi et al., 2015a; Scholz, 2015). In the case of high concentration of BR46, AB113 and the mixture of both dyes, the best reduction percentage rates (p < 0.05) were in the summer season, as shown in Table 5.7, as a result of the higher temperature, as confirmed by several researchers, who stated that the best treatment performance occurred during the higher temperature (Song et al., 2006; Sani et al., 2013).

Characteristics			Summ	ner <sup>(a)</sup>	Autum	n <sup>(b)</sup>	Winte	r <sup>(c)</sup>	Spring <sup>(d)</sup>		
Dye	Type	Wetland	Mean $\pm$ SD	Reduction	Mean $\pm$ SD	Reduction	Mean $\pm$ SD	Reduction	Mean $\pm$ SD	Reduction	
	of	number(s)		(%)		(%)		(%)		(%)	
	flow										
BR46	In	N/A	6.26 ±0.99	N/A	6.07±0.49	N/A	6.22±0.76	N/A	6.06±0.66	N/A	
	Out	1	0.56±0.13	91	$0.67 \pm 0.23$	89	$0.47 \pm 0.22$	92	$0.34 \pm 0.08$	94	
	Out	5	$0.58 \pm 0.18$	91	$0.58 \pm 0.29$	90	$0.54 \pm 0.31$	91	$0.58 \pm 0.16$	90	
	Out	7	$1.00\pm0.39$	84	$0.69 \pm 0.16$	88	$0.66 \pm 0.37$	89	$0.28\pm0.27$	95	
	In	N/A	$202.80 \pm 9.52$	N/A	$207.47 \pm 9.64$	N/A	208.64±11.05	N/A	$204.65 \pm 7.18$	N/A	
	Out	11	95.9±11.94	53	$168.3 \pm 14.18$	19	174.7±4.63	16	128.58±22.27	37	
	Out	15	$30.94 \pm 4.41$	85	$54.18 \pm 8.07$	74	80.01±16.99	62	63.71±26.82	69	
AB113	In	N/A	8.96±1.55	N/A	6.66±1.15	N/A	6.51±1.36	N/A	$7.78 \pm 1.17$	N/A	
	Out	2	$1.98 \pm 1.13$	78	$0.96 \pm 0.65$	86	$0.70\pm0.11$	89	$0.80 \pm 0.18$	90	
	Out	3	$2.24{\pm}1.14$	75	$1.18\pm0.76$	82	$1.03\pm0.15$	84	$0.85 \pm 0.18$	89	
	Out	9	3.1±0.94	65	$1.56\pm0.53$	77	$1.25\pm0.32$	81	$0.69 \pm 0.02$	91	
	In	N/A	192.33±8.14	N/A	207.58±13.31	N/A	$208.00 \pm 9.24$	N/A	213.82±7.04	N/A	
	Out	13	$60.68 \pm 3.90$	68	129.81±27.93	37	172.16±21.45	17	96.49±20.34	55	
	Out	17	49.81±5.65	74	$109.38 \pm 26.88$	47	$146.48 \pm 16.07$	30	71.29±33.29	67	

Table 5.7 Seasonal reduction (mg/l) of dyes from artificial wastewater containing two azo textile dyes.

Table 5.7 cont	inued									
Characteristics	8		Sumn	ner <sup>(a)</sup>	А	utumn <sup>(b)</sup>	Win	ter <sup>(c)</sup>	Sprir	ng <sup>(d)</sup>
Dye	Type	Wetland	Mean $\pm$ SD	Reduction	Mean ±	Reduction	Mean ±	Reduction	Mean $\pm$ SD	Reduction
-	of flow	number(s)		(%)	SD	(%)	SD	(%)		(%)
The mixture	In	N/A	$0.15 \pm 0.02$	N/A	$0.16 \pm 0.02$	N/A	$0.15 \pm 0.02$	N/A	0.15±0.03	N/A
of both dyes	Out	4	$0.07 \pm 0.00$	53	$0.06 \pm 0.01$	63	$0.05 \pm 0.01$	67	$0.09 \pm 0.05$	40
	Out	6	$0.05 \pm 0.01$	67	$0.04 \pm 0.00$	75	$0.04\pm0.00$	73	$0.08 \pm 0.05$	47
	Out	8	$0.07 \pm 0.00$	53	$0.06 \pm 0.01$	63	$0.06 \pm 0.00$	60	$0.09 \pm 0.04$	40
	Out	10	$0.08 \pm 0.00$	47	$0.07 \pm 0.01$	56	$0.06 \pm 0.00$	60	$0.09 \pm 0.03$	40
	In	N/A	$5.35 \pm 0.38$	N/A	$5.24 \pm 0.28$	N/A	5.34±0.26	N/A	$5.44 \pm 0.29$	N/A
	Out	12	$3.60 \pm 0.35$	33	4.57±0.21	13	4.89±0.32	8	4.57±0.41	22
	Out	14	$2.37 \pm 0.44$	56	$3.98 \pm 0.38$	24	4.51±0.43	16	$3.62 \pm 0.59$	33
	Out	16	$2.09 \pm 0.53$	61	$3.75 \pm 0.45$	28	4.32±0.54	19	$3.75 \pm 0.58$	31
	Out	18	$1.57 \pm 0.52$	71	$3.34\pm0.44$	36	$3.96 \pm 0.56$	26	$2.52 \pm 0.0.42$	54
Temperature °	С		22	.8	11	1.6	8.	6	20.	5

Note: <sup>(a)</sup>21/06/2016 to 21/9/2016; <sup>(b)</sup>22/09/2016 to 20/12/2016; <sup>(c)</sup>21/12/2016 to 19/03/2017; <sup>(d)</sup>20/03/2017 to 29/05/2017; BR, basic red; AB, acid blue; N/A, not applicable; SD, standard deviation.

## **5.10 Aromatic Amines Reduction**

Azo dyes decolourization is achieved under various conditions including: aerobic, anaerobic or anoxic (O'neill et al., 2000; Sponza and Işik, 2002; van der Zee, 2002; Davies et al., 2006). In anaerobic conditions, azo bond (N=N) cleaves and this process releases aromatic amine, which resists any further anaerobic treatment (Brown and Hamburger, 1987; Chung and Stevens, 1993). Aromatic amine can be reduced under aerobic treatment (Weber and Wolfe, 1987; Pinheiro et al., 2004; Ong et al., 2011). Amine compounds are a toxic material and have an effect on the individual bacteria, which leads to the biomass being insufficient to degrade dyes (Phugare et al., 2011; Holkar et al., 2014). Each dye has one or more types of aromatic amines (Pielesz et al., 2002; Pinheiro et al., 2004). Constructed wetlands can degrade aromatic amines under aerobic conditions (Mbuligwe, 2005; Ong et al., 2010, Ong et al., 2011). In this study, three types of amines were released as a result of dye AB113 degradation: 3-AminoBenzeneSulfonic Acid (ABSA), 1,4-DiAminoNaphthalene (DAN) and 5-Amino-8-(phenyl amino) Naphthalene-1-Sulfonic Acid (ANSA) (Senthilvelan et al., 2014). The wavelength of maximum absorbance for each is 288 nm, 255 nm and 225 nm, respectively (Koepernik and Borsdorf, 1983; Paul et al., 1990). In the case of BR46, two aromatic amines were released as a result of its degradation: N-benzyi-N-mehtylaniline (NBNMA) and Nbenzyl-N-methylbenzene-1,4-diamine (NBNMD), with the wavelength of maximum absorbance for each being 254 nm and 290 nm, respectively (Fihtengolts, 1969; Küçükgüzel et al., 1999). For the low concentration of AB113, in the case of ABSA (Figure 5.27), Wetland 9 (high resting and low contact times) had a significant (p < 0.05) reduction efficiency when compared with unplanted Wetland 2 and planted Wetland 3 (low resting and high contact times). For the high concentration of AB113, in the case of ABSA amine (Figure 5.28), Wetland 17 (high resting and contact times) had a significant (p < 0.05) reduction efficiency when compared with Wetland 13 (low resting and contact times). It follows that reduction of aromatic amine compounds needs aerobic conditions, as stated previously. Amines DAN and ANSA, for low and high concentration, didn't appear in the UV-spectrophotometer equipment. This is because both of them are unstable and therefore they were not detected in solution, as confirmed by Davies et al. (2005) and Davies et al. (2006) who also found that using HPLC analysis did not detect this type of amines. For the low concentration of BR46, in the case of NBNMD amine, Wetland 7 (high resting and low contact times) had a significant (p < 0.05) reduction efficiency when compared with unplanted Wetland 1 and planted Wetland 5 (low resting and high contact times), as shown in Figure 5.29. NBNMA amine, didn't appear in UV-spectrophotometer equipment regarding unplanted Wetland 1, while for planted Wetlands 5 and 7, it was sometimes detected and sometimes not dependant on the activity of micro-organisms required to degrade this type of amines. For the high concentration BR46, in the case of NBNMD amine, as shown in Figure 5.30, Wetland 15 (high resting and contact times) had a significant (p < 0.05) reduction efficiency when compared with Wetland 11 (low resting and contact times). Furthermore, during the period between (19/12/16; 6 °C) and (3/2/17; 10 °C), NBNMD amine was not detected as a result of the decrease in temperature during this period and its effect on the growth and development of microbial communities (Jerman et al., 2009). NBNMA amine, didn't appear in UV-spectrophotometer equipment for the same reason as stated before.



Figure 5.27 Absorbance of ABSA released for Wetlands 2, 3 and 9 (low concentration of AB113)



Figure 5.28 Absorbance of ABSA released for Wetlands 13 and 17 (high concentration of AB113)



Figure 5.29 Absorbance of NBNMD released for Wetlands 1, 5 and 7 (low concentration of BR46)



Figure 5.30 Absorbance of NBNMD released for Wetlands 11 and 15 (high concentration of BR46)

## 5.11 Concluding Remarks

The vertical-flow constructed wetlands showed a significantly (p<0.05) good denitrification for low and high concentrations of AB113, BR46 and the mixture of both dyes during all seasons under greenhouse conditions. Furthermore, a good PO4-P reduction was noted, especially with wetlands having long contact times. In addition, good dye reductions were achieved in the case of low and high concentrations of BR46, and low concentrations in the case of AB113 and the mixture of both dyes. The results showed VFCW systems have an ability to cleave the nitrogen atoms and reduce aromatic amine compounds, which are released as a result of nitrogen atom cleavage.

# 6. CONCLUSIONS AND RECOMMENDATIONS

## 6.1 Conclusions

Vertical-flow constructed wetland filters were operated to investigate the performance of treating azo dyes and other water quality variables through different operational variables (low and high concentrations as well as resting and contact times) in two experimental phases (two azo textile dyes only during a first phase and artificial wastewater containing two textile dyes in a second phase). The effect of the presence of plants on dye reduction depends on the dye itself (molecular weight and the chemical structure) and its concentration. Gravel plays a major role to support micro-organisms and nutrients reduction. Furthermore, VFCW systems have the ability to treat different types of dyes, mixtures of them with and without artificial wastewater in case of low and high concentrations. The effects of different operational parameters, resting times, contact times and loading rates on the performance of VFCW systems depends on the group of the selected dyes (acid or basic). Regarding the first experimental phase (two azo textile dyes only), the presence of *P. australis* did not affect dye reductions (p>0.05), while it had a minor but insignificant impact on COD reduction (p>0.05). The use of plants concerning the short contact time scenario for NH4-N and a low concentration of AB113 is linked to a better reduction compared to other cases apart from that for all nutrients; particularly PO<sub>4</sub>-P and NO<sub>3</sub>-N. In the case of low dye concentrations, the presence of plants for the long contact time scenario impacted significantly (p < 0.05) positively on the reduction efficiencies of PO<sub>4</sub>-P, NH<sub>4</sub>-N and NO<sub>3</sub>-N. The verticalflow constructed wetlands showed significantly (p < 0.05) good dye reductions for low and high concentrations of BR46, and low concentrations of AB113 during all seasons under greenhouse conditions. The COD reduction was a function of the dye used. This was explained by the complexity of the chemical structure of the dye and seasonal variations in temperature. For high concentrations of BR46, wetlands with a long hydraulic retention time were significantly (p<0.05) better that those with a short hydraulic retention time for the reduction of PO<sub>4</sub>-P. For AB113, wetlands with a shorter hydraulic retention time removed PO<sub>4</sub>-P significantly (p<0.05) well. In the case of the dye AB113, the four-day contact time was not sufficient to remove or reduce TSS and turbidity.

In the second experimental phase (artificial wastewater containing two azo textile dyes), the vertical-flow constructed wetlands showed a significantly (p<0.05) good denitrification process for low and high concentrations of AB113, BR46 and the mixture of both dyes during all seasons under greenhouse conditions. Furthermore, a good PO<sub>4</sub>-P reduction was noted, especially with wetlands having long contact times. In addition, good dye reductions were achieved in the case of low and high concentrations of BR46, and low concentrations in the case of AB113 and the mixture of both dyes. The results showed VFCW systems have an ability to cleave the nitrogen atoms and reduce aromatic amine compounds, which are released as a result of nitrogen atom cleavage.

## **6.2 Recommendations**

Five main recommendations, which should be considered for further research work, are listed below.

- The effects of different plant species and an increase of the contact time on particle and dye reduction processes;
- 2- The effect of pH (low and high) on dye reduction;
- 3- More investigations of aromatic amine compounds by using HPLC combined with FTIR instruments, especially in the case of the mixture of dyes;
- 4- Wetland design implications based on the research findings in terms of investment of area and costs; and
- 5- Working on modelling the results of both controls (without plant) and planted wetlands (dye concentration, nutrients and other pollutants) to help the designer of industrial constructed wetlands in scaling-up tasks.

## References

- Abdelhakeem, S. G., Aboulroos, S. A. & Kamel, M. M. 2015. Performance of a vertical subsurface flow constructed wetland under different operational conditions. *Journal of Advanced Research*.
- Abou-Elela, S. I., Golinielli, G., Abou-Taleb, E. M. & Hellal, M. S. 2013. Municipal wastewater treatment in horizontal and vertical flows constructed wetlands. *Ecological engineering*, 61(Part A), 460-468.
- Abou-Elela, S. I. & Hellal, M. S. 2012. Municipal wastewater treatment using vertical flow constructed wetlands planted with Canna, Phragmites and Cyprus. *Ecological engineering*, 47(209-213.
- Ahmed, S., Alexandra, C., Clark, M. & Tapley, K. 2002. Choosing an Effluent Treatment Plant, *Dhaka, Bangladesh, Genesis (Pvt.) Ltd.*
- Akarslan, F. & Demiralay, H. 2015. Effects of Textile Materials Harmful to Human Health. *Acta Physica Polonica A*, 128(2B).
- Akbari, A., Remigy, J. & Aptel, P. 2002. Treatment of textile dye effluent using a polyamide-based nanofiltration membrane. *Chemical Engineering and Processing: Process Intensification*, 41(7), 601-609.
- Akratos, C. S. & Tsihrintzis, V. A. 2007. Effect of temperature, HRT, vegetation and porous media on removal efficiency of pilot-scale horizontal subsurface flow constructed wetlands. *Ecological engineering*, 29(2), 173-191.
- Al-Isawi, R., Sani, A., Almuktar, S. & Scholz, M. 2015a. Vertical-flow constructed wetlands treating domestic wastewater contaminated by hydrocarbons. *Water Science and Technology*, 71(6), 938-946.
- Al-Isawi, R., Scholz, M., Wang, Y. & Sani, A. 2015b. Clogging of vertical-flow constructed wetlands treating urban wastewater contaminated with a diesel spill. *Environmental Science and Pollution Research*, 22(17), 12779-12803.
- Alavian, V., Qaddumi, H. M., Dickson, E., Diez, S. M., Danilenko, A. V., Hirji, R. F., Puz, G., Pizarro, C., Jacobsen, M. & Blankespoor, B. 2009. Water and climate change: understanding the risks and making climate-smart investment decisions, *Washington, D.C., World Bank*.
- Albalawneh, A., Chang, T.-K., Chou, C.-S. & Naoum, S. 2016. Efficiency of a Horizontal Sub-Surface Flow Constructed Wetland Treatment System in an Arid Area. *Water*, 8(2), 51.
- Allegre, C., Maisseu, M., Charbit, F. & Moulin, P. 2004. Coagulation–flocculation– decantation of dye house effluents: concentrated effluents. *Journal of hazardous materials*, 116(1), 57-64.
- Allen, S. E., Grimshaw, H. M., Parkinson, J. A. & Quarmby, C. 1974. Chemical analysis of ecological materials, *Oxford, UK., Blackwell Scientific Publications*.
- Allende, K. L., Mccarthy, D. & Fletcher, T. 2014. The influence of media type on removal of arsenic, iron and boron from acidic wastewater in horizontal flow wetland microcosms planted with *Phragmites australis*. *Chemical Engineering Journal*, 246, 217-228.
- Almuktar, S., Scholz, M., Al-Isawi, R. & Sani, A. 2015. Recycling of domestic wastewater treated by vertical-flow wetlands for irrigating Chillies and Sweet Peppers. Agricultural Water Management, 149, 1-22.

- APHA 1995. American Public Health Association. Standard Methods for the Examination of Water and Wastewater. Washington, D.C.
- Arcos, J. C., Argus, M. F. & Wolf, G. 2013. Chemical induction of cancer: structural bases and biological mechanisms, *Elsevier*.
- Arias, C. A., Brix, H. & Marti, E. 2005. Recycling of treated effluents enhances removal of total nitrogen in vertical flow constructed wetlands. *Journal of Environmental Science and Health*, 40(6-7), 1431-1443.
- Arias, C. A., Del Bubba, M. & Brix, H. 2001. Phosphorus removal by sands for use as media in subsurface flow constructed reed beds. *Water Research* 35(5), 1159–1168.
- Arivoli, A., Mohanraj, R. & Seenivasan, R. 2015. Application of vertical flow constructed wetland in treatment of heavy metals from pulp and paper industry wastewater. *Environmental Science and Pollution Research*, 22(17), 13336-13343.
- Armstrong, W., Armstrong, J. & Beckett, P. 1990. Measurement and Modeling of Oxygen Release from Roots of *Phragmites Australis*. Pags. 41-52 en Cooper, P. F. & Findlater, B.C. (Eds.),". Proceedings of the Internacional Conference on the Use of Constructed Wetlands in Water Pollution Control". Pergamon Press.
- Arora, P. K. 2015. Bacterial degradation of monocyclic aromatic amines. *Frontiers* in microbiology, 6, 1-14.
- Aslam, M. M., Hassan, S. & Baig, M. 2010. Removal of metals from the refinery wastewater through vertical flow constructed wetlands. *International Journal of Agriculture and Biology*, 12(5), 796-798.
- Austin, G. & Yu, K. 2016. Constructed Wetlands and Sustainable Development, *New York, Routledge*.
- Ávila, C., Nivala, J., Olsson, L., Kassa, K., Headley, T., Mueller, R. A., Bayona, J. M. & García, J. 2014. Emerging organic contaminants in vertical subsurface flow constructed wetlands: influence of media size, loading frequency and use of active aeration. *Science of the total environment*, 494-495(211-217.
- Ávila Martín, C. 2013. Effect of design and operational factors on the removal of emerging organic contaminants in constructed wetlands for wastewater treatment. PhD, Politècnica de Catalunya.
- Ayaz, S. Ç., Aktaş, Ö., Akça, L. & Fındık, N. 2016. Full-scale hybrid constructed wetlands incorporated with an initial anaerobic stage for domestic wastewater treatment in a drinking water catchment area. *Desalination and Water Treatment*, 57(19), 8626-8638.
- Baban, A., Yediler, A. & Ciliz, N. K. 2010. Integrated water management and CP implementation for wool and textile blend processes. *Clean–Soil, Air, Water*, 38(1), 84-90.
- Babel, S. & Kurniawan, T. A. 2004. Cr (VI) removal from synthetic wastewater using coconut shell charcoal and commercial activated carbon modified with oxidizing agents and/or chitosan. *Chemosphere*, 54(7), 951-967.
- Babu, B. R., Parande A.K., Raghu S. & T., P. K. 2007. Textile technology Cotton Textile Processing: Waste Generation and Effluent Treatment. *The Journal* of Cotton Science, 11, 141-153.

- Bachand, P. A. & Horne, A. J. 1999. Denitrification in constructed free-water surface wetlands: II. Effects of vegetation and temperature. *Ecological Engineering*, 14(1), 17-32.
- Ballantine, D. J. & Tanner, C. C. 2010. Substrate and filter materials to enhance phosphorus removal in constructed wetlands treating diffuse farm runoff: a review. *New Zealand Journal of Agricultural Research*, 53(1), 71-95.
- Barakat, M. 2011. New trends in removing heavy metals from industrial wastewater. *Arabian Journal of Chemistry*, 4(4), 361-377.
- Barbour, M. G. & Billings, W. D. 2000. North American terrestrial vegetation, USA, Cambridge University Press.
- Bastian, R. & Hammer, D. 1993. The use of constructed wetlands for wastewater treatment and recycling. *In:* Moshiri, G. A. (ed.) *Constructed wetlands for water quality improvement. Boca Raton, FL: CRC Press.*
- Belal, A. R., Jamali, D., Karam, C. & Blowfield, M. 2015. Social and environmental accountability in developing countries. *In:* Dima, J., Charlotte, K. & Michael, B. (eds.) *Development-Oriented Corporate Social Responsibility: Multinational Corporations and the Global Context. Sheffield, UK: Greenleaf*
- Bell, C. L., Case, D. R. & Brownell, W. F. 2011. Environmental law handbook, *Plymouth, UK, Government Institutes*.
- Bendoricchio, G., Cin, L. & Persson, J. 2000. Guidelines for free water surface wetland design. *EcoSys Bd*, 8, 51-91.
- Bezbaruah, A. N. & Zhang, T. C. 2003. Performance of a constructed wetland with a sulfur/limestone denitrification section for wastewater nitrogen removal. *Environmental science & technology*, 37(8), 1690-1697.
- Białowiec, A., Janczukowicz, W. & Randerson, P. F. 2011. Nitrogen removal from wastewater in vertical flow constructed wetlands containing LWA/gravel layers and reed vegetation. *Ecological Engineering*, 37(6), 897-902.
- Bilotta, G. & Brazier, R. 2008. Understanding the influence of suspended solids on water quality and aquatic biota. *Water research*, 42(12), 2849-2861.
- Bojcevska, H. & Tonderski, K. 2007. Impact of loads, season, and plant species on the performance of a tropical constructed wetland polishing effluent from sugar factory stabilization ponds. *Ecological engineering*, 29(1), 66-76.
- Boon, A. 1986. Report of a Visit by Members and Staff of WRc to Germany (FRG) to Investigate the Root Zone Method of Treatment of Wastewaters. Report 376-s. *Water Research Centre, Stevenage*.
- Boyd, C. E. 2000. Dissolved Oxygen and Redox Potential. Water Quality: An Introduction. Boston, MA: Springer.
- Boyd, C.E., Gautier, D., 2000. Effluent composition and water quality standards. Global Aquaculture Advocate, 3(5), 61–66
- Breaux, A., Farber, S. & Day, J. 1995. Using natural coastal wetlands systems for wastewater treatment: an economic benefit analysis. *Journal of environmental management*, 44(3), 285-291.
- Brillas, E. & Martínez-Huitle, C. A. 2015. Decontamination of wastewaters containing synthetic organic dyes by electrochemical methods. An updated review. *Applied Catalysis B: Environmental*, 166-167, 603-643.

- Brix, H. 1987. Applicability of the wastewater treatment plant in Othfresen as scientific demonstration of the root-zone method. *Water Science and Technology*, 19(10), 19-24.
- Brix, H. 1994a. Constructed wetlands for municipal wastewater treatment in Europe, *Amsterdam, Elsevier*.
- Brix, H. 1994b. Functions of macrophytes in constructed wetlands. *Water Science and Technology*, 29(4), 71-78.
- Brix, H. 1994c. Use of constructed wetlands in water pollution control: historical development, present status, and future perspectives. *Water science and technology*, 30(8), 209-223.
- Brix, H. 1997. Do macrophytes play a role in constructed treatment wetlands? *Water Science and Technology*, 35(5), 11-17.
- Brix, H. Plants used in constructed wetlands and their functions. 1st International Seminar on the use of Aquatic Macrophytes for Wastewater Treatment in Constructed Wetlands, edit. Dias V., Vymazal J. Lisboa, Portugal, 2003. 81-109.
- Brix, H., Arias, C. & Del Bubba, M. 2001. Media selection for sustainable phosphorus removal in subsurface flow constructed wetlands. *Water Science and Technology*, 44(11-12), 47-54.
- Brix, H. & Arias, C. A. 2005. The use of vertical flow constructed wetlands for onsite treatment of domestic wastewater: New Danish guidelines. *Ecological engineering*, 25(5), 491-500.
- Brix, H., Dyhr-Jensen, K. & Lorenzen, B. 2002. Root-zone acidity and nitrogen source affects Typha latifolia L. growth and uptake kinetics of ammonium and nitrate. *Journal of Experimental Botany*, 53(379), 2441-2450.
- Brown, D. & Hamburger, B. 1987. The degradation of dyestuffs: Part III-Investigations of their ultimate degradability. *Chemosphere*, 16(7), 1539-1553.
- Brown, M. A. & De Vito, S. C. 1993. Predicting azo dye toxicity. *Critical reviews in environmental science and technology*, 23(3), 249-324.
- Bulc, T. G. 2006. Long term performance of a constructed wetland for landfill leachate treatment. *Ecological engineering*, 26(4), 365-374.
- Bulc, T. G., Ojstršek, A. & Vrhovšek, D. 2006. The use of constructed wetland for textile wastewater treatment. Proceedings of the 10th International Conference on Wetland Systems for Water Pollution Control. 1667-1675, Lisbon, Portugal
- Bulc, T. G. & Ojtrsek, A. 2008. The use of constructed wetland for dye-rich textile wastewater treatment. *Journal of Hazardous Materials*, 155(1), 76-82.
- Butterworth, E., Richards, A., Jones, M., Dotro, G. & Jefferson, B. 2016. Assessing the potential for tertiary nitrification in sub-surface flow constructed wetlands. *Environmental Technology Reviews*, 5(1), 68-77.
- Cakmak, B. & Apaydin, H. 2010. Review. Advances in the management of the wastewater in Turkey: natural treatments or constructed wetlands. *Spanish Journal of Agricultural Research*, 8(1), 188-201.
- Carias, C. C., Novais, J. M. & Martins-Dias, S. 2008. Are *Phragmites australis* enzymes involved in the degradation of the textile azo dye acid orange 7? *Bioresource Technology*, 99(2), 243-251.

- Carliell, C., Barclay, S., Naidoo, N., Buckley, C., Mulholland, D. & Senior, E. 1994. Anaerobic decolorisation of reactive dyes in conventional sewage treatment processes. *Water S. A.*, 20(4), 341.
- Carmen, Z. & Daniela, S. 2012. Textile organic dyes-characteristics, polluting effects and separation/elimination procedures from industrial effluents-a critical overview. Organic Pollutants Ten Years After the Stockholm Convention-Environmental and Analytical Update. InTech: Croatia, 55-81.
- Carter, V. 1996. Technical aspects of wetlands. Wetland hydrology, water quality, and associated functions. United States Geological Survey Water Supply paper 2425. US Geological Survey, USA.
- Caselles-Osorio, A., Puigagut, J., Segú, E., Vaello, N., Granés, F., García, D. & García, J. 2007. Solids accumulation in six full-scale subsurface flow constructed wetlands. *Water Research*, 41(6), 1388-1398.
- Castro, M. C. R., Belsley, M. & Raposo, M. M. M. 2016. Synthesis and characterization of push–pull bithienylpyrrole NLOphores with enhanced hyperpolarizabilities. *Dyes and Pigments*, 131, 333-339.
- Cerezo, R. G., Suárez, M. & Vidal-Abarca, M. 2001. The performance of a multistage system of constructed wetlands for urban wastewater treatment in a semiarid region of SE Spain. *Ecological Engineering*, 16(4), 501-517.
- Chai, Y., Ding, H., Zhang, Z., Xian, Y., Pan, Z. & Jin, L. 2006. Study on photocatalytic oxidation for determination of the low chemical oxygen demand using a nano-TiO 2–Ce (SO 4) 2 coexisted system. *Talanta*, 68(3), 610-615.
- Chan, E., Bursztynsky, T., Hantzsche, N. & Litwin, Y. 1982. The use of wetlands for water pollution control. U.S. EPA, EPA-600/2-82-086. Municipal environmental laboratory, Cinncinatti, Ohio.
- Chandra, R. 2015. Environmental waste management, Boca Raton, FL, CRC Press.
- Chapanova, G., Jank, M., Schlegel, S. & Koeser, H. 2007. Effect of temperature and salinity on the wastewater treatment performance of aerobic submerged fixed bed biofilm reactors. *Water science and technology*, 55(8-9), 159-164.
- Chen, J., Wang, Q., Hua, Z. & Du, G. 2007. Research and application of biotechnology in textile industries in China. *Enzyme and Microbial Technology*, 40(7), 1651-1655.
- Cheng, B., Hu, C. & Zhao, Y. 2011. Effects of plants development and pollutant loading on performance of vertical subsurface flow constructed wetlands. *International Journal of Environmental Science & Technology*, 8(1), 177-186.
- Cheng, S., Grosse, W., Karrenbrock, F. & Thoennessen, M. 2002. Efficiency of constructed wetlands in decontamination of water polluted by heavy metals. *Ecological engineering*, 18(3), 317-325.
- China Water Risk. 2011. *The Environmental Cost of Clothes* [Online]. Available: <u>http://chinawaterrisk.org/resources/analysis-reviews/the-environmental-cost-of-clothes/</u> 2016].
- Choudhary, A. K., Kumar, S. & Sharma, C. 2011. Constructed wetlands: an approach for wastewater treatment. *Elixir Pollution*, 37(8), 3666-3672.
- Chung, K. T. 2000. Mutagenicity and carcinogenicity of aromatic amines metabolically produced from azo dyes. *Journal of Environmental Science* & *Health Part C*, 18(1), 51-74.

- Chung, K.T. 1983. The significance of azo-reduction in the mutagenesis and carcinogenesis of azo dyes. *Mutation Research/Reviews in Genetic Toxicology*, 114(3), 269-281.
- Chung, K. T., Stevens, S. E. & Cerniglia, C. E. 1992. The Reduction of Azo Dyes by the Intestinal Microflora. *Critical Reviews in Microbiology*, 18(3), 175-190.
- Chung, K. T. & Stevens, S. E. 1993. Degradation azo dyes by environmental microorganisms and helminths. *Environmental Toxicology and Chemistry*, 12(11), 2121-2132.
- Coban, O., Kuschk, P., Wells, N. S., Strauch, G. & Knoeller, K. 2015. Microbial nitrogen transformation in constructed wetlands treating contaminated groundwater. *Environmental Science and Pollution Research*, 22(17), 12829-12839.
- Connor, R. 2015. The United Nations world water development report 2015: water for a sustainable world, *Paris*, *UNESCO Publishing*.
- Cooper, P. & Green, B. 1998. United Kingdom. In: Vymazal, J., Brix, H., Cooper, P., Green, M. & Haberl, R. (eds.) Constructed Wetlands for Wastewater Treatment in Europe. Leiden, The Netherlands: Backhuys Publishers, 315-335.
- Cooper, P. 2001. Constructed wetlands and reed-beds: Mature technology for the treatment of wastewater from small populations. *Water and Environment Journal*, 15(2), 79-85.
- Cooper, P. 2005. The performance of vertical flow constructed wetland systems with special reference to the significance of oxygen transfer and hydraulic loading rates. *Water Science and Technology*, 51(9), 81-90.
- Cooper, P. & Green, B. 1995. Reed bed treatment systems for sewage treatment in the United Kingdom-the first 10 years' experience. *Water Science and Technology*, 32(3), 317-327.
- Cooper, P., Griffin, P., Humphries, S. & Pound, A. 1999. Design of a hybrid reed bed system to achieve complete nitrification and denitrification of domestic sewage. *Water science and technology*, 40(3), 283-289.
- Cooper, P. F. & Findlater, B. 2013. Constructed Wetlands in Water Pollution Control: Proceedings of the International Conference on the Use of Constructed Wetlands in Water Pollution Control, Held in Cambridge, UK, 24–28 September 1990, *Elsevier*.
- Cooper, P. F., Job, G. D., Green, M. B. & Shutes, R. B. E. 1996. Reeds Beds and Constructed Wetlands for Wastewater Treatment *Swindon*, *UK.*, *Water Research Centre*.
- Corcoran, E. 2010. Sick water?: the central role of wastewater management in sustainable development: a rapid response assessment, *United Nation Environmental Programme/Earthprint. Norway, Birkland Trykkeri AS.*
- Correia, V. M., Stephenson, T. & Judd, S. J. 1994. Characterisation of textile wastewaters-a review. *Environmental technology*, 15(10), 917-929.
- Costa, M. C., Mota, F. S. B., Santos, A. B. D., Mendonça, G. L. F. & Nascimento, R. F. D. 2012. Effect of dye structure and redox mediators on anaerobic azo and anthraquinone dye reduction. *Quimica Nova*, 35(3), 482-486.
- Costanza, R., D'arge, R., De Groot, R., Farber, S., Grasso, M., Hannon, B., Limburg, K., Naeem, S., O'neill, R. V. & Paruelo, J. 2016. The Value of the

World's Ecosystem Services and Natural Capital (1997). *The Globalization and Environment Reader*, 117.

- Crini, G. & Badot, P.-M. 2010. Sorption processes and pollution: conventional and non-conventional sorbents for pollutant removal from wastewaters, *Presses* Univ. Franche-Comté.
- Crites, R. W., Middlebrooks, E. J. & Bastian, R. K. 2014. Natural wastewater treatment systems, *CRC Press*.
- Cui, L., Zhu, X., Ma, M., Ouyang, Y., Dong, M., Zhu, W. & Luo, S. 2008. Phosphorus sorption capacities and physicochemical properties of nine substrate materials for constructed wetland. *Archives of environmental contamination and toxicology*, 55(2), 210-217.
- Cumnan, S. & Yimrattanabovorn, J. 2012. The Use of Constructed Wetland for Azo Dye Textile Wastewater. *International Journal of Civil Engineering and Building Materials*, 4(2), 150-158.
- Dąbrowski, W., Karolinczak, B., Gajewska, M. & Wojciechowska, E. 2017. Application of subsurface vertical flow constructed wetlands to reject water treatment in dairy wastewater treatment plant. *Environmental Technology*, 38(2), 175-182.
- Dasgupta, J., Sikder, J., Chakraborty, S., Curcio, S. & Drioli, E. 2015. Remediation of textile effluents by membrane based treatment techniques: a state of the art review. *Journal of environmental management*, 147, 55-72.
- Davies, L., Carias, C., Novais, J. & Martins-Dias, S. 2005. Phytoremediation of textile effluents containing azo dye by using Phragmites australis in a vertical flow intermittent feeding constructed wetland. *Ecological Engineering*, 25(5), 594-605.
- Davies, L. C., Cabrita, G., Ferreira, R., Carias, C., Novais, J. & Martins-Dias, S. 2009. Integrated study of the role of Phragmites australis in azo-dye treatment in a constructed wetland: From pilot to molecular scale. *ecological engineering*, 35(6), 961-970.
- Davies, L. C., Pedro, I. S., Novais, J. M. & Martins-Dias, S. 2006. Aerobic degradation of acid orange 7 in a vertical-flow constructed wetland. *Water Research*, 40(10), 2055-2063.
- Davies, T. & Cottingham, P. 1994. The use of constructed wetlands for treating industrial effluent (textile dyes). *Water Science and Technology*, 29(4), 227-232.
- Dawson, E. & Macklin, M. 1998. Speciation of heavy metals on suspended sediment under high flow conditions in the River Aire, West Yorkshire, UK. *Hydrological Processes*, 12(9), 1483-1494.
- De La Varga, D., Díaz, M., Ruiz, I. & Soto, M. 2013. Avoiding clogging in constructed wetlands by using anaerobic digesters as pre-treatment. *Ecological engineering*, 52, 262-269.
- Debusk, W. F. 1999. Wastewater treatment wetlands: Applications and treatment efficiency, *University of Florida Cooperative Extension Service, Institute of Food and Agriculture Sciences, EDIS.*
- Deniz, F. & Karaman, S. 2011. Removal of Basic Red 46 dye from aqueous solution by pine tree leaves. *Chemical Engineering* 170(1), 67-74.
- Deniz, F. & Saygideger, S. D. 2011. Removal of a hazardous azo dye (Basic Red 46) from aqueous solution by princess tree leaf. *Desalination*, 268(1), 6-11.

- Dey, S. & Islam, A. 2015. A Review on Textile Wastewater Characterization in Bangladesh. *Resources and Environment*, 5(1), 15-44.
- Ding, H. C., Chai, Y. H., Zhang, Z. H., Liu, M. C., Xian, Y. Z., Pan, Z. S. & Jin, L. T. 2005. Photocatalytic Oxidation for Determination of Chemical Oxygen Demand Using Nano-TiO2 Film. *Chinese Journal of Chemistry*, 23(10), 1425-1429.
- Ding, L. & Shen, Y. 2006. The treatment technology of constructed wetland and its research progress. *Jiangsu Environmental Science and Technology*, 19, 34-37.
- Ding, S., Li, Z. & Wangrui 2010. Overview of dyeing wastewater treatment technology. *Water resources protection*, 26, 73-78.
- Ding, Y., Song, X., Wang, Y. & Yan, D. 2012. Effects of dissolved oxygen and influent COD/N ratios on nitrogen removal in horizontal subsurface flow constructed wetland. *Ecological Engineering*, 46, 107-111.
- Dobo, K. L., Greene, N., Cyr, M. O., Caron, S. & Ku, W. W. 2006. The application of structure-based assessment to support safety and chemistry diligence to manage genotoxic impurities in active pharmaceutical ingredients during drug development. *Regulatory Toxicology and Pharmacology*, 44(3), 282-293.
- Dordio, A. V. & Carvalho, A. J. P. 2013. Organic xenobiotics removal in constructed wetlands, with emphasis on the importance of the support matrix. *Journal of Hazardous materials*, 252, 272-292.
- Dos Santos, A. B., Cervantes, F. J. & Van Lier, J. B. 2007. Review paper on current technologies for decolourisation of textile wastewaters: perspectives for anaerobic biotechnology. *Bioresource technology*, 98(12), 2369-2385.
- Drizo, A., Frost, C., Smith, K. & Grace, J. 1997. Phosphate and ammonium removal by constructed wetlands with horizontal subsurface flow, using shale as a substrate. *Water Science and Technology*, 35(5), 95-102.
- Du, X., Shi, C. & Ma, F. 2016. Influence of intermittent aeration and organic loading rate on lab-scale constructed wetland systems treating synthetic wastewater. *Desalination and Water Treatment*, 57(21), 9651-9659.
- Dunbabin, J. S. & Bowmer, K. H. 1992. Potential use of constructed wetlands for treatment of industrial wastewaters containing metals. *Science of the Total environment*, 111(2-3), 151-168.
- Dzurik, A. A. 2003. Water Resources Planning, USA, Rowman & Littlefield Publishers.
- Economopoulou, M. A. & Tsihrintzis, V. A. 2004. Design methodology of free water surface constructed wetlands. *Water resources management*, 18(6), 541-565.
- Edwards, J. C. 2000. *Investigation of color removal by chemical oxidation for three reactive textile dyes and spent textile dye wastewater*. Virginia Polytechnic Institute and State University.
- Ekambaram, S. 2016. Engineering Chemistry: Rare Topics, India, Partridge.
- Eke, P. E. 2008. Hydrocarbon removal with constructed wetlands: Design and operation of experimental hybrid constructed wetlands applied for hydrocarbon treatment, and application of an artificial neural network to support constructed wetlands optimization and management, *University of Edinburgh*.

- Eke, P. E. & Scholz, M. 2008. Benzene removal with vertical-flow constructed treatment wetlands. *Journal of chemical technology and biotechnology*, 83(1), 55-63.
- Ekici, P., Leupold, G. & Parlar, H. 2001. Degradability of selected azo dye metabolites in activated sludge systems. *Chemosphere*, 44(4), 721-728.
- El Qada, E. N., Allen, S. J. & Walker, G. M. 2006. Adsorption of methylene blue onto activated carbon produced from steam activated bituminous coal: a study of equilibrium adsorption isotherm. *Chemical Engineering Journal*, 124(1), 103-110.
- Ellis, J., Shutes, R. & Revitt, D. 2003a. Constructed Wetlands and Links with Sustainable Drainage Systems. Environment Agency, *Bristol*, *Environmental agency*.
- Ellis, J., Shutes, R. & Revitt, D. 2003b. Guidance manual for constructed wetlands, *London, Environment Agency*.
- Ellis, T. G. 2004. Chemistry of wastewater. *Encyclopedia of Life Support System* (EOLSS).
- Elzein, Z., Abdou, A. & Elgawad, I. A. 2016. Constructed Wetlands as a Sustainable Wastewater Treatment Method in Communities. *Procedia Environmental Sciences*, 34, 605-617.
- Erkurt, H. A. 2010. Biodegradation of Azo Dyes, Springer Science & Business Media.
- Evangelou, V. 1998. Environmental soil and water chemistry, NY, John Wiley.
- Fan, J., Liang, S., Zhang, B. & Zhang, J. 2013a. Enhanced organics and nitrogen removal in batch-operated vertical flow constructed wetlands by combination of intermittent aeration and step feeding strategy. *Environmental Science and Pollution Research*, 20(4), 2448-2455.
- Fan, J., Wang, W., Zhang, B., Guo, Y., Ngo, H. H., Guo, W., Zhang, J. & Wu, H. 2013b. Nitrogen removal in intermittently aerated vertical flow constructed wetlands: impact of influent COD/N ratios. *Bioresource technology*, 143, 461-466.
- Fan, J., Zhang, B., Zhang, J., Ngo, H. H., Guo, W., Liu, F., Guo, Y. & Wu, H. 2013c. Intermittent aeration strategy to enhance organics and nitrogen removal in subsurface flow constructed wetlands. *Bioresource technology*, 141, 117-122.
- Fergusson, J. E. 1990. Heavy elements: chemistry, environmental impact and health effects, *Oxford*, *Pergamon Press*.
- Ferreira, R. A., Duarte, J. G., Vergine, P., Antunes, C. D., Freire, F. & Martins-Dias 2014. Phragmites sp. physiological changes in a constructed wetland treating an effluent contaminated with a diazo dye (DR81). *Environmental Science and Pollution Research*, 21(16), 9626-9643.
- Fihtengolts, V. 1969. Atlas of UV Absorption Spectra of Substances Used in Synthetic Rubber Manufacture. NIST Chemistry WebBook, NIST Standard Reference Database, 69.
- Fleming, I., Rowe, R. & Cullimore, D. R. 1999. Field observations of clogging in a landfill leachate collection system. *Canadian Geotechnical Journal*, 36(4), 685-707.
- Foladori, P., Ortigara, A., Ruaben, J. & Andreottola, G. 2012. Influence of high organic loads during the summer period on the performance of hybrid

constructed wetlands (VSSF+ HSSF) treating domestic wastewater in the Alps region. *Water Science and Technology*, 65(5), 890-897.

- Fonder, N. & Headley, T. 2010. Systematic classification, nomenclature and reporting for constructed treatment wetlands. *Water and nutrient management in natural and constructed wetlands. Springer.*
- Fréchette, G. R. & Schotter, A. 2015. Handbook of experimental economic methodology, *Oxford University Press, USA*.
- Fu, F. & Wang, Q. 2011. Removal of heavy metal ions from wastewaters: a review. *Journal of environmental management*, 92(3), 407-418.
- Fu, G., Zhang, J., Chen, W. & Chen, Z. 2013. Medium clogging and the dynamics of organic matter accumulation in constructed wetlands. *Ecological engineering*, 60, 393-398.
- Fuchs, V. J. 2009. Nitrogen Removal and Sustainability in Vertical Flow Conctructed Wetlands for Small Scale Wastewater Treatment. PhD, Michigan Technology University.
- Gadaleta, D., Manganelli, S., Manganaro, A., Porta, N. & Benfenati, E. 2016. A knowledge-based expert rule system for predicting mutagenicity (Ames test) of aromatic amines and azo compounds. *Toxicology*, 370, 20-30.
- Gadd, G. M. 1990. Heavy metal accumulation by bacteria and other microorganisms. *Experientia*, 46(8), 834-840.
- Gajewska, M. 2011. Fluctuation of nitrogen fraction during wastewater treatment in a multistage treatment wetland. *Environment Protection Engineering*, 37(3), 119-128.
- Gambrell, R. 1994. Trace and toxic metals in wetlands—a review. Journal of Environmental Quality, 23(5), 883-891.
- García-Pérez, A., Harrison, M., Chivers, C. & Grant, B. 2015. Recycled Shredded-Tire Chips Used As Support Material in a Constructed Wetland Treating High-Strength Wastewater from a Bakery: Case Study. *Recycling*, 1(1), 3-13.
- Garcia, J., Rousseau, D. P., Morato, J., Lesage, E., Matamoros, V. & Bayona, J. M. 2010. Contaminant removal processes in subsurface-flow constructed wetlands: a review. *Critical Reviews in Environmental Science and Technology*, 40(7), 561-661.
- Gargi Sharma, Priya & Brighu, U. 2014. Performance Analysis of Vertical Up-flow Constructed Wetlands for Secondary Treated Effluent. *APCBEE Procedia*, 10, 110-114.
- Ge, Y., Wang, X. C., Dzakpasu, M., Zheng, Y., Zhao, Y. & Xiong, J. 2016. Characterizing phosphorus removal from polluted urban river water by steel slags in a vertical flow constructed wetland. Water Science and Technology, 73(11), 2644-2653.
- Gessner, T. P., Kadlec, R. H. & Reaves, R. P. 2005. Wetland remediation of cyanide and hydrocarbons. *Ecological engineering*, 25(4), 457-469.
- Ghermandi, A., Van Den Bergh, J. C., Brander, L. M., De Groot, H. L. & Nunes, P. A. 2009. The values of natural and constructed wetlands: a meta-analysis. Tinbergen Institute discussion paper. *Amsterdam*.
- Ghermandi, A., Van Den Bergh, J. C., Brander, L. M., De Groot, H. L. & Nunes, P. A. 2010. Values of natural and human-made wetlands: A meta-analysis. *Water Resources Research*, 46(12).

- Ghosh, D. & Gopal, B. 2010. Effect of hydraulic retention time on the treatment of secondary effluent in a subsurface flow constructed wetland. *Ecological Engineering*, 36(8), 1044-1051.
- Gikas, G., Akratos, C. & Tsihrintzis, V. 2007. Performance monitoring of a vertical flow constructed wetland treating municipal wastewater. *Global nest. The international journal*, 9(3), 277-285.
- Gikas, G. & Tsihrintzis, V. 2010. On-site treatment of domestic wastewater using a small-scale horizontal subsurface flow constructed wetland. *Water Science and Technology*, 62(3), 603-614.
- Golob, V., Vinder, A. & Simonič, M. 2005. Efficiency of the coagulation/flocculation method for the treatment of dyebath effluents. *Dyes and pigments*, 67(2), 93-97.
- Gopinathan, P. & Nair, V. G. 2011. Degradation of Azo Dye Pollutants in the Textile Wastewater Using Advanced Oxidation Process.
- Gorra, R., Coci, M., Ambrosoli, R. & Laanbroek, H. 2007. Effects of substratum on the diversity and stability of ammonia-oxidizing communities in a constructed wetland used for wastewater treatment. *Journal of applied microbiology*, 103(5), 1442-1452.
- Greaney, K. M. 2005. An assessment of heavy metal contamination in the marine sediments of Las Perlas Archipelago, Gulf of Panama. MSc, Heriot-Watt
- Green, M. 1996. Experience with establishment and operation of reed bed treatment for small communities in the UK. *Wetlands Ecology and Management*, 4(3), 147-158.
- Green, M., Friedler, E., Ruskol, Y. & Safrai, I. 1997. Investigation of alternative method for nitrification in constructed wetlands. *Water Science and Technology*, 35(5), 63-70.
- Green, M. & Martin, J. 1996. Constructed reed beds clean up storm overflows on small wastewater treatment works. *Water environment research*, 68(6), 1054-1060.
- Green, M. & Upton, J. 1993. Reed bed treatment for small communities: UK experience. *Constructed wetlands for water quality improvement*, 517-524.
- Greene, J. C. & Baughman, G. L. 1996. Effects of 46 dyes on population growth of freshwater green alga selenastrum capricornutum. *Textile Chemist and Colorist*, 28, 23-30.
- Grekova, M., Popov, I., Vassilev, D. & Topalova, Y. 2009. Isolation and characterisation of microbial strain AZO29 capable of azo dye decolourization. *Biotechnology & Biotechnological Equipment*, 23(sup1), 318-322.
- Gross, A., Shmueli, O., Ronen, Z. & Raveh, E. 2007. Recycled vertical flow constructed wetland (RVFCW)—a novel method of recycling greywater for irrigation in small communities and households. *Chemosphere*, 66(5), 916-923.
- Gudelj, I., Hrenović, J., Dragičević, T., Delaš, F. & Gudelj, H. 2011. Azo dyes, their environmental effects, and defining a strategy for their biodegradation and detoxification. *Archives of Industrial Hygiene and Toxicology*, 62(1), 91-101.
- Guimaraes, G. P., Alves, D. G., Jorge, M. F., Nascentes, A. L., De Pinho, C. F. & Da Silva, L. D. 2016. Removal of nitrogen and phosphorus from cattle

farming wastewater using constructed wetland system. *African Journal of Agricultural Research*, 11(44), 4542-4550.

- Håkanson, L. 2006. The relationship between salinity, suspended particulate matter and water clarity in aquatic systems. Ecological Research, 21(1), 75-90.
- Haberl, R., Grego, S., Langergraber, G., Kadlec, R. H., Cicalini, A.-R., Dias, S. M., Novais, J. M., Aubert, S., Gerth, A. & Thomas, H. 2003. Constructed wetlands for the treatment of organic pollutants. *Journal of soils and sediments*, 3(2), 109.
- Haberl, R., Perfler, R. & Mayer, H. 1995. Constructed wetlands in Europe. *Water Science and Technology*, 32(3), 305-315.
- Hallberg, K. B. & Johnson, D. B. 2005. Biological manganese removal from acid mine drainage in constructed wetlands and prototype bioreactors. *Science of the Total Environment*, 338(1), 115-124.
- Halverson, N. 2004. Review of constructed subsurface flow vs. surface flow wetlands. *Savannah River Site (US)*.
- Hamilton, H., Nix, P. & Sobolewski, A. 1993. An overview of constructed wetlands as alternatives to conventional waste treatment systems.
- Hammer, D. A. & Bastian, R. K. 1989. Wetland ecosystems: natural water purifiers. In: Hammer, D. (ed.) Constructed wetlands for wastewater treatment: municipal, industrial and agricultural. Boca Raton, Florida: Lewis.
- Haroun, M. & Idris, A. 2009. Treatment of textile wastewater with an anaerobic fluidized bed reactor. *Desalination*, 237(1), 357-366.
- Haslam, S. M. 1972. Phragmites communis Trin.(Arundo Phragmites L.,? Phragmites australis (Cav.) Trin. ex Steudel). *The journal of ecology*, 585-610.
- Hassaan, M. A., El Nemr, A. & Madkour, F. F. 2016. Testing the advanced oxidation processes on the degradation of Direct Blue 86 dye in wastewater. *The Egyptian Journal of Aquatic Research*, 43(1), 11-19.
- Hatch, K. L. & Maibach, H. I. 1995. Textile dye dermatitis. *Journal of the American Academy of Dermatology*, 32(4), 631-639.
- Hatice, A. E. 2010. Biodegradation of azo dyes. *Handbook of environmental chemistry. Springer, London.*
- Haule, A., Pratap, H., Katima, H., Mbwette, T. & Njau, K. Nitrogen removal from domestic wastewater in subsurface flow constructed wetlands by indigenous macrophytes in the tropics: a comparative study for six potential macrophytes in Tanzania. Eighth IWA international conference on the use of constructed wetlands in water pollution control, 2000. 16-19.
- Haygarth, P., Bilotta, G., Bol, R., Brazier, R., Butler, P., Freer, J., Gimbert, L., Granger, S., Krueger, T. & Macleod, C. 2006. Processes affecting transfer of sediment and colloids, with associated phosphorus, from intensively farmed grasslands: an overview of key issues. *Hydrological Processes*, 20(20), 4407-4413.
- Heal, K., Dobbie, K., Bozika, E., Mchaffie, H., Simpson, A. & Smith, K. 2005. Enhancing phosphorus removal in constructed wetlands with ochre from mine drainage treatment. *Water Science and Technology*, 51(9), 275-282.
- Hedin, R. S., Nairn, R. W. & Kleinmann, R. L. 1994. Passive treatment of coal mine drainage, US Dept. of the Interior, Bureau of Mines.
- Helmer, R., Hespanhol, I., Supply, W. & Council, S. C. 1997. Water pollution control: a guide to the use of water quality management principles, *London*, *E & FN Spon*.
- Hernández, M. 2013. Ecological engineering for controlling water pollution in Latin America, UK, Wit Press.
- Hoffmann, H., Platzer, C., Winker, M. & Muench, E. V. 2011. Technology review of constructed wetlands Subsurface flow constructed wetlands for greywater and domestic wastewater treatment. *Deutsche Gesellschaft für, Internationale Zusammenarbeit (GIZ) GmbH, Sustainable sanitationecosan program, Postfach,* 5180, p.65726.
- Holkar, C. R., Pandit, A. B. & Pinjari, D. V. 2014. Kinetics of biological decolorisation of anthraquinone based Reactive Blue 19 using an isolated strain of Enterobacter sp. F NCIM 5545. *Bioresource technology*, 173, 342-351.
- Hou, Y., Liu, X., Zhang, X., Hu, X. & Cao, L. 2016. Rhizosphere phytoremediation with Cyperus rotundus for diesel-contaminated wetlands. Water, Air, & Soil Pollution, 227(1), 26.
- Hou, J., Xia, L., Ma, T., Zhang, Y., Zhou, Y. & He, X. 2017. Achieving short-cut nitrification and denitrification in modified intermittently aerated constructed wetland. *Bioresource Technology*, 232, 10-17
- Hua, G.-F., Zhu, W. & Zhang, Y.-H. Effects of Solubilization Treatment and Intermittent Operation on Clogging in Vertical Flow Constructed Wetland. Intelligent System Design and Engineering Applications (ISDEA), 2013 Third International Conference on, 2013. IEEE, 696-699.
- Hua, G., Zhao, Z., Kong, J., Guo, R., Zeng, Y., Zhao, L. & Zhu, Q. 2014. Effects of plant roots on the hydraulic performance during the clogging process in mesocosm vertical flow constructed wetlands. *Environmental Science and Pollution Research*, 21(22), 13017-13026.
- Hua, G., Zhu, W., Zhao, L. & Huang, J. 2010. Clogging pattern in vertical-flow constructed wetlands: Insight from a laboratory study. *Journal of Hazardous Materials*, 180(1), 668-674.
- Hunt, P. & Poach, M. 2001. State of the art for animal wastewater treatment in constructed wetlands. *Water Science and Technology*, 44(11-12), 19-25.
- Hussein, A. & Scholz, M. 2017. Dye wastewater treatment by vertical-flow constructed wetlands. *Ecological Engineering*, 101, 28-38.
- Inamori, R., Gui, P., Dass, P., Matsumura, M., Xu, K.-Q., Kondo, T., Ebie, Y. & Inamori, Y. 2007. Investigating CH 4 and N 2 O emissions from ecoengineering wastewater treatment processes using constructed wetland microcosms. *Process Biochemistry*, 42(3), 363-373.
- Imfeld, G., Braeckevelt, M., Kuschk, P. & Richnow, H. H. 2009. Monitoring and assessing processes of organic chemicals removal in constructed wetlands. *Chemosphere*, 74(3), 349-362.
- India Ministry of Environment, Forest and Environment Change. 1986. *Environmental (Protection) Rules*. http://envfor.nic.in/legis/env/env4.html. [Accessed 09-05- 017].
- Islam, M., Mahmud, K., Faruk, O. & Billah, M. S. 2011. Textile Dyeing Industries in Bangladesh for Sustainable Development. *International Journal of Environmental Science and Development.*, 2(6), p.428

- Jadhav, R. S. & Buchberger, S. G. 1995. Effects of vegetation on flow through free water surface wetlands. *Ecological Engineering*, 5(4), 481-496.
- Järup, L. 2003. Hazards of heavy metal contamination. *British medical bulletin*, 68(1), 167-182.
- Jegatheesan, V. 2011. Evaluating the Performance of Horizontal SubsurfaceFlow Constructed Wetlands Using Natural Zeolite (escott). *International journal* of environmental science and development, 2(4), 311.
- Jerman, V., Metje, M., Mandić-Mulec, I. & Frenzel, P. 2009. Wetland restoration and methanogenesis: the activity of microbial populations and competition for substrates at different temperatures. *Biogeosciences*, 6(6), 1127-1138.
- Ji, B., Wang, H. & Yang, K. 2014. Nitrate and COD removal in an upflow biofilter under an aerobic atmosphere. *Bioresource technology*, 158, 156-160.
- Ji, G., Sun, T. & Ni, J. 2007. Surface flow constructed wetland for heavy oilproduced water treatment. *Bioresource Technology*, 98(2), 436-441.
- Ji, Y., Li, X.-T. & Chen, G.-Q. 2008. Interactions between a poly (3hydroxybutyrate-co-3-hydroxyvalerate-co-3-hydroxyhexanoate) terpolyester and human keratinocytes. *Biomaterials*, 29(28), 3807-3814.
- Jie, H., Daping, L., Qiang, L., Yong, T., Xiaohong, H., Xiaomei, W., Xudong, L. & Ping, G. 2009. Effect of organic carbon on nitrification effciency and community composition of nitrifying biofilms. *Environmental Sciences*, 21, 387-394.
- Johari, N. E., Abdul-Talib, S., Wahid, M. A. & Ghani, A. A. 2016. Trend of Total Phosphorus on Total Suspended Solid Reduction in Constructed Wetland Under Tropical Climate. ISFRAM 2015. *Singapore: Springer*.
- Johnston, C. 1993. Mechanisms of water wetland water quality interation. Contructed wetlands for water quality improvement. London: Lewis.
- Jørgensen, S. E. 2009. Applications in ecological engineering, Amsterdam, Elsevier.
- Joshi, M., Bansal, R. & Purwar, R. 2004. Colour removal from textile effluents. Indian journal of fibre & textile research, 29(2), 239-259.
- Jothimani, P., Kalaichelvan, G., Bhaskaran, A., Selvaseelan, D. A. & Ramasamy, K. 2003. Anaerobic biodegradation of aromatic compounds. *Indian journal* of experimental biology, 41(9), 1046-1067.
- Juang, R. S., Tseng, R. L., Wu, F. C. & Lin, S. J. 1996. Use of chitin and chitosan in lobster shell wastes for color removal from aqueous solutions. *Journal of Environmental Science & Health Part A*, 31(2), 325-338.
- Justin, M. Z., Vrhovšek, D., Stuhlbacher, A. & Bulc, T. G. 2009. Treatment of wastewater in hybrid constructed wetland from the production of vinegar and packaging of detergents. *Desalination*, 246(1-3), 100-109.
- Juwarkar, A., Oke, B., Juwarkar, A. & Patnaik, S. 1995. Domestic wastewater treatment through constructed wetland in India. *Water Science and Technology*, 32(3), 291-294.
- Kabra, A. N., Khandare, R. V. & Govindwar, S. P. 2013. Development of a bioreactor for remediation of textile effluent and dye mixture: A plant-bacterial synergistic strategy. *Water research*, 47(3), 1035-1048.
- Kadlec, R. H. & Knight, R. L. 1996. Treatment Wetlands, Florida, CRC Press.
- Kadlec, R. H. & Reddy, K. 2001. Temperature effects in treatment wetlands. *Water Environment Research*, 73(5), 543-557.

- Kadlec, R. H. & Wallace, S. 2008. Treatment wetlands, *Boca Raton, Florida, CRC Press.*
- Kallner, S. 2006. Nitrogen removal in treatment wetlands: Factors influencing spatial and temporal variations. PhD, Institutionen för fysik, kemi och biologi.
- Kalyani Kenkar & Kadlag, V. 2016. Study of dairy wastewater treatment by using constructed wetland. *International journal of advance research in science and engineering*, 5(12).
- Kalyuzhnyi, S., Sklyar, V., Mosolova, T., Kucherenko, I., Russkova, J. & Degtyaryova, N. 2000. Methanogenic biodegradation of aromatic amines. *Water science and technology*, 42(5-6), 363-370.
- Kamariah, M. 2006. Subsurface flow and free water surface flow constructed wetland with magnetic field for leachate treatment. Thesis of Master of Engineering (Civil-Wastewater): Universiti Teknologi Malaysia.
- Kampf, R. & Claassen, T. 2005. The use of treated wastewater for nature: the Waterharmonica, a sustainable solution as an alternative for separate drainage and treatment. 2nd IWA Leading-Edge on Water and Wastewater Treatment Technologies, 341-350.
- Kant, R. 2012. Textile dyeing industry an environmental hazard. *Natural Science*, 4(1), 22.
- Kantawanichkul, S. & Wannasri, S. 2013. Wastewater treatment performances of horizontal and vertical subsurface flow constructed wetland systems in tropical climate. *Songklanakarin Journal of Science and Technology*, 35(5), 599-603.
- Kanu, I. & Achi, O. 2011. Industrial effluents and their impact on water quality of receiving rivers in Nigeria. *Journal of applied technology in environmental sanitation*, 1(1), 75-86.
- Karathanasis, A., Potter, C. & Coyne, M. S. 2003. Vegetation effects on fecal bacteria, BOD, and suspended solid removal in constructed wetlands treating domestic wastewater. *Ecological engineering*, 20(2), 157-169.
- Karczmarczyk, A. 2004. Phosphorus removal from domestic wastewater in horizontal subsurface flow constructed wetland after 8 years of operation–a case study. *Journal of Environmental Engineering and Landscape Management*, 12(4), 126-131.
- Kaseva, M. 2004. Performance of a sub-surface flow constructed wetland in polishing pre-treated wastewater—a tropical case study. *Water research*, 38(3), 681-687.
- Kasuya, E. 2001. Mann–Whitney U test when variances are unequal. *Animal Behaviour*, 61(6), 1247-1249.
- Kayombo, S., Mbwette, T., Katima, J., Ladegaard, N. & Jørgensen, S. 2005. Waste stabilization ponds and constructed wetlands design manual. *UNEP-IETC* with the Danish International Development Agency (Danida).
- Kayombo, S., Mbwette, T., Katima, J., Ladegaard, N. & Jrgensen, S. 2004. Waste stabilization ponds and constructed wetlands: design manual.
- Keskinkan, O. & Lugal Göksu, M. 2007. Assessment of the dye removal capability of submersed aquatic plants in a laboratory-scale wetland system using anova. *Brazilian Journal of Chemical Engineering*, 24(2), 193-202.

- Kessel, M. A., Speth, D. R., Albertsen, M., Nielsen, P. H., Den Camp, H. J. O., Kartal, B., Jetten, M. S. & Lücker, S. 2015. Complete nitrification by a single microorganism. *Nature*, 528(7583), 555-559.
- Khandare, R. V., Kabra, A. N., Kadam, A. A. & Govindwar, S. P. 2013. Treatment of dye containing wastewaters by a developed lab scale phytoreactor and enhancement of its efficacy by bacterial augmentation. *International Biodeterioration & Biodegradation*, 78, 89-97.
- Khataee, A. R. 2009. Photocatalytic removal of C.I. Basic Red 46 on immobilized TiO2 nanoparticles: Artificial neural network modelling. *Environmental Technology*, 30(11), 1155–1168.
- Khehra, M. S., Saini, H. S., Sharma, D. K., Chadha, B. S. & Chimni, S. S. 2005a. Comparative studies on potential of consortium and constituent pure bacterial isolates to decolorize azo dyes. *Water Research*, 39(20), 5135-5141.
- Khehra, M. S., Saini, H. S., Sharma, D. K., Chadha, B. S. & Chimni, S. S. 2005b. Decolorization of various azo dyes by bacterial consortium. *Dyes and Pigments*, 67(1), 55-61.
- Kickuth, R. 1977. Degradation and incorporation of nutrients from rural wastewaters by plant rhizosphere under limnic conditions. *Utilization of manure by land spreading*, 235-243.
- Kickuth, R. 1978. Elimination gelöster Laststoffe durch Röhrichtbestände. Arb Dtsch Fischereiverbandes, 25, 57-70.
- Kickuth, R., 1981. Abwasserreinigung in Mosaikmatritzen aus aeroben und anaerobenTeilbezirken. In: Moser, F. (Ed.), Grundlagen der Abwasserreinigung. Verlag Oldenburg, M<sup>-</sup>unchen, Wien,630–665.
- Kimani, R. W., Mwangi, B. M. & Gichuki, C. M. 2012. Treatment of flower farm wastewater effluents using constructed wetlands in lake Naivasha, Kenya. *Indian Journal of Science and Technology*, 5(1), 1870-1878.
- Kirandeep, K., Arvinder, K. & Rajvir, K. 2015. Cytogenotoxicity of Azo dye acid blue-113 (ab-113) to Channa punctatus (Bloch). *Environmental Research and Development*, 9(3), 547.
- Kirsten, S. & Brander, L. 2004. The Economic Values of the World's Wetlands. *Living Waters, Amsterdam.*
- Kivaisi, A. K. 2001. The potential for constructed wetlands for wastewater treatment and reuse in developing countries: a review. *Ecological engineering*, 16(4), 545-560.
- Klein, R. & Bateman, I. 1998. The recreational value of Cley Marshes nature reserve: An argument against managed retreat? *Water and Environment Journal*, 12(4), 280-285.
- Klomjek, P. & Nitisoravut, S. 2005. Constructed treatment wetland: a study of eight plant species under saline conditions. *Chemosphere*, 58(5), 585-593.
- Knight, R., Ruble, R., Kadlec, R. & Reed, S. 1993. Wetlands for wastewater treatment: performance database. *Constructed wetlands for water quality improvement*, 35-35.
- Knight, R. L. 1997. Wildlife habitat and public use benefits of treatment wetlands. *Water Science and Technology*, 35(5), 35-43.

- Knight, R. L., Clarke Jr., R. A. & Bastian, R. K. 2001. Surface flow (SF) treatment wetlands as a habitat for wildlife and humans. . Water Science and Technology., 44, 27-37.
- Knight, R. L., Payne, V. W., Borer, R. E., Clarke, R. A. & Pries, J. H. 2000. Constructed wetlands for livestock wastewater management. *Ecological engineering*, 15(1), 41-55.
- Knowles, P. 2012. *Clogging in horizontal subsurface flow constructed wetlands.* PhD, Aston.
- Knowles, P., Dotro, G., Nivala, J. & García, J. 2011. Clogging in subsurface-flow treatment wetlands: occurrence and contributing factors. *Ecological Engineering*, 37(2), 99-112.
- Kodam, K., Soojhawon, I., Lokhande, P. & Gawai, K. 2005. Microbial decolorization of reactive azo dyes under aerobic conditions. *World journal* of microbiology and biotechnology, 21(3), 367-370.
- Koepernik, H. & Borsdorf, R. 1983. Identification of substituted naphthalenesulfonic acids by means of UV-spectroscopy-a contribution to the structure determination of acid azo dyes. *Fur Praktische Chemie*, 325(6), 1002-1010.
- Konnerup, D., Trang, N. T. D. & Brix, H. 2011. Treatment of fishpond water by recirculating horizontal and vertical flow constructed wetlands in the tropics. *Aquaculture*, 313(1), 57-64.
- Koottatep, T. & Polprasert, C. 1997. Role of plant uptake on nitrogen removal in constructed wetlands located in the tropics. *Water Science and Technology*, 36(12), 1-8.
- Kopec, D. 2007. Small Subsurface Flow Constructed Wetlands with Soil Dispersal System. *Permit to Install*, 3, 1-23.
- Korkusuz, E. A., Beklİoğlu, M. & Demirer, G. N. 2004. Treatment efficiencies of the vertical flow pilot-scale constructed wetlands for domestic wastewater treatment. *Turkish Journal of Engineering and Environmental Sciences*, 28(5), 333-344.
- Kosolapov, D., Kuschk, P., Vainshtein, M., Vatsourina, A., Wiessner, A., Kästner, M. & Müller, R. 2004. Microbial processes of heavy metal removal from carbon-deficient effluents in constructed wetlands. *Engineering in Life Sciences*, 4(5), 403-411.
- Kröpfelová, L. 2008. Wastewater Treatment in Constructed Wetlands with Horizontal Sub-Surface Flow. *Czech Republic: Springer Science+ Business Media BV*.
- Küçükgüzel, I., Ülgen, M. & Gorrod, J. W. 1999. In vitro hepatic microsomal metabolism of N-benzyl-N-methylaniline. *Il Farmaco*, 54(5), 331-337.
- Kumar, V. & Chopra, A. 2012. Monitoring of physico-chemical and microbiological characteristics of municipal wastewater at treatment plant, Haridwar City (Uttarakhand) India. *Journal of Environmental Science and Technology*, 5(2), 109.
- Kuschk, P., Wiessner, A., Kappelmeyer, U., Weissbrodt, E., Kästner, M. & Stottmeister, U. 2003. Annual cycle of nitrogen removal by a pilot-scale subsurface horizontal flow in a constructed wetland under moderate climate. *Water Research*, 37(17), 4236-4242.

- Kyambadde, J. 2005. Optimizing processes for biological nitrogen removal in Nakivubo wetland, Uganda. AlbaNova univercity centre.
- Lade, H. S., Waghmode, T. R., Kadam, A. A. & Govindwar, S. P. 2012. Enhanced biodegradation and detoxification of disperse azo dye Rubine GFL and textile industry effluent by defined fungal-bacterial consortium. *International Biodeterioration & Biodegradation*, 72, 94-107.
- Lakherwal, D. 2014. Adsorption of heavy metals: a review. *International Journal* of Environmental Research and Development, 4(1), 41-48.
- Lakatos, G. (1998). Hungary. In: Constructed wetlands for wastewater tretament in Europe, edited by Vymazal, J., Brix, H., Cooper, P., Green, B., and Haberi, R., Backhuys, Leiden, The Netherland, 191-206.
- Langergraber, G. 2008. Modeling of Processes in Subsurface Flow Constructed Wetlands: A Review. *Soil Science Society of Amrica*, 7(2).
- Lantzke, I., Mitchell, D., Heritage, A. & Sharma, K. 1999. A model of factors controlling orthophosphate removal in planted vertical flow wetlands. *Ecological Engineering*, 12(1), 93-105.
- Latif, U. & Dickert, F. L. 2015. Chemical Oxygen Demand. In: Moretto, L. M. & Kalcher, K. (eds.) Environmental analysis by electrochemical sensors and biosensors. New York: Springer.
- Lavanya, C., Dhankar, R., Chhikara, S. & Sheoran, S. 2014. Degradation of toxic dyes: a review. *International Journal of Current Microbiologyand Applied Sciences*, 3(6), 189-199.
- Lavrova, S. & Koumanova, B. 2013. Nutrients and organic matter removal in a vertical-flow constructed wetland. In: Yogesh, B. P. & Prakash, R. (eds.) Applied bioremediation-active and passive approaches. Croatia: InTech.
- Lavrova, S. & Koumanova, B. 2014. Nutrients removal from landfill leachate in "wetland" system. *Chemical Technology and Metallurgy*, 49(2), 143-148.
- Lee, A. H. & Nikraz, H. 2014. BOD: COD ratio as an indicator for Pollutants Leaching from Landfill. *Journal of Clean Energy Technologies*, 2(3), 263-266.
- Lee, B. H. & Scholz, M. 2007. What is the role of Phragmites australis in experimental constructed wetland filters treating urban runoff? *Ecological Engineering*, 29(1), 87-95.
- Lee, C. G., Fletcher, T. D. & Sun, G. 2009. Nitrogen removal in constructed wetland systems. *Engineering in Life Sciences*, 9(1), 11-22.
- Lee, D. J., Kang, S. W., Park, J. H., Kim, S. H., Choi, I. W., Hwang, T. H., Lim, B. J., Jung, S. J., Park, H. N. & Cho, J. S. 2015. Enhancement of nutrient removal in a hybrid constructed wetland utilizing an electric fan air blower with renewable energy of solar and wind power. *Journal of Chemistry*, 2015.
- Lehl, H. K., Ong, S. A., Ho, L. N., Wong, Y. S., Naemah Mohd Saad, F., Oon, Y. L., Oon, Y. S., Yong, C. Y. & Thung, W. E. 2016. Multiple aerobic and anaerobic baffled constructed wetlands for simultaneous nitrogen and organic compounds removal. Desalination and Water Treatment, 57(60), 29160-29167.
- Lesage, E. 2006. *Behaviour of heavy metals in constructed treatment wetlands*. Ghent University.

- Li, C., Wu, S. & Dong, R. 2015. Dynamics of organic matter, nitrogen and phosphorus removal and their interactions in a tidal operated constructed wetland. *Journal of environmental management*, 151, 310-316.
- Li, F., Shan, S., Li, Y., Li, Y. & Wang, Z. 2012. Optimization of aerobic/anaerobic subsurface flow constructed wetlands. Huan jing ke xue= Huanjing kexue/[bian ji, Zhongguo ke xue yuan huan jing ke xue wei yuan hui" Huan jing ke xue" bian ji wei yuan hui.], 33(2), 436-441.
- Li, J., Yu, H. & Luan, Y. 2015. Meta-analysis of the copper, zinc, and cadmium absorption capacities of aquatic plants in heavy metal-polluted water. *International journal of environmental research and public health*, 12(12), 14958-14973.
- Lianfang, Z., Wei, Z. & Wei, T. 2009. Clogging processes caused by biofilm growth and organic particle accumulation in lab-scale vertical flow constructed wetlands. *Journal of Environmental Sciences*, 21(6), 750-757.
- Likens, G. E. 2010. Biogeochemistry of inland waters, Academic Press, Amsterdam.
- Lim, P., Wong, T. & Lim, D. 2001. Oxygen demand, nitrogen and copper removal by free-water-surface and subsurface-flow constructed wetlands under tropical conditions. *Environment International*, 26(5), 425-431.
- Lin, Y. F., Jing, S. R., Lee, D. Y., Chang, Y. F., Chen, Y. M. & Shih, K. C. 2005. Performance of a constructed wetland treating intensive shrimp aquaculture wastewater under high hydraulic loading rate. *Environmental Pollution*, 134(3), 411-421.
- Litchfield, D., & Schatz, 1989. Constructed wetlands for wastewater treatment at Amoco Oil Company's Mandan, North Dakoat Refinery. In: Constructed wetlands for wastewater treatment, Hammer, D., (ed.), Lewis, Chelsea, Michigan, 233-237.
- Liu, D., Ge, Y., Chang, J., Peng, C., Gu, B., Chan, G. & Wu, X. 2009. Constructed wetlands in China: recent developments and future challenges. *Frontiers in Ecology and the Environment*, 7(5), 261-268.
- Liu, J. & Diamond, J. 2005. China's environment in a globalizing world. *Nature*, 435(7046), 1179-1186.
- Liu, J., Zhang, X.-H., You, S.-H., Wu, Q.-X. & Zhou, K.-N. 2015. Function of Leersia hexandra Swartz in constructed wetlands for Cr (VI) decontamination: a comparative study of planted and unplanted mesocosms. *Ecological Engineering*, 81, 70-75.
- Liu, W., M, M. & Huang, X.-H. 2006. Performance of hybrid constructed wetland systems for treating septic tank effluent. *Journal of Environmental Sciences*, 18(4), 665-669.
- Lopez, J. L. A. 2014. Vertical-flow constructed wetlands for the treatment of wastewater and stormwater from combined sewer systems. Doctoral dissertation, INSA de Lyon.
- Lourenço, N., Novais, J. & Pinheiro, H. 2003. Analysis of secondary metabolite fate during anaerobic-aerobic azo dye biodegradation in a sequential batch reactor. *Environmental technology*, 24(6), 679-686.
- Luederitz, V., Eckert, E., Lange-Weber, M., Lange, A. & Gersberg, R. M. 2001. Nutrient removal efficiency and resource economics of vertical flow and

horizontal flow constructed wetlands. *Ecological Engineering*, 18(2), 157-171.

- Luise, J., Robert, E., Lamonte, G., Barry, I., Jeffrey, L., Timonthy, B., Glenn, R., Melanie, S., Charles, T. & Harold, W. 1993. General Consideration. A Handbook of Constructed Wetlands. EPA publications.
- Ma, F., Jiang, L. & Zeng, T. Reversing clogging in vertical-flow constructed wetlands by backwashing treatment. Advanced Materials Research, 2010. Trans Tech Publ, 1064-1068.
- Madigan, M. T., Martinko, J. M., Dunlap, P. V. & Clark, D. P. 2008. Brock Biology of micro-organisms 12th edn. *Int. Microbiol*, 11, 65-73.
- Magdoff, F. & Van Es, H. 2010. Building Soils for Better Crops, Sustainable Agriculture Network. *Beltsville, Maryland*.
- Mahmood, Q., Pervez, A., Zeb, B. S., Zaffar, H., Yaqoob, H., Waseem, M. & Afsheen, S. 2013. Natural treatment systems as sustainable ecotechnologies for the developing countries. *BioMed research international*, 2013, 1-19.
- Maier, U., Debiase, C., Baeder-Bederski, O. & Bayer, P. 2009. Calibration of hydraulic parameters for large-scale vertical flow constructed wetlands. *Journal of Hydrology*, 369(3), 260-273.
- Manios, T., Stentiford, E. & Millner, P. 2003. Removal of total suspended solids from wastewater in constructed horizontal flow subsurface wetlands. *Journal of Environmental Science and Health, Part A*, 38(6), 1073-1085.
- Marchand, L., Mench, M., Jacob, D. & Otte, M. 2010. Metal and metalloid removal in constructed wetlands, with emphasis on the importance of plants and standardized measurements: a review. *Environmental pollution*, 158(12), 3447-3461.
- Marti, E., Arias, C. A., Brix, H. & Johansen, N.-H. 2003. Recycling of treated effluents enhances reduction of total nitrogen in vertical flow constructed wetlands. *Publicationes Instituti Geographici Universitatis Tartuensis*, 94, 150-155.
- Martin, M. 2011. Our Toxic World A guide to hazardous substances in our everyday lives. *Journal of Resources, Energy and Development*, 8(2), 189-190.
- Martínez-Huitle, C. A. & Brillas, E. 2009. Decontamination of wastewaters containing synthetic organic dyes by electrochemical methods: a general review. *Applied Catalysis B: Environmental*, 87(3), 105-145.
- Masi, F., Martinuzzi, N., Bresciani, R., Giovannelli, L. & Conte, G. 2007. Tolerance to hydraulic and organic load fluctuations in constructed wetlands. *Water science and technology*, 56(3), 39-48.
- Matagi, S., Swai, D. & Mugabe, R. 1998. A review of heavy metal removal mechanisms in wetlands. *Afr. J. Trop. Hydrobiol. Fish.*, 8(1), 13-25.
- Mays, P. & Edwards, G. 2001. Comparison of heavy metal accumulation in a natural wetland and constructed wetlands receiving acid mine drainage. *Ecological Engineering*, 16(4), 487-500.
- Mazumder, D. 2013. Scope of BOD, Nitrogen and Phosphorous Removal through Plant-Soil Interaction in the Wetland. *International Journal of Environmental, Chemical, Ecological, Geological and Geophysical Engineering*, 7(2), 82-91.

- Mbuligwe, S. E. 2005. Comparative treatment of dye-rich wastewater in engineered wetland systems (EWSs) vegetated with different plants. *Water Research*, 39(2), 271-280.
- Mburu, N., Thumbi, G. & Mayabi, A. 2008. Removal of Bacteria Pathogens from Domestic Wastewater in a Tropical Subsurface Horizontal Flow Constructed Wetland. Proceedings of Taal2007: The 12th World Lake Conference, 2008. p.1015.
- Melián, J. H., Rodríguez, A. M., Arana, J., Díaz, O. G. & Henríquez, J. G. 2010. Hybrid constructed wetlands for wastewater treatment and reuse in the Canary Islands. *Ecological Engineering*, 36(7), 891-899.
- Mena, J., Rodriguez, L., Nuñez, J., Fernández, F. & Villaseñor, J. 2008. Design of horizontal and vertical subsurface flow constructed wetlands treating industrial wastewater. WIT Transactions on Ecology and the Environment, 111, 555-564.
- Meng, P., Pei, H., Hu, W., Shao, Y. & Li, Z. 2014. How to increase microbial degradation in constructed wetlands: influencing factors and improvement measures. *Bioresource technology*, 157, 316-326.
- Mengzhi, C., Yingying, T., Xianpo, L. & Zhaoxiang, Y. 2009. Study on the heavy metals removal efficiencies of constructed wetlands with different substrates. *Journal of water Resource and Protection*, 1, 22-28.
- Merino-Solís, M. L., Villegas, E., De Anda, J. & López-López, A. 2015. The effect of the hydraulic retention time on the performance of an ecological wastewater treatment system: An anaerobic filter with a constructed wetland. *Water*, 7(3), 1149-1163.
- Midhun, G., Divya, L., George, J., Jayakumar, P. & Suriyanarayanan, S. 2016. Wastewater Treatment Studies on Free Water Surface Constructed Wetland System. *Integrated Waste Management in India. Springer*.
- Mietto, A., Politeo, M., Breschigliaro, S. & Borin, M. 2015. Temperature influence on nitrogen removal in a hybrid constructed wetland system in Northern Italy. *Ecological Engineering*, 75, 291-302.
- Mimis, S. & Gaganis, P. 2007. Vertical flow constructed wetlands for wastewater treatment: A pilot scale study. Proceedings of the 10 th conference on environmental science and technology, Kos island, Greece. 501-505.
- Mitsch, W. J. & Wise, K. M. 1998. Water quality, fate of metals, and predictive model validation of a constructed wetland treating acid mine drainage. Water research, 32(6), 1888-1900.
- Mohan, S. V., Rao, N. C. & Sarma, P. 2007. Simulated acid azo dye (Acid black 210) wastewater treatment by periodic discontinuous batch mode operation under anoxic–aerobic–anoxic microenvironment conditions. *ecological engineering*, 31(4), 242-250.
- Molle, P., Prost-Boucle, S. & Lienard, A. 2008. Potential for total nitrogen removal by combining vertical flow and horizontal flow constructed wetlands: a full-scale experiment study. *Ecological engineering*, 34(1), 23-29.
- Monjeau, C. 1901. Purifying water. Google Patents.
- Moosvi, S., Keharia, H. & Madamwar, D. 2005. Decolourization of textile dye Reactive Violet 5 by a newly isolated bacterial consortium RVM 11.1. *World Journal of Microbiology and Biotechnology*, 21(5), 667-672.

- Moshiri, G. A. 1993. Construction Wetlands for Water Quality Improvement, *Boca Raton, Florida, USA., Lewis Publishers, CRC Press.*
- Münch, C., Kuschk, P. & Röske, I. 2005. Root stimulated nitrogen removal: only a local effect or important for water treatment? *Water Science and Technology*, 51(9), 185-192.
- Murry, M. & John, E. 1992. Organic Chemistry, 3<sup>rd</sup> edition. Wadsworth, Belmont. *ISBN 0-534-16218-5*.
- Mustafa, A. 2010. Nutrient removal with integrated constructed wetlands: microbial ecology and treatment performance evaluation of full-scale integrated constructed wetlands. doctoral dissertation, Edinburgh, UK.
- Mustapha, H. I., Van Bruggen, J. & Lens, P. 2015. Vertical subsurface flow constructed wetlands for polishing secondary Kaduna refinery wastewater in Nigeria. *Ecological Engineering*, 84, 588-595.
- Nagajyoti, P., Lee, K. & Sreekanth, T. 2010. Heavy metals, occurrence and toxicity for plants: a review. *Environmental Chemistry Letters*, 8(3), 199-216.
- Naushad, M. & Al-Othman, Z. A. 2012. Ion exchange materials and environmental remediation. *Ion Exchange Technology II. Springer*.
- Nawab, B., Esser, K. B., Jenssen, P. D., Nyborg, I. L. P. & Baig, S. A. 2016. Technical Viability of Constructed Wetland for Treatment of Dye Wastewater in Gadoon Industrial Estate, Khyber Pakhtunkhwa, Pakistan. *Wetlands*, 36(1), 1-9.
- Nilratnisakorn, S., Thiravetyan, P. & Nakbanpote, W. 2009. A constructed wetland model for synthetic reactive dye wastewater treatment by narrow-leaved cattails (Typha angustifolia Linn.). *Water Science and Technology*, 60(6), 1565-1574.
- Nivala, J. & Rousseau, D. P. 2009. Reversing clogging in subsurface-flow constructed wetlands by hydrogen peroxide treatment: two case studies. *Water Science and Technology*, 59(10), 2037-2046.
- Nivala, J., Wallace, S., Headley, T., Kassa, K. & Brix, H. 2012. Oxygen transfer and consumption in subsurface flow treatment wetlands. *Ecological Engineering*, 61, 544-554.
- Noller, B., Woods, P. & Ross, B. 1994. Case studies of wetland filtration of mine waste water in constructed and naturally occurring systems in Northern Australia. *Water Science and Technology*, 29(4), 257-265.
- Noonpui, S. & Thiravetyan, P. 2011. Treatment of reactive azo dye from textile wastewater by burhead (Echinodorus cordifolius L.) in constructed wetland: effect of molecular size. *Journal of Environmental Science and Health Part A*, 46(7), 709-714.
- Novitzki, R., Smith, R. & Fretwell, J. 2002. Restoration, Creation and Recovery of Wetlands. Wetland Functions, Values, and Assessment. National Water Summary on Wetland Resources. United States Geological Survey Water Supply. Paper 2425
- O'neill, C., Lopez, A., Esteves, S., Hawkes, F., Hawkes, D. & Wilcox, S. 2000. Azo-dye degradation in an anaerobic-aerobic treatment system operating on simulated textile effluent. *Applied microbiology and biotechnology*, 53(2), 249-254.

- O'neill, C., Hawkes, F. R., Hawkes, D. L., Esteves, S. & Wilcox, S. J. 2000. Anaerobic–aerobic biotreatment of simulated textile effluent containing varied ratios of starch and azo dye. *Water Research*, 34(8), 2355-2361.
- Obarska, H. & Gajewska, M. 2003. The removal of nitrogen compounds in constructed wetlands in Poland. *Polish Journal of Environmental Studies*, 12(6), 739-746.
- Ojstršek, A., Fakin, D. & Vrhovšek, D. 2007. Residual dyebath purification using a system of constructed wetland. *Dyes and pigments*, 74(3), 503-507.
- Oki, L. R. & White, S. A. 2011. Ecological approaches used in nurseries to treat water. UCNFA News, 15(2), 1-6.
- Okurut, T. O. 2000. A pilot study on municipal wastewater treatment using a constructed wetland in Uganda, *Netherlands*, *CRC Press*.
- Olejnik, D. & Wojciechowski, K. 2012. The conception of constructed wetland for dyes removal in water solutions. *CHEMIK nauka-technika-rynek*, 1(66), 611-614.
- Olgun, A. & Atar, N. 2009. Equilibrium and kinetic adsorption study of Basic Yellow 28 and Basic Red 46 by a boron industry waste. *Hazardous Materials*, 161(1), 148-156.
- Olsson, L. 2011. Effect of design and dosing regime on the treatment performance of vertical flow constructed wetlands. Master dissertation, Linköpings.
- Ong, S. A., Ho, L. N., Wong, Y. S., Dani Leonard, D. & Hafizah, S. 2011. Semibatch operated constructed wetlands planted with phragmites australis for treatment of dying wastewater. *Engineering Science and Technology*, *Taylor's University*, 6(5), 623-631.
- Ong, S. A., Katsuhiro, U., Daisuke, I. & Kazuaki, Y. 2009. Simultaneous removal of color, organic compounds and nutrients in azo dye-containing wastewater using up-flow constructed wetland. *Hazardous Materials*, 165(1-3), 696-703.
- Ong, S. A., Katsuhiro, U., Daisuke, I., Yuji, I. & Kazuaki, Y. 2010. Treatment of azo dye Acid Orange 7 containing wastewater using up-flow constructed wetland with and without supplementary aeration. *Bioresources Technology*, 101(23), 9049-9057.
- Öövel, M., Tooming, A., Mauring, T. & Mander, Ü. 2007. Schoolhouse wastewater purification in a LWA-filled hybrid constructed wetland in Estonia. *Ecological Engineering*, 29(1), 17-26.
- Ottova, V., Balcarová, J. & Vymazal, J. 1997. Microbial characteristics of constructed wetlands. *Water Science and Technology*, 35(5), 117-123.
- Oyaro, N., Juddy, O., Murago, E. N. & Gitonga, E. 2007. The contents of Pb, Cu, Zn and Cd in meat in Nairobi, Kenya. *International journal of food, agriculture and environment,* 5(3-4), 119-121.
- Pagter, M., Bragato, C. & Brix, H. 2005. Tolerance and physiological responses of Phragmites australis to water deficit. *Aquatic Botany*, 81(4), 285-299.
- Paing, J., Guilbert, A., Gagnon, V. & Chazarenc, F. 2015. Effect of climate, wastewater composition, loading rates, system age and design on performances of French vertical flow constructed wetlands: a survey based on 169 full scale systems. *Ecological Engineering*, 80, 46-52.

- Pandey, A., Singh, P. & Iyengar, L. 2007. Bacterial decolorization and degradation of azo dyes. *International Biodeterioration & Biodegradation*, 59(2), 73-84.
- Park, W. & Polprasert, C. 2008. Roles of oyster shells in an integrated constructed wetland system designed for P removal. *Ecological Engineering*, 34(1), 50-56.
- Parshetti, G., Kalme, S., Saratale, G. & Govindwar, S. 2006. Biodegradation of malachite green by Kocuria rosea MTCC 1532. Acta Chimica Slovenica, 53(4), 492.
- Patrick, W. & Verloo, M. 1998. Distribution of soluble heavy metals between ionic and complexed forms in a saturated sediment as affected by pH and redox conditions. *Water Science and Technology*, 37(6-7), 165-171.
- Paul, A., Sarpal, R. S. & Dogra, S. K. 1990. Effects of solvent and acid concentration on the absorption and fluorescence spectra of α, αdiaminonaphthalenes. *Journal of the Chemical Society, Faraday Transactions*, 86(12), 2095-2101.
- Paul, E. A. 2006. Soil microbiology, ecology and biochemistry, Academic press.
- Paulino, A. T., Minasse, F. A., Guilherme, M. R., Reis, A. V., Muniz, E. C. & Nozaki, J. 2006. Novel adsorbent based on silkworm chrysalides for removal of heavy metals from wastewaters. *Journal of colloid and interface science*, 301(2), 479-487.
- Pereira, L. & Alves, M. 2012. Dyes-Environmental Impact and Remediation. In: Malik, A. & Grohmann, E. (eds.) Environmental Protection Strategies for Sustainable Development. New York: Springer.
- Pervez, A., Headley, A. & Terzis, E. 1999. The treatment of azo dyes using reedbed treatment systems. In Wetlands & Remediation: An International Conference, 187-194.
- Philippi, L., Sezerino, P., Bento, A. & Magri, M. Vertical flow constructed wetlands for nitrification of anaerobic pond effluent in southern Brazil under different loading rates. 10th International Conference on Wetland System for Water Pollution Control. Lisboa: IWA, 2006. 631-639.
- Phugare, S. S., Kalyani, D. C., Patil, A. V. & Jadhav, J. P. 2011. Textile dye degradation by bacterial consortium and subsequent toxicological analysis of dye and dye metabolites using cytotoxicity, genotoxicity and oxidative stress studies. *Journal of Hazardous Materials*, 186(1), 713-723.
- Pielesz, A., Baranowska, I., Rybak, A. & Włochowicz, A. 2002. Detection and determination of aromatic amines as products of reductive splitting from selected azo dyes. *Ecotoxicology and environmental safety*, 53(1), 42-47.
- Pillai, J. S. & Vijayan, N. Decentralized greywater treatment for nonpotable reuse in a vertical flow constructed wetland. Green Technologies (ICGT), 2012 International Conference on, 2012. IEEE, 058-063.
- Pinheiro, H. M., Touraud, E. & Thomas, O. 2004. Aromatic amines from azo dye reduction: status review with emphasis on direct UV spectrophotometric detection in textile industry wastewaters. *Dyes and pigments*, 61(2), 121-139.
- Platzer, C. 1999. Design recommendations for subsurface flow constructed wetlands for nitrification and denitrification. *Water Science and Technology*, 40(3), 257-263.

- Podczeck, F. & Jones, B. E. 2004. Pharmaceutical capsules, UK, Pharmaceutical Press.
- Postolache, O. A., Girão, P., Pereira, J. D. & Ramos, H. M. G. 2007. Multibeam optical system and neural processing for turbidity measurement. *Sensors Journal, IEEE*, 7(5), 677-684.
- Priya, G. S. & Urmila, B. 2013. Comparison of different types of media for nutrient removal efficiency in vertical upflow constructed wetlands. *International Journal of Environmental Engineering and Management*, 4, 405-416.
- Prochaska, C. & Zouboulis, A. 2006. Removal of phosphates by pilot vertical-flow constructed wetlands using a mixture of sand and dolomite as substrate. *Ecological Engineering*, 26(3), 293-303.
- Prochaska, C., Zouboulis, A. & Eskridge, K. M. 2007. Performance of pilot-scale vertical-flow constructed wetlands, as affected by season, substrate, hydraulic load and frequency of application of simulated urban sewage. *Ecological engineering*, 31(1), 57-66.
- Prost-Boucle, S. & Molle, P. 2012. Recirculation on a single stage of vertical flow constructed wetland: treatment limits and operation modes. *Ecological Engineering*, 43, 81-84.
- Quayle, W. C., Fattore, A., Zandona, R., Christen, E. W. & Arienzo, M. 2009. Evaluation of organic matter concentration in winery wastewater: a case study from Australia. *Water Science and Technology*, 60(10), 2521-2528.
- Ragkos, A., Psychoudakis, A., Christofi, A. & Theodoridis, A. 2006. Using a functional approach to wetland valuation: the case of Zazari–Cheimaditida. *Regional Environmental Change*, 6(4), 193-200.
- Rai, P. K. 2008. Heavy metal pollution in aquatic ecosystems and its phytoremediation using wetland plants: an ecosustainable approach. *International journal of phytoremediation*, 10(2), 133-160.
- Ramesh, S. 2011. *Poll fervour paints water channels blue* [Online]. *THE HINDU*. Available: <u>http://www.thehindu.com/news/cities/Coimbatore/poll-fervour-paints-water-channels-blue/article2482677.ece</u> 2016].
- Ramsar, I. 1971. Convention on Wetlands of International Importance, Especially as Waterfowl Habitat. *Ramsar (Iran)*, 2.
- Ratna & Padhi, B. 2012. Pollution due to synthetic dyes toxicity & carcinogenicity studies and remediation. *International Journal of Environmental Sciences*, 3(3), 940.
- Razali, N. M. & Wah, Y. B. 2011. Power comparisons of shapiro-wilk, kolmogorov-smirnov, lilliefors and anderson-darling tests. *Journal of Statistical Modeling and Analytics*, 2(1), 21-33.
- Reddy, K., Kadlec, R., Flaig, E. & Gale, P. 1999. Phosphorus retention in streams and wetlands: a review. *Critical reviews in environmental science and technology*, 29(1), 83-146.
- Reddy, K. R. & Delaune, R. D. 2008. Biogeochemistry of wetlands: science and applications, *CRC press*.
- Reuter, J. E., Djohan, T. & Goldman, C. R. 1992. The use of wetlands for nutrient removal from surface runoff in a cold climate region of California—results from a newly constructed wetland at Lake Tahoe. *Journal of Environmental Management*, 36(1), 35-53.

- Richardson, C. & Craft, C. 1993. Effective phosphorus retention in wetlands: fact or fiction, *Lewis Publishers: Boca Raton, FL, USA*.
- Riu, J., Schönsee, I., Barcelo, D. & Rafols, C. 1997. Determination of sulphonated azo dyes in water and wastewater. *TrAC Trends in Analytical Chemistry*, 16(7), 405-419.
- Rizzo, S., Bartolini, M., Ceccarini, L., Piazzi, L., Gobbi, S., Cavalli, A., Recanatini, M., Andrisano, V. & Rampa, A. 2010. Targeting Alzheimer's disease: Novel indanone hybrids bearing a pharmacophoric fragment of AP2238. *Bioorganic & medicinal chemistry*, 18(5), 1749-1760.
- Robinson, T., Chandran, B. & Nigam, P. 2002. Removal of dyes from a synthetic textile dye effluent by biosorption on apple pomace and wheat straw. *Water Research*, 36, 2824-2830.
- Robinson, T., Mcmullan, G., Marchant, R. & Nigam, P. 2001. Remediation of dyes in textile effluent: a critical review on current treatment technologies with a proposed alternative. *Bioresource technology*, 77(3), 247-255.
- Robles-Molina, J., Gilbert-López, B., García-Reyes, J. F. & Molina-Díaz, A. 2014. Monitoring of selected priority and emerging contaminants in the Guadalquivir River and other related surface waters in the province of Jaén, South East Spain. *Science of the Total Environment*, 479, 247-257.
- Rosolen, S. 2000. An Evaluation of Adsorptive Media For Improving Phosphorus Removal In Constructed Treatment Wetlands. M. Sc. thesis in Civil Engineering, Queen's University, Kingston, Ontario, Canada.
- Rousseau, D. P., Vanrolleghem, P. A. & De Pauw, N. 2004a. Constructed wetlands in Flanders: a performance analysis. *Ecological Engineering*, 23(3), 151-163.
- Rousseau, D. P., Vanrolleghem, P. A. & De Pauw, N. 2004b. Model-based design of horizontal subsurface flow constructed treatment wetlands: a review. *Water research*, 38(6), 1484-1493.
- Roy, T. 1998. Economic reforms and textile industry in India. *Economic and Political Weekly*, 33(32), 2173-2182.
- Royal Commission on Sewage Disposal. 1915. Final report of the commissioners appointed to inquire and report what methods of treating and disposing of sewage may properly adopted. His Majesty's Stationary Office, London.
- Rozema, E. R., Vanderzaag, A. C., Wood, J. D., Drizo, A., Zheng, Y., Madani, A. & Gordon, R. J. 2016. Constructed Wetlands for Agricultural Wastewater Treatment in Northeastern North America: A Review. *Water*, 8(5), 173.
- Rustige, H., Tomac, I. & Höner, G. 2003. Investigations on phosphorus retention in subsurface flow constructed wetlands. *Water science and technology*, 48(5), 67-74.
- Saeed, T. & Sun, G. 2012. A review on nitrogen and organics removal mechanisms in subsurface flow constructed wetlands: dependency on environmental parameters, operating conditions and supporting media. *Journal of environmental management*, 112, 429-448.
- Sahu, O. 2014. Reduction of heavy metals from waste water by wetland. *International Letters of Natural Sciences*, 7, 35-43.
- Salomons, W., Förstner, U. & Mader, P. 2012. Heavy metals: problems and solutions, *Springer Science & Business Media*.

- Sani, A. 2015. Treatment performance assessments of different wetland mesocosms. PhD, University of Salford.
- Sani, A., Scholz, M. & Bouillon, L. 2013. Seasonal assessment of experimental vertical-flow constructed wetlands treating domestic wastewater. *Bioresource technology*, 147, 585-596.
- Saratale, R. G., Saratale, G. D., Chang, J. & Govindwar, S. 2011. Bacterial decolorization and degradation of azo dyes: a review. *Journal of the Taiwan Institute of Chemical Engineers*, 42(1), 138-157.
- Sathasivan, A. 2009. Biological Phosphorus Removal Processes for Wastewater Treatment, in Water and Wastewater Treatment Technologies. [Ed.
- Saravanamuthu (Vigi) Vigneswaran], in Encyclopedia of Life Support Systems (EOLSS), Developed under the Auspices of the UNESCO, Eolss Publishers, Oxford, UK.
- Sayadi, M., Kargar, R., Doosti, M. & Salehi, H. 2012. Hybrid constructed wetlands for wastewater treatment: A worldwide review. *Proceedings of the international academy of ecology and environmental sciences*, 2(4), 204.
- Schaechter, M. 2009. Encyclopedia of microbiology, Oxford, UK, Elesvier.
- Schmidt, I., Zart, D. & Bock, E. 2001. Effects of gaseous NO2 on cells of Nitrosomonas eutropha previously incapable of using ammonia as an energy source. *Antonie van Leeuwenhoek*, 79(1), 39-47.
- Scholz, M. 2004. Treatment of gully pot effluent containing nickel and copper with constructed wetlands in a cold climate. *Journal of chemical technology and biotechnology*, 79(2), 153-162.
- Scholz, M. 2006. Wetland systems to control urban runoff, Amsterdam, Elesevier.
- Scholz, M. 2010. Wetland systems: storm water management control, *Berlin*, *Springer Science & Business Media*.
- Scholz, M. 2011. Wetland Systems, Storm Water Management Control, London, Springer.
- Scholz, M. 2015. Wetlands for Water Pollution Control, Amsterdam, Elsevier.
- Scholz, M. & Hedmark, Å. 2010. Constructed wetlands treating runoff contaminated with nutrients. *Water, Air, and Soil Pollution*, 205(1-4), 323-332.
- Scholz, M., Höhn, P. & Minall, R. 2002. Mature experimental constructed wetlands treating urban water receiving high metal loads. *Biotechnology Progress*, 18(6), 1257-1264.
- Scholz, M. & Lee, B. H. 2005. Constructed wetlands: a review. International journal of environmental studies, 62(4), 421-447.
- Scholz, M. & Xu, J. 2002a. Comparison of constructed reed beds with different filter media and macrophytes treating urban stream water contaminated with lead and copper. *Ecological Engineering*, 18(3), 385-390.
- Scholz, M. & Xu, J. 2002b. Performance comparison of experimental constructed wetlands with different filter media and macrophytes treating industrial wastewater contaminated with lead and copper. *Bioresource Technology*, 83(2), 71-79.
- Scholz, M., Xu, J. & Dodson, H. I. 2001. Comparison of filter media, plant communities and microbiology within constructed wetlands treating wastewater containing heavy metals. *Journal of Chemical Technology and Biotechnology*, 76(8), 827-835.

- Schwarzenbach, R. P., Escher, B. I., Fenner, K., Hofstetter, T. B., Johnson, C. A., Von Gunten, U. & Wehrli, B. 2006. The challenge of micropollutants in aquatic systems. *Science*, 313(5790), 1072-1077.
- Sehar, S., Aamir, R., Naz, I., Ali, N. & Ahmed, S. 2013. Reduction of contaminants (physical, chemical, and microbial) in domestic wastewater through hybrid constructed wetland. *ISRN microbiology*, 2013(
- Sehar, S., Naz, I., Khan, S., Naeem, S., Perveen, I., Ali, N. & Ahmed, S. 2016. Performance Evaluation of Integrated Constructed Wetland for Domestic Wastewater Treatment. *Water Environment Research*, 88(3), 280-287.
- Şen, S. & Demirer, G. 2003. Anaerobic treatment of real textile wastewater with a fluidized bed reactor. *Water Research*, 37(8), 1868-1878.
- Senthilvelan, T., Kanagaraj, J. & Panda, R. 2014. Enzyme-mediated bacterial biodegradation of an azo dye (CI Acid Blue 113): reuse of treated dye wastewater in post-tanning operations. *Applied biochemistry and biotechnology*, 174(6), 2131-2152.
- Seo, D. C., Cho, J. S., Lee, H. J. & Heo, J. S. 2005. Phosphorus retention capacity of filter media for estimating the longevity of constructed wetland. *Water Research*, 39(11), 2445-2457.
- Shapiro, S. S. & Wilk, M. B. 1965. An analysis of variance test for normality (complete samples). *Biometrika*, 52(3/4), 591-611.
- Sharma, G. & Brighu, U. 2014. Performance analysis of vertical up-flow constructed wetlands for secondary treated effluent. *APCBEE Procedia*, 10, 110-114.
- Sharma, S. K. 2014. Heavy Metals in Water: Presence, Removal and Safety, *Royal Society of Chemistry*.
- Shaw, E. A. & Richardson, J. S. 2001. Direct and indirect effects of sediment pulse duration on stream invertebrate assemblages and rainbow trout (Oncorhynchus mykiss) growth and survival. *Canadian Journal of Fisheries and Aquatic Sciences*, 58(11), 2213-2221.
- Shehzadi, M., Afzal, M., Khan, M. U., Islam, E., Mobin, A., Anwar, S. & Khan, Q. M. 2014. Enhanced degradation of textile effluent in constructed wetland system using Typha domingensis and textile effluent-degrading endophytic bacteria. *water research*, 58, 152-159.
- Seidel, K., 1955. Die Flechtbinse Scirpus lacustris. In: O" kologie, Morphologie und Entwicklung, ihre Stellung bei den Volkern und ihre wirtschaftliche Bedeutung. Schweizerbart ische Verlagsbuchnadlung, Stuttgart, Germany, pp. 37–52.
- Seidel, K., 1961. Zur Problematik der Keim- und Pflanzengewasser. Verh. Internat. Verein. Limnol. 14, 1035–1039.
- Seidel, K., 1965a. Phenol-Abbau in Wasser durch Scirpus lacustris L. wehrend einer versuchsdauer von 31 Monaten. Naturwissenschaften 52, 398–406.
- Seidel, K., 1965b. Neue Wege zur Grundwasseranreicherung in Krefeld, vol. II. Hydrobotanische Reinigungsmethode. GWF Wasser/Abwasser, 831–833.
- Seidel, K., 1966. Reinigung von Gew"assern durch h"ohere Pflanzen. Naturwissenschaften 53, 289–297.
- Seidel, K., 1976. Macrophytes and water purification. In: Tourbier, J., Pierson, R.W. (Eds.), Biological Control of Water Pollution. Pennsylvania University Press, Philadelphia, 109–122.

- Shen, Z., Zhou, Y., Liu, J., Xiao, Y., Cao, R. & Wu, F. 2015. Enhanced removal of nitrate using starch/PCL blends as solid carbon source in a constructed wetland. *Bioresource technology*, 175, 239-244.
- Sheoran, A. & Sheoran, V. 2006. Heavy metal removal mechanism of acid mine drainage in wetlands: a critical review. *Minerals engineering*, 19(2), 105-116.
- Shi, E., She, Z. L., Xie, T., Wu, J. & Fu, X. H. Nutrient Removal from Polluted River Water by Using Vertical and Horizontal Subsurface Flow Constructed Wetlands. Advanced Materials Research, 2012. Trans Tech Publ, 1304-1307.
- Shilton, A., Pratt, S., Drizo, A., Mahmood, B., Banker, S., Billings, L., Glenny, S. & Luo, D. 2005. 'Active'filters for upgrading phosphorus removal from pond systems. *Water Science and Technology*, 51(12), 111-116.
- Shirzad-Siboni, M., Jafari, S. J., Giahi, O., Kim, I., Lee, S.-M. & Yang, J.-K. 2014. Removal of acid blue 113 and reactive black 5 dye from aqueous solutions by activated red mud. *Journal of Industrial and Engineering Chemistry*, 20(4), 1432-1437.
- Shon, H., Vigneswaran, S., Kandasamy, J. & Cho, J. 2007. Characteristics of effluent organic matter in wastewater.
- Siedlecka, A. 1995. Some aspects of interactions between heavy metals and plant mineral nutrients. *Acta Societatis Botanicorum Poloniae*, 64(3), 265-272.
- Singh, S. N. 2014. Microbial Degradation of Synthetic Dyes in Wastewaters, London, *Springer*.
- Singh, N., Kumar, D. & Sahu, A. P. 2007. Arsenic in the environment: effects on human health and possible prevention. *Journal of Environmental Biology*, 28(2), 359.
- Sinicrope, T. L., Langis, R., Gersberg, R. M., Busnardo, M. J. & Zedler, J. B. 1992. Metal removal by wetland mesocosms subjected to different hydroperiods. *Ecological Engineering*, 1(4), 309-322.
- Sivakumar, D., Shankar, D., Vijaya Prathima, A. & Valarmathi, M. 2013. Constructed wetland treatment of textile industry wastewater using aquatic macrophytes. *Inernational journal of environmental sciences*, 3(4).
- Snyderwine, E. G., Sinha, R., Felton, J. S. & Ferguson, L. R. 2002. Highlights of the eighth international conference on carcinogenic/mutagenic Nsubstituted aryl compounds. *Mutation Research/Fundamental and Molecular Mechanisms of Mutagenesis*, 506, 1-8.
- Song, X., Ding, Y., Wang, Y., Wang, W., Wang, G. & Zhou, B. 2015. Comparative study of nitrogen removal and bio-film clogging for three filter media packing strategies in vertical flow constructed wetlands. *Ecological Engineering*, 74, 1-7.
- Song, Z., Zheng, Z., Li, J., Sun, X., Han, X., Wang, W. & Xu, M. 2006. Seasonal and annual performance of a full-scale constructed wetland system for sewage treatment in China. Ecological Engineering, 26(3), 272-282.
- Songliu, L., Hongying, H., Yingxue, S. & Jia, Y. 2009. Effect of carbon source on the denitrification in constructed wetlands. *Environmental sciences*, 21(8), 1036-1043.
- Sousa, W. T. Z., Panitz, C. M. N. & Thomaz, S. M. 2011. Performance of pilotscale vertical flow constructed wetlands with and without the emergent

macrophyte Spartina alterniflora treating mariculture effluent. *Brazilian* Archives of Biology and Technology, 54(2), 405-413.

- Specht, K. & Platzek, T. 1995. Textile dyes and finishes-Remarks to toxicological and analytical aspects. *Deutsche Lebensmittel-Rundschau*, 91(352-359.
- Spiegel, J. & Maystre, L. Y. 1998. Environmental pollution control and prevention. Encyclopedia of Occupational Health and Safety. 4th ed. Geneva: International Labour Office, 2.
- Sponza, D. & Işik, M. 2002. Decolorization and azo dye degradation by anaerobic/aerobic sequential process. *Enzyme and Microbial Technology*, 31(1), 102-110.
- Sri Lanka Central Environmental Authority. 2008. National Environmental (Protection and Quality) Regulations. http://www.cea.lk/web/index.php/en. [Accessed 09, 05, 2017].
- Stansbury, J. & Branigan, J. 2013. Development of a Field Test Method for Total Suspended Solids Analysis. *university of Nebraska- Lincoln*.
- Stefanakis, A., Akratos, C. S. & Tsihrintzis, V. A. 2014. Vertical flow constructed wetlands: Eco-engineering systems for wastewater and sludge treatment, *China*, *Newnes*.
- Stefanakis, A. I. & Tsihrintzis, V. A. 2012. Effects of loading, resting period, temperature, porous media, vegetation and aeration on performance of pilotscale vertical flow constructed wetlands. *Chimical Engineering*, 181-182, pp.416-430.
- Stoline, M. R. 1981. The status of multiple comparisons: simultaneous estimation of all pairwise comparisons in one-way ANOVA designs. *The American Statistician*, 35(3), 134-141.
- Stolz, A., Nörtemann, B. & Knackmuss, H. 1992. Bacterial metabolism of 5aminosalicylic acid. Initial ring cleavage. *Biochemical Journal*, 282(3), 675-680.
- Stottmeister, U., Buddhawong, S., Kuschk, P., Wiessner, A. & Mattusch, J. 2006. Constructed wetlands and their performance for treatment of water contaminated with arsenic and heavy metals. *Soil and water pollution monitoring, protection and remediation. Springer.*
- Stottmeister, U., Wiessner, A., Kuschk, P., Kappelmeyer, U., Ka<sup>°</sup>Stner, M., Bederski, O., Mu<sup>°</sup> Ller, R. A. & Moormann, H. 2003. Effects of plants and micro-organisms in constructed wetlands for wastewater treatment. *Biotechnology Advances*, 22, 93-117.
- Strickland, T. & Korleski, C. 2007. Guidance Document for Small Subsurface Flow Constructed Wetlands with Soil Dispersal System. *State of Ohio Environmental Protection Agency Division of Surface Water*.
- STWI. 2012. Guidelines for Sustainable Water Use in the Production and manufacturing Processes of Textiles [Online]. Available: http://stwi.se/guidelines/ [Accessed 09-05-2017].
- Sukumaran, D. 2013. Phytoremediation of heavy metals from industrial effluent using constructed wetland technology. *Applied Ecology and Environmental Sciences*, 1(5), 92-97.
- Sultana, M.-Y. 2014. *Treatment of industrial and agro-industrial wastewater using constructed wetlands*. Doctoral dissertation, Patras.
- Sumner, M. E. 1999. Handbook of soil science, USA, CRC press.

- Sun, Q. & Yang, L. 2003. The adsorption of basic dyes from aqueous solution on modified peat–resin particle. *Water research*, 37(7), 1535-1544.
- Suthersan, S. S. 2001. Natural and enhanced remediation systems, *Washington*, *Lewis Publisher*.
- Taiwan Environmental Protection Administration. 2003. Taiwan Effluent Standards.
- Tanner, C. C. 1996. Plants for constructed wetland treatment systems—a comparison of the growth and nutrient uptake of eight emergent species. *Ecological engineering*, 7(1), 59-83.
- Tanner, C. C., Adams, D. D. & Downes, M. T. 1997. Methane emissions from constructed wetlands treating agricultural wastewaters. *Journal of environmental quality*, 26(4), 1056-1062.
- Tanner, C. C., Clayton, J. S. & Upsdell, M. P. 1995. Effect of loading rate and planting on treatment of dairy farm wastewaters in constructed wetlands— II. Removal of nitrogen and phosphorus. *Water research*, 29(1), 27-34.
- Taylor, M. D., White, S. A., Chandler, S. L., Klaine, S. J. & Whitwell, T. 2006. Nutrient management of nursery runoff water using constructed wetland systems. *HortTechnology*, 16(4), 610-614.
- Tee, H. C., Lim, P. E., Seng, C. E., Nawi, M. a. M. & Adnan, R. 2015. Enhancement of azo dye Acid Orange 7 removal in newly developed horizontal subsurface-flow constructed wetland. *Environmental Management*, 147, 349-355.
- Thurston, R. V., Russo, R. C. & Vinogradov, G. 1981. Ammonia toxicity to fishes. Effect of pH on the toxicity of the unionized ammonia species. *Environmental science & technology*, 15(7), 837-840.
- Tiedje, J. M. 1982. Denitrification. *Methods of Soil Analysis. Part 2. Chemical and Microbiological Properties*, methodsofsoilan2), 1011-1026.
- Tietz, A., Kirschner, A., Langergraber, G., Sleytr, K. & Haberl, R. 2007. Characterisation of microbial biocoenosis in vertical subsurface flow constructed wetlands. *Science of the Total Environment*, 380(1), 163-172.
- Tilley, E., Ulrich, L., Lüthi, C., Reymond, P. & Zurbrügg, C. 2014. Compendium of sanitation systems and technologies, *Eawag*.
- Timmons, M. B., Ebeling, J., Wheaton, F., Summerfelt, S. & Vinci, B. 2002. Mass balances, loading rates, and fish growth. *Recirculating Aquaculture Systems, 2nd ed. Cayuga Aqua Ventures, New York, USA*, 89-116.
- Toet, S., Van Logtestijn, R. S., Kampf, R., Schreijer, M. & Verhoeven, J. T. 2005. The effect of hydraulic retention time on the removal of pollutants from sewage treatment plant effluent in a surface-flow wetland system. Wetlands, 25(2), 375-391.
- Tousignant, E., Eng, P., Fankhauser, O. & Hurd, H. 1999. Guidance manual for the design, construction and operations of constructed wetlands for rural applications in Ontario, Stantec Consulting Ltd. *Research and Technology Transfer Group, Alfred College (University of Guelph), South Nation Conservation*, 5-6.
- Tsihrintzis, V., Akratos, C., Gikas, G., Karamouzis, D. & Angelakis, A. 2007. Performance and cost comparison of a FWS and a VSF constructed wetland system. *Environmental technology*, 28(6), 621-628.

- Tumelo Seadira, Jeffrey Baloyi, Mpfunzeni Raphulu, Richard Moutloali & Ochieng, A. 2014. Acid Mine Drainage Treatment Using Constructed Wetland Intl' Conf. on Chemical, Integrated Waste Management & Environmental Engineering. Johannesburg
- Turner, K. 1991. Economics and wetland management. Ambio, 59-63.
- Tyler, E. Hydraulic wastewater loading rates to soil. On-Site Wastewater Treatment, 2001. American Society of Agricultural and Biological Engineers, 80.
- USEPA, 2000a. A Handbook of Constructed Wetlands: A Guide to Creating Wetlands for: Agricultural Wastewater, Domestic Wastewater, Coal Mine Drainage Stormwater in the Mid - Atlantic Region: Volume 1: General Considerations. United States Environmental Protection Agency, EPA report number 843B00005, ISBN 0-16-052999-9. Washington, D.C.
- USEPA, 2000b. Free water surface wetlands. Wastewater technology fact sheet. United State Environmental Protection Agency, EPA report number 832-F-00-024. Office of water. Washington, D.C.
- Van, A. J. 1995. Nitrogen removal in constructed wetlands treating nitrified meat processing effluent. Water Science and Technology, 32(3), 137-147.
- Van Der Zee, F. P. 2002. *Anaerobic azo dye reduction*. doctoral dissertation, Wageningen University.
- Van Der Zee, F. P., Lettinga, G. & Field, J. A. 2001. Azo dye decolourisation by anaerobic granular sludge. *Chemosphere*, 44(5), 1169-1176.
- Veenstra, S., 1998. The Netherlands. In: Constructed wetlands for wastewater treatment in Europe, Vymazal, J., Brix, H., Cooper, P., Green, B. and Haberi, R., eds., Backhuys, Leiden, The Netherlands, 289-314.
- Vera, L., Martel, G. & Marquez, M. 2013. Two years monitoring of the natural system for wastewater reclamation in Santa Lucía, Gran Canaria Island. *Ecological engineering*, 50, 21-30.
- Verma, M., Bakshi, N. & Nair, R. P. 2001. Economic valuation of Bhoj Wetland for sustainable use. Unpublished project report for World Bank assistance to Government of India, Environmental Management Capacity-Building. Bhopal: Indian Institute of Forest Management.
- Villa, J. A., Mitsch, W. J., Song, K. & Miao, S. 2014. Contribution of different wetland plant species to the DOC exported from a mesocosm experiment in the Florida Everglades. *Ecological Engineering*, 71(118-125.
- Vogt, P. F. & Gerulis, J. J. 2000. Ullmann's Encyclopedia of Industrial Chemistry, *Wiley-VCH*.
- Vohla, C., Kõiv, M., Bavor, H. J., Chazarenc, F. & Mander, Ü. 2011. Filter materials for phosphorus removal from wastewater in treatment wetlands— A review. *Ecological Engineering*, 37(1), 70-89.
- Von Muench, E. & Winker, M. 2011. Technology review on urine diversion components-Overview of urine diversion components such as waterless urinals, urine diversion toilets, urine storage and reuse systems. *Gesellschaft für Internationale Zusammenarbeit (GIZ) GmbH*.
- Vymazal, J. 1996. Subsurface horizontal-flow constructed wetlands for wastewater treatment: the Czech experience. *Wetlands Ecology and Management*, 4(3), 199-206.

- Vymazal, J. 2002. The use of sub-surface constructed wetlands for wastewater treatment in the Czech Republic: 10 years experience. *Ecological Engineering*, 18(5), 633-646.
- Vymazal, J. 2005a. Constructed wetlands for wastewater treatment in Europe. Nutrient management in agricultural watersheds: a wetland solution. Amsterdam, The Netherland: Wageningen Academic.
- Vymazal, J. 2005b. Horizontal sub-surface flow and hybrid constructed wetlands systems for wastewater treatment. *Ecological Engineering*, 25(5), 478-490.
- Vymazal, J. 2007. Removal of nutrients in various types of constructed wetlands. *Science of the total environment*, 380(1), 48-65.
- Vymazal, J. 2008a. Constructed wetlands for wastewater treatment: A review. Proceedings of Taal2007: The 12th World Lake Conference, 965, 980.
- Vymazal, J., 2008b. Constructed wetlands, surface flow. In: Jørgensen, S.E., Fath, B. (Eds.), Encyclopedia of Ecology, Vol. 1. *Elsevier, Amsterdam, Netherlands*, 765-776
- Vymazal, J. 2009. The use constructed wetlands with horizontal sub-surface flow for various types of wastewater. *Ecological engineering*, 35(1), 1-17.
- Vymazal, J. 2010a. Constructed Wetlands for Wastewater Treatment. *Water*, 2(3), 530-549.
- Vymazal, J. 2010b. Constructed wetlands for wastewater treatment: five decades of experience. *Environmental science & technology*, 45(1), 61-69.
- Vymazal, J. 2013. The use of hybrid constructed wetlands for wastewater treatment with special attention to nitrogen removal: a review of a recent development. *Water research*, 47(14), 4795-4811.
- Vymazal, J. 2014. Constructed wetlands for treatment of industrial wastewaters: a review. *Ecological Engineering*, 73(724-751.
- Vymazal, J., Brix, H., Cooper, P. F., Haberl, R., Perfler, R. & Laber, J. 1998. Removal mechanisms and types of constructed wetlands. Constructed wetlands for wastewater treatment in Europe. Leiden, The Netherland: Backhuys.
- Vymazal, J., Greenway, M., Tonderski, K., Brix, H. & Mander, Ü. 2006. Constructed wetlands for wastewater treatment. *Wetlands and natural resource management. Springer*.
- Vymazal, J. & Kröpfelová, L. 2009a. Removal of nitrogen in constructed wetlands with horizontal sub-sureface flow: a review. *Wetlands*, 29(4), 1114-1124.
- Vymazal, J. & Kröpfelová, L. 2009b. Removal of organics in constructed wetlands with horizontal sub-surface flow: a review of the field experience. *Science of the total environment*, 407(13), 3911-3922.
- Vymazal, J. & Kröpfelová, L. 2011. A three-stage experimental constructed wetland for treatment of domestic sewage: first 2 years of operation. *Ecological Engineering*, 37(1), 90-98.
- Wagener, S. M. & Laperriere, J. D. 1985. Effects of placer mining on the invertebrate communities of interior Alaska streams. *Freshwater Invertebrate Biology*, 4(4), 208-214.
- Walker, D. J. & Hurl, S. 2002. The reduction of heavy metals in a stormwater wetland. *Ecological Engineering*, 18(4), 407-414.
- Wallace, P. 2004. Engineered wetlands lead the way. Land and Water, 48(5).

- Wallace, S. D. & Knight, R. L. 2006. Small-scale constructed wetland treatment systems: feasibility, design criteria, O & M requirements, London, *IWA Publishing*.
- Walter, B. P. & Heimann, M. 2000. A process-based, climate-sensitive model to derive methane emissions from natural wetlands: Application to five wetland sites, sensitivity to model parameters, and climate. *Global Biogeochemical Cycles*, 14(3), 745-765.
- Wang, Z., Huang, K., Xue, M. & Liu, Z. 2011. Textile dyeing wastewater treatment, *Rijeka, InTech.*
- Ward, P. J., Strzepek, K. M., Pauw, W. P., Brander, L. M., Hughes, G. A. & Aerts, J. C. 2010. Partial costs of global climate change adaptation for the supply of raw industrial and municipal water: a methodology and application. *Environmental Research Letters*, 5(4).
- Weber, E. J. & Wolfe, L. N. 1987. Kinetic studies of the reduction of aromatic azo compounds in anaerobic sediment/water systems. *Environmental Toxicology and Chemistry*, 6(12), 911-919.
- Weedon, C. 2003. Compact vertical flow constructed wetland systems-first two years' performance. *Water Science and Technology*, 48(5), 15-23.
- Weedon, C. 2010. A decade of compact vertical flow constructed wetlands. *Water Science and Technology*, 62(12), 2790-2800.
- Wei, H. 2003. Officials failing to stop textile factories dumping waste in Qiantang River. *Time Weekly*, 8/1/2003.
- Wieder, R. K. 1989. A survey of constructed wetlands for acid coal mine drainage treatment in the eastern United States. Wetlands, 9(2), 299-315.
- Wetzel, R. 1993. Constructed Wetlands: Scientific Foundations Are Critical, in G. Moshiri (ed.), Constructed Wetlands for Water Quality Improvement, CRC Press, Boca Raton, Florida.
- Wießner, A., Kappelmeyer, U., Kuschk, P. & Kästner, M. 2005. Influence of the redox condition dynamics on the removal efficiency of a laboratory-scale constructed wetland. *Water Research*, 39(1), 248-256.
- Williams, J. B. 2002. Phytoremediation in wetland ecosystems: progress, problems, and potential. *Critical Reviews in Plant Sciences*, 21(6), 607-635.
- Winkler, M.-K., Le, Q. H. & Volcke, E. I. 2015. Influence of partial denitrification and mixotrophic growth of NOB on microbial distribution in aerobic granular sludge. *Environmental Science & Technology*, 49(18), 11003-11010.
- Winter, M. & Kickuth, R. 1989. Elimination of sulphur compounds from wastewater by the root zone process—I. Performance of a large-scale purification plant at a textile finishing industry. *Water Research*, 23(5), 535-546.
- Wong, M. & Bradshaw, A. 1982. A comparison of the toxicity of heavy metals, using root elongation of rye grass, Lolium perenne. *New Phytologist*, 91(2), 255-261.
- Wood, C. 2010. *Cotton and China* [Online]. *Miller-McCune*. Available: https://oecotextiles.wordpress.com/2010/03/24/cotton-and-china) 2016].
- Wu, H., Fan, J., Zhang, J., Ngo, H. H., Guo, W., Liang, S., Hu, Z. & Liu, H. 2015. Strategies and techniques to enhance constructed wetland performance for

sustainable wastewater treatment. *Environmental Science and Pollution Research*, 22(19), 14637-14650.

- Wu, H., Zhang, J., Li, P., Zhang, J., Xie, H. & Zhang, B. 2011. Nutrient removal in constructed microcosm wetlands for treating polluted river water in northern China. *Ecological Engineering*, 37(4), 560-568.
- Wu, S., Wallace, S., Brix, H., Kuschk, P., Kirui, W. K., Masi, F. & Dong, R. 2015. Treatment of industrial effluents in constructed wetlands: Challenges, operational strategies and overall performance. *Environmental Pollution*, 201, 107-120.
- Wu, S., Zhang, D., Austin, D., Dong, R. & Pang, C. 2011. Evaluation of a lab-scale tidal flow constructed wetland performance: oxygen transfer capacity, organic matter and ammonium removal. *Ecological Engineering*, 37(11), 1789-1795.
- Xianfa, L. & Chuncai, J. 1995. Constructed wetland systems for water pollution control in North China. *Water Science and Technology*, 32(3), 349-356.
- Xu, D., Xu, J., Wu, J. & Muhammad, A. 2006. Studies on the phosphorus sorption capacity of substrates used in constructed wetland systems. *Chemosphere*, 63(2), 344-352.
- Xu, Q., Cui, L., Zhang, L., Li, Z. & Yu, G. 2013. The Effect of Two Factor Combination of Three Kinds of Loading on the Soil Clogging in Vertical Flow Constructed Wetland. *Frontier of Environmental Science*, 2(4).
- Yadav, A. K., Dash, P., Mohanty, A., Abbassi, R. & Mishra, B. K. 2012. Performance assessment of innovative constructed wetland-microbial fuel cell for electricity production and dye removal. *Ecological Engineering*, 47, 126-131.
- Yalcuk, A. & Dogdu, G. 2014. Treatment of azo dye Acid Yellow 2G by using labscale vertical-flow intermittent feeding constructed wetlands. *Journal of Selcuk University Natural and Applied Science*, 355-368.
- Yalcuk, A. & Ugurlu, A. 2009. Comparison of horizontal and vertical constructed wetland systems for landfill leachate treatment. *Bioresource Technology*, 100(9), 2521-2526.
- Yamashita, T. & Yamamoto-Ikemoto, R. 2014. Nitrogen and phosphorus removal from wastewater treatment plant effluent via bacterial sulfate reduction in an anoxic bioreactor packed with wood and iron. *International journal of environmental research and public health*, 11(9), 9835-9853.
- Yang, W., Chang, J., Xu, B., Peng, C. & Ge, Y. 2008. Ecosystem service value assessment for constructed wetlands: A case study in Hangzhou, China. *Ecological Economics*, 68(1), 116-125.
- Yang, Y., Wyatt, I., Travis, D. & Bahorsky, M. 1998. Decolorization of Dyes Using UV/H 2 O 2 Photochemical Oxidation. *Textile Chemist & Colorist*, 30(4).
- Yang, Y., Zhan, X., Wu, S., Kang, M., Guo, J. & Chen, F. 2016. Effect of hydraulic loading rate on pollutant removal efficiency in subsurface infiltration system under intermittent operation and micro-power aeration. *Bioresource* technology, 205, 174-182.
- Yardley, J. 2013. Bangladesh Pollution, Told in Colors and Smells. *The New York Times*, 15/07/2013.
- Yaseen, D. A. & Scholz, M. 2016. Shallow pond systems planted with Lemna minor treating azo dyes. *Ecological Engineering*, 94, 295-305.

- Ye, F. & Li, Y. 2009. Enhancement of nitrogen removal in towery hybrid constructed wetland to treat domestic wastewater for small rural communities. *Ecological Engineering*, 35(7), 1043-1050.
- Ye, J., Wang, L., Li, D., Han, W. & Ye, C. 2012. Vertical oxygen distribution trend and oxygen source analysis for vertical-flow constructed wetlands treating domestic wastewater. *Ecological Engineering*, 41, 8-12.
- Ye, J., Xu, Z. & Li, H. 2008. Clogging mechanism in vertical-flow constructed wetland: clogging cause and accumulation distribution characteristics]. *Huan jing ke xue = Huanjing kexue/[bian ji, Zhongguo ke xue yuan huan jing ke xue wei yuan hui" Huan jing ke xue" bian ji wei yuan hui.]*, 29(6), 1508-1512.
- Yeddou, M., Hamadi, A., Kaddour, S., Bensaadi, Z. & Bensmaili, A. 2012. Biosorption behavior of Basic Red 46 and Violet 3 by dead Pleurotus mutilus from single-and multicomponent systems. *Journal of Chemistry*, 2013.
- Yeh, T. 2008. Removal of metals in constructed wetlands: review. *Practice periodical of hazardous, toxic, and radioactive waste management,* 12(2), 96-101.
- Yemashova, N., Telegina, A., Kotova, I., Netrusov, A. & Kalyuzhnyi, S. 2004. Decolorization and partial degradation of selected azo dyes by methanogenic sludge. *Applied biochemistry and biotechnology*, 119(1), 31-40.
- Yirong, C. & Puetpaiboon, U. 2004. Performance of constructed wetland treating wastewater from seafood industry. *Water Science and Technology*, 49(5-6), 289-294.
- Yu, J., Wang, X. & Yue, P. L. 2001. Optimal decolorization and kinetic modeling of synthetic dyes by Pseudomonas strains. *Water research*, 35(15), 3579-3586.
- Zanieri, L., Bracali, M., Bresciani, R. & Masi, F. 2010. Multi-stage CW systems for winery wastewater treatment. th IWA International Conference on Wetland Systems for Water Pollution Control, held in Venice, Italy, 2010. 1139-1144.
- Zeb, B. S., Mahmood, Q., Jadoon, S., Pervez, A., Irshad, M., Bilal, M. & Bhatti, Z. A. 2013. Combined industrial wastewater treatment in anaerobic bioreactor posttreated in constructed wetland. *BioMed research international*, 2013.
- Zhang, M.-M., Chen, W.-M., Chen, B.-Y., Chang, C.-T., Hsueh, C.-C., Ding, Y., Lin, K.-L. & Xu, H. 2010. Comparative study on characteristics of azo dye decolorization by indigenous decolorizers. *Bioresource technology*, 101(8), 2651-2656.
- Zhao, Y., Sun, G. & Allen, S. 2004. Purification capacity of a highly loaded laboratory scale tidal flow reed bed system with effluent recirculation. *Science of the Total Environment*, 330(1), 1-8.
- Zheng, H., Yang, H., Long, D. & Hua, J. 2017. Monitoring surface water quality using social media in the context of citizen science. *Hydrology and Earth System Sciences*, 21(2), 949.
- Zidan, A. R. A., El-Gamal, M. M., Rashed, A. A. & Eid, M. a. a. E.-H. 2015. Wastewater treatment in horizontal subsurface flow constructed wetlands using different media (setup stage). *Water Science*, 29(1), 26-35.

- Zissi, U. & Lyberatos, G. 1996. Axo-dye biodegradation under anoxic conditions. *Water Science and Technology*, 34(5-6), 495-500.
- Zollinger, H. 2003. Color chemistry: syntheses, properties, and applications of organic dyes and pigments, *John Wiley & Sons*.
- Zurayk, R., Nimah, M., Geha, Y. & Rizk, C. 1997. Phosphorus retention in the soil matrix of constructed wetlands. *Communications in Soil Science & Plant Analysis*, 28(6-8), 521-535.

## **APPENDIX A**

## **Published Journal Paper**

Ecological Engineering 101 (2017) 28-38



Full research paper

## Dye wastewater treatment by vertical-flow constructed wetlands

## Amjad Hussein, Miklas Scholz\*

Civil Engineering Research Group, School of Computing, Science and Engineering, The University of Salford, Newton Building, Greater Manchester, M5 4WT, UK

#### ARTICLE INFO

#### Article history: Received 9 June 2016 Received in revised form 6 January 2017 Accepted 10 January 2017

Keywords; Acid Blue 113 Basic Red 46 Chemical oxygen demand Phragmites australis Reed bed filter Textile wastewater ABSTRACT

Wetlands have long played an important role as natural purification systems. Textile industry processes are among the most environmentally unsustainable industrial processes, because they produce coloured effluents in large quantities polluting water resources. In this study, two different azo dyes (Acid Blue 113 (AB113) and Basic Red 46 (BR46)) have been fed as part of synthetic wastewater recipes to a laboratory-scale vertical-flow construction wetland set-up comprising wetlands with gravel media as controls and wetlands planted with *Phragmites australis* (Cav.) Trin. ex Steud. (Common Reed) for each dye. Two different concentrations (7 mg/l and 215 mg/l) were used for each dye at two different hydraulic retention times (48 h and 96 h). According to results for the low concentration of BR46, there is no significant (p > 0.05) difference between wetlands (unplanted and planted) in terms of dye removal. The use of plants concerning the short contact time scenario for ammonia-nitrogen (NH4-N) and a low concentration of AB113 is linked to good removal. In case of low dye concentrations, the presence of plants for the long contact time scenario impacted significantly (p < 0.05) positive on the removal efficiencies were S4% and 80% and 43% and 43% of the long and short treention times, respectively. All reductions were statistically significant (p < 0.05). For the high concentration of BR46, the removal percentages for this dye and COD were 94% and 82% and 480% and 480% the COD removals for the dye were 71%, 68% and 80%. The COD removals were 4%, 7% and 15% for the control, and the short and long retention times, respectively. For the low concentration of AB113, the percentage corresponding removals for the dye and COD were 71% and 73%, and 50% for the 64% hand 96 he relation times, respectively. For the low concentration and 52% for the control, and the short and long retention times, respectively. For the low concentration of AB113, the percentage corresponding removals for the dye and COD were 71% and 73

© 2017 Elsevier B.V. All rights reserved.

CrossMark

#### 1. Introduction

Azo dyes are important colouring agents in the textile, food and pharmaceutical industries (Tee et al., 2015; Yaseen and Scholz, 2016), and are linked to a relatively high toxicity, mutagenicity and carcinogenicity. Azo dyes and their corresponding breakdown products are difficult to treat in traditional wastewater treatment systems according to Erkurt (2010). Many technological solutions

<sup>6</sup> Corresponding author. E-mail address: m.scholz@salford.ac.uk (M. Scholz).

http://dx.doi.org/10.1016/j.ecoleng.2017.01.016 0925-8574/© 2017 Elsevier B.V. All rights reserved. such as coagulation-flocculation and advanced oxidation process (Sivakumar et al., 2013), which have been applied to treat dyes, are not feasible in practice, because of too high costs and complex processes involved. Furthermore, some developing countries such as China and Bangladesh (Chen et al., 2007; Islam et al., 2011), which have a strong textile industry, but unreliable energy sources, may benefit from less-energy demanding methods of wastewater treatment.

Textile wastewater causes considerable environmental pollution. The main challenges are high concentrations of organic matter and particular persistent colorants (dyes) that have to be resistant to the effects of sweat, soap, water, light and oxidants (Olejnik and Wojciechowski, 2012). The azo dyes AB113 and BR46 are, therefore, widely used in the textile industry (Pervez et al., 2000; Olgun and Atar, 2009; Deniz and Karaman, 2011; Yaseen and Scholz, 2016). Typically, textile industry-processing effluents contain dyes in the range between 10 and 200 mg/l (Yassen and Scholz, 2016). Most textile dyes at a rather low concentration of even >1 mg/l can be detected by the human eye (Pandey et al., 2007).

Abbreviations: AB, acid blue; ANOVA, analysis of variance; AO, acid orange; AY, acid yellow; BR, basic red; CASRN, chemical abstracts survey registry number; COD, chemical oxygen demand; DO, dissolved oxygen; DY, disperse yellow; EC, electric conductivity; HF, horizontal-flow; N, hitroger; NIA=N, anmoina-hitroger; n/a, not applicable; NO<sub>2</sub>-N, nitrite-nitrogen; NO<sub>2</sub>-N, nitrate-nitrogen; PO<sub>4</sub>-P, orthophosphate-phosphorus; TSR, reactive black; SD, standard deviation; SE, standard error; TDS, total dissolved solids; T-N, total nitrogen; TOC, total organic carbon; T-P, total phosphorus; TSS, total suspended solids; VF, vertical-flow; VY, vat yellow; A<sub>max</sub>, maximum absorbance.

Vertical-flow constructed wetlands are engineered ecosystems designed to remove pollutants from wastewater (Kadlec and Wallace, 2009). These systems mimic the treatment that occurs in natural wetlands by relying on heterotrophic microorganisms, aquatic plants and a combination of naturally occurring processes (Scholz, 2015).

Researchers previously investigated the performance of ponds (Vaseen and Scholz, 2016) and, in particular, constructed wetlands to treat textile wastewater (Table 1). However, results rarely cover all seasons (Vymazal, 2014). Pervez et al. (2000) used aerated vertical-flow wetlands to remove two textile Azo dyes (RB171 and AB113) from synthetic wastewater for a period of 70 days. The percentage removal of 98% was high for both dyes. However, the results did not assess the removal of COD, phosphorus and nitrogen.

Davies et al. (2006) used vertical-flow wetlands for aerobic degradation of Acid Orange 7 (AO7) in a short-term study (Table 1). The results showed colour and COD removals of 99% and 93%, respectively. However, phosphorus and nitrogen removal were not recorded.

Yalcuk and Dogdu (2014) used vertical-flow constructed wetlands to treat Acid Yellow 2G E107 Dye-containing wastewater (Table 1). The constructed wetland consisted of three vertical wetland filters, which were filled with fine gravel, sand and zeolite. One wetland was kept unplanted as a control and the other two were planted with *Canna idica* L. and *Typha angustifolia* L., respectively. The period of operation was only three months. The results showed that the average colour removal percentages were 87% for the control and 98% for the other wetlands. For NH<sub>4</sub>-N, the average percentages removals were 43%, 61% and 46% for the control, *C. idica* and *T. angustifolia*, respectively. The average percentages of ortho-phosphate-phosphorus (PO<sub>4</sub>-P) for the control, *C. indica* and *T. angustifolia* were 84%, 87% and 88% respectively. Only 90 days was the time of operation.

The presence of *Phragmites australis* in vertical-flow constructed wetlands has a significant impact on the removal of organic matters, aromatic amines and NH<sub>4</sub>-N (Ong et al., 2011). The growth cycle of *P. australis* traditionally completes between May and September in Britain (Haslam, 1972). Ferreira et al. (2014) detected normal growth for *Phragmites* with absence of toxic signs or depletion of leaf nitrogen content, when they used vertical-flow wetlands to treat an effluent comprising Diazo dye (DR81).

The aim of this study is to assess the efficiency of vertical-flow constructed wetlands in removing dye, COD,  $PO_4$ -P and other nutrients. The corresponding objectives are to assess (a) the role of *P*. *australis* on dye removal; (b) the influence of two groups of dyes (Acid and Direct) on the performance of constructed wetlands on dyes removal; and (c) the influence of operational parameters such as contact time, resting time and loading rate on dye removal.

#### 2. Materials and methods

#### 2.1. Wetland rig and operation

The study has been conducted between 1 May 2015 and 31 May 2016. The first month may be viewed as the start-up period. An experimental constructed wetland rig (Fig. 1) treating textile wastewater has been operated within a greenhouse located on top of the Newton Building (The University of Salford). The rig has been designed to assess the system performance by simulating processes occurring within full-scale constructed wetlands. The rig comprises twenty-two vertical-flow wetland filters, allowing wastewater to drain vertically, enhancing aerobic biodegradation of organic matter and nitrogen (Fuchs 2009). The experimental rig to minimise random



Fig. 1. Experimental vertical-flow constructed wetland rig located within a greenhouse at the beginning of the experiment (Picture taken by Mr. Amjad Hussein on 15 May 2015).

impacts of parameters such as sunlight direction and temperature differences on the wetland performances.

Operational parameters such as contact time, retention time and hydraulic loading rate on dye removal were assessed. Contact time is defined as the duration the wastewater is in contact with the wetland filter content. In comparison, resting time is the duration when the wetland filter is empty (i.e. no wastewater input). The dyes AB113 and BR46 were tested at two different concentrations (target concentrations of 7 mg/l and 215 mg/l) for different retention times (approximately 48 h and 94 h) to assess their influence on the performance of vertical-flow constructed wetlands.

Round and black plastic drainage pipes were used to construct the vertical-flow wetlands (Fig. 1). All twenty-two wetlands were designed according to the following dimensions: height of 100 cm and diameter of 10 cm. One wetland was filled with water, another one was filled to a depth of 90 cm with unwashed gravel and the other wetlands were filled to a depth of 90 cm with washed gravel (Table 2). Two different layers of gravel were used as filter media. Large gravel with a diameter of 10–20 mm was applied as the bottom layer to prevent clogging of the outlet. Pea gravel with a diameter of 5–10 mm was located at the top layer. The outlet valves were located at the centre of the bottom plate of each wetland. The internal diameter of the vinyl outlet tubing was 10 mm. Selected wetlands were planted with *P. australis* (Table 2). The

Selected wetlands were planted with *P. australis* (Table 2). The growth of *P. australis* was monitored. Dead above-ground plant parts were cut down to about 13 cm height. The cuttings were recycled by placing them into their corresponding wetland filters.

able 1
evious studies (listed in order of date) on textile wastewater treatment by constructed wetlands.

Dye used	Type of wetland	Design characteristics	Plants used	Removal performance	Duration (days)	Country of operation	References
AB113, *RB171	VF	Gravel-sand	P. australis	98% colour	70	USA	Pervez et al. (2000)
A07	VF	Gravel-sandy clay soil	P. australis	74% colour, 64% COD and 71% TOC	77	Portugal	Davies et al. (2005)
Various dyes in real wastewater	HF	Gravel-sand	Typha and cocoyam	77% colour, 72% COD and 59% sulphate	84	Tanzania	Mbuligwe (2005)
*RB5, DY211, VY46	VF-HV	Gravel-sand-tuff	P. australis	90% colour, 84% COD, 93% TSS, 52% T-N, 87% Norganic, -331% NH <sub>4</sub> -N, 88% sulphate, 80% anion surfactant and 93% TSS	60	Slovenia	Bulc and Ojstršek (2008)
A07	VF	Gravel-sludge	P. australis	94% colour, 95% COD and 86% NH4-N	27	n/a	Ong et al. (2011)
AY 2G E107	VF	Gravel-sand- zeolite	Canna and Typha	95% colour, 64% COD, 94% PO <sub>4</sub> -P and 77% NH <sub>4</sub> -N	90	Turkey	Yalcuk and Dogdu (2014)

Note: AB, acid blue; \*RB, reactive blue; AO, acid orange; VF, vertical flow; COD, chemical oxygen demand; TOC, total organic carbon; \*\*RB, reactive black; DY, disperse yellow; VY, vat yellow; HV, horizontal flow; TSS, total suspended solid; T-N, total nitrogen; N, nitrogen; NH<sub>4</sub>-N, annmonium nitrogen; PO<sub>4</sub>-P, Ortho-phosphate-phosphorus; AY, acid yellow;

#### Table 2

Packing order of the experimental constructed wetland set-up. Samples were taken at the end of each contact time period, just before the start of the resting time period between 1 May 2015 and 31 May 2016.

Wetland	Media	Plants	Dye			Resting	Contact
number			Туре	Mean (mg/l)	SD (mg/l)	time (h)	time (h)
1	Only Water	No	None	n/a	n/a	2	94
2	Unwashed gravel	No	None	n/a	n/a	2	94
3	Washed gravel	No	None	n/a	n/a	2	94
4	Washed gravel	No	BR46	6.9	1.33	2	94
5	Washed gravel	No	AB113	6.6	1.82	2	94
6,7	Washed gravel	Yes	AB113	6,6	1,82	2	94
8,10	Washed gravel	Yes	BR46	6.9	1.33	2	94
11.12	Washed gravel	Yes	BR46	6.9	1.33	48	48
13.14	Washed gravel	Yes	AB113	6.6	1.82	48	48
15.16	Washed gravel	Yes	BR46	209.3	4.48	48	48
17,18	Washed gravel	Yes	AB113	221.5	31.86	48	48
19,20	Washed gravel	Yes	BR46	209.3	4.48	96	96
21,22	Washed gravel	Yes	AB113	221.5	31.86	96	96

Note: SD, standard deviation; n/a, not applicable; BR, basic red; AB, acid blue.

The dyes AB113 and BR46 were used at two inflow concentrations (low and high) to assess the performance of the vertical-flow wetland systems to treat dye at the presence of fertiliser only. No other contaminants were added to allow for a statistically sound assessment of the impact of dye. The fact that apart from some high dye concentrations all other concentrations in the inflow were relatively low may naturally result in low outflow concentrations as well, indicating good treatment performance, but not necessarily good removal performances.

Details for each dye can be found in Tables 2 and 3. The textile dye BR46 had a wavelength for the maximum absorbance ( $\lambda_{max}$ ) of 530 nm (Khatae, 2009), and was obtained from Dystar (Am Prime Park, Raunheim, Germany). The dye AB113 had a  $\lambda_{max}$  of 566 nm (Shirzad-Siboni et al., 2014), and was obtained from Sigma Aldrich (The Old Brickyard, New Road Gillingham, Dorset, United Kingdom). Dye stock solutions were prepared by dissolving 10 g of each dye in 1000 ml of distilled water. The experimental solutions were obtained by diluting stock solution samples to the required concentrations.

The fertiliser TNC Complete, which is an aquatic plant supplement bought from TNC Limited (Spotland Bridge Mill, Mellor Street, Rochdale, United Kingdom), was used in the experimental work as a nutrient and trace element source for the plants. The corresponding ingredient composition was as follows: nitrogen (1.5%), phosphorus (0.2%), potassium (5%), magnesium (0.8%), iron (0.08%), manganese (0.018%), copper (0.002%), zinc (0.01%), boron (0.01%) and molybdenum (0.001%). Ethylene diamine tetra acetic acid (EDTA), which is used as a source for copper, iron, manganese and zinc, is also provided by TNC Complete. For ten litre of freshly prepared influent, one millilitre fertiliser was added.

The wetland system has been designed to operate in batch flow mode to avoid expenses such as pumping and automatic control costs. Wastewater was poured directly into the wetland filter from the top. The wastewater stayed within the wetland for the duration of the contact time. The treated wastewater was released from the wetland through an outlet pipe located in the centre of the wetland bottom. The duration the liquid stayed within the wetland is the resting time (see also above).

### 2.2. Wetland filter set-up

Table 4 outlines the application of the simplified statistical wetland filter set-up design (Table 2) to assess the impact of individual key variables. Wetland 1 was deliberately left empty (without media and plants; Table 2). Only tap water was used as the influent. The mean tap water values for dissolved oxygen (DO), pH, electric conductivity (EC), redox potential, total suspended solids (TSS) and turbidity were 10.1 mg/l, 7.6, 85.1  $\mu$ S, -32 mV, 1 mg/l and 1.3 NTU, respectively.

Wetland 2 was filled with unwashed gravel, while Wetland 3 comprised washed gravel and tap water. Wetlands 4, 8/10, and 11/12 were filled with the same media as Wetland 3, and planted with *P. australis*, except for Wetland 4, which was used as a control (Table 2). The rhizomes of the plants were washed free of sediment

#### A. Hussein, M. Scholz / Ecological Engineering 101 (2017) 28-38

 Table 3

 Details of dyes used in the experimental constructed wetlands.

Dye	BR46	AB113
Molecular weight	401.3	681.65
Molecular formula	C <sub>18</sub> H <sub>21</sub> BrN <sub>6</sub>	C32H21N5Na2O6S2
Source	Dystar	Sigma Aldrich
CASRN	12221-69-1	3351-05-1
Purity of dye (%)	70-80	Approximately 50
Nutrient content for 200 mg of dye (mg/l)	50.0 ortho-phosphate-phosphorus,	137.0 ortho-phosphate-phosphorus,
	7.5 ammonia-nitrogen and 0.93 nitrate-nitrogen	7.7 ammonia-nitrogen and 1.8 nitrate-nitrogen
Chemical structure	$\{ \begin{array}{c} \sum_{i=1}^{i} CH_i \\ \sum_{i \in I_i} \sum_{i \in I_i} \sum_{j \in I_i} \sum_{i \in I_i} \sum_$	NMXA

Note: CASRN, chemical abstracts survey registry number; BR, basic red; AB, acidic blue.

Table 4

Application of the simplified statistical wetland filter set-up design (Table 2) to assess the impact of individual key variables.

Comparison of two wetland systems wit	h each other	Impact to be assessed
First wetland with number	Second wetland with number	
1	2	Unwashed gravel
2	3	Washed gravel
3	4	BR46
3	5	AB113
4	5	Difference between BR46 and AB113
5	6,7	Phragmites australis
6,7	8,10	Difference between BR46 and AB113
8,10	11,12	Decrease in contact time(or increase in resting time)
11,12	13,14	Difference between BR46 and AB113
11,12	15,16	Increased BR46 concentration
13,14	17,18	Increased AB113 concentration
15,16	19,20	Increased contact and resting times
17,18	21,22	Increased contact and resting times
15,16	17,18	Difference between BR46 and AB113
19,20	21,22	Difference between BR46 and AB113

and planted in selected gravel-filled wetlands. The influent was tap water mixed with BR46 ( $6.9 \pm 1.33$  mg/l).

The contact time for Wetlands 8 and 10 was 94 h, and the resting time was 2 h, while for Wetlands 11 and 12, the contact time was 48 h and the resting time was 48 h (Table 2). The purpose for different contact and resting times was to assess the performance of wetlands in removing dyes under different design conditions, and compare the results with those for Wetland 4 to assess the importance of the presence of plants on dye removal and other parameters. The same previously explained set-up was used for Wetlands 5, 6/7, and 13/14, but with the dye AB113 at a concentration of about  $6.6 \pm 1.82 \text{ mg/l}$ .

The Wetlands 15/16, and 19/20 contained the same media, plants and dye (BR46) as Wetlands 8/10, and 11/12, but the dye concentration was approximately 209 mg/l. The Wetlands 15 and 16 are replicates with 48-h contact time and 48-h resting time, while the Wetlands 19 and 20 are replicates with 96-h contact time and a resting time of 96h. Wetlands 17, 18, 21 and 22 are similar to Wetlands 15, 16, 19 and 20, but the influent dye was AB113 at a concentration of about 216 mg/l (Table 2).

### 2.3. Experimental and statistical analysis

The water quality analysis was performed according to APHA (1995), unless stated otherwise, to assess the annual and seasonal treatment performance. Both influent and effluent were analysed. The effluent was obtained from the bottom of each wetland filter. The inflow wastewater was freshly prepared before it was poured into the wetland from the top.

The spectrophotometer DR 2800 Hach Lange (www.hach.com) was applied for the water quality analysis for variables including dyes, COD, NH<sub>4</sub>-N, NO<sub>3</sub>-N, PO<sub>4</sub>-P and TSS. Turbidity was determined with a Turbicheck Turbidity Meter (Lovibond Water Testing, Tintometer Group, Division Street, Chicago, IL, USA). The pH and redox potential for all samples was measured using a VARIO pH meter (Wissenschaftlich-Technische Werkstätten, Weilheim, Germany). This meter was calibrated with standard buffer solutions of pH 4, 7 and 9 each two weeks or whenever required. The EC and total dissolved solids (TDS) were measured using a Mettler Toledo Education Line Conductivity Meter (Boston Road, Leicester, UK). The DO was measured using a Hach Lange HQ30d Flexi Meter (Pacific Way, Salford, UK). Water samples were taken regularly as shown in Table 2, and they were brought to the university-based laboratory for testing.

Water samples were always taken at the same time (10:00 to 11:00 am) to ensure that the various environmental boundary conditions, which are variable throughout the day, such as diurnal variations of pH and DO, would not impact on the results. Laboratory tests were performed to measure dyes, COD, PO<sub>4</sub>-P, NO<sub>3</sub>-N, NH<sub>4</sub>-N, EC, DO, redox potential, pH, TSS, TDS and turbidity. The temperature at the site is recorded each week using a thermometer placed alongside the constructed wetlands.

In order to investigate statistically significant differences, the Shapiro-Wilk's test (Shapiro and Wilk, 1965; Razali and Wah, 2011) was used to assess the normality of data. A one-way analysis of variance (ANOVA) test using Statistical Package for the Social Science software was applied to analyse normal distributed data, while the Mann-Whitney test was used to analyse non-normal data

(Stoline, 1981; Kasuya, 2001). These ANNOVA and Mann-Whitney tests were used to compare means between different treatments such as the ones highlighted in Table 4. Significant findings have been outlined and discussed. Somehow surprising and/or important insignificant findings have occasionally been highlighted as well.

## 3. Results and discussion

3.1. Test of normality for plant and water quality variables

Tests of normality results for dimensions of *P. australis* and for effluent water quality characteristics regarding general physical and chemical variables are discussed in the supplementary material S1

### 3.2. Plant growth assessment

When temperatures started to decrease in winter (see Supplementary material S2), the plants in this experiment began to yellow. The above-ground parts died, were cut and returned to the wetland filters. The authors followed common practice to cut the aboveground plant parts down to between 10 and 15 cm height according to Stefanakis et al. (2014).

Based on the experimental results, the most critical observation is that for all considered concentrations and contact times, the plants that were subjected to the dye AB113 revealed more effective growth compared to those that were fed by the dye BR46. Taking into consideration the potential impact of retention time, it has been noted that plants associated with a long contact time showed much better growth than those plants linked to a short one (Table 5). These findings support those by Pagter et al. (2005), who investigated the effect of water stress tolerance of *P. australis* grown in the laboratory by examining effects of different levels of required water. The results showed that a water deficit reduces the leaf biomass per plant and the leaf area.

The ANOVA and Mann-Whitney tests were applied for normal and non-normal distributed data, respectively (see above). Regarding the growth of plants, there is no significant (p > 0.05) difference with respect to the length and diameter at the low concentration of the dye AB113, while for the high concentration, there is a significant (p > 0.05) difference for the diameter but no significant (p > 0.05) difference for the diameter but no significant (p > 0.05) difference for the length. In case of plants grown in the presence of the dye BR46, there is a significant (p < 0.05) difference for both parameters was noted at the high concentration of dye, while no significance (p > 0.05) for both parameters was noted at the high concentration of dye. Readers may wish to consult Table S1 and the corresponding discussion in the supplementary material S1 for more details. However, no generic interpretation of the data can be made for both dyes, indicating further research needs by ecologists.

## 3.3. Dissolved oxygen and redox potential

Redox potentials of less than -100 mV indicate anaerobic environments, while values greater than 100 mV indicate aerobic environments (Suthersan, 2001). Redox potential and DO values for the effluents of both dyes were in the range between -32 and - 6 mV and between 5.4 and 8.2 mg/l, respectively, for low dye concentrations, and between -32 and - 4 mV and between -32 and - 6 mV respectively, for high dye concentrations (Tables 2 and 2.5 mg/l, respectively, for high dye concentrations (Tables 2 and 2.5 mg/l, respectively, for high dye concentrations (Tables 2 and 2.5 mg/l, respectively, for high dye concentrations (Tables 2 and 2.5 mg/l, here is the possibility that the effluent samples got aerated between taking the samples and corresponding measurements. Nevertheless, these findings indicate that dye degradation may have taken place in both aerobic and anaerobic environments.

## 3.4. Electrical conductivity, total suspended solids and turbidity

Electrical conductivity can be used as an indirect measure of the charge (or the ion-carrying species) in the wetland outflow (Islam et al., 2011). For the low concentration of both dyes (BR46 and AB113), as shown in Table 6, an increase was noted for Wetlands 11/12 and Wetlands 13/14 (contact times of 48 h for each), respectively, when compared to Wetlands 8/10 and 4 as well as Wetlands 6/7 and 5 (contact times of 94 h for each), respectively. For the high concentration in case of dye AB113, the average value of EC in the effluent of Wetlands 21/22 (high resting and contact times) is the same as for the influent, while for Wetlands 17/18 (low resting and contact times), there is a decrease in the average value for the effluent of Wetlands 15/16 (low resting and contact times;  $389 \pm 99.73 \, \mu$ S/cm), and Wetlands 19/20 ( $385 \pm 83.21 \, \mu$ S/cm), if compared to the influent ( $357 \pm 112.3 \, \mu$ S/cm).

Yalcuk and Dogdu (2014) have reported similar findings. The measurement of TSS is important for the design of water treatment facilities (Dzurik, 2003). For the dye BR46 in case of low concentration, there is an increase in the effluent for Wetlands 4, 11 and 12 (Tables 2 and 6), while there is no increase in the average value for Wetlands 8 and 10 (Tables 2 and 6), if compared with the influent.

For the high concentration scenario, the removal in Wetlands 19 and 20 (high resting and contact times) is better than that in Wetlands 15 and 16 (low resting and contact times). For the dye AB113 in case of low concentration, increased values of TSS in the effluent were noted for the Wetlands 5, 6/7, and 13/14 (Tables 2 and 6). Only a slight increase was noted for Wetlands 6/7 (low resting and high contact times), while a higher increase was recorded for Wetlands 13/14 (high resting and low contact times). In case of the high concentration scenario, a statistically insignificant (p > 0.05) value was noted for the Wetlands 17/18 (low resting and contact times) and 21/22 (high resting and contact times). Increased values for Wetlands 21/22 were noted (392 ± 253.13 mg/l), if compared to the Wetlands 17/18 (371.7 ± 177.57 mg/l).

Turbidity is often used to interpret the degree of clarity of water. It is a variable often applied as an indicator of the amount of suspended sediments and/or larger micro-organisms in water. A high turbidity of surface water may also indicate elevated concentrations of TSS, reduced algal populations, and potential harm to fish and other aquatic fauna (Postolache et al., 2007).

For the dye BR46 (p < 0.05) in case of the low concentration scenario, removal was noted for the planted Wetlands  $8/10 (3.0 \pm 1.33 \text{ mg/l})$ , while there was an increase linked to the unplanted Wetlands 4 (8.8 ± 7.12 mg/l; p > 0.05) and planted Wetlands 11/12 ( $3.6 \pm 1.89$  mg/l) (see also Tables 2 and 6), if compared to the influent ( $3.5 \pm 2.89$  mg/l). In case of high concentrations, good removal was noted for the Wetlands 15/16 (low resting and contact times;  $6.6\pm2.20\,mg/l;\,p\!>\!0.05)$  and the Wetlands 19/20 (high resting and contact times; 6.8 ± 3.90 mg/l; p > 0.05), when compared with the influent ( $16.6 \pm 3.10 \text{ mg/l}$ ). For the dye AB113 in case of low concentration (p < 0.05), increased values in the effluent were noted for the unplanted Wetland 5 ( $39.2 \pm 27.40 \text{ mg/l}$ ), the planted Wetlands 6/7 (19.1  $\pm$  16.11 mg/l) and the planted Wetland  $13/14(46.8 \pm 32.67 \text{ mg/l})(\text{see also Tables 2 and 6})$ , if compared to the influent ( $7.4 \pm 5.67 \text{ mg/l}$ ). A greater increase was recorded for Wetlands 13/14 (high resting and low contact times), if compared to Wetlands 6/7 (low resting and high contact times). In case of the high concentration scenario, a removal was noted for Wetlands 17/18 (low resting and contact times;  $208.3 \pm 196.67 \text{ mg/l}$ ; p < 0.05), while there was an increase in the average value for Wetlands 21/22 (high resting and contact times;  $224 \pm 202.69 \text{ mg/l}$ ; p > 0.05), if compared to the influent (213.4 ± 157.62 mg/l; p < 0.05).

Characte	ristics		Length (cm)			Diameter (mm	ı)	
Dye	Wetland number	Number of steams	Mini-mum	Maxi-mum	Mean ±SD	Mini-mum	Maxi-mum	Mean ±SD
BR46	8 and 10	40	40	110	$86 \pm 25.0$	2.5	3.8	$3.6 \pm 0.16$
	11 and 12	28	17	44	$30 \pm 9.0$	1.1	1.8	$1.5 \pm 0.30$
	15 and 16	34	14	50	$27 \pm 13.0$	1.0	1.2	$1.1 \pm 0.07$
	19 and 20	23	12	34	$23 \pm 8.1$	0.9	1.5	$1.2 \pm 0.25$
AB113	6 and 7	45	110	141	$131 \pm 9.5$	3.4	3.8	$3.6 \pm 0.16$
	13 and 14	30	23	41	$32 \pm 4.5$	2.1	2.4	$2.2 \pm 0.10$
	17 and 18	36	28	46	$35 \pm 6.0$	2.1	2.3	$2.2 \pm 0.06$
	21 and 22	28	22	52	$39 \pm 10.7$	1.8	2.5	$2.1 \pm 0.25$

Table 5 Dimensions of Phragmites australis (Cav.) Trin. ex Steud. (Common Reed) planted in the experimental wetlands (Tables 2 and 4).

Note: BR, basic red; AB, acidic blue; SD, standard deviation.



Fig. 2. Inflow and outlow concentrations of Basic Red 46 for Wetlands 4, 8/10 and 11/12 (see also Tables 2 and 4).

Many researchers confirmed that vertical-flow wetlands have a relatively poor ability to remove TSS and turbidity (Lin et al., 2005; Bulc and Ojstršek, 2008). Regarding this study, for the low concentration of BR46, a long contact time (94 h) was better than a short (48 h) one in removal of TSS and turbidity. While for high concentrations, the short contact time was better than the long contact time. In case of low and high concentrations for AB113, the four-day contact time was not enough to reduce TSS and turbidity.

## 3.5. pH value

For the low concentration of the dye BR46, the average pH inflow value was 7.2. A slight increase in the effluent pH values of 0.16 and 0.03 were noted for the unplanted Wetland 4 and the planted Wetlands 11/12 (see also Tables 2 and 6), respectively, while there was a decrease of 0.31 for Wetlands 8/10 (replicates), as shown in Table 6. For the high concentration, the average influent value was 6.49. Increases of 0.19 and 0.57 were noted for Wetlands 15/16 (low contact and resting times) and Wetlands 19/20 (high contact and resting times), respectively. In case of the dye AB113 for the low concentration scenario, the average influent pH value was 7.32. A slight pH increase of 0.11 was noted for the unplanted Wetland 5, while the pH value decreased for the planted Wetlands 6/7 and the planted Wetlands 13/14 (see also Tables 2 and 6) by 0.44 and 0.15, respectively. For the high concentration, the average influent pH value was 8.39. The pH value decreased for the Wetlands 17/18 (low  $resting \, and \, contact \, times) \, and \, the \, Wetlands \, 21/22 \, (high \, resting \, and \,$ contact times) by 1.10 and 1.06, respectively, as shown in Table 6.

These results indicate the ability of macrophytes such as *P. australis* to modify pH conditions in the rhizosphere, confirming results by Brix et al. (2002). However, these researches used a different wetland plant (*Typha angustifolia* L.).

oInflow △Wetland (5) ◇Wetlands (6/7) ★Wetlands (13/14)



Fig. 3. Inflow and outlow concentrations of Acid Blue 113 for Wetlands 5, 6/7 and 13/14 (see also Tables 2 and 4).

### 3.6. Dye removal and chemical oxygen demand removal

The percentage removal of dyes is shown in Table 7. The degradation of azo dyes takes place both in aerobic and anaerobic conditions via various processes involving enzymes and/or chemical reduction (Pandey et al., 2007). BR46 is theoretically easier to degrade than AB113 due to a lower molecular weight (Table 3).

There is no significant (*p* > 0.05) difference in dye removal for the low concentration of BR46 between the unplanted control Wetland 4, the planted Wetlands 8/10 and the planted Wetlands 11/12 (Tables 2 and 6, and Fig. 2). It follows that the presence of plants does not affect dye removal. In comparison, for AB113, the dye removal in the planted Wetlands 6 and 7 was significantly (*p* < 0.05) better

Charact	eristics			pH			Redox potent	ial (mV)		Dissolved oxy	/gen (mg/l)	
Dye	Type of flow	Wetland number	No, of samples	Mini-mum	Maxi-mum	Mean ±SD	Mini-mum	Maxi-mum	Mean ±SD	Mini-mum	Maxi-mum	Mean ±SD
BR46	In	n/a	70	6.80	7.54	$7.2 \pm 0.147$	-39	5	$-25 \pm 7.7$	9.0	10.9	$9.7 \pm 0.46$
	Out	4	70	7.11	7.75	$7.36 \pm 0.137$	-43	-14	$-29 \pm 6.35$	5.5	10.0	$8.2 \pm 1.12$
	Out	8 and 10	70	6.67	7.21	$6.89 \pm 0.148$	-22	8	$-5 \pm 6.79$	3.7	8.1	$5.4 \pm 0.92$
	Out	11 and 12	70	6,85	8.18	$7,23 \pm 0.212$	-40	-7	$-22 \pm 9.25$	2.9	9.5	$7.5 \pm 1.54$
	In	n/a	70	6.04	6.83	$6.49 \pm 0.179$	-6	28	$12 \pm 8.06$	8.7	10.7	$9.6 \pm 0.41$
	Out	15 and 16	70	6.35	7.11	$6.68 \pm 0.217$	-17	17	$4 \pm 7.53$	3.8	9.2	$6.3 \pm 1.34$
	Out	19 and 20	35	6.65	7.37	$7.06 \pm 0.163$	-24	4	$-13 \pm 7.58$	3.2	8.7	$6.0 \pm 1.29$
AB113	In	n/a	70	7.14	7.65	$7.32 \pm 0.122$	-43	-21	$-30 \pm 5.4$	8.9	11.2	$9.6 \pm 0.55$
	Out	5	70	7.20	7.77	$7.43 \pm 0.165$	-51	-19	$-32 \pm 6.5$	5.1	10.1	$8.3 \pm 1.38$
	Out	6 and 7	70	6.62	7.4	$6.88 \pm 0.184$	-33	17	$-6 \pm 8.2$	3.7	6.9	$5.4 \pm 0.77$
	Out	13 and 14	70	6.78	7.57	$7.17 \pm 0.199$	-40	5	$-23 \pm 8.2$	3.5	10.4	$8.1 \pm 1.45$
	In	n/a	70	7.25	9.04	$8.39 \pm 0.432$	-129	-46	$-88 \pm 18.2$	8.2	10.7	$9.5 \pm 0.49$
	Out	17 and 18	70	7.04	7.6	$7.29 \pm 0.118$	-46	-10	$-28 \pm 9.6$	5.4	9.9	$7.5 \pm 1.19$
	Out	21 and 22	35	7.10	7.99	$7.33 \pm 0.202$	-74	-12	$-32 \pm 14$	3.9	10.0	$7.1 \pm 1.76$
haract	eristics			Total suspend	led solids (mg/l)		Turbidity (NI	U)		Electric condo	uctivity (µS/cm)	
BR46	In	n/a	70	0	5.1	$2.0 \pm 1.51$	1.6	18.5	$3.5 \pm 2.89$	90	270	$159 \pm 51.6$
	Out	4	70	0	23.0	$6.8 \pm 5.81$	1.9	32	$8.8 \pm 7.12$	70	483	$193 \pm 79.1$
	Out	8 and 10	70	0	6.2	$2.0 \pm 1.88$	1.5	6	$3.0 \pm 1.33$	103	749	$257 \pm 97.4$
	Out	11 and 12	70	0	10.2	$3.4 \pm 3.33$	1.9	12	$3.6 \pm 1.89$	82	419	$180 \pm 62.4$
	In	n/a	70	37.1	59.0	$47.9 \pm 5.11$	8.9	24	$16.6 \pm 3.1$	147	701	$357 \pm 112.3$
	Out	15 and 16	70	1.1	21.3	$7.6 \pm 4.70$	2.9	13	$6.6 \pm 2.2$	253	647	$389 \pm 99.73$
	Out	19 and 20	35	2.3	41.0	$6.9 \pm 8.41$	2,9	20,1	$6.8 \pm 3.9$	251	535	$385 \pm 83.2$
AB113	In	n/a	70	5.5	25.0	$14.1 \pm 5.40$	1.3	21.0	$7.4 \pm 5.67$	70	350	$220 \pm 104.7$
	Out	5	70	3.4	85.0	$41.9 \pm 23.07$	5.8	134.0	$39.2 \pm 27.40$	75	396	$242 \pm 87.9$
	Out	6 and 7	70	1.4	64.0	$18.8 \pm 15.92$	2.8	57.0	$19.1 \pm 16.11$	114	448	$294 \pm 85.4$
	Out	13 and 14	70	3.5	133.0	$46.5 \pm 33.60$	2.3	128.0	$46.8 \pm 32.67$	83	430	$235 \pm 99.6$
	In	n/a	70	119,0	843.0	$364.8 \pm 195.47$	21	660	$213.4 \pm 157.62$	465	976	$635 \pm 123.3$
	Out	17 and 18	70	121.0	706.0	$371.7 \pm 177.57$	12	650	$208.3 \pm 196.67$	456	854	$624 \pm 99.1$
	Out	21 and 22	35	67.0	863.0	392 + 253.13	9	675	224 - 202.69	502	844	$635 \pm 93.3$

Characte	naracteristics			Dye concentra	tion (mg/l)		Dye loading rate	e (mg/l/d)			
Dye	Type of flow	Wetland number	No. of samples	Mini-mum	Maxi-mum	Mean ±SD	Removal (%)	Mini-mum	Maxi-mum	Mean ±5D	Removal (%
BR46	In	n/a	70	4.4	10.0	$6.9 \pm 1.33$	n/a	0.42	0.94	$0.65 \pm 0.12$	n/a
	Out	4	70	0.1	0.9	$0.3 \pm 0.2$	96	0.006	0.08	$0.03 \pm 0.02$	95
	Out	8 and 10	70	0.1	0.7	$0.2 \pm 0.1$	97	0.006	0.07	$0.02 \pm 0.01$	97
	Out	11 and 12	70	0.1	0.5	$0.2 \pm 0.1$	97	0.005	0.02	$0.02 \pm 0.01$	97
	In	n/a	70	200.0	218.0	$209.3 \pm 4.48$	n/a	19.4	21.1	$20.3 \pm 0.39$	n/a
	Out	15 and 16	70	2.0	56	$37.7 \pm 19$	82	0.19	5.4	$3.7 \pm 1.8$	82
	Out	19 and 20	35	2.0	29.8	$12.5 \pm 9$	94	0.12	1.74	$0.7 \pm 0.5$	89
AB113	In	n/a	70	3.5	10.5	$6.6 \pm 1.82$	n/a	0.27	0.82	$0.51 \pm 0.11$	n/a
	Out	5	70	0.1	4.6	$1.9 \pm 1.0$	71	0.007	0,33	$0.14 \pm 0.07$	73
	Out	6 and 7	70	0.1	4.5	$1.3 \pm 1.1$	80	0.006	0.32	$0.09 \pm 0.08$	82
	Out	13 and 14	70	0.3	4.5	$2.1 \pm 1$	68	0.216	0.32	$0.15 \pm 0.07$	71
	In	n/a	70	155.0	292.0	221,5 ± 31,86	n/a	12,77	22,6	$17.23 \pm 2.5$	n/a
	Out	17 and 18	70	17.1	191,5	$63 \pm 48$	71	1.23	13,8	$4,54 \pm 3,45$	74
	Out	21 and 22	35	26.91	187.5	$59 \pm 40$	73	0.97	6.77	$2.12 \pm 1.44$	75
Characte	ristics			COD concentra	ation (mg/l)		COD loading rat	te (mg/l/d)			
BR46	In	n/a	20	11	32	$22 \pm 6$	n/a	0.88	2,56	$1.76 \pm 0.48$	n/a
	Out	4	20	2,8	19,5	$11 \pm 5.4$	50	0,21	1,44	$0.81 \pm 0.4$	54
	Out	8 and 10	20	0,3	13.8	$7.2 \pm 4.3$	67	0.02	1.02	$0.53 \pm 0.32$	70
	Out	11 and 12	20	0.7	16.3	$9.1 \pm 5.1$	59	0.05	1.2	$0.67 \pm 0.38$	62
	In	n/a	20	249	280	$261 \pm 11$	n/a	19,9	22.4	$20.88 \pm 0.8$	n/a
	Out	15 and 16	20	6	100	$69 \pm 34$	74	0,44	7.4	$5,1 \pm 2,52$	76
	Out	19 and 20	20	14	40	$28 \pm 9$	89	0.52	1.48	$1.04 \pm 0.33$	95
AB113	In	n/a	20	36.9	86.4	$48.2 \pm 12.2$	n/a	2.95	6.91	$3.84 \pm 1.04$	n/a
	Out	5	20	27.6	73	$46 \pm 14.7$	5	2.04	5.4	$3.4 \pm 1.11$	11
	Out	6 and 7	20	14	78	$40.6 \pm 19.4$	16	1.04	5,77	$3.03 \pm 1.55$	21
	Out	13 and 14	20	24	72	$44.6 \pm 12.9$	7	1.78	5.32	$3.18 \pm 0.96$	17
	In	n/a	20	543	610	$584 \pm 34$	n/a	43.44	48.8	$47.84 \pm 1.3$	n/a
	Out	17 and 18	20	117	454	$271 \pm 103$	54	8.66	33.6	$22.34 \pm 7.7$	53
	Out	21 and 22	20	126	362	$262\pm86$	55	9.32	26.78	$21.46 \pm 6.3$	55



Fig. 4. Inflow and outlow concentrations of Basic Red 46 for Wetlands 15/16 and 19/20 (see also Tables 2 and 4).

than the one for the unplanted control Wetland 5 (Fig. 3) due to the presence of plants (Keskinkan and Lugal Göksu, 2007).

For high concentrations, the best percentage removals for BR46 were noted for Wetlands 19/20 (high resting and contact times), which were statistically significantly (p < 0.05) different, if compared to those of Wetlands 15/16 (low resting and contact times). Wetlands with a low loading rate performed better than those wetlands with a high loading rate (Fig. 4). For the dye AB113, there was no significant (p > 0.05) difference in the removal of dyes between Wetlands 17 and 18 (low resting and contact times), and Wetlands 21 and 22 (high resting and contact times) as shown in Table 7.

For COD removal concerning low concentrations of dyes (AB113 and BR46), wetlands with a long retention time had the best COD removal (regardless of planting regime), if compared to the control and other wetlands having short retention times. Cheng et al. (2011) operated two series of vertical-flow constructed wetlands (one planted and other one unplanted) under different C: N: P ratios. Their results obtained showed that the removal of COD depended on the presence and development of plants. The removal rate was higher in the planted wetlands (Cheng et al., 2011). Concerning high concentrations of dye BR46, the best removal

Concerning high concentrations of dye BR46, the best removal percentages for COD were noted for Wetlands 19/20 (high resting and contact times), which were statistically significantly (p < 0.05) different from Wetlands 15/16 (low resting and contact times). Wetlands with a low loading rate performed better than those wetlands with a high loading rate (Table 7). In case for the dye AB113, there was no significant (p > 0.05) difference in the removal of COD between Wetlands 17/18 (low resting and contact times) and Wetlands 21/22 (high resting and contact times) as shown in Table 7.

### 3.7. Nutrient removal

The removal of phosphorous is controlled by chemical and physical adsorption, sedimentation, plant uptake and precipitation in constructed wetland systems (Brix, 1997). For the low concentrations of BR46 and AB113, the removal percentages in Wetlands 8/10 (89%) as well as Wetlands 6/7 (30%) were significantly (p < 0.05) better than those of the control Wetlands 4 (78%) and 5 (27%) as well as Wetlands 11/12 (68%) and 13/14 (14%), which have short and long retention times in that order (Table 8). For the high concentrations of BR46, the removal percentages of Wetlands 19/20 (high resting and contact times; 81%) were significantly (p < 0.05) better than those for Wetlands 15/16 (low resting and contact times; 70%), while for AB113, the removal percentages of Wetlands 17/18 (low resting and contact times; 78%) (p < 0.05) were also significantly better than those for Wetlands 21/22 (high resting and contact times; 71%) as shown in Fig. 5.

Characteristics			Ammonia-ni	itrogen (mg/l)			Nitrate-nitro	ogen (mg/l)			Ortho-phosp	hate-phosph	orus (mg/l)	
Dye Type of flow	Wetland number	No. of samples	Mini-mum	Maxi-mum	Mean ± SD	Removal (%)	Mini-mum	Maxi-mum	Mean ± SD	Removal (%)	Mini-mum	Maxi-mum	Mean ± SD	Removal (%
BR46 In	n/a	20	0.54	0.75	$0.62 \pm 0.065$	n/a	17.9	18.0	$18.0\pm0.03$	n/a	5.3	6.3	$5.9 \pm 0.33$	n/a
Out	4	20	0.45	0.68	$0.55 \pm 0.086$	11	4.5	16.9	$9.8 \pm 3.98$	45	0.9	1.7	$1.3 \pm 0.27$	78
Out	8 and 10	20	0.21	0.51	$0.36 \pm 0.111$	42	2.5	13.9	$7.5 \pm 3.69$	58	0.19	1.3	$0.7 \pm 0.4$	89
Out	11 and 12	20	0.49	06.0	$0.68 \pm 0.118$	-10	3.5	16.0	$8.9 \pm 3.83$	50	0.5	2.8	$1.9 \pm 0.83$	68
Ц	n/a	20	0.98	1.04	$1.02 \pm 0.016$	n/a	19.3	20.0	$19.7 \pm 0.18$	n/a	55.0	87.0	$64.0 \pm 8.50$	n/a
Out	15 and 16	20	0.25	0.47	$0.35 \pm 0.075$	66	1.5	16.2	$6.2 \pm 5.43$	69	6.5	47.0	$19.3 \pm 13.20$	70
Out	19 and 20	20	0.21	0.46	$0.31 \pm 0.078$	70	0.6	12.1	$4.9 \pm 4.46$	75	1.9	43.6	$12.1 \pm 14.60$	81
AB113In	n/a	20	0.63	0.70	$0.66 \pm 0.022$	n/a	8.4	08.5	$8.5 \pm 0.03$	n/a	6.9	8.3	$7.6 \pm 0.43$	n/a
Out	5	20	0.37	0.62	$0.48 \pm 0.079$	27	10.5	16.3	$12.1 \pm 1.38$	-42	4.8	6.4	$5.6 \pm 0.49$	27
Out	6 and 7	20	0.35	0.81	$0.52 \pm 0.143$	21	3.9	6.5	$4.9 \pm 0.98$	42	4.1	6.4	$5.3 \pm 0.58$	30
Out	13 and 14	20	0.35	0.56	$0.47 \pm 0.062$	29	10,0	15.1	$11.6 \pm 1.45$	-36	5.7	7.7	$6.6 \pm 0.51$	14
ll	n/a	20	2.21	2.77	$2.33 \pm 0.159$	n/a	19.4	20.0	$19.8 \pm 0.21$	n/a	148.0	154.0	$151.0 \pm 2.10$	n/a
Out	17 and 18	20	0.73	1.54	$1.00 \pm 0.236$	57	0.9	15.2	$6.0 \pm 5.8$	70	15.0	50.4	33.1±11.30	78
Out	21 and 22	20	0.80	1.97	$1.10 \pm 0.315$	53	0.8	15.2	$5.3 \pm 5.10$	73	23.3	70.3	435+1530	71



Fig. 5. Inflow and outlow concentrations of ortho-phosphate-phosphorus for Wet-lands 17/18 and 21/22 (see also Tables 2 and 4).

As indicated in Table 8, the concentrations of NH<sub>4</sub>-N for both dyes are very low due to their chemical structures (Yalcuk and Dogdu, 2014). For the high concentration of dyes, there are no significant (p>0.05) differences between wetlands. For the low concentration of the dye BR46, the removals for the planted Wetlands 8 and 10 were significantly (p < 0.05) better than those for the corresponding removal of the unplanted Wetland 4, while there are increases in removal for planted Wetlands 11 and 12 (Table 8). For the low concentration of the dye AB113, the removal percentages with respect to planted Wetlands 13 and 14 were better than those for the unplanted Wetland 4 and the planted Wetlands 6 and 7 (Table 8), which was statistically insignificant (p > 0.05).

The removal of NO3-N is a relatively time-consuming anaerobic process, and naturally takes place mostly in the sediment (Mustafa et al., 2009). For the high concentration of both dves (BR46 and AB113), the removal percentages for wetlands with a long retention times are better than those for wetlands having short retention times, confirming findings by Van Loosdrecht and Clement (2005). Regarding the low concentration of BR46, the removal percentages for the planted Wetlands 8 and 10 were better than those for the unplanted Wetland 4 and the planted Wetlands 11 and 12 (Table 8). For the low concentrations of AB113, there were significantly (p < 0.05) better removal percentages for the Wetlands 6 and 7 (low resting and high contact times), but there were increases in the effluents regarding the NO3-N removal of Wetlands 13 and 14 (high resting and low contact times) due to the presence of carbon under aerobic conditions (Hu et al., 2009). The removal percentages for the high concentrations of dyes were better than the ones for the low concentration of dyes due to a corresponding increase in carbon percentage.

#### 4. Conclusions and recommendations for further research

The presence of P. australis did not affect dye removal (p > 0.05), while it had a minor but insignificant impact on COD removal (p > 0.05). The use of plants concerning the short contact time scenario for NH<sub>4</sub>-N and a low concentration of AB113 is linked to a better removal compared to other cases apart from that for all nutrients; particularly PO4-P and NO3-N. In case of low dye concentrations, the presence of plants for the long contact time scenario impacted significantly (p<0.05) positive on the removal efficiencies of PO4-P, NH4-N and NO3-N.

The vertical-flow constructed wetlands showed significantly (p < 0.05) good dye removals for low and high concentrations of BR46, and low concentrations of AB113 during all seasons under greenhouse conditions. The COD removal was a function of the dye used. This was explained by the complexity of the chemical structure of the dye and seasonal variations in temperature

For high concentrations of BR46, wetlands with a long hydraulic retention time were significantly (p < 0.05) better that those with a short hydraulic retention time for the removal of PO<sub>4</sub>-P. For AB113. wetlands with a shorter hydraulic retention time removed PO4-P significantly (p < 0.05) well. In case of the dye AB113, the four-day contact time was not enough to remove or reduce TSS and turbidity.

The authors recommend to assess (a) the effects of different plant species and an increase of the contact time on particle and dye removal processes; (b) the effect of pH (low and high) on dye removal; (c) the treatment of synthetic textile wastewater contaminated with different dyes, matching the typical characteristics of real effluent from a textile factory; (d) the impact of different dyes mixed with other contaminants and plant growth parameters such as stem diameter; and (e) wetland design implications based on the research findings in terms of investment of area and costs.

#### **Competing interest**

Authors have no competing interests.

## Supplementary material

Note that on-line supplementary material is associated with this article. Supplementary material S1 comprises a discussion on tests of normality supported by two tables. Supplementary Material S2 shows the temporal variation of temperature throughout the experimental time period.

#### Acknowledgements

The lead author received a PhD Studentship from the Ministry of Higher Education and Scientific Research of the Iraqi Government via Al-Muthanna University. The authors acknowledge the technical support provided by Dina Ali Yaseen.

### Appendix A. Supplementary data

Supplementary data associated with this article can be found, in the online version, at http://dx.doi.org/10.1016/j.ecoleng.2017.01. 016.

#### References

- APHA, 1995. Standard Methods for the Examination of Water and Wastewater. American Public Health Association (APHA), Baltimore.
  Brix, H., Dyhr-Jensen, K., Lorenzen, B., 2002. Root-zone acidity and nitrogen source affects Typha latifolia L. growth and uptake kinetics of ammonium and nitrate. J. Experimental Bot. 53 (379), 2441–2450.
  Brix, H., 1997. Do macrophytes play a role in constructed treatment wetlands? Water Sci. Technol. 35 (5), 11–17.
  Bulc, T.G., Ojstršek, A., 2008. The use of constructed wetland for dye-rich textile wastewater treatment. J. Hazard. Mater. 155 (1–2), 76–82.
  Chen, J., Wang, Q., Hua, Z., Du, G., 2007. Research and application of biotechnology in textile industries in China. Enzyme Microb. Technol. 40 (7), 1651–1655.
  Cheng, B., Hu, C., Zhao, Y., 2011. Effects of plants development and pollutant loading on performance of vertical subsurface flow constructed wetlands. Int. J. Environ. Sci. Technol. 81 (1), 177–186.
- Environ. Sci. Technol. 8 (1), 177–186. Davies, L.C., Carias, C.C., Novais, J.M., Martins-Dias, S., 2005. Phytoremediation of
- textile effluents containing azo dye by using *Phragmites australis* in a vertical flow intermittent feeding constructed wetland. Ecol. Eng. 25 (5), 594–605. Davies, L.C., Pedro, I.S., Novais, J.M., Martins-Dias, S., 2006. Aerobic degradation of acid orange 7 in a vertical-flow constructed wetland. Water Res. 40 (10),
- Deniz, F., Karaman, S., 2011. Removal of Basic Red 46 dye from aqueous solution by , Chem, Eng. 170 (1), 67-74.
- Dzurik, A.A., 2003. Water Resources Planning, USA. Rowman & Littlefield rs. New York. Erkurt, H.A., 2010, Biodegradation of Azo Dyes, Springer, Berli
- Ferreira, R.A., Duarte, J.G., Vergin, P., Antunes, C.D., Freire, F., Martins-Dias, S., 2014. Phragmites sp. physiological changes in a constructed wetland treat ting

an effluent contaminated with a diazo dye (DR81). Environ. Sci. Pollut. Res. 21 (16) 9626-9643

- (16), 9626–9643, Fuchs, V.J., 2009. Nitrogen Removal and Sustainability in Vertical Flow Conctructed Wetlands for Small Scale Wastewater Treatment. PhD Thesis. Michigan Technology University, Houghton. Haslam, S.M., 1972. Phragmites communis Trin. (Arundo Phragmites L., Phragmites australis (Cav.) Trin. ex Steudel). J. Ecol. 60 (2), 585–610.Hu, J., Li, D.-P., Liu, Q., Tao, Y., He, X.-H., Wang, X.-M., Li, X.-D., Ping, G., 2009. Effect of organic carbon on prification effective and community composition of
- nitrifying biofilms. Environ, Sci. 21 (3), 387–394. Islam, M.M., Mahmud, K., Faruk, O., Billah, M.S., 2011. Textile dyeing industries in Bangladesh for sustainable development, Int. J. Environ. Sci. Dev. 2 (6),
- 428-450. Kadlec, R.H., Wallace, S., 2009. Treatment Wetlands, second ed. CRC Press, Boca
- Raton, Florida. Kasuya, E., 2001. Mann–Whitney U test when variances are unequal. Anim. Behav.
- Kestye, J., 2007. In Semini-winner yourses when variances are unequal. Alum. Behav. 61 (6), 1247–1249.
  Keskinkan, O., Lugal Coksu, M., 2007. Assessment of the dye removal capability of submersed aquatic plants in a laboratory-scale wetland system using anova. Braz, J. Chem. Eng. 24 (2), 193–202.
  Khatae, AR, 2009. Photoccatalytic removal of C.I. Basic Red 46 on immobilized TiO2 nanoparticles: artificial neural network modelling. Environ. Technol. 30 (11), 1155–1165.
- 1155-1168
- Lin, Y.F., Jing, S.R., Lee, D.Y., Chang, Y.F., Chen, Y.M., Shih, K.C., 2005, Performance of
- a constructed wetland treating intensive shrimg anguaculture wastewater under high hydraulic loading rate. Environm, Pollut, 134 (3), 411-421, Mbuligwe, SE, 2005. Comparative treatment of dye-rich wastewater in engineered wetland systems (EWSs) vegetated with different plants. Water Per: 30(3), 271-280.
- 271-280 Res. 39 (2), 271–280. Mustafa, A., Scholz, M., Harrington, R., Carrol, P., 2009. Long-term performance of representative integrated constructed wetland treating farmyard runoff. Ecol. nce of a
- Yellow
- Ong, S.-A., Ho, L.-N., Wong, Y.-S., Dugil, D.L., Samad, H., 2011. Semi-batch operated constructed wetlands planted with *Phraqmites asutralis* for treatment of dying constructed wetlands planted with Phragmites asutralis for treatment of dy wastewater. Eng. Sci. Technol. 6 (5), 619–627.Pagter, M., Bragato, C., Brix, H., 2005. Tolerance and physiological responses of Phragmites australis to water deficit. Aquat. Bot. 81 (4), 285–299.

- Pandey, A., Singh, P., Iyengar, L., 2007. Bacterial decolorization and degradation of azo dyes. Int. Biodeterior. Biodegrad. 59 (2), 73–84.
  Pervez, A., Headley, A.D., Terzis, E., 2000. The treatment of azo dyes using reed bed treatment systems. In: Means, J.L., Hinchee, R.E. (Eds.), Wetlands and
- Remediation. Battelle Press, Columbus, pp. 187–194. Postolache, O.A., Girão, P., Pereira, J.D., Ramos, H.M.G., 2007. Multibeam optical system and neural processing for turbidity measurement. Sens. J. 7 (5).
- 677-684. Razali, NM, Wah, Y.B., 2011. Power comparisons of shapiro-wilk, kolmogorov-smirnov, lilliefors and anderson-darling tests. J. Statist. Model, Analyt. 2 (1), 21-33. Scholz, M., 2015. Wetlands for Water Pollution Control, second ed. Elsevier,
- Shapiro, S.S., Wilk, M.B., 1965. An analysis of variance test for normality (complete
- Shapho, S.J., Wita, J.S.J., and Mary Sol Virality feet of the informative (compreters samples). Biometrika 52 (3–4), 591–611.
   Shirzad-Siboni, M., Jafari, S.J., Giahi, O., Kim, I., Lee, S.-M., Yang, J.-K., 2014. Removal of acid blue 113 and reactive black 5 dye from aqueous solutions by activated red mud. J. Ind. Eng. Chem. 20 (4), 1432–1437.
   Sivakumar, D., Shankar, D., Vijaya Prathima, A., Valarmathi, M., 2013. Constructed industry wastewater using aquatic macrophytes.
- nd treatment of te Environ, Sci. 3 (4), 1223-1232 Stefanakis, A., Akratos, C.S., Tsihinitzis, V.A., 2014. Vertical Flow Constructed Wetlands: Eco-engineering Systems for Wastewater and Sludge Treatment.
- sterdam Stoline, M.R., 1981. The status of multiple comparisons: simultaneous estin
- of all pair wise comparisons in one-way ANOVA designs. Am. Statistician 35 (3), 134-141
- Suthersan, S.S., 2001, Natural and Enhanced Remediation Systems, Lewis
- Publisher, Washington, DC. Tee, H.-C., Lim, P.-E., Seng, C.-E., Nawi, M.A.M., Adnan, R., 2015. Enhancement of azo dye Acid Orange 7 removal in newly developed horizontal subsurface-fit
- azo dye Acid Orange 7 removal in newly developed horizontal subsurface-flow constructed wetland, Environ, Manage. 147, 349–355. Van Loosdrecht, M., Clement, J., 2005. 2nd IWA leading-edge conference on water and wastewater treatment technologies. Water Intell. Online 4, 344–345. Vymazal, J., 2014. Constructed wetlands for treatment of industrial wastewaters: a review. Ecol. Eng. 73, 724–751. Valcuk, A., Dodu, G., 2014. Teasure
- review. Ecol. Eng. 73, 724-751. Yalcuk, A., Dogdu, G., 2014. Treatment of azo dye Acid Yellow 2G by using lab-scale vertical-flow intermittent feeding constructed wetlands. Selcuk Univ. Nat.
- Yaseen, D.A., Scholz, M., 2016, Shallow pond systems planted with Lemna minor treating azo dyes. Ecol. Eng. (in press).

## **APPENDIX B**

# Test of normality for plant and water quality variables

## **B.1 Azo textile Dyes**

The corresponding results for the plant dimensions are shown in Table B.1. The lengths were approximately normally distributed for Wetlands, 12 and 13, 16 and 17, and 18 and 19, while the opposite was the case for Wetlands 6 and 7, 8 and 9, 10 and 11, 14 and 15, and 20 and 21. The diameters were approximately non-normally distributed for all wetlands as shown in Table B.1 and Figures B1 and B2 as s sample

Sieuu. (C	John Keeu	) Telated to Table	t 4.1	
	Characteris	stic	Test of normality (if p	$\geq 0.05$ , data are normally
			distributed; if $p < 0.05$	, data are not normally
			distri	buted)
Dye	Wetland	Number of	Length (mm)	Diameter (mm)
	numbers	stems		
BR46	8 and 9	40	0.001	0.000
	10 and 11	28	0.024	0.011
	14 and 15	34	0.000	0.006
	18 and 19	23	0.158	0.007
AB113	6 and 7	45	0.000	0.004
	12 and 13	30	0.291	0.002
	16 and 17	36	0.518	0.038
	20 and 21	28	0.017	0.049

Table B.1 Test of normality for dimensions of *Phragmites australis* (Cav.) Trin. ex Steud. (Common Reed) related to Table 4.1

Note: BR, basic red; AB, acid blue; Sig., p value depending on the Shapiro-Wilk test.


Figure B1 Normal Q-Q plot for ortho-phosphate-phosphorus for Wetlands 20/21



Figure B2. Normal Q-Q plot for ammonia-nitrogen for Wetlands 20/21

Moreover, the corresponding results for AB113 are shown in Table B.2. Wetland 5 was approximately normally distributed for COD and NH<sub>4</sub>-N, while the opposite was the case for dye concentration, pH, redox potential, EC, TSS, turbidity, DO, PO<sub>4</sub>-P and NO<sub>3</sub>-N. For Wetlands 6 and 7, the results were approximately normally distributed for DO, COD, PO<sub>4</sub>-P and NH<sub>4</sub>-N, while findings were not normally distributed for dye concentration, pH, redox potential, EC, TSS, turbidity and NO<sub>3</sub>-N. For Wetlands 12 and 13, the obtained results were approximately normally distributed for dye concentration, pH, redox potential, EC, TSS, turbidity and NO<sub>3</sub>-N. For Wetlands 12 and 13, the obtained results were approximately normally distributed for dye concentration, COD and NH<sub>4</sub>-N, while not normally distributed for pH, redox potential, EC, DO, TSS, turbidity, PO<sub>4</sub>-P and NO<sub>3</sub>-N. For the high concentration, the obtained results for Wetlands 16 and 17 showed approximately normally distributed for dye concentration, pH, DO, EC, TSS, turbidity, NH<sub>4</sub>-N and NO<sub>3</sub>-N. For Wetlands 20 and 21, the obtained results showed data that were normally distributed for redox potential, EC, DO, and PO<sub>4</sub>-P, while they were not normally distributed for PO, DO, NH<sub>4</sub>-N and NO<sub>3</sub>-N.

For the dye BR46, in the case of low concentration, the obtained results for Wetland 4 (Table B.2) showed normally distributed data for pH, redox potential, COD, NH<sub>4</sub>-N, NO<sub>3</sub>-N and PO<sub>4</sub>-P, while not normally distributed findings were recorded for dye concentration, DO, TSS, turbidity and EC. For Wetlands 8 and 9, the obtained results showed normally distributed data for COD, NH<sub>4</sub>-N, NO<sub>3</sub>-N and PO<sub>4</sub>-P, while not normally distributed for dye concentration, pH, redox potential, DO, TSS, turbidity and EC. For Wetlands 10 and 11, the results showed normal distributions for redox potential, COD, NH<sub>4</sub>-N and NO<sub>3</sub>-N, while not normal distributed data for dye

concentration, pH, DO, TSS, turbidity, EC and PO<sub>4</sub>-P. In the case of high concentration, for Wetlands 14 and 15, the obtained results showed normal distributions for pH and DO, while not normal distributed output for dye concentration, redox potential, TSS, turbidity, EC, COD, PO<sub>4</sub>-P, NO<sub>3</sub>-N and NH<sub>4</sub>-N. For Wetlands 18 and 19, the obtained results showed normal distributions for pH, redox potential, DO, EC and COD, while not normally distributed data were noted for dye concentration, TSS, turbidity, PO<sub>4</sub>-P, NH<sub>4</sub>-N and NO<sub>3</sub>-N as shown in Table B.2.

Characteristic		Test of normality (if $p \ge 0.05$ , data are normally distributed; if $p < 0.05$ , data are not normally distributed)				
Dye	Wetland	No. of	Dye concentration (mg/l)	рН (–)	Redox potential (mV)	Dissolved oxygen
	number(s)	samples				(mg/l)
BR46	4	70	0.000	0.009	0.532	0.029
	8 and 9	70	0.000	0.019	0.006	0.020
	10 and 11	70	0.002	0.000	0.110	0.000
	14 and 15	70	0.000	0.071	0.034	0.287
	18 and 19	35	0.029	0.930	0.381	0.955
AB113	5	70	0.000	0.005	0.031	0.000
	6 and 7	70	0.000	0.000	0.000	0.277
	12 and 13	70	0.068	0.029	0.019	0.000
	16 and 17	70	0.000	0.050	0.302	0.017
	20 and 21	35	0.000	0.004	0.051	0.100
			Total suspended solids	Turbidity (NTU)	Electrical conductivity	
			(mg/l)		(µS/cm)	
BR46	4	70	0.000	0.000	0.021	
	8 and 9	70	0.000	0.000	0.000	
	10 and 11	70	0.000	0.000	0.006	
	14 and 15	70	0.000	0.000	0.027	
	18 and 19	35	0.000	0.000	0.238	
AB113	5	70	0.047	0.000	0.033	
	6 and 7	70	0.000	0.000	0.001	
	12 and 13	70	0.000	0.003	0.011	
	16 and 17	70	0.006	0.000	0.041	
	20 and 21	35	0.029	0.021	0.113	

Table B.2 Test of normality for effluent water quality characteristics regarding general physical and chemical variables related to Tables 4.2, 4.3 and 4.4

Dye	Wetland	No. of	Chemical oxygen demand	Ortho-phosphate-phosphorus	Ammonia-nitrogen	Nitrate-nitrogen
	number(s)	samples	(mg/l)	(mg/l)	(mg/l)	(mg/l)
BR46	4	20	0.327	0.196	0.122	0.107
	8 and 9	20	0.578	0.416	0.075	0.275
	10 and 11	20	0.850	0.017	0.243	0.096
	14 and 15	20	0.029	0.017	0.012	0.015
	18 and 19	20	0.420	0.000	0.010	0.031
AB113	5	20	0.261	0.000	0.352	0.001
	6 and 7	20	0.197	0.069	0.360	0.036
	12 and 13	20	0.466	0.000	0.449	0.011
	16 and 17	20	0.283	0.862	0.012	0.018
	20 and 21	20	0.015	0.857	0.001	0.006

Table B.2 continued

Note: BR, basic red; AB, acid blue; Sig., p value depending on the Shapiro-Wilk test.

## **B.2** Artificial wastewater containing two azo textile dyes

The corresponding results for the plants' dimension are shown in Table B.3. The lengths were approximately normally distributed for Wetlands 3, 4, 7, 8, 9, 10, 11, 12, 13, 14, 15, 16, 17 and 18 while the opposite was the case for Wetlands 5 and 6. The diameters were approximately normally distributed for Wetlands 3, 4, 6, 8, 9, 10, 11, 13, 14, 16, 17 and 18 while the opposite was the case for Wetlands 7, 12 and 15.

Regarding water quality variables in case of AB113, as shown in Table B.4, Wetlands 2, 3 and 9 were approximately no normally distribution for all variables (dye concentration, pH, redox, EC, TSS, turbidity, DO, colour, COD, PO<sub>4</sub>-P, NH<sub>3</sub>-N and NO<sub>3</sub>-N). Wetland 13 was approximately normally distribution for redox, NH<sub>4</sub>-N and COD, while no normally distribution for dye concentration, pH, EC, TSS, turbidity, DO, PO<sub>4</sub>-P, NO<sub>3</sub>-N and colour. Wetland 17 was normally distribution for NO<sub>3</sub>-N, while the opposite for dye concentration, pH, redox, EC, DO, TSS, turbidity, PO<sub>4</sub>-P, NH<sub>4</sub>-N, COD and colour.

In case of BR46, Wetland 1 was approximately normally distribution for COD, while the opposite for dye concentration, pH, redox, EC, TSS, turbidity, DO, colour, NH<sub>4</sub>-H, NO<sub>3</sub>-N and PO<sub>4</sub>-P. Wetland 5 was approximately normally distribution for pH and DO, while the opposite for dye concentration, redox, EC, TSS, turbidity, colour, NH<sub>4</sub>-N, NO<sub>3</sub>-N, PO<sub>4</sub>-P and COD. Wetland 7 was approximately normally distribution for DO, PO<sub>4</sub>-P and COD. Wetland 7 was approximately normally distribution for DO, PO<sub>4</sub>-P and COD, while the opposite for dye concentration, pH, redox, EC, TSS, turbidity, colour, NH<sub>4</sub>-N and NO<sub>3</sub>-N. Wetland 11 was approximately normally distribution for DO, while the opposite for dye concentration, pH, redox, EC, TSS, turbidity, colour, NH<sub>4</sub>-N, NO<sub>3</sub>-N, PO<sub>4</sub>-P and COD. Wetland 11 was approximately normally distribution for DO, while the opposite for dye concentration, pH, redox, EC, TSS, turbidity, colour, NH<sub>4</sub>-N, NO<sub>3</sub>-N, PO<sub>4</sub>-P and COD. Wetland 11 was approximately normally distribution for DO, while the opposite for dye concentration, pH, redox, EC, TSS, turbidity, colour, NH<sub>4</sub>-N, NO<sub>3</sub>-N, PO<sub>4</sub>-P and COD. Wetland 11 was approximately normally distribution for DO, while the opposite for dye concentration, pH, redox, EC, TSS, turbidity, colour, NH<sub>4</sub>-N, NO<sub>3</sub>-N, PO<sub>4</sub>-P and COD. Wetland 15 was approximately

normally distribution for PO<sub>4</sub>-P and COD, while the opposite for dye concentration, pH, redox, TSS, turbidity, DO, colour, EC, NH<sub>4</sub>-N and NO<sub>3</sub>-N.

In case of the mixture between both of previous dyes, Wetlands 4 and 12 were approximately no normally distribution for all variables (dye concentration, pH, redox, DO, TSS, turbidity, EC, colour, NH<sub>4</sub>-N, NO<sub>3</sub>-N, PO<sub>4</sub>-P and COD. Wetlands 6 was approximately normally distribution for PO<sub>4</sub>-P, while the opposite for the dye concentration, pH, redox, TSS, turbidity, EC, colour, DO, NH<sub>4</sub>-N, NO<sub>3</sub>-N and COD. Wetland 8 was approximately distribution for redox, DO and PO<sub>4</sub>-P, while the opposite for dye concentration, pH, EC, TSS, turbidity, colour, COD, NH<sub>4</sub>-N and NO<sub>3</sub>-N. Wetland 10 was approximately normally distribution for dye concentration, while the opposite for pH, redox, EC, TSS, turbidity, DO, colour, COD, NH<sub>4</sub>-N, NO<sub>3</sub>-N and PO<sub>4</sub>-P. Wetland 14 was approximately normally distribution for redox, while the opposite for dye concentration, pH, EC, TSS, DO, turbidity, colour, COD, NH<sub>4</sub>-N, NO<sub>3</sub>-N and PO<sub>4</sub>-P. Wetland 16 was approximately normally distribution for COD, NH<sub>4</sub>-N, NO<sub>3</sub>-N and PO<sub>4</sub>-P, while the opposite for dye concentration, pH, redox, TSS, EC, DO, turbidity and colour. Wetland 18 was approximately normally distribution for dye concentration, turbidity, DO, COD, PO<sub>4</sub>-P, NH<sub>4</sub>-N and NO<sub>3</sub>-N, while the opposite for pH, redox, EC, TSS and colour.

	Characteristic	C	Test of normality (if $p \ge 0.05$ , data are normally distributed; if $p < 0.05$ , data are not normally			
			distribut	ed)		
Dye	Wetland	Number	Length (mm)	Diameter (mm)		
	number(s)	of stems				
BR46	5	68	0.042	0.015		
	7	20	0.513	0.000		
	11	17	0.720	0.062		
	15	14	0.131	0.025		
AB113	3	40	0.060	0.402		
	9	15	0.082	0.547		
	13	20	0.113	0.937		
	17	10	0.335	0.824		
The	4	48	0.287	0.364		
mixture	6	55	0.004	0.545		
of the	8	7	0.729	0.970		
dyes	10	10	0.592	0.822		
	12	10	0.655	0.004		
	14	16	0.548	0.972		
	16	7	0.685	0.996		
	18	9	0.122	0.790		

Table B.3 Test of normality for dimensions of *Phragmites australis* (Cav.) Trin. ex Steud. (Common Reed) related to Table 5.1

Note: BR, basic red; AB, acid blue.

Characteristic		Test of normality (if $p \ge 0.05$ , data are normally distributed; if $p < 0.05$ , data are not normally				
				distri	buted)	
Dye	Wetland	No. of	Dye concentration	рН (–)	Redox potential (mV)	Dissolved oxygen
	number(s)	samples	(mg/l)			(mg/l)
BR46	1	82	0.003	0.000	0.038	0.000
	5	82	0.002	0.916	0.004	0.050
	7	82	0.000	0.000	0.033	0.564
	11	81	0.000	0.000	0.000	0.445
	15	41	0.010	0.000	0.000	0.014
AB113	2	82	0.000	0.000	0.000	0.000
	3	82	0.000	0.000	0.000	0.000
	9	82	0.000	0.000	0.000	0.000
	13	81	0.000	0.000	0.110	0.000
	17	41	0.001	0.000	0.002	0.000
The mixture	4	82	0.015	0.000	0.000	0.000
of both of	6	82	0.021	0.000	0.000	0.000
dyes	8	82	0.000	0.001	0.053	0.585
	10	82	0.059	0.000	0.000	0.024
	12	81	0.000	0.000	0.000	0.000
	14	81	0.000	0.000	0.194	0.000
	16	41	0.000	0.000	0.014	0.000
	18	41	0.051	0.021	0.033	0.411

 Table B.4 Test of normality for effluent water quality characteristics regarding general physical and chemical variables related to Tables 5.2,

 5.3 and 5.6

Dye	Wetland	No. of	Total suspended solids	Turbidity	Electric conductivity	Colour (Pt Co.)
	number(s)	samples	(mg/l)	(NTU)	(µS/cm)	
BR46	1	82	0.006	0.043	0.000	0.000
	5	82	0.002	0.000	0.000	0.000
	7	82	0.000	0.000	0.000	0.000
	11	81	0.009	0.042	0.000	0.001
	15	41	0.000	0.002	0.000	0.005
AB113	2	82	0.001	0.000	0.000	0.000
	3	82	0.006	0.000	0.000	0.014
	9	82	0.000	0.002	0.000	0.005
	13	81	0.012	0.004	0.000	0.000
	17	41	0.004	0.003	0.000	0.000
The	4	82	0.000	0.000	0.000	0.000
mixture	6	82	0.000	0.000	0.000	0.000
of both	8	82	0.000	0.000	0.000	0.000
of dyes	10	82	0.000	0.000	0.000	0.000
	12	81	0.000	0.000	0.000	0.000
	14	81	0.000	0.000	0.000	0.021
	16	41	0.000	0.000	0.002	0.015
	18	41	0.026	0.062	0.000	0.000

Table B.4 continued

Dye	Wetland	No. of	Chemical oxygen	Ortho-phosphate-	Ammonia-nitrogen	Nitrate-nitrogen
	number(s)	samples	demand (mg/l)	phosphorus (mg/l)	(mg/l)	(mg/l)
BR46	1	30	0.357	0.018	0.026	0.006
	5	30	0.000	0.015	0.000	0.000
	7	30	0.210	0.324	0.000	0.000
	11	30	0.001	0.000	0.038	0.009
	15	30	0.318	0.151	0.012	0.004
AB113	2	30	0.000	0.000	0.021	0.000
	3	30	0.017	0.009	0.002	0.000
	9	30	0.035	0.000	0.006	0.000
	13	30	0.112	0.003	0.073	0.002
	17	30	0.000	0.001	0.000	0.587
The	4	30	0.000	0.000	0.037	0.000
mixture	6	30	0.000	0.079	0.000	0.000
of both	8	30	0.000	0.055	0.000	0.000
of dyes	10	30	0.000	0.000	0.008	0.000
	12	30	0.000	0.000	0.010	0.000
	14	30	0.000	0.000	0.016	0.000
	16	30	0.175	0.612	0.075	0.109
	18	30	0.990	0.149	0.092	0.006

Tab	ole B	4 co	ontini	ıed

Note: BR; basic red, AB; acid blue